

User's Guide NWA/WAC/WAX Series

802.11 a/b/g/n/ac/ax Access Point

Default Login Details

Management IP Address	http://DHCP-assigned IP OR http://192.168.1.2
User Name	admin
Password	1234

Version 6.10-6.20 Edition 1, 5/2021



Copyright $\ensuremath{\textcircled{\odot}}$ 2021 Zyxel and/ or its affiliates. All rights reserved.

IMPORIANT!

READ CAREFULLY BEFORE USE

KEEP THIS GUIDE FOR FUTURE REFERENCE.

This is a User's Guide for a series of products. Not all products support all firmware features. Screenshots and graphics in this book may differ slightly from your product due to differences in your product hardware, firmware, or your computer operating system. Every effort has been made to ensure that the information in this manual is accurate.

Some screens or options in this book may not be available for your product (see the product feature tables in Section 1.4 on page 18).

Related Documentation

• Quick Start Guide

The Quick Start Guide shows how to connect the Zyxel Device and access the Web Configurator.

CLI Reference Guide

The CLI Reference Guide explains how to use the Command-Line Interface (CLI) and CLI commands to configure the Zyxel Device.

Note: It is recommended you use the Web Configurator to configure the Zyxel Device.

• Web Configurator Online Help

Click the help icon in any screen for help in configuring that screen and supplementary information.

• Nebula Control Center User's Guide

This User's Guide shows how to manage the Zyxel Device remotely. The features of these devices can be managed through Nebula Control Center. It also offers features that are not available when the Zyxel Device is in standalone mode (see Section 2.1.2 on page 26).

• NXC Series User's Guide

See this User's Guide for instructions on using the NXC as an AP Controller (AC) for the Zyxel Device. This is used when the Zyxel Device is set to be managed by a Zyxel AC.

More Information

Go to **support.zyxel.com** to find other information on the Zyxel Device.



Document Conventions

Warnings and Notes

These are how warnings and notes are shown in this guide.

Warnings tell you about things that could harm you or your device.

Note: Notes tell you other important information (for example, other things you may need to configure or helpful tips) or recommendations.

Syntax Conventions

- All models in this series may be referred to as the "Zyxel Device" in this guide.
- Product labels, screen names, field labels and field choices are all in **bold** font.
- A right angle bracket (>) within a screen name denotes a mouse click. For example, Configuration > Network > IP Setting means you first click Configuration in the navigation panel, then the Network sub menu and finally the IP Setting tab to get to that screen.

Icons Used in Figures

Figures in this guide may use the following generic icons. The Zyxel Device icon is not an exact representation of your device.

Zyxel Device	Router	Switch	Internet
Server	Desktop		AP Controller
Printer	Nebula Switch	Nebula Gateway	Smart T.V.
IP Phone			

Contents Overview

Introduction	
AP Management	
Hardware	
Web Configurator	
Standalone Configuration	41
Standalone Configuration	
Dashboard	
Setup Wizard	
Monitor	
Network	
Wireless	
User	
AP Profile	
MON Profile	
WDS Profile	
Certificates	
System	
Log and Report	
File Manager	
Diagnostics	
LEDs	
Reboot	
Shutdown	
Local Configuration in Cloud Mode	
Cloud Mode	
Network	
Maintenance	
Appendices and Trouble shooting	
Troubleshooting	

Table of Contents

Document Conventions	
Contents Overview	4
Table of Contents	5
Chapter 1 Introduction	12
1.1 Overview	
1.2 Zyxel Device Roles	
1.2.1 Root AP	
1.2.2 Wireless Repeater	
1.2.3 Radio Frequency (RF) Monitor	
1.3 Sample Feature Applications	
1.3.1 MBSSID	
1.3.2 Dual-Radio	
1.4 Zyxel Device Product Feature	
Chapter 2 AP Management	
2.1 Management Mode	
2.1.1 Standalone	
2.1.2 Nebula Control Center	
2.2 Switching Management Modes	
2.3 Zyxel One Network (ZON) Utility	
2.3.1 Requirements	
2.3.2 Run the ZON Utility	
2.4 Ways to Access the Zyxel Device	
2.5 Good Habits for Managing the Zyxel Device	
Chapter 3 Hardware	
3.1 Zyxel Device Single LED	
3.1.1 Zyxel Device LED	
Chapter 4 Web Configuration	20
Web Configurator	
4.1 Overview	
4.2 Accessing the Web Configurator	
4.3 Navigating the Web Configurator	

4.3.1 Title Bar	
4.3.2 Navigation Panel	
4.3.3 Standalone Mode Navigation Panel Menus	
4.3.4 Cloud Mode Navigation Panel Menus	
4.3.5 Tables and Lists	
Part I: Standalone Configuration	
Chapter 5	
Standalone Configuration	42
5.1 Overview	
5.2 Starting and Stopping the Zyxel Device	
Chapter 6 Dashboard	44
6.1 Overview	
6.1.1 CPU Usage	
6.1.2 Memory Usage	
Chapter 7	
Se tup Wiza rd	49
7.1 Accessing the Wizard	40
7.1 Accessing the Wizard	
7.2.1 Step 1 Time Settings	
7.2.2 Step 2 Password and Uplink Connection	
7.2.3 Step 3 Radio	
7.2.4 Step 4 SSID	
7.2.5 Summary	
Chapter 8	
Monitor	
8.1 Overview	
8.1.1 What You Can Do in this Chapter	
8.2 What You Need to Know	
8.3 Network Status	
8.3.1 Port Statistics Graph	
8.4 Radio List	
8.4.1 AP Mode Radio Information	
8.5 Station List	61
8.6 WDS Link Info	
8.7 Detected Device	

8.8 View Log	
Chapter 9	
Ne two rk	69
9.1 Overview	
9.1.1 What You Can Do in this Chapter	
9.2 IP Setting	
9.3 VLAN	
9.4 NCC Discovery	
Chapter 10	
Wire less	75
10.1 Overview	
10.1.1 What You Can Do in this Chapter	
10.1.2 What You Need to Know	
10.2 AP Management	
10.3 Rogue AP	
10.3.1 Add/Edit Rogue/Friendly List	
10.4 DCS	
10.5 Technical Reference	
Chapter 11	
Use r	86
11.1 Overview	
11.1.1 What You Can Do in this Chapter	
11.1.2 What You Need To Know	
11.2 User Summary	
11.2.1 Add/Edit User	
11.3 Setting	
11.3.1 Edit User Authentication Timeout Settings	
Chapter 12	
AP Pro file	93
12.1 Overview	
12.1.1 What You Can Do in this Chapter	
12.1.2 What You Need To Know	
12.2 Radio	
12.2.1 Add/Edit Radio Profile	
12.3 SSID	
12.3.1 SSID List	
12.3.2 Add/Edit SSID Profile	
12.4 Security List	
12.4.1 Add/Edit Security Profile	

12.5 MAC Filter List	115
12.5.1 Add/Edit MAC Filter Profile	115
Chapter 13 MON Profile	
13.1 Overview	
13.1.1 What You Can Do in this Chapter	
13.2 MON Profile	
13.2.1 Add/Edit MON Profile	118
Chapter 14	
WDS Pro file	120
14.1 Overview	120
14.1.1 What You Can Do in this Chapter	
14.2 WDS Profile	120
14.2.1 Add/Edit WDS Profile	121
Observe 15	
Chapter 15 Certific a tes	122
15.1 Overview	
15.1.1 What You Can Do in this Chapter	
15.1.2 What You Need to Know	
15.1.3 Verifying a Certificate	
15.2 My Certificates	
15.2.1 Add My Certificates	
15.2.2 Edit My Certificates	
15.2.3 Import Certificates	
15.3.1 Edit Trusted Certificates	
15.3.2 Import Trusted Certificates	
15.4 Technical Reference	
	10/
Chapter 16	100
Syste m	138
16.1 Overview	138
16.1.1 What You Can Do in this Chapter	138
16.2 Host Name	138
16.3 Date and Time	
16.3.1 Pre-defined NTP Time Servers List	
16.3.2 Time Server Synchronization	
16.4 WWW Overview	
16.4.1 Service Access Limitations	
16.4.2 System Timeout	143

	16.4.3 HTTPS	
	16.4.4 Configuring WWW Service Control	144
	16.4.5 HTTPS Example	
	16.5 SSH	
	16.5.1 How SSH Works	
	16.5.2 SSH Implementation on the Zyxel Device	
	16.5.3 Requirements for Using SSH	
	16.5.4 Configuring SSH	
	16.5.5 Examples of Secure Telnet Using SSH	
	16.6 FTP	
Chaj	pter 17	
Log a	and Report	157
	17.1 Overview	
	17.1.1 What You Can Do In this Chapter	
	17.2 Log Setting	
	17.2.1 Log Setting Screen	
	17.2.2 Edit System Log Settings	
	17.2.3 Edit Remote Server	
	17.2.4 Active Log Summary	
Chaj	pter 18	
File N	Manager	
	18.1 Overview	
	18.1.1 What You Can Do in this Chapter	
	18.1.2 What you Need to Know	
	18.2 Configuration File	
	18.2.1 Example of Configuration File Download Using FTP	
	18.3 Firmware Package	
	18.3.1 Example of Firmware Upload Using FTP	
	18.4 Shell Script	
	pter 19	
Dia g	no stic s	
	19.1 Overview	
	19.1.1 What You Can Do in this Chapter	
	19.2 Diagnostics	
	19.3 Remote Capture	
	pter 20	
LEDs		
	20.1 Overview	
	20.1.1 What You Can Do in this Chapter	

20.2 Suppression Screen	
20.3 Locator Screen	
Chapter 21 Reboot	183
21.1 Overview	
21.1.1 What You Need To Know	
21.2 Reboot	
Chapter 22	
Shutdown	
	104
22.1 Overview	
22.1.1 What You Need To Know	
22.2 Shutdown	
	105
Part II: Local Configuration in Cloud Mode	
Chapter 23	
Cloud Mode	
23.1 Overview	
23.2 Cloud Mode Web Configurator Screens	
23.3 Dashboard	
Chapter 24	
Ne two rk	
24.1 Overview	189
24.1.1 What You Can Do in this Chapter	
24.2 IP Setting	
24.3 VLAN	
Chapter 25	
Maintenance	
25.1 Overview	
25.1.1 What You Can Do in this Chapter	
25.2 Shell Script	
25.3 Diagnostics	
25.4 View Log	

Part III: Appendices and Trouble shooting1	97	'
--	----	---

Cha j	p te r	26
--------------	--------	-----------

The uble shooting	
26.1 Overview	
26.2 Power, Hardware Connections, and LED	
26.3 Zyxel Device Management, Access, and Login	
26.4 Internet Access	
26.5 WiFi Network	
26.6 Resetting the Zyxel Device	205
26.7 Getting More Troubleshooting Help	
Appendix A Importing Certificates	
Appendix B IPv6	230
Appendix C Customer Support	
Appendix D Legal Information	
Index	251

C HAPTER 1 Introduction

1.1 Overview

This User's Guide covers the models listed in the following table. They can be managed in one of the following methods: remote management through Nebula Control Center (NCC) or an AP Controller (AC) such as the NXC, or local management in Standalone Mode. Each Zyxel Device runs in standalone mode by default, but it is recommended to use NCC management if it is available for your device.

NCC, AC or Standalone (Nebula Flex PRO)

• NWA5123-AC HD

- WAC6103D-I
- WAC6303D-S
- WAC6502D-E
- WAC6502D-S
- WAC6503D-S
- WAC6552D-S
- WAC6553D-E
- WAC500
- WAC500H
- WAC5302D-Sv2
- WAX510D
- WAX610D
- WAX650S

For more information about Access Point (AP) management, see Section 2.1 on page 25.

Use the Zyxel Device to set up a wireless network with other IEEE 802.11a/b/g/n/ac/ax compatible devices in either 2.4 GHz and 5 GHz networks or both at the same time.

When two or more APs are interconnected, this network is called a Wireless Distribution System (WDS). See Section 1.2.2 on page 14 for more information on root and repeater APs and how to set them up.

1.2 Zyxel Device Roles

This section describes some of the different roles that your Zyxel Device can take up within a network. Not all roles are supported by all models (see Section 1.4 on page 18). The Zyxel Device can serve as a:

Access Point (AP) - This is used to allow wireless clients to connect to the Internet.

AC or Standalone

- NWA5123-AC
- - WAC5302D-S
- NWA1123-AC HD • NWA1302-AC

NCC or Standalone

• NWA1123-ACv2

• NWA1123-AC PRO

• NWA110AX

(Nebula Flex)

- NWA210AX
 - NWA1123ACv3

- Radio Frequency (RF) monitor An RF monitor searches for rogue APs to help eliminate network threats if it supports monitor mode and rogue APs detection/containment. An RF monitor cannot simultaneously act as an AP.
- Root AP A root AP connects to the gateway or switch through a wired Ethernet connection and has wireless repeaters connected to it to extend its range.
- Wireless repeater A wireless repeater wirelessly connects to a root AP and extends the network's wireless range.

The following figure shows a network setup that uses these different roles to create a secure Wireless Distribution System (WDS). The root AP (\mathbf{Y}) is connected to a network with Internet access and has a wireless repeater (\mathbf{X}) connected to it to expand the wireless network's range. Clients (\mathbf{A} , \mathbf{B} , and \mathbf{C}) can access the wired network through the wireless repeater and/or root AP.

If a client (**D**) tries to set up his own AP (**R**) with weak security settings, the network becomes exposed to threats. The RF monitor (**M**) scans the area to detect all APs, which can help the network administrator discover these rogue APs and remove them or use the NXC to quarantine them.



Figure 1 Sample Network Setup

1.2.1 RootAP

In Root AP mode, you can have multiple SSIDs active for regular wireless connections and one SSID for the connection with a repeater (repeater SSID). Wireless clients can use either SSID to associate with the Zyxel Device in Root AP mode. A repeater must use the repeater SSID to connect to the Zyxel Device in Root AP mode.

When the Zyxel Device is in Root AP mode, repeater security between the Zyxel Device and other repeaters is independent of the security between the wireless clients and the AP or repeater. When repeater security is enabled, both APs and repeaters must use the same pre-shared key. See Section 10.2 on page 76 and Section 14.2 on page 120 for more details.

Unless specified, the term "security settings" refers to the traffic between the wireless clients and the AP. At the time of writing, repeater security is compatible with the Zyxel Device only.

1.2.2 Wireless Repeater

Using Repeater mode, your Zyxel Device can extend the range of the WLAN. In the figure below, the Zyxel Device in Repeater mode (Z) has a wireless connection to the Zyxel Device in Root AP mode (X) which is connected to a wired network and also has a wireless connection to another Zyxel Device in Repeater mode (Y) at the same time. Z and Y act as repeaters that forward traffic between associated wireless clients and the wired LAN. Clients A and B access the AP and the wired network behind the AP through repeaters Z and Y.





When the Zyxel Device is in Repeater mode, repeater security between the Zyxel Device and other repeater is independent of the security between the wireless clients and the AP or repeater. When repeater security is enabled, both APs and repeaters must use the same pre-shared key. See Section 10.2 on page 76 and Section 14.2 on page 120 for more details.

For NCC managed devices, you only need to enable **AP Smart Mesh** to automatically create wireless links between APs. See the NCC User's Guide for more details.

To set up a WDS in standalone mode APs, do the following steps. You should already have the root AP set up (see the Quick Start Guide for hardware connections).

- 1 Go to Configuration > Object > WDS Profile in your root AP Web Configurator and click Add.
- 2 Enter a profile name, an SSID for the WDS, and a pre-shared key.
- **3** Do steps 1 and 2 for the wireless repeater using the same SSID and pre-shared key.
- 4 Once the security settings of peer sides match one another, the connection between the root and repeater Zyxel Devices is made.

To set up a WDS in NXC managed Zyxel Devices, see the NXC User's Guide.

1.2.3 Radio Frequency (RF) Monitor

The Zyxel Device can be set to work as an RF monitor to discover nearby Access Points. The information it obtains from other APs is used to tag possible rogue APs and quarantine them if the Zyxel Device is managed by the NXC (see Section 2.1.3 on page 27). If the Zyxel Device's radio setting is set to **MON Mode** (RF Monitor mode), it will serve as a dedicated RF monitor and its AP clients are disconnected.

The models that do not support **MON Mode** support **Rogue AP Detection** (see Section 10.3 on page 79). **Rogue AP Detection** allows the AP to scan all channels similar to **MON Mode** except that the Zyxel Device still works as an AP while it scans the environment for wireless signals. To see which Zyxel Devices support the RF Monitor feature, see Section 1.4 on page 18.

The Zyxel Device in **MON Mode** scans a range of WiFi channels that you specify in a **MON Profile**, either in the 2.4 GHz or 5 GHz band. To scan both bands, you need to set both radio 1 and radio 2 in **MON Mode**. Once a rogue AP is detected, the network administrator can manually change the network settings to limit its access to the network using its MAC address or have the device physically removed. If the Zyxel Device is managed by an NXC, the network administrator can also use **Rogue AP Containment** through the NXC.

MON Mode in Standalone Mode

To use an RF monitor in standalone mode, do the following steps:

- 1 Create a **MON Profile** in **Configuration > Object > MON Profile > Add**. Specify a **Channeldwell time** to determine how long the RF monitor scans a specific channel before moving to the next one.
- 2 To scan all 2.4 GHz and 5 GHz channels, select auto in Scan Channel Mode. Make sure that the Activate check box is selected and click OK.
- 3 Go to the Configuration > Wireless > AP Management screen and set Radio 1 OP Mode (2.4 GHz) and/or Radio 2 OP Mode (5 GHz) to MON Mode.
- 4 Select the Radio 1(2) Profile that you created in the previous step. Make sure that the Radio 1(2) Activate check box is selected and click Apply.
- 5 Go to Monitor > Wire less > Detected Device to see a list of APs scanned by the RF monitor.
- 6 Select an AP or APs in the list and click **Mark as Rogue AP** or **Mark as Friendly AP**.

MON Mode in NXC-Managed Zyxel Devices

For NXC-managed Zyxel Devices, do the following steps in the NXC Web Configurator:

- 1 Create a MON Profile in CONFIGURATION > Object > MON Profile > Add. Specify a Channel dwell time to determine how long the RF monitor scans a specific channel before moving to the next one.
- 2 To scan all 2.4 GHz and 5 GHz channels, select auto in Scan Channel Mode. Make sure that the Activate check box is selected and click OK.
- 3 Go to the CONFIGURATION > Wireless > AP Management > Mgmt. AP List > Edit screen and/or set Radio 1 OP Mode (2.4 GHz) and Radio 2 OP Mode (5 GHz) to MON Mode.

- 4 Select the Radio 1(2) Profile that you created in the previous step. Select Ovenide Group Radio Setting and click OK.
- 5 Go to MONTIOR > Wire less > Detected Device to see a list of APs scanned by the RF monitor.
- 6 Select an AP or APs in the list and click Mark as Rogue AP or Mark as Friendly AP.
- 7 To quarantine a rogue AP, go to CONFIGURATION > Wire less > Rogue AP, select the APs you want to quarantine, and click Containment. Make sure the Enable Rogue AP Containment check box is selected, and click Apply.

1.3 Sample Feature Applications

This section describes some possible scenarios and topologies that you can set up using your Zyxel Device.

1.3.1 MBSSID

A Basic Service Set (BSS) is the set of devices forming a single wireless network (usually an access point and one or more wireless clients). The Service Set IDentifier (SSID) is the name of a BSS. In Multiple BSS (MBSSID) mode, the Zyxel Device provides multiple virtual APs, each forming its own BSS and using its own individual SSID profile.

You can configure multiple SSID profiles, and have all of them active at any one time.

You can assign different wireless and security settings to each SSID profile. This allows you to compartmentalize groups of users, set varying access privileges, and prioritize network traffic to and from certain BSSs.

To the wireless clients in the network, each SSID appears to be a different access point. As in any wireless network, clients can associate only with the SSIDs for which they have the correct security settings.

For example, you might want to set up a wireless network in your office where Internet telephony (VoIP) users have priority. You also want a regular wireless network for standard users, as well as a 'guest' wireless network for visitors. In the following figure, **VoIP_SSID** users have QoS priority, **SSID01** is the wireless network for standard users, and **Guest_SSID** is the wireless network for guest users. In this example, the guest user is forbidden access to the wired Land Area Network (LAN) behind the AP and can access only the Internet.



1.3.2 Dual-Radio

Some of the Zyxel Device models are equipped with dual wireless radios. This means you can configure two different wireless networks to operate simultaneously.

Note: A different channel should be configured for each WLAN interface to reduce the effects of radio interference.

You could use the 2.4 GHz band for regular Internet surfing and downloading while using the 5 GHz band for time sensitive traffic like high-definition video, music, and gaming.



Figure 4 Dual-Radio Application

1.4 Zyxel Device Product Feature Comparison

The following tables show the differences between each Zyxel Device model.

FEATURES	NWA1123- ACv2	NWA1123 -AC PRO	NWA1123- AC HD	NWA1302 -AC	NWA5123 -AC	NWA5123 -AC HD	WAC 5302 D-S
Supported Wireless Standards	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac
Supported Frequency Bands	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz
Available Security Modes	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK- MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK- MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK- MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK- MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK- MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK- MIX
Number of SSID Profiles	64	64	64	64	64	64	64
Number of Wireless Radios	2	2	2	2	2	2	2
Monitor Mode & Rogue APs Containment ^A	No	No	No	No	Yes	No	No
Rogue AP Detection	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table 1 Zyxel Device 1000/5000 Series Comparison Table

Table 1	Zyxel Device	1000/5000 Series	Comparison	Table (continued)
---------	--------------	------------------	------------	-------------------

FEATURES	NWA1123- ACv2	NWA1123 -AC PRO	NWA1123- AC HD	NWA1302 -AC	NWA5123 -AC	NWA5123 -AC HD	WAC 5302 D-S
WDS (Wireless Distribution System) - Root AP & Repeater Modes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Tunnel Forwarding Mode	No	No	No	No	No	Yes	No
Layer-2 Isolation	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Supported PoE Standards	IEEE 802.3af IEEE 802.3at	IEEE 802.3at	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at
Power Detection	No	No	Yes	Yes	No	Yes	Yes
External Antennas	No	No	No	No	No	No	No
Internal Antennas	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Antenna Switch	No	Yes (per radio + physical switch)	No	No	No	No	No
Console Port	4-Pin Serial	4-Pin Serial	4-Pin Serial	4-Pin Serial	4-Pin Serial	4-Pin Serial	4-Pin Serial
LED Locator	Yes	Yes	Yes	Yes	Yes	Yes	Yes
LED Suppression	Yes	Yes	Yes	Yes	Yes	Yes	Yes
AC (AP Controller) Discovery	No	No	No	No	Yes	Yes	Yes
NebulaFlex PRO	No	No	No	No	No	Yes	No
NCC Discovery	Yes	Yes	Yes	Yes	No	Yes	No
802.11r Fast Roaming Support	Yes	Yes	Yes	Yes	Yes	Yes	Yes
802.11k/v Assisted Roaming	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Bluetooth Low Energy (BLE)	No	No	No	No	No	No	Yes
USB Port for BLE	No	No	No	No	No	No	Yes
Ethernet Storm Control	No	No	Yes	No	No	Yes	No
Grounding	No	No	Yes	No	No	Yes	No
Power Jack	Yes	No	Yes	No	Yes	Yes	No
Firmware Version	6.10	6.20	6.20	6.20	6.10	6.20	6.10
Maximum number of log messages			512 event	logs			256 event logs

A. For NXC managed devices only. See the NXC User's Guide for details.

The following tables show the differences between each Zyxel Device model.

FEATURES	WAC 500/ WAC 500H	NWA1123-ACv3	WAC 5302D-Sv2
Supported Wireless Standards	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac
Supported Frequency Bands	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz
Available Security Modes	None WEP WPA2-MIX WPA2-PSK WPA2-PSK WPA2-PSK-MIX Enhanced-open WPA3-enterprise WPA3-personal	None WEP WPA2-MIX WPA2-PSK WPA2-PSK-MIX Enhanced-open WPA3-enterprise WPA3-personal	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX
Number of SSID Profiles	64	64	64
Number of Wireless Radios	2	2	2
Monitor Mode & Rogue APs Containment ^A	No	No	No
Rogue AP Detection	Yes	Yes	Yes
WDS (Wireless Distribution System) - Root AP & Repeater Modes	Yes	Yes	Yes
Tunnel Forwarding Mode	Yes	No	Yes
Layer-2 Isolation	Yes	Yes	Yes
Supported PoE Standards	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at
Power Detection	No	No	Yes
External Antennas	No	No	No
Internal Antennas	Yes	Yes	Yes
Antenna Switch	No	No	No
Console Port	4-Pin Serial	4-Pin Serial	4-Pin Serial
LED Locator	Yes	Yes	Yes
LED Suppression	Yes	Yes	Yes
AC (AP Controller) Discovery	Yes	No	Yes
NebulaFlex PRO	Yes	No	Yes

Table 2 Zyxel Device 1000/5000 Series Comparison Table

NWA/WAC/WAX Series User's Guide

,	WAC500/		
FEATURES	WAC500H	NWA1123-ACv3	WAC 5302D-Sv2
NCC Discovery	Yes	Yes	Yes
802.11r Fast Roaming Support	Yes	Yes	Yes
802.11k/v Assisted Roaming	Yes	Yes	Yes
Bluetooth Low Energy (BLE)	No	No	No
USB Port for BLE	No	No	No
Ethernet Storm Control	Yes	Yes	No
Grounding	No	No	No
Power Jack	Yes	Yes	No
Firmware Version	6.20	6.20	6.20
Maximum number of log messages		512	<u>.</u>

Table 2	7vxel Device	1000/5000 Series	Comparison	Table	(continued)
		1000/0000 001103	Companyon	IGDIO	

A. For NXC managed devices only. See the NXC User's Guide for details.

The following tables show the differences between each Zyxel Device model.

FEATURES	WAC6103D-I	WAC 6303D-S	WAC 6502D-E WAC 6553D-E	WAC 6502D-S WAC 6503D-S WAC 6552D-S
Supported Wireless Standards	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac			
Supported Frequency Bands	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz
Available Security Modes	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX
Number of SSID Profiles	64	64	64	64
Number of Wireless Radios	2	2	2	2
Monitor Mode & Rogue APs Containment ^A	Yes	No	Yes	Yes
Rogue AP Detection	Yes	Yes	Yes	Yes

Table 3 WAC 6000 Series Comparison Table

FEA TURES	WAC6103D-I	WAC6303D-S	WAC 6502D-E WAC 6553D-E	WAC 6502D-S WAC 6503D-S WAC 6552D-S
WDS (Wireless Distribution System) - Root AP & Repeater Modes	Yes	Yes	Yes	Yes
Tunnel Forwarding Mode	Yes	Yes	Yes	Yes
Layer-2 Isolation	Yes	Yes	Yes	Yes
Supported PoE Standards	IEEE 802.3at	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at	IEEE 802.3af IEEE 802.3at
Power Detection	No	Yes	Yes	Yes
External Antennas	No	No	Yes	No
Internal Antennas	Yes	Yes	No	Yes
Antenna Switch	Yes (per radio + physical switch)	No	No	No
Console Port	4-Pin Serial	4-Pin Serial	RJ-45 serial	RJ-45 serial
LED Locator	Yes	Yes	Yes	Yes
LED Suppression	Yes	Yes	Yes	Yes
AC (AP Controller) Discovery	Yes	Yes	Yes	Yes
NebulaFlex PRO	Yes	Yes	Yes	Yes
NCC Discovery	Yes	Yes	Yes	Yes
802.11r Fast Roaming Support	Yes	Yes	Yes	Yes
802.11k/v Assisted Roaming	Yes	Yes	Yes	Yes
Bluetooth Low Energy (BLE)	No	Yes	No	No
USB Port for BLE	No	No	No	No
Ethernet Storm Control	No	Yes	No	No
Grounding	No	Yes	Yes	Yes
Power Jack	No	Yes	WAC6502D-E: Yes WAC6553D-E: No	WAC6502D-S: Yes WAC6503D-S: Yes WAC6552D-S: No
Firmware Version	6.20	6.20	6.20	6.20
Maximum number of log messages		512	event logs	

Table 3 WAC 6000 Series Comparison Table (continued)

A. For NXC managed devices only. See the NXC User's Guide for details.

The following tables show the differences between each Zyxel Device model.

Table 4 802.11ax Series Co	WAX510D		NWA110AX
FEATURES	WAX610D	WAX650S	NWA210AX
Supported Wireless Standards	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac IEEE 802.11ax	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11n IEEE 802.11ac IEEE 802.11ax	IEEE 802.11a IEEE 802.11b IEEE 802.11g IEEE 802.11a IEEE 802.11ac IEEE 802.11ax
Supported Frequency Bands	2.4 GHz 5 GHz	2.4 GHz 5 GHz	2.4 GHz 5 GHz
Available Security Modes	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX Enhanced-open WPA3-enterprise WPA3-personal	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX Enhanced-open WPA3-enterprise WPA3-personal	None WEP WPA2 WPA2-MIX WPA2-PSK WPA2-PSK-MIX Enhanced-open WPA3-enterprise WPA3-personal
Number of SSID Profiles	64	64	64
Number of Wireless Radios	2	2	2
Monitor Mode & Rogue APs Containment ^A	No	No	No
Rogue AP Detection	Yes	Yes	Yes
WDS (Wireless Distribution System) - Root AP & Repeater Modes	Yes	Yes	Yes
Tunnel Forwarding Mode	Yes	Yes	No
Layer-2 Isolation	Yes	Yes	Yes
Supported PoE Standards	IEEE 802.3af IEEE 802.3at	IEEE 802.3at IEEE 802.3bt	IEEE 802.3af IEEE 802.3at
Power Detection	Yes	Yes	Yes
External Antennas	No	No	No
Internal Antennas	Yes	Yes	Yes
Antenna Switch	Yes (per AP)	No	No
Console Port	4-Pin Serial	4-Pin Serial	4-Pin Serial
LED Locator	Yes	Yes	Yes
LED Suppression	Yes	Yes	Yes
AC (AP Controller) Discovery	Yes	Yes	No
NebulaFlex PRO	Yes	Yes	No
NCC Discovery	Yes	Yes	Yes
802.11r Fast Roaming Support	Yes	Yes	Yes
802.11k/v Assisted Roaming	Yes	Yes	Yes
Bluetooth Low Energy (BLE)	No	Yes	No
USB Port for BLE	No	No	No
Ethernet Storm Control	Yes	Yes	Yes
Grounding	Yes	Yes	Yes

Table 4 802.11ax Series Comparison Table

Table 4	802.11 ax Series Comparison Table	(continued)
	ouz. That series comparison rable	

FEATURES	WAX510D WAX610D	WAX650S	NWA110AX NWA210AX	
Power Jack	Yes	Yes	Yes	
Firmware Version	6.20	6.20	6.20	
Maximum number of log messages	512 event logs			

A. For NXC managed devices only. See the NXC User's Guide for details.

C HAPTER 2 AP Management

2.1 Management Mode

The Zyxel Device is a unified AP and can be managed by the NCC or an AP controller (AC), or work as a standalone device. We recommend you use NCC to manage multiple APs (see the NCC User's Guide). An AP Controller such as the NXC can only manage multiple APs in the same location.

Note: Not all models can be managed by NCC or an AC. See Section 1.4 on page 18 to check whether your product supports these.

The following table shows the default IP addresses and firmware upload methods for different management modes.

MANAGEMENTMODE	DEFAULT IP ADDRESS	UPLO AD FIRM WARE VIA
MANAGEMENTMODE	DEFAULI IF ADDRESS	OPIDAD FIRM WARE VIA
Nebula Control Center	Dynamic	NCC Portal
AP Controller	Dynamic	AP Controller using CAPWAP
Standalone	Dynamic or Static (192.168.1.2)	Built-in Web Configurator

 Table 5
 Zyxel Device Management Mode Comparison

When the Zyxel Device is in standalone mode and connects to a DHCP server, it uses the IP address assigned by the DHCP server. Otherwise, the Zyxel Device uses the default static management IP address (192.168.1.2). You can use the **NCC Discovery** or **AC Discovery** screen to allow the Zyxel Device to be managed by the NCC or an AC, respectively.

When the Zyxel Device is managed by the NCC or an AC, it acts as a DHCP client and obtains an IP address from the NCC/AC. It can be configured ONLY by the NCC/AC. To change the Zyxel Device back to standalone mode, use the **Re set** button to restore the default configuration. Alternatively, you need to check the NCC/AC for the Zyxel Device's IP address and use FTP to upload the default configuration file at conf/system-default.conf to the Zyxel Device and reboot the device.

2.1.1 Standalone

When working in standalone mode, the Zyxel Device is configured mainly with its built-in Web Configurator. You can only connect to and set up one Zyxel Device at a time in this mode.

See Chapter 5 on page 42 for detailed information about the standalone Web Configurator screens.

2.1.2 Nebula Control Center

In this mode, which is also called cloud mode, you can manage and monitor the Zyxel Device through the Zyxel Nebula cloud-based network management system. This means you can manage devices remotely without the need of connecting to each device directly. It offers many features to better manage and monitor not just the Zyxel Device, but your network as a whole, including supported switches and gateways. Your network can also be managed through your smartphone using the Nebula Mobile app. See Section 23.1 on page 186 for an example NCC managed network topology.

NCC allows different levels of management. You can configure each device on its own or configure a set of devices together as a site. You can also monitor groups of sites called organizations, as shown below.

	Table 6 NCC Management Levels					
	Organization					
Site A				Site	e B	
	Device	A-1	Device A-2	Device B-1	Device B-2	

Table (NCC Man

It graphically presents your device/network statistics and shows an overview of your network topology, as shown in the following figure. It also sends reports, alerts, and notifications for events, such as when a site goes offline.





See the NCC (Nebula Control Center) User's Guide for how to configure Nebula managed devices. See Chapter 24 on page 189 if you want to change the Zyxel Device's VLAN setting or manually set its IP address.

Note: Make sure your network firewall allows TCP ports 443, 4335, and 6667 as well as UDP port 123 so the device can connect to and sync with the NCC.

2.1.3 APController(AC)

If the Zyxel Device supports management using an AC (see Section 9.1.1 on page 69) such as the NXC2500 or NXC5500, and you have this AC in the same subnet, it will be managed by the controller automatically. To set the Zyxel Device to be managed by an AC in a different subnet or change between management modes, use the AC Discovery screen (see Section 9.4 on page 73 and Section 9.1.1 on page 69). You can use the AC to manage multiple Zyxel Devices. See Section 9.1.1 on page 69 for an example AC managed network topology.

Note: If the Zyxel Device is already registered to NCC, the controller will be unable to manage it.

An AC uses Control And Provisioning of Wireless Access Points (CAPWAP, see RFC 5415) to discover and configure multiple managed APs.

2.2 Switching Management Modes

The Zyxel Device is in standalone mode by default, with NCC and/or AC discovery enabled.

Standalone-to-NCC

Register the Zyxel Device at the NCC website and then turn on the Zyxel Device. Make sure that NCC **Discovery** is enabled (see Section 9.4 on page 73). The NCC manages the Zyxel Device automatically when it is discovered.

Standalone-to-AC (NXC)

By default, the Zyxel Device must be in the same subnet as the NXC. See Section 9.1.1 on page 69 for setting it up in a different subnet. Make sure **AC Disc overy** is enabled (see Section 9.4 on page 73). The NXC manages the Zyxel Device automatically when it is discovered.

NXC-to-NCC

Register the Zyxel Device at the NCC website. Make sure that **NCC Discovery** is enabled on your Zyxel Device (see Section 9.4 on page 73). In the NXC Web Configurator, select the Zyxel Device and press the **Nebula** button. The NCC manages the Zyxel Device automatically when it is discovered.

NCC-to-NXC

Unregister the Zyxel Device at the NCC portal. By default, the Zyxel Device must be in the same subnet as the NXC. See Section 9.1.1 on page 69 for setting it up in a different subnet. Make sure **AC Disc overy** is enabled (see Section 9.4 on page 73). The NXC manages the Zyxel Device automatically when it is discovered.

NCC-to-Standalone

Unregister the Zyxel Device from the NCC organization/site. Reset the Zyxel Device to factory defaults (see Section 26.6 on page 205).

AC-to-Standalone

Use the **Reset** button to return the Zyxel Device to its factory default settings (see Section 26.6 on page 205).

2.3 ZyxelOne Network (ZON) Utility

ZON Utility is a program designed to help you deploy and manage a network more efficiently. It detects devices automatically and allows you to do basic settings on devices in the network without having to be near it.

The ZON Utility issues requests via Zyxel Discovery Protocol (ZDP) and in response to the query, the device responds back with basic information including IP address, firmware version, location, system and model name in the same broadcast domain. The information is then displayed in the ZON Utility screen and you can perform tasks like basic configuration of the devices and batch firmware upgrade in it. You can download the ZON Utility at www.zyxel.com and install it on your computer (Windows operating system).

2.3.1 Requirements

Before installing the ZON Utility on your PC, please make sure it meets the requirements listed below.

Operating System

At the time of writing, the ZON Utility is compatible with:

- Windows 7 (both 32-bit / 64-bit versions)
- Windows 8 (both 32-bit / 64-bit versions)
- Windows 8.1 (both 32-bit / 64-bit versions)
- Window 10 (both 32-bit / 64-bit versions)

Note: To check for your Windows operating system version, right-click on **My Computer> Properties**. You should see this information in the **General** tab.

Note: It is suggested that you install Npcap, the packet capture library for Windows operating systems, and remove WinPcap or any other installed packet capture tools before you install the ZON utility.

Hard ware

Here are the minimum hardware requirements to use the ZON Utility on your PC.

- Core i3 processor
- 2 GB RAM
- 100 MB free hard disk
- WXGA (Wide XGA 1280x800)

2.3.2 Run the ZON Utility

- **1** Double-click the ZON Utility to run it.
- 2 The first time you run the ZON Utility, you will see if your device and firmware version support the ZON Utility. Click the **OK** button to close this screen.

KEL		
Her to the top	le no en ute sour device l'im	wore a supporting the 2014 utility.
ivent	Second And Reven	- Bernstein Berne
		From V9.38
	МАДККО и мака	WARESTED & WELLS WARESTED WARESTED & WELLS WARESTED WARESTED
	7936117991	From VS45
	w(30)-0.5974	Fine(42) • NATESALE 400-1
	K#6200-Kases	From V2.23 • Reversion of care a • Reversion of care a • Reversion of care a • Reversion of care a

Figure 6 Supported Devices and Versions

If you want to check the supported models and firmware versions later, you can click the **Show information about ZON** icon in the upper right hand corner of the screen. Then select the **Supported model and firmware version** link. If your device is not listed here, see the device release notes for ZON utility support. The release notes are in the firmware zip file on the Zyxel web site.



ZYX	EL								521	8 0
Θ	0	0 (8	0	1.53	158	12	:e)	1997	(B)	(B
P.S.				and a	-					-
	, sinter	V428666	11. APR	ant.	11,100.27	(196,00				
	i innista	Vagendari	01. 8416	entre :	11000	No.21	0.15	6		
-	3000000	e veliter	611.1410	4046.)	104630	1046,04	-810	а.,		

3 Select a network adapter to which your supported devices are connected.

Figure 8	Netw	vork Ad	apter
----------	------	---------	-------

Se leite faithe	Bictory.
ZYXEL	
Professionel is 2004 states. Principal of a supervision of produce states and the distance of the state in the firm. Places shows the interface for distance ing de connected nervisit, and state, Gef suffer to a	
	3

4 Click the Go button for the ZON Utility to discover all supported devices in your network.

Figure 9	Discovery
Circovény	
ZYX	EL:
	oliceway pangross
	1
	Curve

5 The ZON Utility screen shows the devices discovered.

el Conchester (Ch	N Utility Screen			S. 183 44
	ZYXEL			B 8 0
	10 20 20	46 50 60 7	a e s e s	IQ- 110 125 185
			tensi Tapatropan	
		Wellighted a service man from	6.1.1 (DA)#	
A	🚓 musik	wannang, sasarat, ngu	all that as	0
R -		VERSION OF THE PERSON NEWSFILM	ALL 194594 11	00.
2	and transmission			MARAN I

6 Select a device and then use the icons to perform actions. Some functions may not be available for your devices.

Note: You must know the selected device admin password before taking actions on the device using the ZON utility icons.

Fig ure	11	Password	Prompt
1,5 410	**	1 03311010	i i Oli i Pi

ZYXEL	
é	Neces enter the potent shotof poptword to proceed.
131	Detroe :
	Pransord

The following table describes the icons numbered from left to right in the ZON Utility screen.

ICON	DESC RIPTIO N
1 IP Configuration	Change the selected device's IP address.
2 Renew IP Address	Update a DHCP-assigned dynamic IP address.
3 Reboot Device	Use this icon to restart the selected device(s). This may be useful when troubleshooting or upgrading new firmware.
4 Reset Configuration to Default	Use this icon to reload the factory-default configuration file. This means that you will lose all previous configurations.
5 Locator LED	Use this icon to locate the selected device by causing its Locator LED to blink.
6 Web GUI	Use this to access the selected device Web Configurator from your browser. You will need a username and password to log in.
7 Firmware Upgrade	Use this icon to upgrade new firmware to selected device(s) of the same model. Make sure you have downloaded the firmware from the Zyxel website to your computer and unzipped it in advance.
8 Change Password	Use this icon to change the admin password of the selected device. You must know the current admin password before changing to a new one.
9 Configure NCC Discovery	You must have Internet access to use this feature. Use this icon to enable or disable the Nebula Control Center (NCC) discovery feature on the selected device. If it is enabled, the selected device will try to connect to the NCC. Once the selected device device is connected to and has registered in the NCC, it will go into the Nebula cloud management mode.
10 ZAC	Use this icon to run the Zyxel AP Configurator of the selected AP.
11 Clear and Rescan	Use this icon to clear the list and discover all devices on the connected network again.
12 Save Configuration	Use this icon to save configuration changes to permanent memory on a selected device.
13 Settings	Use this icon to select a network adapter for the computer on which the ZON utility is installed, and the utility language.

Table 7 ZON Utility Icons

The following table describes the fields in the ZON Utility main screen.

Table 8	70N	Utility	Fields
	2011	Ommy	110103

LABEL	DESC RIPTIO N
Туре	This field displays an icon of the kind of device discovered.
Model	This field displays the model name of the discovered device.
Firmware Version	This field displays the firmware version of the discovered device.

IABEL	DESC RIPTIO N
MAC Address	This field displays the MAC address of the discovered device.
IP Address	This field displays the IP address of an internal interface on the discovered device that first received an ZDP discovery request from the ZON utility.
System Name	This field displays the system name of the discovered device.
Location	This field displays where the discovered device is.
Status	This field displays whether changes to the discovered device have been done successfully. As the Zyxel Device does not support IP Configuration , Renew IP address and Flash Locator LED , this field displays "Update failed", "Not support Renew IP address" and "Not support Flash Locator LED" respectively.
NCC Discovery	This field displays if the discovered device supports the Nebula Control Center (NCC) discovery feature. If it's enabled, the selected device will try to connect to the NCC. Once the selected device is connected to and has registered in the NCC, it'll go into the Nebula cloud management mode.
Serial Number	Enter the admin password of the discovered device to display its serial number.
Hardware Version	This field displays the hardware version of the discovered device.

Table 8 ZON Utility Fields

2.4 Ways to Access the Zyxel Device

You can use the following ways to configure the Zyxel Device.

Web Configurator

The Web Configurator allows easy Zyxel Device setup and management using an Internet browser. If your Zyxel Device is managed by the NCC or an AC, use this only for troubleshooting if you cannot connect to the Internet. This User's Guide provides information about the Web Configurator.

NCC

This is the primary means by which you manage the Zyxel Device in cloud (NCC) mode. With the NCC, you can remotely manage and monitor the Zyxel Device through a cloud-based network management system. See the NCC User's Guide for more information.

ZON Utility

Zyxel One Network (ZON) Utility is a utility tool that assists you to set up and maintain network devices in a simple and efficient way. You can download the ZON Utility at www.zyxel.com and install it on your computer (Windows operating system). For more information on ZON Utility see Section 2.3 on page 29.

Command-Line Interface (CLI)

The CLI allows you to use text-based commands to configure the Zyxel Device. You can access it using remote management (for example, SSH or Telnet) or via the console port. See the Command Reference Guide for more information.

File Transfer Protocol (FIP)

This protocol can be used for firmware upgrades and configuration backup and restore.

Simple Network Management Protocol (SNMP)

The Zyxel Device can be monitored by an SNMP manager. See the SNMP chapter in this User's Guide.

2.5 Good Habits for Managing the Zyxel Device

Do the following things regularly to make the Zyxel Device more secure and to manage it more effectively.

- Change the password often. Use a password that's not easy to guess and that consists of different types of characters, such as numbers and letters.
- Write down the password and put it in a safe place.
- Back up the configuration (and make sure you know how to restore it). Restoring an earlier working configuration may be useful if the Zyxel Device becomes unstable or even crashes. If you forget your password, you will have to reset the Zyxel Device to its factory default settings. If you backed up an earlier configuration file, you will not have to totally re-configure the Zyxel Device; you can simply restore your last configuration.

C HA PTER 3 Hardware

See the Quick Start Guide for hardware installation and connections.

3.1 Grounding (WAC6552D-S and WAC6553D-E)

Earth grounding helps protect against lightning and interference.

Note: The power installation must be performed by qualified service personnel and should conform to the National Electrical Code.

The Zyxel Device must be connected to earth ground to adequately ground the Zyxel Device and protect the operator from electrical hazards.

Qualified service personnel must confirm that the protective earthing terminal of the building is a valid terminal.

Before connecting the ground, ensure that a qualified service personnel has attached an appropriate ground lug to the ground cable.

- 1 Remove one of the ground screws from the Zyxel Device's rear panel.
- 2 Secure a green/yellow ground cable (18 AWG or smaller) to the Zyxel Device's rear panel using the ground screw.
- 3 Attach the other end of the cable to the ground, either to the same ground electrode as the pole you installed the Zyxel Device on or to the main grounding electrode of the building.
 - Note: Follow your country's regulations and safety instructions to electrically ground the Zyxel Device properly. If you are uncertain that suitable grounding is available, contact the appropriate electrical inspection authority or an electrician.

Warning! Connect the ground cable before you connect any other cables or wiring.

The figure below illustrates how the ground cable (A) is attached to the Zyxel Device and goes to the earth ground (B).





3.2 Zyxel Device Models With Single LEDs

The LEDs of some Zyxel Device models can be controlled by using the suppression feature such that the LEDs stay lit (ON) or OFF after the Zyxel Device is ready. Some Zyxel Device models also has Locator LED which allows you to see the actual location of the Zyxel Device among several devices in the network. See Section 1.4 on page 18 to check which models support these features. Refer to Chapter 20 on page 180 for the LED **Suppression** and **Locator** menus in standalone mode.

The following models have single LEDs: NWA1123-ACv2, NWA1123ACv3, NWA1123-AC HD, NWA5123AC, NWA5123-AC HD, WAC6303D-S, NWA110AX, NWA210AX, WAX510D, WAX610D and WAX650S.

3.3 Zyxel Device Single LED

The LED of the Zyxel Device can be controlled by using the suppression feature such that the LEDs stay lit (ON) or OFF after the Zyxel Device is ready. Refer to Chapter 20 on page 180 for the LED **Suppression** and **Locator** menus in standalone mode.
3.3.1 NWA1123-ACv2

Figure 13 NWA1123-ACv2 LED



The following are the LED descriptions for your NWA1123-ACv2.

COLOR		STATUS	DESC RIPTIO N
-	Amber	Blinks amber for 1 second	The LED blinks amber and green alternatively when the Zyxel
-	Green	and green for 1 second alternatively.	Device is booting up or is connecting to the NCC.
	Amber	Blinks amber and green	The Zyxel Device is discovering the NCC.
ţ	Green	alternatively 3 times and then turns solid green for 3 seconds.	
	Green	On	The Zyxel Device is ready for use and its wireless interface is activated.
***		Slow Blinking (On for 1s, Off for 1s)	The wireless module of the Zyxel Device is disabled or failed, the Zyxel Device is using default wireless settings, or the Zyxel Device is connected to the NCC but is unregistered with the NCC.
N de		Fast Blinking (On for 50 ms, Off for 50 ms)	The Locator LED is on.
	Amber	On	The Zyxel Device is powered up.

		STZ EED (SSTIILISSA)	
COLOR		STATUS	DESC RIPIIO N
	Red	Steady On	The Zyxel Device failed to boot up or is experiencing system failure.
		Slow Blinking (Blink for 3 times, Off for 3s)	The Uplink interface is down.
		Fast Blinking (On for 50 ms, Off for 50 ms)	The Zyxel Device is undergoing firmware upgrade.

Table 9 NWA1123-ACv2 LED (continued)

3.3.2 WAC6303D-S, NWA1123-AC HD and NWA5123-AC HD

Figure 14 WAC6303D-S LED



The following are the LED descriptions for your WAC6303D-S, NWA1123-AC HD or NWA5123-AC HD.

COLOR		STATUS	DESC RIPIIO N
	Amber	Blinks amber for 1 second and green for 1 second	The Zyxel Device is booting up or is connecting with NCC.
	Green	alternatively.	
	Amber	Blinks amber and green	The Zyxel Device is discovering the NCC or an AC.
ţ	Green	alternatively 3 times and then turns solid green for 3 seconds.	
	Amber	Blinks amber and green	The Zyxel Device is managed by an AC but the uplink is
ţ	Green	alternatively 2 times and then turns solid green for 3 seconds.	disconnected.

Table 10 WAC6303D-S, NWA1123-AC HD and NWA5123-AC HD LED

COLOR		STATUS	DESC RIPTIO N
	Green	Slow Blinking (On for 1 second, Off for 1 second)	The wireless module of the Zyxel Device is disabled or fails, the Zyxel Device is using default wireless settings, or the Zyxel Device is configured to be managed by NCC but is not yet registered with the NCC.
	Green	Steady On	The Zyxel Device is ready for use, the Zyxel Device's wireless interface is activated, and/or wireless clients are connected to the Zyxel Device in full power mode (see Table 14 on page 44).
	Amber	Steady On	The Zyxel Device is ready for use, the Zyxel Device's wireless interface is activated, and/or wireless clients are connected to the Zyxel Device in limited power mode (see Table 14 on page 44).
	Bright Blue	Steady On	The Zyxel Device's wireless interface is activated, but there are no wireless clients connected when it is in full power mode (see Table 14 on page 44).
-	White	Steady On	The Zyxel Device's wireless interface is activated, but there are no wireless clients connected when it is in limited power mode (see Table 14 on page 44). Note: The color of the white LED may have slight differences (for example, very light purple) on different models.
		Slow Blinking (On for 100ms per second)	Locator LED is on. It switches off automatically after the configured amount of time (1-60 min). Default duration is 10 minutes. Note: The color of the white LED may have slight differences (for example, very light purple) on different models.
	Blue	Slow Blinking (Blink for 1 time, Off for 1 second)	The Zyxel Device is performing a Channel Availability Check (CAC) with Dynamic Frequency Selection (DFS) to monitor a channel for radar signals.
	Red	On	The Zyxel Device failed to boot up or is experiencing system failure.
-		Fast Blinking (On for 50 milliseconds, Off for 50 milliseconds)	The Zyxel Device is undergoing firmware upgrade.
2.00		Slow Blinking (Blink for 3 times, Off for 3 seconds)	The Uplink port of the Zyxel Device in standalone mode is disconnected.

Table 10 WAC6303D-S, NWA1123-AC HD and NWA5123-AC HD LED (continued)

3.3.3 NWA5123-AC

Figure 15 NWA5123-AC LED



The following are the LED descriptions for your NWA5123-AC.

TODIE IT INWASTZS-AC LED	Table 11	NWA5123-AC LED
--------------------------	----------	----------------

COLOR		STATUS	DESC RIPTIO N
	Amber	Blinks amber for 1 second	The Zyxel Device is booting up.
Ę	Green	and green for 1 second alternatively.	
	Amber	Blinks amber and green	The Zyxel Device is discovering an AC.
t	Green	alternatively 3 times and then turns solid green for 3 seconds.	
	Amber	Blinks amber and green	The Zyxel Device is managed by an AC and the uplink
t	Green	alternatively 2 times and then turns solid green for 3 seconds.	interface is down.
	Green	On	The Zyxel Device is ready for use and its wireless interface is activated.
		Slow Blinking (On for 1s, Off for 1s)	The wireless module of the Zyxel Device is disabled or failed, or the Zyxel Device is using default wireless settings.
¥67		Fast Blinking (On for 50ms, Off for 50ms)	The Locator LED is on.
	Amber	On	The Zyxel Device is powered up.

COLOR		STATUS	DESC RIPTIO N
-	Red	Steady On	The Zyxel Device failed to boot up or is experiencing system failure.
		Slow Blinking (Blink for 3 times, Off for 3s)	The Uplink interface is down.
		Fast Blinking (On for 50 ms, Off for 50 ms)	The Zyxel Device is undergoing firmware upgrade.

Table 11 NWA5123-AC LED (continued)

3.3.4 WAC500, WAC500H, NWA1123ACv3, NWA110AX, NWA210AX, WAX510D, WAX610D and WAX650S

Figure 16 WAC500, NWA1123Acv3, NWA110AX, NWA210AX, WAX510D, WAX610D and WAX650S LED



Figure 17 WAC500H LED



The following are the LED descriptions for your WAC500, WAC500H, NWA1123ACv3, NWA110AX, NWA210AX, WAX510D, WAX610D and WAX650S.

COLOR		STATUS	DESC RIPIIO N
1	Amber	Blinks amber for 1 second and green for 1 second	The Zyxel Device is booting up or is connecting with NCC.
	Green	alternatively.	
_	Amber	Blinks amber and green	The Zyxel Device is discovering the NCC or an AC.
ţ,	Green	alternatively 3 times and then turns solid green for 3 seconds.	
	Amber	Blinks amber and green	The Zyxel Device is managed by an AC but the uplink is
ţ,	Green	alternatively 2 times and then turns solid green for 3 seconds.	disconnected.
	Green	Slow Blinking (On for 1 second, Off for 1 second)	The wireless module of the Zyxel Device is disabled or fails, the Zyxel Device is using default wireless settings, or the Zyxel Device is configured to be managed by NCC but is not yet registered with the NCC.
			Note: WiFi networks on the WAX650S are turned off automatically when it is connected to a device that supplies power using IEEE 802.3af PoE.

Table 12 WAC500, WAC500H, NWA1123ACv3, NWA110AX, WAX510D and WAX650S LED

COLOR		STATUS	DESC RIPTIO N
	Green	Steady On	The Zyxel Device is ready for use, the Zyxel Device's wireless interface is activated, and/or wireless clients are connected to the Zyxel Device in full power mode (see Table 14 on page 44).
	Amber	Steady On	The Zyxel Device is ready for use, the Zyxel Device's wireless interface is activated, and/or wireless clients are connected to the Zyxel Device in limited power mode (see Table 14 on page 44).
	Bright Blue	Steady On	The Zyxel Device's wireless interface is activated, but there are no wireless clients connected when it is in full power mode (see Table 14 on page 44).
-	White	Steady On	The Zyxel Device's wireless interface is activated, but there are no wireless clients connected when it is in limited power mode (see Table 14 on page 44).
			Note: The color of the white LED may have slight differences (for example, very light purple) on different models.
		Slow Blinking (On for 100ms per second)	Locator LED is on. It switches off automatically after the configured amount of time (1-60 min). Default duration is 10 minutes.
			Note: The color of the white LED may have slight differences (for example, very light purple) on different models.
<u> </u>	Blue	Slow Blinking (Blink for 1 time, Off for 1 second)	The Zyxel Device is performing a Channel Availability Check (CAC) with Dynamic Frequency Selection (DFS) to monitor a channel for radar signals.
	Red	On	The Zyxel Device failed to boot up or is experiencing system failure.
		Fast Blinking (On for 50 milliseconds, Off for 50 milliseconds)	The Zyxel Device is undergoing firmware upgrade.
		Slow Blinking (Blink for 3 times, Off for 3 seconds)	The Uplink port of the Zyxel Device in standalone mode is disconnected.

Table 12	WAC500, WAC500H	NWA1123ACv3.	NWA110AX.	WAX510D	and WAX650S LED (continued)
		,	1 1 1 1 1 0 1 0 1 0 1	117 010101	

3.4 Zyxel Device Models With Multiple LEDs

The LEDs of some Zyxel Device models can be controlled by using the suppression feature such that the LEDs stay lit (ON) or OFF after the Zyxel Device is ready. Some Zyxel Device models also has Locator LED which allows you to see the actual location of the Zyxel Device among several devices in the network. See Section 1.4 on page 18 to check which models support these features. Refer to Chapter 20 on page 180 for the LED **Suppression** and **Locator** menus in standalone mode.

The following models have multiple LEDs: NWA1123-AC PRO, NWA1302-AC, WAC6103D-I, WAC5302D-S, WAC6502D-E, WAC6502D-S, WAC6503D-S.

3.4.1 NWA1123-AC PRO

Figure 18 NWA1123-AC PRO LEDs



Table 13 NWA1123-AC PRO LEDs

LED	COLOR	STATUS	DESC RIPIIO N
PWR/SYS	Amber	Blinks amber for 1	The LED blinks amber and green alternatively when the Zyxel Device is booting up.
¢.	Green	second and green for 1 second alternatively.	
	Green	On	The Zyxel Device is ready for use.
		Slow Blinking (On for 1 sec, Off for 1 sec)	The wireless module of the Zyxel Device is disabled or failed.
	Red	On	There is a system error and the Zyxel Device cannot boot up, or the Zyxel Device suffered a system failure.
		Fast Blinking (On for 50 ms, Off for 50 ms)	The Zyxel Device is undergoing firmware upgrade.
		Slow Blinking (Blink for 3 times, Off for 3 sec)	The Uplink interface is down.

IED	COLOR	STATUS	DESC RIPIIO N
Management	Green	On	The Zyxel Device is managed by the NCC.
<u>s</u>		Slow Blinking (On for 1 sec, Off for 1 sec)	The Zyxel Device is using default wireless settings, or the Zyxel Device is connected to the NCC but is unregistered with the NCC.
	Amber	Blinks amber for 1	The Zyxel Device is searching for (discovering) the NCC.
	Green	second and green for 1 second alternatively	
	Amber	Blinks amber and	The NCC is connecting to the registered Zyxel Device.
	Green	green alternatively 3 times and then turns solid green for 3 seconds.	
		Off	The Zyxel Device is in standalone mode.
WLAN	Green	On	The 2.4 GHz radio is set to "Ceiling" and is active
隐念的	Amber	On	The 2.4 GHz radio is set to "Wall" and is active
1000 IN 101		Off	The 2.4 GHz WLAN is not active.
WLAN	Green	On	The 5 GHz radio is set to "Ceiling" and is active
56	Amber	On	The 5 GHz radio is set to "Wall" and is active
201 - 201		Off	The 5 GHz WLAN is not active.
UPLINK	Amber	On	The port is operating as a 100 Mbps connection.
ς, Έ		Blinking	The Zyxel Device is sending/receiving data through the port at 100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The Zyxel Device is sending/receiving data through the port at 1 Gbps.
		Off	The port is not connected.
LAN	Amber	On	The port is operating as a 100 Mbps connection.
10		Blinking	The LAN port is sending/receiving data at 100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The LAN port is sending/receiving data at 1 Gbps.
		Off	The LAN port is not connected.
Locator	White	Blinking	The Locator is activated and will blink to show the actual location of the Zyxel Device between several devices in the network.
		Off	The Locator function is off.

Table 13 NWA1123-AC PRO LEDs (continued

3.4.2 NWA1302-AC/WAC5302D-Sv2

By default, the LEDs automatically turn on when the NWA1302-AC/ WAC5301D-Sv2 is ready. If the LEDs are turned off by the NCC, you can press the **LED ON** button for one second to turn on the LEDs again. The LEDs will blink and turn off after two minutes.



	P.P.P.P	NO.
	0 C C	

LED	COLOR	STATUS	DESC RIPTIO N
PWR/SYS	Amber	Blinks amber for 1	The LED blinks amber and green alternatively when the Zyxel Device is booting up.
రు	Green	second and green for 1 second alternatively.	
	Green	On	The Zyxel Device is ready for use.
		Slow Blinking (On for 1 sec, Off for 1 sec)	The wireless module of the Zyxel Device is disabled or failed.
		Fast Blinking (On 50 ms, Off 50 ms)	The Locator LED is on.
	Red	On	There is a system error and the Zyxel Device cannot boot up, or the Zyxel Device suffered a system failure.
		Fast Blinking (On for 50 ms, Off for 50 ms)	The Zyxel Device is doing firmware upgrade.
		Slow Blinking (Blink for 3 times, Off for 3 sec)	The Uplink interface is down.
Management	Green	On	The Zyxel Device is managed by the NCC.
		Slow Blinking (On for 1 sec, Off for 1 sec)	The Zyxel Device is using default wireless settings, or the Zyxel Device is connected to the NCC but is unregistered with the NCC.
	Amber	Blinks amber for 1	The Zyxel Device is searching for (discovering) the NCC.
	Green	second and green for 1 second alternatively	
	Amber	Blinks amber and green	The NCC is connecting to the registered Zyxel Device.
	Green	alternatively 3 times and then turns solid green for 3 seconds.	
	Off		The Zyxel Device is in standalone mode.

Table 14 NWA1302-AC/ WAC5301D-Sv2 LEDs

IED	COLOR	STATUS	DESC RIPIIO N
UPLINK	Amber	On	The port is operating as a 10/100 Mbps connection.
i ר		Blinking	The Zyxel Device is sending/receiving data through the port at 10/100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The Zyxel Device is sending/receiving data through the port at 1 Gbps.
		Off	The port is not connected.
WLAN ((‡))	Green	On	The 2.4 GHz WLAN is active.
		Off	The 2.4 GHz WLAN is not active.
WLAN ((‡))	Green	On	The 5 GHz WLAN is active.
		Off	The 5 GHz WLAN is not active.
LAN	Amber	On	The port is operating as a 10/100 Mbps connection.
		Blinking	The LAN port is sending/receiving data through the port at 10/100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The LAN port is sending/receiving data through the port at 1 Gbps.
	Off		The LAN port is not connected.

Table 14 NWA1302-AC/ WAC5301D-Sv2 LEDs (continued)

3.4.3 WAC6502D-E, WAC6502D-S, and WAC6503D-S

Figure 20 WAC6502D-E, WAC6502D-S, or WAC6503D-S LEDs



IED	COLOR	STATUS	DESC RIPTIO N
PWR/SYS	Amber	Blinks amber for 1	The Zyxel Device is booting up or is connecting to the NCC or to an AC.
ట	Green	second and green for 1 second alternatively.	
	Green	On	The Zyxel Device is ready for use.
		Slow Blinking (On for 1s, Off for 1s)	The wireless module of the Zyxel Device is disabled or failed.
	Red	On	There is system error and the Zyxel Device cannot boot up, or the Zyxel Device suffered a system failure.
		Fast Blinking (On for 50ms, Off for 50 ms)	The Zyxel Device is doing firmware upgrade.
		Slow Blinking (Blink for 3 times, Off for 3s)	The Uplink interface is down.
		Slow Blinking (Blink for 2 times, Off for 3s)	The Zyxel Device is managed by an AC and the uplink is disconnected.

IED	COLOR	STATUS	DESC RIPIIO N
Management	Green	On	The Zyxel Device is managed by a the NCC or an AC.
R.		Slow Blinking (Blink for 3 times, Off for 3s)	The Zyxel Device is searching (discovery) for an AC.
		Slow Blinking (On for 1s, Off for 1s)	The Zyxel Device is using default wireless settings, or the Zyxel Device is connected to the NCC but is unregistered with the NCC.
		Off	The Zyxel Device is in standalone mode.
	Amber	Blinks amber for 1	The Zyxel Device is searching (discovery) for the NCC.
	Green	second and green for 1 second alternatively	
	Amber	Blinks amber and green	The NCC is connecting to the registered Zyxel Device.
	Green	alternatively 3 times and then turns solid green for 3 seconds.	
WLAN	Green	On	The 2.4 GHz WLAN is active.
((和)) 225		Off	The 2.4 GHz WLAN is not active.
WLAN	Green	On	The 5 GHz WLAN is active.
((\$)) SC		Off	The 5 GHz WLAN is not active.
	Amber	On	The port is operating as a 100 Mbps connection.
		Blinking	The Zyxel Device is sending/receiving data through the port at 100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The Zyxel Device is sending/receiving data through the port at 1 Gbps.
		Off	The port is not connected.
LAN	Amber	On	The port is operating as a 100 Mbps connection.
12		Blinking	The LAN port is sending/receiving data through the port at 100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The LAN port is sending/receiving data through the port at 1 Gbps.
		Off	The LAN port is not connected.
Locator	White	Blinking	The Locator is activated and will blink to show the actua location of the Zyxel Device between several devices in the network.
-		Off	The Locator function is off.

3.4.4 WAC6103D-I

Figure 21 WAC6103D-I LEDs



Table 16	WAC6103D-I LEDs

LED	COLOR	STATUS	DESC RIPTIO N
PWR/SYS	Amber	Blinks amber for 1	The Zyxel Device is booting up.
t)	Green	second and green for 1 second alternatively.	
	Green	On	The Zyxel Device is ready for use.
		Slow Blinking (On for 1s, Off for 1s)	The wireless module of the Zyxel Device is disabled or failed.
	Red	On	There is system error and the Zyxel Device cannot boot up, or the Zyxel Device suffered a system failure.
		Fast Blinking (On for 50 ms, Off for 50 ms)	The Zyxel Device is doing firmware upgrade.
		Slow Blinking (Blink for 3 times, Off for 3s)	The Uplink port is disconnected.

Table 16 WAC	COLOR	STATUS	DESC RIPIIO N
Management	Green	On	The Zyxel Device is managed by an AC or the NCC.
	Gleen	-	
<u>er</u>		Slow Blinking (Blink for 3 times, Off for 3s)	The Zyxel Device is searching (discovery) for an AC.
		Slow Blinking (On for 1s, Off for 1s)	The Zyxel Device is using default wireless settings, or the Zyxel Device is connected to the NCC but is unregistered with the NCC.
		Off	The Zyxel Device is in standalone mode.
	Amber	Blinks amber for 1	The Zyxel Device is searching (discovery) for the NCC.
	Green	second and green for 1 second alternatively	
	Amber	Blinks amber and green	The NCC is connecting to the registered Zyxel Device.
	Green	alternatively 3 times and then turns solid green for 3 seconds.	
WLAN	Green	On	The 2.4 GHz radio is set to "Ceiling" and is active.
2,68	Amber	On	The 2.4 GHz radio is set to "Wall" and is active.
		Off	The 2.4 GHz WLAN is not active.
WLAN	Green	On	The 5 GHz radio is set to "Ceiling" and is active.
医髓	Amber	On	The 5 GHz radio is set to "Wall" and is active.
100 NO		Off	The 5 GHz WLAN is not active.
UPLINK	Amber	On	The port is operating as a 100 Mbps connection.
1 7		Blinking	The Zyxel Device is sending/receiving data through the port at 100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The Zyxel Device is sending/receiving data through the port at 1 Gbps.
		Off	The port is not connected.
LAN	Amber	On	The port is operating as a 100 Mbps connection.
12		Blinking	The LAN port is sending/receiving data through the port at 100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The LAN port is sending/receiving data through the port at 1 Gbps.
		Off	The LAN port is not connected.
Locator	White	Blinking	The Locator is activated and will blink to show the actual location of the Zyxel Device between several devices in the network.
-		Off	The Locator function is off.

Table 16 WAC6103D-I LEDs (continued)

3.4.5 WAC5302D-S

The LEDs automatically turn off when the WAC5302D-S is ready. You can press the **LED ON** button for one second to turn on the LEDs again. The LEDs will blink and turn off after two minutes.

Figure 22 WAC5302D-S LEDs

		Contraction of the second seco	
Course			

Table 17	WAC5302D-S LEDs

LED	COLOR	STATUS	DESC RIPIIO N	
PWR/SYS	Amber	Blinks amber for 1	The LED blinks amber and green alternatively when the WAC is booting up.	
¢	Green	second and green for 1 second alternatively.		
	Green	On	The Zyxel Device is ready for use.	
		Slow Blinking (On for 1s, Off for 1s)	The wireless module of the Zyxel Device is disabled or failed.	
		Fast Blinking (On 50 ms, Off 50 ms)	The Locator LED is on.	
Red		On	There is system error and the Zyxel Device cannot boot up, or the Zyxel Device suffered a system failure.	
		Fast Blinking (On for 50ms, Off for 50 ms)	The Zyxel Device is doing firmware upgrade.	
		Slow Blinking (Blink for 3 times, Off for 3s)	The Uplink interface is down.	
		Slow Blinking (Blink for 2 times, Off for 3s)	The Zyxel Device is managed by an AC and the uplink is disconnected.	
Management	Green	On	The Zyxel Device is managed by a controller.	
ί.		Slow Blinking (Blink for 3 times, Off for 3s)	The Zyxel Device is searching (discovery) for a controller.	
		Slow Blinking (On for 1s, Off for 1s)	The Zyxel Device is using default wireless settings, or the Zyxel Device is connected to the NCC but is unregistered with the NCC.	
		Off	The Zyxel Device is in standalone mode.	

IED	COLOR	STATUS	DESC RIPTIO N
UPLINK	Amber	On	The port is operating as a 10/100 Mbps connection.
ί.		Blinking	The Zyxel Device is sending/receiving data through the port at 10/100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The Zyxel Device is sending/receiving data through the port at 1 Gbps.
		Off	The port is not connected.
WLAN Green	On	The 2.4 GHz WLAN is active.	
		Off	The 2.4 GHz WLAN is not active.
	Green	On	The 5 GHz WLAN is active.
		Off	The 5 GHz WLAN is not active.
LAN	Amber	On	The port is operating as a 10/100 Mbps connection.
_		Blinking	The LAN port is sending/receiving data through the port at 10/100 Mbps.
	Green	On	The port is operating as a Gigabit connection (1000 Mbps).
		Blinking	The LAN port is sending/receiving data through the port at 1 Gbps.
		Off	The LAN port is not connected.

Table 17 WAC5302D-S LEDs (continued)

C HAPTER 4 Web Configurator

4.1 Overview

The Web Configurator is an HTML-based management interface that allows easy system setup and management via internet browser. Use a browser that supports HTML5, such Internet Explorer 11, Mozilla Firefox, or Google Chrome. The recommended screen resolution is 1024 by 768 pixels.

In order to use the Web Configurator you need to allow:

- Web browser pop-up windows from your device.
- JavaScript (enabled by default).
- Java permissions (enabled by default).

4.2 Accessing the Web Configurator

- 1 Make sure your Zyxel Device hardware is properly connected. See the Quick Start Guide.
- 2 If the Zyxel Device and your computer are not connected to a DHCP server, make sure your computer's IP address is in the range between "192.168.1.3" and "192.168.1.254".
- 3 Browse to the Zyxel Device's DHCP-assigned IP address or http://192.168.1.2. The Log in screen appears. If you are in NCC mode, check the NCC's AP> Monitor> Access Point screen for the Zyxel Device's LAN IP address.

ZYXEL		G English 🔹
Enter I	WAX510D Jser Name/Password and Click Login	
	8 9	
	Login	

If a Zyxel Device is in standalone mode and supports NCC, the login page displays as shown in the following figure.

ZYXEL		Q Togers
NWA	1123-AC-HD	
to the Over Plan	wheneved proright (Login	-
0		
	and the second	Contractor of
	Login	Made 2

Click **Nebula Mode** to show the following screen. Here, you can watch a tutorial for using the Zyxel Nebula Control Center (NCC) or access the link to the NCC, as shown in the following figure. Otherwise, continue with the next step. The NCC is a cloud-based network management system that allows you to remotely manage and monitor the Zyxel Device (see Section 2.1.2 on page 26).



If you want to return to the login page, click Standalone Mode and follow the next steps.

- 4 Enter the user name (default: "admin") and password (default: "1234"). If the Zyxel Device is being managed or has been managed by the NCC, check the NCC's **Site-Wide > Configure > General setting** screen for the Zyxel Device's current password.
- 5 Select the language you prefer for the Web Configurator. Click Login.
- 6 The wizard screen opens when the Zyxel Device is accessed for the first time or when you reset the Zyxel Device to its default factory settings.
- 7 If you logged in using the default user name and password, the **Update Admin Info** screen appears. Otherwise, the dashboard appears.

ZYXEL		
	WA	с
	Update Ac	desin into
As a security preces	rion. It is highly re camin po	commended that you change the sword.
New Passward		
Confirm Password		
	(max, 85 cipt); (11000;101,01	anomiate, perilate dino 300.001
	Apply	Ignore :

The **Update Admin Info** screen appears every time you log in using the default user name and default password. If you change the password for the default user account, this screen does not appear anymore.

4.3 Navigating the Web Configurator

The following summarizes how to navigate the Web Configurator from the **Dashboard** screen. The following figures show the **Dashboard** screen for standalone mode and for cloud (NCC) mode. The screen is different for standalone mode and cloud (NCC) mode and may vary slightly for different models.



albinuari (Distant States
C Index Morrorities	Sector	Classica.	
Contraction Contraction System Contraction		When Colors 200 00 Constitution 200 000000 Constitution 200 0000000	0x85124-0.0000
Select Revenue Constructions One Address Recipies Michael Office Michael Conference International Address Conference Conference Conference International Address Conference Conference Conference Conference Conference Conference Conference Conference Conference Conference International Address Conference Conf		Contribution VC. Accession Value - Hendelstein - Accession Value - Parl	
Last Terreste Hellon: Trans California (approx. 1993) Last Terreste (approx. 1993) Last Menezae Lapprox. 20100;35(1980)	С	E Indentione Date of Lawrence	
E lycky scores	100.51	EN KOWF. 1 EXECTORE	
Churchage 17		El el adorection El el adorection (Ordened Al) E Repuis / P E Mandy al E Mandy al	-
C. Physical Street Street	Taxable	Martina have the let offers with	Contration of the
NO.2 NO.2 <th< td=""><td></td><td>f Vertevaletiks Universe (2011 C+1, 00)</td><td>2004).Wood 1.#1</td></th<>		f Vertevaletiks Universe (2011 C+1, 00)	2004).Wood 1.#1

Figure 14 The Web Configurator's Main Screen for Cloud Mode

al proper sease	1.642	
McC Addynam	100 XX4/2803020	
and Menther	contraction and compare	
shiri Milah 🔬 🖓	AVOR III	1 March 1997
45 Decembric March 199	Criteriel 5 (2137) harteelt parents 25 altra	C
G Orichael Interfacility	Conversion (Conception and Conversion processing) -	
No.4	The supervised in the supervis	
end:	Its broase point is costraiched to the internal.	
Acces Connectivate Altern	All and a product the second s	
anova Carros Carros Indeploy Decar	The access part from Coll Science of the Science Scien	
ka Bury ni Amur. CC	96.	

The Web Configurator's main screen is divided into these parts:

- A Title Bar
- **B** Navigation Panel
- **C** Main Window

4.3.1 Title Bar

The title bar provides some useful links that always appear over the screens below, regardless of how deep into the Web Configurator you navigate. If your Zyxel Device is in NCC mode, not all icons will be available in the Title Bar.

Figure 15 Title Bar



The icons provide the following functions.

IABEL	DESC RIPHO N
Wizard	Click this to open the wizard. See Chapter 7 on page 49 for more information.
Help	Click this to open the help page for the current screen.
Forum	Click this to go to Zyxel Biz User Forum, where you can get the latest Zyxel Device information and have conversations with other people by posting your messages.
Site Map	Click this to see an overview of links to the Web Configurator screens.
CLI	Click this to open a popup window that displays the CLI commands sent by the Web Configurator.
Logout	Click this to log out of the Web Configurator.
nebula	Click this to open the NCC web site login page in a new tab or window.

Site Map

Click Site MAP to see an overview of links to the Web Configurator screens. Click a screen's link to go to that screen.

Figure 16 Site Map

🐣 Site Map			28
C Monitor			-
Network Status	Wireles: • AP Information. • Station info • Defected Device	Log	
Configuration			Œ
T Maintenance			Œ

СЦMessages

Click \mathbf{CII} to look at the CLI commands sent by the Web Configurator. These commands appear in a popup window, such as the following.

Figure 17 CLI Messages

li cu	18 18
Clear	1
### CLUIDH	
	Concel

Click \mathbf{Clear} to remove the currently displayed information.

Note: See the Command Reference Guide for information about the commands.

4.3.2 Navigation Panel

Use the menu items on the navigation panel to open screens to configure Zyxel Device features. Click the arrow in the middle of the right edge of the navigation panel to hide the navigation panel menus or drag it to resize them. The following sections introduce the Zyxel Device's navigation panel menus and their screens.



4.3.3 Standalone Mode Navigation Panel Menus

The following are the screens available in standalone mode. Note that some screens may not be available for your Zyxel Device model. See Section 1.4 on page 18 to see which features your Zyxel Device model supports.

Dashboard

The dashboard displays general device information, system status, system resource usage, and interface status in widgets that you can re-arrange to suit your needs.

For details on the Dashboard's features, see Chapter 6 on page 44.

Monitor Menu

The monitor menu screens display status and statistics information.

FO LDER O R LINK	TAB	FUNC TIO N
Network Status	Network Status	Display general LAN interface information and packet statistics.
Wireless		
AP Information	Radio List	Display information about the radios of the connected APs.
Station Info	Station List	Display information about the connected stations.
WDS Link Info	WDS Link Info	Display statistics about the Zyxel Device's WDS (Wireless Distribution System) connections.
Detected Device	Detected Device	Display information about suspected rogue APs.
Log	View Log	Display log entries for the Zyxel Device.

Table 8 Monitor Menu Screens Summary

Configuration Menu

Use the configuration menu screens to configure the Zyxel Device's features.

FO LDER OR LINK	TAB	FUNC TIO N
Network	IP Setting	Configure the IP address for the Zyxel Device Ethernet interface.
	VLAN	Manage the Ethernet interface VLAN settings.
	Storm Control	Enable or disable the broadcast/multicast storm control feature.
	AC Discovery	Configure the Zyxel Device's AP Controller settings.
	NCC Discovery	Configure proxy server settings to access the NCC.
Wireless		•
AP Management	WLAN Setting	Manage the Zyxel Device's general wireless settings.
Rogue AP	Rogue/Friendly AP List	Configure how the Zyxel Device monitors for rogue APs.
Load Balancing	Load Balancing	Configure load balancing for traffic moving to and from wireless clients.
DCS	DCS	Configure dynamic wireless channel selection.
Bluetooth	Advertising Settings	Configure the beacon ID(s) to be included in the Bluetooth advertising packet.
Object		
User	User	Create and manage users.
	Setting	Manage default settings for all users, general settings for user sessions, and rules to force user authentication.

Table 9 Configuration Menu Screens Summary

FO LDER OR LINK	TAB	FUNC TIO N
AP Profile	Radio	Create and manage wireless radio settings files that can be associated with different APs.
	SSID	Create and manage wireless SSID, security, MAC filtering, and layer-2 isolation files that can be associated with different APs.
MON Profile	MON Profile	Create and manage rogue AP monitoring files that can be associated with different APs.
WDS Profile	WDS	Create and manage WDS profiles that can be used to connect to different APs in WDS.
Certificate	My Certificates	Create and manage th e Zyxel Device's certificates.
	Trusted Certificates	Import and manage certificates from trusted sources.
System		
Host Name	Host Name	Configure the system and domain name for the Zyxel Device.
Power Mode	Power Mode	Configure the Zyxel Device's power settings.
Date/Time	Date/Time	Configure the current date, time, and time zone in the Zyxel Device.
WWW	Service Control	Configure HTTP, HTTPS, and general authentication.
SSH	SSH	Configure SSH server and SSH service settings.
TELNET	TELNET	Configure telnet server settings for the Zyxel Device.
FTP	FTP	Configure FTP server settings.
SNMP	SNMP	Configure SNMP communities and services.
Log & Report	•	
Email Daily Report	Email Daily Report	Configure where and how to send daily reports and what reports to send.
Log Setting	Log Setting	Configure the system log, e-mail logs, and remote syslog servers.

 Table 9
 Configuration Menu Screens Summary (continued)

Maintenance Menu

Use the maintenance menu screens to manage configuration and firmware files, run diagnostics, and reboot or shut down the Zyxel Device.

FO LDER O R LINK	ТАВ	FUNCTIO N	
File Manager	Configuration File	Manage and upload configuration files for the Zyxel Device.	
	Firmware Package	View the current firmware version and to upload firmware.	
	Shell Script	Manage and run shell script files for the Zyxel Device.	
Diagnostics	Diagnostics	Collect diagnostic information.	
LEDs	Suppression	Enable this feature to keep the LEDs off after the Zyxel Device starts.	
	Locator	Enable this feature to see the actual location of the Zyxel Device between several devices in the network.	
Antenna	Antenna Switch	Change antenna orientation for the radios.	
Reboot	Reboot	Restart the Zyxel Device.	
Shutdown	Shutdown	Turn off the Zyxel Device.	

 Table 10
 Maintenance Menu Screens Summary

4.3.4 Cloud Mode Navigation Panel Menus

If your Zyxel Device is in NCC mode, you only need to use the Web Configurator for troubleshooting if your Zyxel Device cannot connect to the Internet.

Da shbo a rd

The dashboard displays general Zyxel Device information, and AP information in widgets that you can re-arrange to suit your needs.

For details on the Dashboard's features, see Chapter 23 on page 187.

Configuration Menu

Use the configuration menu screens to configure the Zyxel Device's features.

FOLDERORLINK	TAB	FUNC TIO N
Network	IP Setting	Configure the IP address for the Zyxel Device Ethernet interface.
	VLAN	Manage the Ethernet interface VLAN settings.

Table 11 Configuration Menu Screens Summary

4.3.5 Tables and Lists

The Web Configurator tables and lists are quite flexible and provide several options for how to display their entries.

4.3.5.1 Manipulating Table Display

Here are some of the ways you can manipulate the Web Configurator tables.

1 Click a column heading to sort the table's entries according to that column's criteria.

🖕 🖗 🕹 🖓 🖓 📲 Remover 🔮 According 🖓 Resservation 🗃 Orban Conservation of				
	STOLL	CHURSNE 2->	niáquatics cana	
1	8	Mic_Rudic_u432	2.04	
e	1	Pile_Road Ac_2010	22	
۳.,	9	dela in	2.05	
8	0	defo. #2	2.5	

- 2 Click the down arrow next to a column heading for more options about how to display the entries. The options available vary depending on the type of fields in the column. Here are some examples of what you can do:
 - Sort in ascending alphabetical order
 - Sort in descending (reverse) alphabetical order
 - Select which columns to display
 - Group entries by field
 - Show entries in groups

• Filter by mathematical operators (<, >, or =) or searching for text.

	the second s		Opeintal by N	
	Wa_Racio_24G	11 Lort Ascending	Messib	
t 📿	Wb_Radio_3G	11 Sert Descending	MISSOD	
Q	delault	and the second se	W Stotus	
9	eelaut2 + + 2toe = e liens	Group by this field	12 Profile Hame	

3 Select a column heading cell's right border and drag to re-size the column.

Q Add 2111	Firmine QAction Practs	ohe die Okaised Herberstein	
# Shaha	Hede Name +	And Jency Band	Operating Made
1	Wiz.Rodio.24G	45	M85SID
i 👋	Wiz,Rodio_SG	G	MBSND
a 👋	detault	1.43	MBSSID
1	clefoult2	DG.	MBSSID

4 Select a column heading and drag and drop it to change the column order. A green check mark displays next to the column's title when you drag the column to a valid new location.

fernine PAthon Phot	Store Canadi Series	
Profile Name +	Treducerica dana	Ciperating Mode
defoult2	35	" Ny Partie tears
defourt	2.4G	MBSSID-
Wit_Rock_SG	90	MBSSID
Wit_Roolo_24G	2.45	MRSSID
	defoult2 defoult2 defoult wit_Radio_SG	default2 5G default 2.4G Wb_Rado_SG 5G

5 Use the icons and fields at the bottom of the table to navigate to different pages of entries and control how many entries display at a time.

) Kas 🛛 🖬 🖬	ersena 🗣 Activititi 🖓 Heer	felds 🗟 Classific Information	
Dona -	Prolle Morre	Important Band	Cleaning Mon
9	defoolt2	3G-	MISSID
9	default.	2.40	MBSSID
	Wit_Rocko_5G	5G	MESSID
	Wit_Roolo_24G	2.45	MBS9D

4.3.5.2 Working with Table Entries

The tables have icons for working with table entries. A sample is shown next. You can often use the [Shift] or [Ctrl] key to select multiple entries to remove, activate, or deactivate.

Figure 1	19	Common	Table	Icons

🗘 Add 🖉 Edli 🍯	Remove 🍯 Activate 📲 Inac	tivote Coject Reference	
# 3104.s +	Profile Instrum	Property frant	Operating Model
	Wit_Rocko_24G	2.45	MRSSID
1. 👰	Wit_Rodo_SG	5G	MBSSID
5 Q	delaut	2.4G	M653ID
1 1	Chucken	36	MASSID
el (V	Test	50	MESSID

Here are descriptions for the most common table icons.

Table 12 Common Table Icons

LABEL	DESC RIPIIO N
Add	Click this to create a new entry. For features where the entry's position in the numbered list is important (features where the Zyxel Device applies the table's entries in order like the firewall for example), you can select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings. In some tables you can just click a table entry and edit it directly in the table. For those types of tables small red triangles display for table entries with changes that you have not yet applied.
Remove	To remove an entry, select it and click \mathbf{Remove} . The Zyxel Device confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Ac tiva te .
Inactivate	To turn off an entry, select it and click Inactivate .
Object Reference	Select an entry and click Object Reference to open a screen that shows which settings use the entry.

PART I Standalone Configuration

C HAPTER 5 Standalone Configuration

5.1 Overview

The Zyxel Device is in standalone mode by default. Use the web configurator to manage and configure the Zyxel Device directly. As shown in the following figure, wireless clients can connect to the Zyxel Device (A) to access network resources.



5.2 Starting and Stopping the Zyxel Device

Here are some of the ways to start and stop the Zyxel Device.

Always use Maintenance > Shutdown or the shutdown command before you turn off the Zyxel Device or remove the power. Not doing so can cause the firm ware to become corrupt.

MEIHO D	DESC RIPIIO N
Turning on the power	A cold start occurs when you turn on the power to the Zyxel Device. The Zyxel Device powers up, checks the hardware, and starts the system processes.
Rebooting the Zyxel Device	A warm start (without powering down and powering up again) occurs when you use the Reboot button in the Reboot screen or when you use the reboot command. The Zyxel Device writes all cached data to the local storage, stops the system processes, and then does a warm start.

Table 13 Starting and Stopping the Zyxel Device

Table 13	Starting and S	topping the	Zvxel Device
	orannig ana o	iopping inio	2,001 001100

MEIHO D	DESC RIPTIO N	
Using the RESET button	If you press the RESET button on the back of the Zyxel Device, the Zyxel Device sets the configuration to its default values and then reboots. See Section 26.6 on page 205 for more information.	
Clicking Maintenance > Shutdown > Shutdown or Using the shutdown command	Clicking Maintenance > Shutdown > Shutdown or using the shutdown command writes all cached data to the local storage and stops the system processes. Wait for the Zyxel Device to shut down and then manually turn off or remove the power. It does not turn off the power.	
Disconnecting the power	Power off occurs when you turn off the power to the Zyxel Device. The Zyxel Device simply turns off. It does not stop the system processes or write cached data to local storage.	

The Zyxel Device does not stop or start the system processes when you apply configuration files or run shell scripts although you may temporarily lose access to network resources.

C HA PTER 6 Da shb o a rd

6.1 Overview

This screen displays general device information, system status, system resource usage, and interface status in widgets that you can re-arrange to suit your needs. You can also collapse, refresh, and close individual widgets.

Search A				A-	- Children berlinge
Selectionshes Selections Height Interve Selections Height Anterve Resource Selection	Sector Science State Ladvas 22,722,403 Units (Science) ender Science (Science) Science (Science) Company Company	- 31	 Automotifiation Sustain Latinace Cummin Deterliting Cummin Deterliting<th>ter rate antiposto roduci Scientification construit (antipotetication) res res restance Nationality</th><th>B</th>	ter rate antiposto roduci Scientification construit (antipotetication) res res restance Nationality	B
tor freuens approx Rate tur friesow pyrose Dövereniecource	NUMBER OF STREET	-	Windows Date Larway	the second s	and a subsection of the local states and the local states of the l
CPU Riceso Mentary Longe			RC-W advances Advances Devine (In: Tradition AP Expanse) Newly A	ж. Е	.kGta
(Renari Hengi tan Antonio (Rengi tan Antonio (Rengi tan			E VELLON DAL		1971 (1995) (1991) (1995)
Wat Motor Dela Lavera Lina del Galicia das 214 della Constante della	79 1925 - 1939 - October 2000 - 19 1 - 2029 - Africa - 19 1 - Africa - 2010 - 1	area Th			

The following table describes the labels in this screen.

Table 14 Dashboard

IABEL	DESC RIPTIO N
Widget Settings (A)	Use this link to re-open closed widgets. Widgets that are already open appear grayed out.
Refresh Time Setting (B)	Set the interval for refreshing the information displayed in the widget.
Refresh Now (C)	Click this to update the widget's information immediately.
Close Widget (D)	Click this to close the widget. Use Widget Settings to re-open it.
Device Information	
System Name	This field displays the name used to identify the Zyxel Device on any network. Click the icon to open the screen where you can change it.
System Location	This field displays the location of the Zyxel Device. Click the icon to open the screen where you can change it.
Model Name	This field displays the model name of this Zyxel Device.

NWA/WAC/WAX Series User's Guide

IABEL	DESC RIPIIO N
Serial Number	This field displays the serial number of this Zyxel Device.
MAC Address Range	This field displays the MAC addresses used by the Zyxel Device. Each physical port or wireless radio has one MAC address. The first MAC address is assigned to the Ethernet LAN port, the second MAC address is assigned to the first radio, and so on.
Firmware Version	This field displays the version number and date of the firmware the Zyxel Device is currently running. Click the icon to open the screen where you can upload firmware.
Last Firmware Upgrade Status	This field displays whether the latest firmware update was successfully completed.
Last Firmware Upgrade	This field displays the date and time when the last firmware update was made.
System Resources	
CPU Usage	This field displays what percentage of the Zyxel Device's processing capability is currently being used. Hover your cursor over this field to display the Show CPU Usage icon that takes you to a chart of the Zyxel Device's recent CPU usage.
Memory Usage	This field displays what percentage of the Zyxel Device's RAM is currently being used. Hove your cursor over this field to display the Show Memory Usage icon that takes you to a chart of the Zyxel Device's recent memory usage.
Flash Usage	This field displays what percentage of the Zyxel Device's onboard flash memory is currently being used.
Ethernet Neighbor	
Local Port (Description)	This field displays the port of the Zyxel Device, on which the neighboring device is discovered.
Model Name	This field displays the model name of the discovered device.
System Name	This field displays the system name of the discovered device.
FW Version	This field displays the firmware version of the discovered device.
Port (Description)	This field displays the discovered device's port which is connected to the Zyxel Device.
IP	This field displays the IP address of the discovered device. Click the IP address to access and manage the discovered device using its Web Configurator.
MAC	This field displays the MAC address of the discovered device.
WDS (Wireless Distribut	ion System) Uplink/Downlink Status
MAC Address	This field displays the MAC address of the root AP or repeater to which the Zyxel Device is connected using WDS.
Radio	This field displays the radio number on the root AP or repeater to which the Zyxel Device is connected using WDS.
Channel	This field displays the channel number on the root AP or repeater to which the Zyxel Device is connected using WDS.
SSID	This field displays the name of the wireless network to which the Zyxel Device is connected using WDS.
Security Mode	This field displays which secure encryption methods is being used by the Zyxel Device to connect to the root AP or repeater using WDS.
Link Status	This field displays the RSSI (Received Signal Strength Indicator) and transmission/reception rate of the wireless connection in WDS.
System Status	· · · · · · · · · · · · · · · · · · ·
System Uptime	This field displays how long the Zyxel Device has been running since it last restarted or was turned on.
Current Date/ Time	This field displays the current date and time in the Zyxel Device. The format is yyyy-mm-dd hh:mm:ss.

Table 14	Dashboard	(continued)
	Dashboara	

LABEL	DESC RIPIIO N
Current Login User	This field displays the user name used to log in to the current session, the amount of reauthentication time remaining, and the amount of lease time remaining.
Boot Status	This field displays details about the Zyxel Device's startup state.
	OK - The Zyxel Device started up successfully.
	Firm ware update OK - A firmware update was successful.
	Problematic configuration after firm ware update - The application of the configuration failed after a firmware upgrade.
	System default configuration - The Zyxel Device successfully applied the system default configuration. This occurs when the Zyxel Device starts for the first time or you intentionally reset the Zyxel Device to the system default settings.
	Fallback to lastgood configuration - The Zyxel Device was unable to apply the startup- config.conf configuration file and fell back to the lastgood.conf configuration file.
	Fallback to system default configuration - The Zyxel Device was unable to apply the lastgood.conf configuration file and fell back to the system default configuration file (system-default.conf).
	Booting in progress - The Zyxel Device is still applying the system configuration.
Management Mode	This shows whether the Zyxel Device is set to work as a stand alone AP.
Power Mode	This displays the Zyxel Device's power status.
	Full - the Zyxel Device receives power using a power adapter and/or through a PoE switch/ injector using IEEE 802.3at PoE plus or IEEE 802.3bt (WAX650S only at the time of writing).
	Lim ite d - the Zyxel Device receives power through a PoE switch/injector using IEEE 802.3af PoE or IEEE 802.3at PoE plus (WAX650S only at the time of writing) even when it is also connected to a power source using a power adapter.
	When the Zyxel Device is in limited power mode, the Zyxel Device throughput decreases and has just one transmitting radio chain.
	It always shows Full if the Zyxel Device does not support power detection. See Section 1.5 on page 19.
Bluetooth	This field displays the Zyxel Device's Bluetooth Low Energy (BLE) capability. Bluetooth Low Energy, which is also known as Bluetooth Smart, transmits less data over a shorter distance and consumes less power than classic Bluetooth. The Zyxel Device communicates with other BLE enabled devices using advertisements.
	N/A displays if the Zyxel Device does not support BLE.
	Unavailable displays if the Zyxel Device supports Bluetooth, but there is no BLE USB dongle connected to the USB port of the Zyxel Device. Some Zyxel Devices, such as the WAC5302D-S, need to have a supported BLE USB dongle attached to act as a beacon to broadcast packets.
	Available displays if the Zyxel Device supports Bluetooth and detects a BLE device but advertising is inactive.
	Advertising displays if the Zyxel Device supports Bluetooth, detects a BLE device, and advertising is activated, which means the Zyxel Device can broadcast packets to every BLE device around it.
Interface Status Summary	If an Ethernet interface does not have any physical ports associated with it, its entry is displayed in light gray text. Click the De ta il icon to go to a (more detailed) summary screen of interface statistics.
Name	This field displays the name of each interface.

Table 14 Dashboard (continued)

IABEL	DESC RIPTIO N	
Status	This field displays the current status of each interface. The possible values depend on what type of interface it is.	
	Inactive - The Ethernet interface is disabled.	
	\mathbf{Down} - The Ethernet interface is enabled but not connected.	
	Speed / Duplex - The Ethernet interface is enabled and connected. This field displays the port speed and duplex setting (Full or Half).	
VID	This field displays the VLAN ID to which the interface belongs.	
IP Addr/Netmask	This field displays the current IP address and subnet mask assigned to the interface. If the IP address is 0.0.0.0, the interface is disabled or did not receive an IP address and subnet mask via DHCP.	
IP Assignment	This field displays how the interface gets its IP address.	
	Static - This interface has a static IP address.	
	DHCPClient - This interface gets its IP address from a DHCP server.	
Action	If the interface has a static IP address, this shows \mathbf{n}/\mathbf{a} .	
	If the interface has a dynamic IP address, use this field to get or to update the IP address for the interface. Click \mathbf{Renew} to send a new DHCP request to a DHCP server.	
WLAN Interface Status Summary	This displays status information for the WLAN interface.	
Status	This displays whether or not the WLAN interface is activated.	
MAC Address	This displays the MAC address of the radio.	
Radio	This indicates the radio number on the Zyxel Device.	
Band	This indicates the wireless frequency band currently being used by the radio.	
	This shows - when the radio is in monitor mode.	
OP Mode	This indicates the radio's operating mode. Operating modes are AP (MBSSID), MON (monitor), Root AP or Repeater .	
Channel	This indicates the channel number the radio is using.	
Antenna	This indicates the antenna orientation for the radio (Wall or Ceiling).	
	This field is not available if the Zyxel Device does not allow you to adjust antenna orientation for the Zyxel Device's radio(s) using the web configurator or a physical switch. Refer to Section 1.4 on page 18 to see if your Zyxel Device has an antenna switch.	
Station	This displays the number of wireless clients connected to the Zyxel Device.	
AP Information	This shows a summary of connected wireless Access Points (APs).	
All Sensed Device	This sections displays a summary of all wireless devices detected by the network. Click the link to go to the Monitor > Wireless > Detected Device screen.	
Un-Classified AP	This displays the number of detected unclassified APs.	
Rogue AP	This displays the number of detected rogue APs.	
Friendly AP	This displays the number of detected friendly APs.	

Table 14 Dashboard (continued)

6.1.1 CPU Usage

Use this screen to look at a chart of the Zyxel Device's recent CPU usage. To access this screen, click **CPU Usage** in the dashboard.



The following table describes the labels in this screen.

Table 15 Dashboard > CPU Usage

LABEL	DESC RIPIIO N
%	The y-axis represents the percentage of CPU usage.
time	The x-axis shows the time period over which the CPU usage occurred.
Refresh Interval	Enter how often you want this window to be automatically updated.
Refresh Now	Click this to update the information in the window right away.

6.1.2 Memory Usage

Use this screen to look at a chart of the Zyxel Device's recent memory (RAM) usage. To access this screen, click **Memory Usage** in the dashboard.



Figure 22 Dashboard > Memory Usage
The following table describes the labels in this screen.

IABEL	DESC RIPTIO N
%	The y-axis represents the percentage of RAM usage.
time	The x-axis shows the time period over which the RAM usage occurred
Refresh Interval	Enter how often you want this window to be automatically updated.
Refresh Now	Click this to update the information in the window right away.

Table 16 Dashboard > Memory Usage

C HA PTER 7 Se tup Wiza rd

7.1 Accessing the Wizard

When you log into the Web Configurator for the first time or when you reset the Zyxel Device to its default configuration, the wizard screen displays.

Note: If you have already configured the wizard screens and want to open it again, click the **Wizard** icon on the upper right corner of any Web Configurator screen.

7.2 Using the Wizard

This wizard helps you configure the Zyxel Device IP address, change time zone, daylight saving and radio settings, and edit an SSID profile to change general wireless and wireless security settings.

7.2.1 Step 1 Time Settings

Use this screen to configure the Zyxel Device's country code, time zone and daylight saving time.

- Country Code: Select the country where the Zyxel Device is located.
- Note: The country code field is not available and you cannot change the country code if the Zyxel Device products comply with the U.S. laws, policies and regulations and are to be sold to the U.S. market.
- Time Zone: Select the time zone of your location. This will set the time difference between your time zone and Greenwich Mean Time (GMT).
- Enable Daylight Saving: Select the option if you use Daylight Saving Time. Configure the day and time when Daylight Saving Time starts and ends.
- Offset allows you to specify how much the clock changes when daylight saving begins and ends. Enter a number from 1 to 5.5 (by 0.5 increments).

Click Next to proceed. Click Cancel to close the wizard without saving.

Higher Set		
stop 1		the Setup Waterd
-	Time Settings	
550.2	The series	jGMT 02000 bei ng Hong Kong Perin Singoslore Nobel 💿
2104	T Choole Coy	/ she Saving
2224	wartfolgen medition	Data and Annual Professional State State
war.	and boot	The second secon
Seat		
		Next Cancel

7.2.2 Step 2 Password and Uplink Connection

Use this screen to configure the Zyxel Device's system password and IP address.

Change Password: Enter a new password and retype it to confirm.

Uplink Connection: Select **Auto** (**DHCP**) if the Zyxel Device is connected to a router with the DHCP server enabled. You then need to check the router for the IP address assigned to the Zyxel Device in order to access the Zyxel Device's Web Configurator again.

Otherwise, select **Static IP** when the Zyxel Device is NOT connected to a router or you want to assign it a fixed IP address. You will need to manually enter:

- the Zyxel Device's IP address and subnet mask.
- the IP address of the router that helps forward traffic.
- a DNS server's IP address. The Domain Name System (DNS) maps a domain name to an IP address and vice versa. The DNS server is extremely important because without it, you must know the IP address of a computer before you can access it.

Click **Prev** to return to the previous screen. Click **Next** to proceed. Click **Cancel** to close the wizard without saving.

Weard Set	ling		
366.6	Change Researd		
	Rever Converted	0.000	
Sep 1) wattern - pissessiti		
	Uplink Connection:		
1463	() Auto(0(127))	a State 🕅	
		P sectores:	0.003 0
		Success Marks	0.0.0.2
Sac 4		insteam/	pupid
		End Server	0.0.0.0
Rec III		58 (200) 2 01	

Figure 24 Wizard: Change Password and Uplink Connection

7.2.3 Step 3 Radio

Use this screen to configure the Zyxel Device's radio transmitter(s).

- Channel Selection: Select Auto to have the Zyxel Device automatically choose a radio channel that has least interference. Otherwise, select Manual and specify a channel the Zyxel Device will use in the 2.4 GHz or 5 GHz wireless LAN. The options vary depending on the frequency band and the country you are in.
- Maximum Output Power: Enter the maximum output power of the Zyxel Device. If there is a high density of APs in an area, decrease the output power of the Zyxel Device to reduce interference with other APs.

Note: Reducing the output power also reduces the Zyxel Device's effective broadcast radius.

Click **Prev** to return to the previous screen. Click **Next** to proceed. Click **Cancel** to close the wizard without saving.

Wittend set	Wizard: Radio	
29,01	Radio	
5 45 2 36 p.3	Minet Chornel Kildth Chornel Salevillar Modimum Curpan Sowen	2.49 F 2049 s © 244 v (t) 100-24 (t) (t) [30] = 20 m(t)-30 (t)
Dep 1	Benck Of orned Width Channel Selection Marth on Carpor Power:	осни 20/40/30/VHL — — Ф. Албо — М. Малиа 20 — Цётлагай
9100	Norman de partemar	Trev Nect Concel

Figure 25 Wizard Radic

7.2.4 Step 4 SSID

Use this screen to enable, disable or edit an SSID profile.

Select an SSID profile and click the **Status** switch to turn it on or off. To change an SSID profile's settings, such as the SSID (WiFi network name) and WiFi password, double-click the SSID profile entry from the list. See Section 7.2.4.1 on page 52 for more information.

Note: You cannot add or remove an SSID profile after running the setup wizard.

993	880					
	13	1.4.1	1945	Sciently More	Barns Marine	4.010
ine 4	1.1	000	24000	WTA2/Personal	Due berd	1
	2	CC)	27.04	WT920Peteorid	Dee Send	5.7
1.00	5	000	2,000	WTVAG4Petronic	Due Send	1
	1	CID	2,004	MTAD/Persona	Dec Gond	1
- 22	<u>.</u>	CED	27/201	WPx2-Report	Due Bend	1.1
84	4	CED	Żywa	WE COMMENT	Dec Rota	- 23
	10	(3)	24000	MRA2 Promote	Film Road	1
ilian -	11	CED	3400	KARAS Reference	This bench	0.0

7.2.4.1 Edit SSID Profile

Use this screen to configure an SSID profile.

The screen varies depending on the security type you selected.

- SSID: Enter a descriptive name of up to 32 printable characters for the wireless LAN.
- VIAN ID: Enter a VLAN ID for the Zyxel Device to use to tag traffic originating from this SSID.

Band Mode: Select the wireless band which this profile should use. 2.4 GHz is the frequency used by IEEE 802.11b/g/n wireless clients. 5 GHz is the frequency used by IEEE 802.11ac/a/n wireless clients.

- Security Type: Select WPA2 to add security on this wireless network. Otherwise, select OPEN to allow any wireless client to associate this network without authentication.
- **Personal:** If you set **Security Type** to **WPA2** and select **Personal**, enter a pre-shared key of between 8 and 63 case-sensitive ASCII characters (including spaces and symbols) or 64 hexadecimal characters.
- Enterprise: Select this option and the Primary / Secondary RADIUS Server check box to have the Zyxel Device use the specified RADIUS server. You have to enter the IP address, port number and shared secret password of the RADIUS server to be used for authentication.

Click OK to proceed. Click Cancel to close the screen without saving.

SITX 1	2;0ef			
Storus:	Active	in .		
VEAND:	11/2-31 (b)	-40.94		
Band Moder	Dual Sona	15		
Secondly Types	WP/A			
 Descrut 				
Secret	12345678			
e bilepise				
			OK	Ca
	SID: Edit (WPA2-Ente	erprise)	OK	Cai
ure 28 Wizard: S IIt 5510 Prattic	SID: Edit (WPA2-Ente	erprise)	OK	Ca
	SID: Edit (WPA2-Ente	erprise)	OK	Ca
ilt SSID Profile		erprise)	OK	Ca
III SSID Profile	7yari Active	l	OK	Ca
lit SSID Profile SAD: Status	7yari Active		OK	Ca
lit SSID Profile SID: SAba MLAN D:	Zyart Active		OK	Ca
III SSID Profile SAD: Saba MJAN D: Band Mode:	7yard Active Duicel Sense		OK	Ca
III SSID Profile SSID: S	7ysrd Active 1		OK	Ca
III SSID Profile SID: SSID: SSID: VLAN D: Bond Mode: Security Type: Personal Personal Dimension Primaly RAC	7yard Active 7a.cl Sane 7a.cl Sane 74.cl Sane		OK	Ca
III SSID Profile SID: SSID: SSID: VLAN D: Bond Mode: Security Type: Personal Personal Dimension Primaly RAC	7ysrd Active Tatol Sone W (A2 303 Server sent? Address		-0	Ca
III SSID Profile CID: Notes VLAN D: Rend Mode: Security Type: Personal Directorice El Prima y RAC RACIUS cen	7yard Active 74 of Sone 74 of Sone 74 of Sone 74 of Sone 74 of Sone 74 of Sone 75 of Sone 76 of Sone 77 of Sone 76 of Son		0	Ca
III SSID Profile SID: Notes VLAN D: Rend Moder Security Type: Presonal Encerptice III Primary RAC RACIUS Sen RACIUS Sen	7ysci Active Taloi Sone W 142 303 Server venti? Adonest ver Porti		0	Ca
III SSID Profile SGD: SGD: SGDS VLAN D: Bond Moder Security Type: Personal Personal Personal Primary RAC RACIUS Cen RACIUS Cen RACIUS Cen RACIUS Cen	7ysci Active Active Michael Or Michael Wick2 SUIS Server Ver IP Aukoness Ver Port Ver Secret: Rodius Server		0	Co
III SSID Profile SGD: SGD: SGDS VLAN D: Bond Moder Security Type: Personal Personal Personal Primary RAC RACIUS Cen RACIUS Cen RACIUS Cen RACIUS Cen	7yard Active 7a.dl Sano 7a.dl San		0	Ca

7.2.5 Summary

Use this screen to check whether what you have configured is correct. Click **Save** to apply your settings and complete the wizard setup. Otherwise, click **Prev** to return to the previous screen or click **Cancel** to close the wizard without saving.

Figure 29 Wizard: Summary

Waard Ard	ling					1.0
(165.)	Summery					1
	tilleon cones	929	1998-00 1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (1997 (19	a ya tta hi Saywara	0.000	
1942 (2.)	Lovight Loving:	1000	(A)			
	Noncontrast Pr	(teres	6#			
1912 1	P Antiness	0.0.5	÷.			
	Curve Mark	10.14	41			
22.67	a continuently	.0.0.0	Ø)			1993
ster 4	Designment:	0.00	41			
	2403030	Alth	6 - C			
and the second	(desado)	AV-D				
Step 2	.0.1					
	a Statut	840	Security Revealed	Francisco estas	1126-61	
		1.0	1000 C	1000	÷.	
				Tree	Save	Carpel

C HAPTER 8 Monitor

8.1 Overview

Use the Monitor screens to check status and statistics information.

8.1.1 What You Can Do in this Chapter

- The Network Status screen (Section 8.3 on page 55) displays general LAN interface information and packet statistics.
- The **AP Inform a tion** > **Ra dio List** screen (Section 8.4 on page 57) displays statistics about the wireless radio transmitters in the Zyxel Device.
- The Station Info screen (Section 8.5 on page 61) displays statistics pertaining to the associated stations.
- The **WDS Link Info** screen (Section 8.6 on page 62) displays statistics about the Zyxel Device's WDS (Wireless Distribution System) connections.
- The **Detected Device** screen (Section 8.7 on page 63) displays information about suspected rogue APs.
- The **View Log** screen (Section 8.8 on page 66) displays the Zyxel Device's current log messages. You can change the way the log is displayed, you can e-mail the log, and you can also clear the log in this screen.

8.2 What You Need to Know

The following terms and concepts may help as you read through the chapter.

Rogue AP

Rogue APs are wireless access points operating in a network's coverage area that are not under the control of the network's administrators, and can open up holes in a network's security. See Chapter 13 on page 117 for details.

Friendly AP

Friendly APs are other wireless access points that are detected in your network, as well as any others that you know are not a threat (those from neighboring networks, for example). See Chapter 13 on page 117 for details.

8.3 Network Status

Use this screen to look at general Ethernet interface information and packet statistics. To access this screen, click **Monitor > Ne twork Status**.

Figure 28 Monitor > Network Status

eriope Sus	mmary								
N DTOIC		storus	MD.	IF Addish	letro cel		P Assent	CW.	40100
UN INK		ICCOM/INF	1	172.1640		352.C	DI CP die	r I	La new
s Interles	e Summary								
Name I	Stotu		ΡA	do es				Actio	0
UNINK		MyRul	IN	CICCAI – HS	t denot ki titler	Sidew Cily Sil		1991	
UCTINK el Stockeller Voltanterec Switch Tert	s Table Ib Crophic View	5 N	econds <mark>Sel</mark>	interval	stoft		ľŕ		llo Ime
ur INK distvikile) Volintered	s Table It	s a				Coll : ons	DC 5	093 54	Up Ime S12531

The following table describes the labels in this screen.

IABEL	DESC RIPTIO N
Interface Summary IPv6 Interface Summary	Use the Interface Summary section for IPv4 network settings. Use the IPv6 Interface Summary section for IPv6 network settings if you connect your Zyxel Device to an IPv6 network. Both sections have similar fields as described below.
Name	This field displays the name of the physical Ethernet port on the Zyxel Device.
Status	This field displays the current status of each physical port on the Zyxel Device.
	Down - The port is not connected.
	Speed / Duplex - The port is connected. This field displays the port speed and duplex setting (Full or Half).
VID	This field displays the VLAN ID to which the port belongs.
IP Addr/Netmask IP Address	This field displays the current IP address (and subnet mask) of the interface. If the IP address is 0.0.0.0 (in the IPv4 network) or :: (in the IPv6 network), the interface does not have an IP address yet.
IP Assignment	This field displays how the interface gets its IPv4 address.
	Static - This interface has a static IPv4 address.
	DHCP Client - This interface gets its IPv4 address from a DHCP server.
Action	Use this field to get or to update the IP address for the interface. Click $\mathbf{Re} \mathbf{ne} \mathbf{w}$ to send a new DHCP request to a DHCP server. If the interface cannot use one of these ways to get or to update its IP address, this field displays \mathbf{n}/\mathbf{a} .
Port Statistics Table))
Poll Interval	Enter how often you want this window to be updated automatically, and click Set Interval.

Table 17 Monitor > Network Status

LABEL	DESC RIPTIO N
Set Interval	Click this to set the Poll Interval the screen uses.
Stop	Click this to stop the window from updating automatically. You can start it again by setting the Poll Interval and clicking Set Interval .
Switch to Graphic View	Click this to display the port statistics as a line graph.
Name	This field displays the name of the interface.
Status	This field displays the current status of the physical port.
	\mathbf{Down} - The physical port is not connected.
	Speed / Duplex - The physical port is connected. This field displays the port speed and duplex setting (Full or Half).
TxPkts	This field displays the number of packets transmitted from the Zyxel Device on the physical port since it was last connected.
RxPkts	This field displays the number of packets received by the Zyxel Device on the physical port since it was last connected.
Tx Bcast	This field displays the number of broadcast packets transmitted from the Zyxel Device on the physical port since it was last connected.
Rx Bcast	This field displays the number of broadcast packets received by the Zyxel Device on the physical port since it was last connected.
Collisions	This field displays the number of collisions on the physical port since it was last connected.
Tx	This field displays the transmission speed, in bytes per second, on the physical port in the one- second interval before the screen updated.
Rx	This field displays the reception speed, in bytes per second, on the physical port in the one- second interval before the screen updated.
Up Time	This field displays how long the physical port has been connected.
System Up Time	This field displays how long the Zyxel Device has been running since it last restarted or was turned on.

Table 17 Monitor > Network Status (continued)

8.3.1 Port Statistics Graph

Use the port statistics graph to look at a line graph of packet statistics for the Ethernet port. To view, click **Monitor > Ne twork Status** and then the **Switch to Graphic View** button.



Figure 29 Monitor > Network Status > Switch to Graphic View

The following table describes the labels in this screen.

LABEL	DESC RIPIIO N
Refresh Interval	Enter how often you want this window to be automatically updated.
Refresh Now	Click this to update the information in the window right away.
Port Selection	Select the Ethernet port for which you want to view the packet statistics.
Switch to Grid View	Click this to display the port statistics as a table.
Kbps/Mbps	The y-axis represents the speed of transmission or reception.
Time	The x-axis shows the time period over which the transmission or reception occurred.
TX	This line represents traffic transmitted from the Zyxel Device on the physical port since it was last connected.
RX	This line represents the traffic received by the Zyxel Device on the physical port since it was last connected.
Last Update	This field displays the date and time the information in the window was last updated.

... . .

8.4 Radio List

Use this screen to view statistics for the Zyxel Device's wireless radio transmitters. To access this screen, click Monitor > Wire less > AP Information > Radio List.

Figure 30	Monitor > Wireless >	AP Information > Radio List ((for Zyxel Device that supports WDS)
-----------	----------------------	-------------------------------	--------------------------------------

_	e è itter	hinni	International Academic Science		1000	Uprocut	Present.	MAC ADD	-	Air / More worthin
9	-	2.40	1	-25	0	0	670810	60:31:97:0	AP. M.	
Q -		33	16121	120	ų.	0	665410	6031-97.0 2	AP p.	detaut2 / det
6.0	Page	1.00	e 56	10 W	ry he	99				Showing 1-2 of 2



	eschuya Refe	130.30	i dan	Transfer	212	$A \times I$	Lond.	241 5 6	N. 14	Crimole_		Survey Hill
2	2	242		1.1	ų.		U.	benswich.		AP) as	. cerect	12%
2		1	1220	24	0	3	0	1112-001	×.	88 YAG	rieto 17	15
10	19230	L SHO	30	cira -	e ins m		0.545	111000.015		100005302	11000000	Deputyted 1923

The following table describes the labels in this screen.

Table 19	Monitor > Wireless >	> AP Information >	Radio List

IABEL	DESC RIPIIO N
More Information	Click this to view additional information about the selected radio's wireless traffic and station count. Information spans a 24 hour period.
Status	This displays whether or not the radio is enabled.
Loading	This indicates the AP's load balance status (Underload or Overload) when load balancing is enabled on the Zyxel Device. Otherwise, it shows - when load balancing is disabled or the radio is in monitor mode.
MAC Address	This displays the MAC address of the radio.
Radio	This indicates the radio number on the Zyxel Device to which it belongs.
OP Mode	This indicates the radio's operating mode. Operating modes are AP (MBSSID), MONITOR , Root AP or Repeater .
AP/WDS Profile	This indicates the AP profile name and WDS profile name to which the radio belongs.
	This field is available only on the Zyxel Device that supports WDS.
Profile	This indicates the AP profile name to which the radio belongs.
	This field is available only on the Zyxel Device that does not support WDS.
Frequency Band	This indicates the wireless frequency band currently being used by the radio.
	This shows - when the radio is in monitor mode.
Channel	This indicates the radio's channel ID.
Transmit Power	This displays the output power of the radio.
Station	This displays the number of wireless clients connected to this radio on the Zyxel Device.

IABEL	DESC RIPTIO N
Upload	This displays the total number of packets received by the radio.
Download	This displays the total number of packets transmitted by the radio.
Channel Utilization	This indicates how much IEEE 802.11 traffic the radio can receive on the channel. It displays what percentage of the radio's channel is currently being used.

Table 19 Monitor > Wireless > AP Information > Radio List (continued)

8.4.1 AP Mode Radio Information

This screen allows you to view a selected radio's SSID details, wireless traffic statistics and station count for the preceding 24 hours. To access this window, select a radio and click the **More Information** button in the **Radio List** screen.



Figure 32 Monitor > Wireless > AP Information > Radio List > More Information

The following table describes the labels in this screen.

Table 20 Monitor	Second Se Second Second Sec
IABEL	DESC RIPIIO N
SSID Detail	This list shows information about all the wireless clients that have connected to the specified radio over the preceding 24 hours.
#	This is the items sequential number in the list. It has no bearing on the actual data in this list.

Table 20 Monitor > Wireless > AP Information > Radio List > More Information

NWA/WAC/WAX Series User's Guide

IABEL	DESC RIPTIO N
SSID Name	This displays an SSID associated with this radio. There can be up to eight maximum.
BSSID	This displays a BSSID associated with this radio. The BSSID is tied to the SSID.
Security Mode	This displays the security mode in which the SSID is operating.
VLAN	This displays the VLAN ID associated with the SSID.
Traffic Statistics	This graph displays the overall traffic information of the radio over the preceding 24 hours.
Kbps/Mbps	This y-axis represents the amount of data moved across this radio in megabytes per second.
Time	This x-axis represents the amount of time over which the data moved across this radio.
Station Count	This graph displays the connected station information of the radio over the preceding 24 hours
Stations	The y-axis represents the number of connected stations.
Time	The x-axis shows the time period over which a station was connected.
Last Update	This field displays the date and time the information in the window was last updated.
ОК	Click this to close this window.
Cancel	Click this to close this window.

 Table 20
 Monitor > Wireless > AP Information > Radio List > More Information (continued)

8.5 Station List

Use this screen to view statistics pertaining to the associated stations (or "wireless clients"). Click **Monitor** > Wireless > Station Info to access this screen.

Figure 33	Monitor > Wireless > Station Info
-----------	-----------------------------------

	1000	274. 601	- 76.00 X I	COLUMN ST	184 H F25-1	23.25 416	265 V	36.000	000 (X) (C)	TR KORD	Constant I	
1	27.0	50159cm.	Ť.	505 Hb/g	N/A	7704-1700	Cp46	destrin	8901	- 590	12,50010	-
e -4	Prig.	11911	1.11.2	De la stal	972			039630763			radon'	

The following table describes the labels in this screen.

IABEL	DESC RIPTIO N
#	This is the station's index number in this list.
IP Address	This is the station's IP address.
MAC Address	This is the station's MAC address.
Radio	This is the radio number on the Zyxel Device to which the station is connected.
Capability	This displays the supported standard currently being used by the station or the standards supported by the station.
802.11 Features	This displays whether the station supports IEEE802.11r, IEEE 802.11k, IEEE 802.11v or none of the above (N/A) .

Table 21 Monitor > Wireless > Station Info

LABEL	DESC RIPTIO N
SSID Name	This indicates the name of the wireless network to which the station is connected. A single AP can have multiple SSIDs or networks.
Security Mode	This indicates which secure encryption methods is being used by the station to connect to the network.
Signal Strength	This is the RSSI (Received Signal Strength Indicator) of the station's wireless connection.
Tx Rate	This is the maximum transmission rate of the station.
Rx Rate	This is the maximum reception rate of the station.
Association Time	This displays the time the station first associated with the Zyxel Device's wireless network.
Refresh	Click this to refresh the items displayed on this page.

Table 21 Monitor > Wireless > Station Info (continued)

8.6 WDS Link Info

Use this screen to view the WDS traffic statistics between the Zyxel Device and a root AP or repeaters. See Section 1.2 on page 12 to know more about WDS. Click **Monitor > Wire less > WDS Link Info** to access this screen.

a MAC Account . Rotan 1	Stillenen Security Man.	Syncif Shiers	Alexandra Bra
H + Page Left) + H 3	une 💷 👘 been		Tio data to digitaly
03 Downlink Info			
	Nama Security Securi	k kaitune (es	- Assession trus

Figure 34 Monitor > Wireless > WDS Link Info

The following table describes the labels in this screen.

Table 22 Monitor > Wireless > WDS Link Info

LABEL	DESC RIPTIO N
WDS Uplink Info	Uplink refers to the WDS link from the repeaters to the root AP.
WDS Downlink	Downlink refers to the WDS link from the root AP to the repeaters.
Info	When the Zyxel Device is in root AP mode and connected to a repeater, only the downlink information is displayed.
	When the Zyxel Device is in repeater mode and connected to a root AP directly or via another repeater, the uplink information is displayed.
	When the Zyxel Device is in repeater mode and connected to a root AP and other repeater(s), both the uplink and downlink information would be displayed.
#	This is the index number of the root AP or repeater in this list.

LABEL	DESC RIPTIO N
MAC Address	This is the MAC address of the root AP or repeater to which the Zyxel Device is connected using WDS.
Radio	This is the radio number on the root AP or repeater to which the Zyxel Device is connected using WDS.
SSID Name	This indicates the name of the wireless network to which the Zyxel Device is connected using WDS.
Security Mode	This indicates which secure encryption methods is being used by the Zyxel Device to connect to the root AP or repeater using WDS.
Signal Strength	This is the RSSI (Received Signal Strength Indicator) of the wireless connection in WDS.
Tx Rate	This is the maximum transmission rate of the root AP or repeater to which the Zyxel Device is connected using WDS.
Rx Rate	This is the maximum reception rate of the root AP or repeater to which the Zyxel Device is connected using WDS.
Association Time	This displays the time the Zyxel Device first associated with the wireless network using WDS.
Refresh	Click this to refresh the items displayed on this page.

Table 22 Monitor > Wireless > WDS Link Info (continued)

8.7 Detected Device

Use this screen to view information about surrounding APs which you could mark as Rogue or Friendly. Click **Monitor > Wire less > Detected Device** to access this screen. Not all Zyxel Devices support monitor mode (see Section 1.4 on page 18). For more information about Rogue APs, see Section 10.3 on page 79.

Note: If the Zyxel Device supports monitor mode, the radio or at least one of the Zyxel Device's radio must be set to monitor mode (in the **Wire less > AP Management** screen) in order to detect other wireless devices in its vicinity.

If the Zyxel Device does not support monitor mode, turn on rogue AP detection in the **Configuration > Wire less > Rogue AP** screen to detect other APs.

D)	line o.	Roger /P 🗿	Monics	Sicult OF					
•	Stat.	Tradad	Rain	MARK Address	SS To North	station	002	See Descal	neizen -
	8	Section		0002010200055	VIDCORCEL	10	$ 077\rangle \langle z \rangle$	60°	Max Ju-
2	9	history		00.02.07.40.00.00	SDD1 05502	д.	$\Gamma(T) =$	n17	Mon.36
1	4	Infection		0000123 11:5600	ty plante	<i>A</i>	1777	602	Vor.M
6	9	Tradaic		0003-0211-0050	1042-004-2	1	inn.	4P	Size Ar.
		Stradout-		65(17)(844:33/70	success?	10	1777 - N	60	Mog.M
ŧ.,	9	Pfindad-++		05:19:02:01:28:00:	sign.	12	100	TE	Mog. ht
6	4	-Minchiet		00.05 05 200.05 70	2108 v2	(¢)		ATT	Marshi
		- Arrely		COLUMN IN THE ASSOCI	1,077 0.077		1777	47	0.91.50
	R.	- Maxing-	-	05:0000002040	Jacobi (AP)	£	1777 (see	AP-	Mbd.du
ġ.	9	hereita e		00011020004000	089500 XXI	1	CTT	m17	9.56 Mill.
ñ.,	4	Directory		62:17(18:42:33(7))	20001000000000	361 I.	1777	60 ····	Mort M
(a)	9	Treduc		02(8/000)125(42)	17.27^{+1}	8	intra-	Nama	Mar him
13		Stridag		02050100-21,2042	17.322	6	1777	1000	Moe.ht
14	1	Principal		02050101001.2042	(0.257	1	per l	North	9.65 hr
13	4	Predict		0400 0025400000	MISCORCE	<u>.</u>	077	60	Max hi
ŝ	9	Produk		1001102/001405	GG GG 7Y	a	inn.	- 6F	Mort Man
7	1	Production		10/2010/05/06/05	Tim 999995	0	075	AP.	Moet in
ŵ,	19	hereita e		14:91:02:1(22:94)	(G. F.)	39. 19	ст.,	άP	9.5e. /u
6	4	Directory		12:91:02:01:446:21	Kole/58,5%)	÷.	1777	6.2	Vor.M
ю.,	9	Interdate		14910010-0099	K-#/58.550	1	inin	4P	See hi
jć.	1100	di lette ki	N	2 . N. B. B. M.			******	CON16111	Scaleing a sharefutiy

Figure 35	Monitor > Wireless >	Detected Device	(for 7yxel Device th	at supports Monitor mode)
ing une oo	101011101 - 111101033 -			

	ed Afte							
Eco.w	AP:	D						
S.upec	red toque Al							
There is	- 	10						
0 vice	offed AP.	216						
Delec	t blass							
)electer	Device							
0 /4.5		P 🔕 Mark un Fr	Colorest and a second statements					
- S	N-0	functions.	diferences.	and the second se	where a	1002-	Show Description	Louiseen -
			constantin o			arr	N/P	TRAILER
22			AC 74-29-25 59-25	V DECERCIÓN.	185		305°	Min Id.,
20			10:02:02:02:02:02:02:02:02:02:02:02:02:02	156 KS		intra-	W97	Meo hit
74			POINT STATE 27:51	1565.55	157	irrr.	10°	Mondal.
27			10/00/75/26-05	7945 20000	(33)	ETC.		Mon Jul.
24			5458745691519766	0.559×0.01903	9		43°	$(A, n, b, 1, 1, \dots, n, M, n, n,$
÷7		an a	90:02:97:10:07:15	Topisticous	2	1000	835°	750.355
70 - 5	epictod c	References and	1026550000000000000000000000000000000000		155	intri-	- MP	4456.4.4
29 7	kepeliy AP		90:01.97.77x 55:51	50303770-0	1	STC.	107 co.	Alco Jul-
70			102405000000	40116-59	(† 6)	1000	A85.00	Mon Jul.
71			90:01:97.00x18:27.	5951	46	intra-	NAME	sico bit
19- C			41/00/07/07/2010	7/01/030	2.0	inn-	Mi*	Min hit.
25			02:00:02:70:70:09	$5_{\rm e}(1)$ (35.3	£ 1	ITT.	W7	Mon Jul.,
$\pi c = N$	sported t	Heldrig SUS	7200203010038		197	intr.	93°	The latter
	operte di Li	04468.505	10740505000		201	1005	199°	2466.14
26			2007129911298	$\mathrm{Releves } \mathrm{Releves}$	1	100	wr	$\{0100,101,10\}$
P. 1. 27	genetively	Helden SAR	102495-00244		363	STC.	1987 cc.	$120.137\ldots$
26			101200-0021545	100000000000000000000000000000000000000	3	1000	A95.0	Men Id
78			42.91.97.70:35.92	h Nebah	42	intra-	100 cm	dice tot
49			-70 (97.75, 0.2577) (9)	$r_{\rm e}(0.3577\ldots)$	148	irrr	WP	The later
14:00	Rej 2 - 649	a e Martes	A. 1. 10 10				1997	syngus seudure

Figure 36 Monitor > Wireless > Detected Device (for Zyxel Device that does not support Monitor mode)

The following table describes the labels in this screen.

IABEL	DESC RIPTIO N
Discovered APs	
Rogue AP	This shows how many devices are detected as rogue APs.
Suspected rogue AP	This shows how many devices are detected as possible rogue APs based on the classification rule(s) in Section 10.3 on page 79.
Friendly AP	This shows how many devices are detected as friendly APs.
Un-classified AP	This shows how many devices are detected, but have not been classified as either Rogue or Friendly by the Zyxel Device.
Detect Now	Click this button for the Zyxel Device to scan for APs in the network.
Detected Device	

Table 23 Monitor > Wireless > Detected Device

NWA/WAC/WAX Series User's Guide

LABEL	DESC RIPTIO N
Mark as Rogue AP	Click this button to mark the selected AP as a rogue AP. For more on managing rogue APs, see the Configuration > Wire less > Rogue AP screen (Section 10.3 on page 79).
Mark as Friendly AP	Click this button to mark the selected AP as a friendly AP. For more on managing friendly APs, see the Configuration > Wire less > Rogue AP screen (Section 10.3 on page 79).
#	This is the detected device's index number in this list.
Status	This indicates the detected device's status.
Device	This indicates the type of device detected.
Role	This indicates the detected device's role (such as friendly or rogue).
Classified by	This indicates the detected device's classification rule.
MAC Address	This indicates the detected device's MAC address.
SSID Name	This indicates the detected device's SSID.
Channel ID	This indicates the detected device's channel ID.
802.11 Mode	This indicates the 802.11 mode (a/b/g/n/ac/ax) transmitted by the detected device.
Security	This indicates the encryption method (if any) used by the detected device.
Description	This displays the detected device's description. For more on managing friendly and rogue APs, see the Configuration > Wire less > Rogue AP screen (Section 10.3 on page 79).
Last Seen	This indicates the last time the device was detected by the Zyxel Device.
Refresh	Click this to refresh the items displayed on this page.

Table 23 Monitor > Wireless > Detected Device (continued)

8.8 View Log

Log messages are stored in two separate logs, one for regular log messages and one for debugging messages. In the regular log, you can look at all the log messages by selecting **All Logs**, or you can select a specific category of log messages (for example, user). You can also look at the debugging log by selecting **Debug Log**. All debugging messages have the same priority.

To access this screen, click **Monitor > Log**. The log is displayed in the following screen.

Note: When a log reaches the maximum number of log messages, new log messages automatically overwrite existing log messages, starting with the oldest existing log message first.

Events that generate an alert (as well as a log message) display in red. Regular logs display in black. Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

The Web Configurator saves the filter settings once you click Search. If you leave the View Log screen and return to it later, the last filter settings would still apply.

de Miter						
24						
Asplay:	Al Logn	4	Priority:		any.	*
solice Avdrare	a faith and a second		Desirio tori A	AN WALLS	1000	
aloree nicitaeel	City.	×	Desiration	ferfoerer	witt	1
wennet.	ere.	1	Revealed.			
Satesuta	1.7.13.62.57	10.777				
Trad on Now	👷 Roberth 🛷 Clear I					
* - 2004		2		Marka -	Swithotish	INSIA -
1 2012-02-02-03	tan in 11 Aspekiwa	ater except team	http://www.bea.ics.ic/	172274.1	172.363.4	Accession
2 2012/07/05 0	G., Station: P	NSNADA REDAK	hós Ár-tert by St			
5 257-67-63-6	taan ah	atos acimin from	http:/http:/http:/http://http://	172-201	177.163.4	-account
4. 0012-02-02-0-	u Station i	sesses an an an an	2 hos and oth by 5.			19.00.00 20070911
5 0012-02/05/04	ian an an Station B	net to A Collected a	her availated o			
1 2012/02/02/02	(n roofi shishin	hete weterin by St			
2-2017-02-02-02	taanaa ay Station 2	cito:4380 et da	has associated on			
B 2012-02-02-0	iana an Station 2	c. t0:42.90 St st2	Non A solution by \$1			
9.0012-02.03.01	tenen in Station 2	C FC:43-50 ST 87	here a weight by STate			
10.0012-02-03.00	1	ning Anishistian	here Award here D			
11 2017-02-02-03		outowit octopicat	her availated			
11-0012-02-02-01	haan aan daar Starford (alatezakot dalett	now descerts by SL.			
10.0012-02405-01	and the second	n rolan na stuar	here a trausic by 3.1.			
11.00.902.03.02	1	n receiver so et ar	has avaidated o			
14 .05 7-07-02 03		C 50:47/20 55:07	has ceautinity D			
16, 2017-07-07-01		0.00:42.80.51 <i>9</i> 7	bas availates a			
17 2012-02-02-07	2	nava katana	we denote by $5\dots$			
10.00.545469.00		natež konstava	hos associated a			
19 00:7-02-02-02	ter Stationer	E 75-23-91-11-64	Not describ by $\mathbb{Z}_{\mathrm{res}}$			
70-2012-02-03-01	haran ni Statorell	0.07-20-0141-04	NEW PLANARS BY SLOP			
Star Same I d	4 1 1 1 1 Mar 25 1991	POR:				Displaying 3 - 30 ef 6

Figure 37 Monitor > Log > View Log

The following table describes the labels in this screen.

Table 24 Monitor > Log > View Log

IABEL	DESC RIPIIO N
Show Filter / Hide	Click this button to show or hide the filter settings.
Filter	If the filter settings are hidden, the Display , Email Log Now , Refresh , and Clear Log fields are available.
	If the filter settings are shown, the Display, Priority, Source Address, Destination Address, Source Interface, Destination Interface, Protocol, Keyword, and Search fields are available.
Display	Select the category of log message(s) you want to view. You can also view All Logs at one time, or you can view the Debug Log .
Priority	This displays when you show the filter. Select the priority of log messages to display. The log displays the log messages with this priority or higher. Choices are: any , emerg , alert , crit , emor , wam , notice , and info , from highest priority to lowest priority. This field is read-only if the Category is Debug Iog .
Source Address	This displays when you show the filter. Type the source IP address of the incoming packet that generated the log message. Do not include the port in this filter.

IABEL	DESC RIPIIO N
Destination Address	This displays when you show the filter. Type the IP address of the destination of the incoming packet when the log message was generated. Do not include the port in this filter.
Source Interface	This displays when you show the filter. Select the source interface of the packet that generated the log message.
Destination Interface	This displays when you show the filter. Select the destination interface of the packet that generated the log message.
Protocol	This displays when you show the filter. Select a service protocol whose log messages you would like to see.
Keyword	This displays when you show the filter. Type a keyword to look for in the Message , Source , Destination and Note fields. If a match is found in any field, the log message is displayed. You can use up to 63 alphanumeric characters and the underscore, as well as punctuation marks ()' ,:;?! +-*/= #\$% @ ; the period, double quotes, and brackets are not allowed.
Search	This displays when you show the filter. Click this button to update the log using the current filter settings.
Email Log Now	Click this button to send log messages to the Active e-mail addresses specified in the Send Log To field on the Configuration > Log & Report > Log Settings screen.
Refresh	Click this to update the list of logs.
Clear Log	Click this button to clear the whole log, regardless of what is currently displayed on the screen.
#	This field is a sequential value, and it is not associated with a specific log message.
Time	This field displays the time the log message was recorded.
Priority	This field displays the priority of the log message. It has the same range of values as the Priority field above.
Category	This field displays the log that generated the log message. It is the same value used in the Display and (other) Category fields.
Message	This field displays the reason the log message was generated. The text "[count= x]", where x is a number, appears at the end of the Message field if log consolidation is turned on and multiple entries were aggregated to generate into this one.
Source	This field displays the source IP address and the port number in the event that generated the log message.
Source Interface	This field displays the source interface of the packet that generated the log message.
Destination	This field displays the destination IP address and the port number of the event that generated the log message.
Destination Interface	This field displays the destination interface of the packet that generated the log message.
Protocol	This field displays the service protocol in the event that generated the log message.
Note	This field displays any additional information about the log message.

Table 24 Monitor > Log > View Log (continued)

C HAPTER 9 Network

9.1 Overview

This chapter describes how you can configure the management IP address and VLAN settings of your Zyxel Device.

The Internet Protocol (IP) address identifies a device on a network. Every networking device (including computers, servers, routers, printers, etc.) needs an IP address to communicate across the network. These networking devices are also known as hosts.





The figure above illustrates one possible setup of your Zyxel Device. The gateway IP address is 192.168.1.1 and the managed IP address of the Zyxel Device is 192.168.1.2 (default), but if the Zyxel Device is assigned an IP address by a DHCP server, the default (192.168.1.2) will not be used. The gateway and the Zyxel Device must belong in the same IP subnet to be able to communicate with each other.

9.1.1 APControllerManagement

This discusses using the Zyxel Device with an AP Controller. AP Controllers, such as the NXC, use Control And Provisioning of Wireless Access Points (CAPWAP) to push firmware and/or configurations to the APs that they manage.

The following figure illustrates a wireless network managed by an AC. You (U) configure the AC (C), which then automatically updates the configurations of the managed APs ($M1 \sim M4$).

69



Figure 39 AC managed Network Example

Note: The Zyxel Device can be a standalone device or be managed by an AC.

AC Discovery and Management

The link between AC Discovery-enabled access points proceeds as follows:

- 1 An Zyxel Device with AC Discovery enabled joins a wired network (receives a dynamic IP address).
- 2 The Zyxel Device sends out a discovery request, looking for an AC.
- 3 If there is an AC on the network, it receives the discovery request. If the AC, such as NXC, is in Manual mode it adds the details of the Zyxel Device to its Unmanaged Access Points list, and you decide which available APs to manage. If the AC is in Always Accept mode, it automatically adds the Zyxel Device to its Managed Access Points list and provides the managed Zyxel Device with default configuration information, as well as securely transmitting the DTLS pre-shared key. The managed Zyxel Device is ready for association with wireless clients.

Managed AP Finds the Controller

A managed Zyxel Device can find the controller in one of the following ways:

- Manually specify the controller's IP address in the Web Configurator's AC Discovery screen.
- Get the controller's IP address from a DHCP server with the controller's IP address configured as option 138.
- Get the controller's IP address from a DNS server SRV (Service) record.
- Broadcasting to discover the controller within the broadcast domain.

Note: The AC needs to have a static IP address. If it is a DHCP client, set the DHCP server to reserve an IP address for the AC.

AC management and IP Subnets

By default, CAPWAP works only between Zyxel Devices with IP addresses in the same subnet.

However, you can configure the Zyxel Device and the AC to use CAPWAP with IP addresses in different subnets by doing the following.

- Activate DHCP. Your network's DHCP server must support option 138 defined in RFC 5415.
- Configure DHCP option 138 with the IP address of the AC on your network.

DHCP Option 138 allows the management request (from the Zyxel Device) to reach the AC in a different subnet, as shown in the following figure.



Figure 40 CAPWAP and DHCP Option 138

Notes on AC Management

This section lists some additional features of Zyxel's implementation of the CAPWAP protocol.

- When the AC uses its internal Remote Authentication Dial In User Service (RADIUS) server, managed Zyxel Devices also use the AC's authentication server to authenticate wireless clients.
- If an Zyxel Device's link to the ACis broken, the Zyxel Device continues to use the wireless settings with which it was last provided.

9.1.2 What You Can Do in this Chapter

- The IP Setting screen (Section 9.2 on page 69) configures the Zyxel Device's LAN IP address.
- The VIAN screen (Section 9.3 on page 71) configures the Zyxel Device's VLAN settings.
- The Storm Control screen (Section 9.4 on page 110) turns on or off the traffic storm control feature on the Zyxel Device.
- The AC Discovery screen (Section 9.5 on page 111) configures the Zyxel Device's AP Controller (AC) settings.

• The NCC Discovery screen (Section 9.4 on page 73) configures the Zyxel Device's Nebula Control Center (NCC) discovery settings.

9.2 IP Setting

Use this screen to configure the IP address for your Zyxel Device. To access this screen, click **Configuration > Network > IP Setting**.

ie sotting	NINI .	JC. Discovery	MCC History	
P Address Amigram	Mt.			
a Gét Automótica	all.			
in the Fred P Act	1996			
#70cma		1012612		
TRACT OF BRIDE		11.885,200.11		
dermit		DOCTORY DOCT	TEDADARIUS.	
The Surger Pro-	and the second		100000	
	10			
Pes Address Astign				
		kuto carrigurations	NAME AND ADDRESS OF A DECK	
Instanti Address	s	HILLS AND ALL		
Pas Actores/Park	ater bin	1	(Dpiloral)	
Solevice"			100,00001	
Metro:		18-15)		
DIOW Cleff				
2,181		manualse		
The particular and the	1000			
DHCPss Reque	of Option			
Distance.				
25 hit was				

Apply Tweet

Each field is described in the following table.

TUDIE ZO CONIIQUIUIION > NEIWOR > IF SEIIING	Table 25	Configuration > Network > IP Setting
--	----------	--------------------------------------

LABEL	DESC RIPTIO N
IP Address Assignment	t
Get Automatically	Select this to make the interface a DHCP client and automatically get the IP address, subnet mask, and gateway address from a DHCP server.
Use Fixed IP Address	Select this if you want to specify the IP address, subnet mask, and gateway manually.
IP Address	Enter the IP address for this interface.
Subnet Mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
Gateway	Enter the IP address of the gateway. The Zyxel Device sends packets to the gateway when it does not know how to route the packet to its destination. The gateway should be on the same network as the interface.
DNS Server IP Address	Enter the IP address of the DNS server.
IPv6 Address Assignme	ent

Table 25	Configuration :	> Natwork >	IP Sattina	(continued)
	Conngoranon		n senng	

IABEL	DESC RIPTIO N				
Enable Stateless Address Auto- configuration (SLAAC)	Select this to enable IPv6 stateless auto-configuration on the Zyxel Device. The Zyxel Device will generate an IPv6 address itself from a prefix obtained from an IPv6 router in the network.				
Link-Local AddressThis displays the IPv6 link-local address and the network prefix that the Zyxel generates itself for the LAN interface.					
IPv6 Address/ Prefix Length	Enter the IPv6 address and the prefix length for the LAN interface if you want to use a static IP address. This field is optional.				
	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.				
Gateway	Enter the IPv6 address of the default outgoing gateway using colon (:) hexadecimal notation.				
Metric	Enter the priority of the gateway (if any) on the LAN interface. The Zyxel Device decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the Zyxel Device uses the one that was configured first. Enter zero to set the metric to 1024 for IPv6.				
DHCPv6 Client	Select this option to set the Zyxel Device to act as a DHCPv6 client.				
DUID	This field displays the DHCP Unique IDentifier (DUID) of the Zyxel Device, which is unique and used for identification purposes when the Zyxel Device is exchanging DHCPv6 messages with others. See Appendix B on page 230 for more information.				
Request Address	Select this option to get an IPv6 address from the DHCPv6 server.				
DHCPv6 Request Options	Select this option to determine what additional information to get from the DHCPv6 server.				
DNS Server	Select this option to obtain the IP address of the DNS server.				
NTP Server	Select this option to obtain the IP address of the NTP server.				
Apply	Click Apply to save your changes back to the Zyxel Device.				
Reset	Click Reset to return the screen to its last-saved settings.				

9.3 VIAN

This section discusses how to configure the Zyxel Device's VLAN settings.

Note: Mis-configuring the management VLAN settings in your Zyxel Device can make it inaccessible. If this happens, you will have to reset the Zyxel Device.





In the figure above, to access and manage the Zyxel Device from computer **A**, the Zyxel Device and switch **B**'s ports to which computer **A** and the Zyxel Device are connected should be in the same VLAN.

A Virtual Local Area Network (VLAN) allows a physical network to be partitioned into multiple logical networks. Devices on a logical network belong to one group. A device can belong to more than one group. With VLAN, a device cannot directly talk to or hear from devices that are not in the same group(s); the traffic must first go through a router.

VLAN also increases network performance by limiting broadcasts to a smaller and more manageable logical broadcast domain. In traditional switched environments, all broadcast packets go to each and every individual port. With VLAN, all broadcasts are confined to a specific broadcast domain.

IEEE 802.1Q Tag

The IEEE 802.1Q standard defines an explicit VLAN tag in the MAC header to identify the VLAN membership of a frame across bridges. A VLAN tag includes the 12-bit VLAN ID and 3-bit user priority. The VLAN ID associates a frame with a specific VLAN and provides the information that devices need to process the frame across the network.

Use this screen to configure the VLAN settings for your Zyxel Device. To access this screen, click **Configuration > Network > VLAN**.

The screen varies depending on whether the Zyxel Device has an extra Ethernet port (except the uplink port).

P Setting Via	AR AC Decovery	NOC Discovery		
LAN Settings				
Монаранная У Ис	40001	1.0000		
🗖 zenci ve vlas	0			
AR setting				
an Selling				
PTOR PACINGS	@-0008-010			
e acte	741		1942	
1 💡	1001		areas.	
in a lesse The	60 1 1 20 (1 <u>- 6</u>	449 (a)		Displaying 7 1 of
LAN Configuration				
and the second	ana na sa			
 sche 	i Norm	AR1	NR.OKCT -	
1 P	Sun -	enes II.	Kn3(2) -	
	M Chimmed and M	INT:		Deckyman - rod.

Figure 43 Configuration > Network > VLAN (for Zyxel Device with multiple Ethernet ports)

Figure 44 Configuration > Network > VLAN (for Zyxel Device with one Ethernet port)

P Setting VLAN	
VLAN Settings	
Monogeneral VIANID	1 (1×100)
🕅 Za NCIVE VLAN 🛛 😗	
	Appry Reset

Each field is described in the following table.

	Table 26	Configuration > Network >	VLAN
--	----------	---------------------------	------

LABEL	DESC RIPTIO N			
VLAN Settings				
Management VLAN ID	Enter a VLAN ID for the Zyxel Device.			
As Native VLAN	Select this option to treat this VLAN ID as a VLAN created on the Zyxel Device and not one assigned to it from outside the network.			
LAN Setting				
Port Setting				
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings. In some tables you can just click a table entry and edit it directly in the table. For those types of tables small red triangles display for table entries with changes that you have not yet applied.			
Activate/ Inactivate	To turn on an entry, select it and click Activate. To turn off an entry, select it and click Inactivate.			
#	This is the index number of the port.			
Status	This field indicates whether the port is enabled (a yellow bulb) or not (a gray bulb).			
Port	This field displays the name of the port.			

LABEL	DESC RIPIIO N				
PVID	This field displays the port number of the VLAN ID.				
VLAN Configuration					
Add	Click this to create a new entry. For features where the entry's position in the numbered list is important (features where the Zyxel Device applies the table's entries in order like the SSID for example), you can select an entry and click Add to create a new entry after the selected entry.				
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings. In some tables you can just click a table entry and edit it directly in the table. For those types of tables small red triangles display for table entries with changes that you have not yet applied.				
Remove	To remove an entry, select it and click Remove . The Zyxel Device confirms you want to remove it before doing so.				
Activate/ Inactivate	To turn on an entry, select it and click Ac tiva te . To turn off an entry, select it and click Inac tiva te .				
#	This is the index number of the VLAN ID.				
Status	This field indicates whether the VLAN is enabled (a yellow bulb) or not (a gray bulb).				
Name	This field displays the name of each VLAN.				
VID	This field displays the VLAN ID.				
Member	This field displays the VLAN membership to which the port belongs.				
Apply	Click Apply to save your changes back to the Zyxel Device.				
Reset	Click Reset to return the screen to its last-saved settings.				

Table 26 Configuration > Network > VLAN (continued)

9.4 Storm Control

Traffic storm control limits the number of broadcast and/or multicast packets the Zyxel Device receives on the ports. When the maximum number of allowable broadcast and/or multicast packets is reached, the subsequent packets are discarded. Enable this feature to reduce broadcast and/or multicast packets in your network.

Note: The maximum traffic rate can be changed using the CLI (see the CLI Reference Guide).

To access this screen, click **Configuration** > **Network** > **Storm Control**.

	911					
IP Setting	VIAN	Storm Control	AC Discove	ey b	ACC Discovery	
Starm Control 5	artino.					
🗉 Broasbart	Storm Canhol					
El Mutticost 3	Storm Control					
			Aupty	flexel -		

Figure 45 Configuration > Network > Storm Control

Each field is described in the following table.

LABEL	DESC RIPIIO N			
Broadcast Storm Control	Select the check box to enable broadcast storm control on the Zyxel Device. Enabling this will drop ingress broadcast traffic in the physical Ethernet port if it exceeds the maximum traffic rate.			
Multicast Storm Control	Select the check box to enable multicast storm control on the Zyxel Device. Enabling this will drop ingress multicast traffic in the physical Ethernet port if it exceeds the maximum traffic rate.			
Apply	Click Apply to save your changes back to the Zyxel Device.			
Reset	Click Reset to return the screen to its last-saved settings.			

Table 27 Configuration > Network > Storm Control

9.5 AC (APController) Discovery

This section discusses how to configure the Zyxel Device's AC Discovery settings. You can have the Zyxel Device managed by an AC on your network. When you do this, the Zyxel Device can be configured ONLY by the AC. See Section 8.9.1 on page 102 for more information on AC management.

Note: The AC Discovery settings are not available in all Zyxel Devices. See Section 0.4 on page 19 for more information.

If you want to return the Zyxel Device to function in standalone mode, you can do one of the two following options:

- Press the Reset button.
- Check the AC for the Zyxel Device's IP address and use FTP to upload the default configuration file to the Zyxel Device. You can get the configuration file at conf/system-default.conf. You must reboot the Zyxel Device after uploading the configuration file.

To access the Controller Discover screen, click Configuration > Network > AC Discovery.

Figure 46 Configuration > Network > AC Discovery

PSelling	VLAN	Storm Control	AC Discovery	НСС Въссияну
Discovery Setti	ng			
aj Auto				
 Access 				
		000.000		
Secondary data ACTP		000.000	(Optional)	
 Dode 				
			Apply	Repet

Each field is described in the following table.

LABEL	DESC RIPTIO N			
Discovery Setting				
Auto	Select this option to use DHCP option 138/DNS SRV record/Broadcast to get the AC's IP address. If the Zyxel Device and a Zyxel AC, such as the NXC2500 or NXC5500, are in the same subnet, it will be managed by the controller automatically.			
Manual	Select this option and enter the IP address of the AC manually. This is necessary when the AP Controller is not in the same subnet and you want it to manage the Zyxel Device.			
Primary / Secondary Static AC IP	Specify the primary and secondary IP address of the AC to which the Zyxel Device connects.			
Disable	Select this to manage the Zyxel Device using its own Web Configurator, neither managing nor being managed by other devices. Please note if an AP Controller is in the same subnet, you will need to click Disa ble if you do not want the Zyxel Device to be managed.			
Apply	Click Apply to save the information entered in this screen. If you select Auto or Manual , the AC uploads the firmware package for managed AP mode to the Zyxel Device and you cannot log in as the web configurator is disabled; you must manage the Zyxel Device through the AC on your network.			
Reset	Click \mathbf{Reset} to return the screen to its last-saved settings.			

Table 28 Configuration > Network > AC Discovery

9.6 NCC Discovery

You can manage the Zyxel Device through the Zyxel Nebula Control Center (NCC). Use this screen to configure the proxy server settings if the Zyxel Device is behind a proxy server.

To access this screen, click Configuration > Network > NCC Discovery.

Figure 47 Configuration > Network > NCC Discovery

a takey	NAM	them Control	AT DECEMP	MCC Discovery
Hebelo Contro	Center Sich	ME STATE OF THE		
House con	97979		new on in the star Complete the s	nad. And and product of the state of
Prosvi Prosvi PLA2	kort Ivini oction Ivini	n HCC		

Each field is described in the following table.

Table 29 C	Configuration > Network > NCC Discovery
------------	---

IABEL	DESC RIPTIO N		
Nebula Control Center Status			
Internet	This field displays whether the Zyxel Device can connect to the Internet.		
Nebula Connectivity	This field displays whether the Zyxel Device can connect to the Zyxel Nebula Control Center (NCC).		
Nebula Control Center Dis	covery Setting		
Enable	Select this option to turn on NCC discovery on the Zyxel Device. The Zyxel Device will try to discover the NCC and go into NCC management mode when it is connected to the Internet and has been registered in the NCC.		
	If NCC discovery is disabled, the Zyxel Device will not discover the NCC and remain in standalone operation.		
Use Proxy to Access NCC	If the Zyxel Device is behind a proxy server, you need to select this option and configure the proxy server settings so that the Zyxel Device can access the NCC through the proxy server.		
Proxy Server	Enter the IP address of the proxy server.		
Proxy Port	Enter the service port number used by the proxy server.		
Authentication	Select this option if the proxy server requires authentication before it grants access to the NCC.		
User Name	Enter your proxy user name.		
Password	Enter your proxy password.		
Apply	Click Apply to save your changes back to the Zyxel Device.		
Reset	Click $\operatorname{\mathbf{Reset}}$ to return the screen to its last-saved settings.		

C HAPTER 10 Wire less

10.1 Overview

This chapter discusses how to configure the wireless network settings in your Zyxel Device.

The following figure provides an example of a wireless network.





The wireless network is the part in the blue circle. In this wireless network, devices **A** and **B** are called wireless clients. The wireless clients use the access point (AP) to interact with other devices (such as the printer) or with the Internet. Your Zyxel Device is the AP.

10.1.1 What You Can Do in this Chapter

- The **AP Management** screen (Section 10.2 on page 76) allows you to manage the Zyxel Device's general wireless settings.
- The **Rogue AP** screen (Section 10.3 on page 79) allows you to assign APs either to the rogue AP list or the friendly AP list.
- The Load Balancing screen (Section 10.4 on page 83) allows you to configure network traffic load balancing between the APs and the Zyxel Device.
- The DCS screen (Section 10.4 on page 83) allows you to configure dynamic radio channel selection.
10.1.2 What You Need to Know

The following terms and concepts may help as you read this chapter.

Station / Wire less Client

A station or wireless client is any wireless-capable device that can connect to an AP using a wireless signal.

Dynamic Channel Selection (DCS)

Dynamic Channel Selection (DCS) is a feature that allows an AP to automatically select the radio channel which it broadcasts. For more information, see Section 10.5 on page 84.

Load Balancing (Wireless)

Wireless load balancing is the process where you limit the number of connections allowed on an wireless access point (AP) or you limit the amount of wireless traffic transmitted and received on it so the AP does not become overloaded.

10.2 APManagement

Use this screen to manage the Zyxel Device's general wireless settings. Click **Configuration > Wire less > APManagement** to access this screen.

	eting							
1 STATE	non objecte							
Barlio 11	and in							
1000	lo i Activate							
100 C 100 C 100 C	CY Hout 2			nti 42 🔘 Par				
Sales	i Swilk	det:	ang -	the second second second	0 1 0	8		
Mex C	utour Novae	30		Jahr Fr	94. 1			
MROID :	officia							
100	stic Make							
	dy eu i	OE						
and the second se	decisie	0						
	dischie -	0						
	dia dia	0						
Contraction of the second second	dar de	0						
	devide	0						
	decole	0						
- A - S	dischie	0						
10000	Acture		a mere	ost 42 👘 Res	0.000 0.000			
Sada	n Cith Nachdais		1.1.1	College College College College College				
Sada	CRACENSE Postlec	del s	1.1.1	the second second second	0 1 0			
Sado Sado			1.1.1	la Ubritza				
Sado Sado	PRWIEC	del s	1.1.1	the second second second				
Rado Rado Mox D Mick D	PRWIEC	del s	1.1.1	the second second second				
Rador Sador Vick D Maxant :	n Provilee Utole: Provien Schliegs	0 H	1.1.1	the second second second				
Saver Saver Mex D Mexado L L L	Former utour Review tolfing: Dic Nokin de ou i decole	дм. 30 О Н О	1.1.1	the second second second				
Save Save Mox D Mox D Maxim 1 2 1	Formier utour Rowern tolfing: Dic Molen de ou decole decole	он он о	1.1.1	the second second second				
Saver Saver Mex D Mex D	Provide utour Provide adding: dic holine de ou i diccole diccole diccole diccole	эм. ээ о н о о	1.1.1	the second second second				
Sate Sate Mor D Mor D 1 1 2 3 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	Provide utour Provide adding: Distribute de out descele descele descele descele	эм. ээ о н о о	1.1.1	the second second second				
Sado Sado Mac D Mac D S	Povrtex utour Novem adfing: Stat Notes de cui decole decole decole decole decole decole	ж. ээ	1.1.1	the second second second				
Sector Sector Morito Sector Se	Provide utour Provide adding: Distribute de out descele descele descele descele	эм. ээ о н о о о	1.1.1	the second second second				

Figure 44 Configuration > Wireless > AP Management

Table 28 Configuration > Wireless > AP Management

IABEL	DESC RIPTIO N
Radio 1 Setting	
Radio 1 Activate	Select the check box to enable the Zyxel Device's first (default) radio.

IABEL	DESC RIPIIO N
Radio 1 OP Mode	Select the operating mode for radio 1.
	APMode means the radio can receive connections from wireless clients and pass their data traffic through to the Zyxel Device to be managed (or subsequently passed on to an upstream gateway for managing).
	MON Mode means the radio monitors the broadcast area for other APs, then passes their information on to the Zyxel Device where it can be determined if those APs are friendly or rogue. If a radio is set to this mode it cannot receive connections from wireless clients (see Section 1.2.3 on page 15).
	Root AP means the radio acts as an AP and also supports the wireless connections with other APs (in repeater mode) to form a WDS (Wireless Distribution System) to extend its wireless network.
	Repeater means the radio can establish a wireless connection with other APs (in either root AP or repeater mode) to form a WDS.
Radio 1 Profile	Select the radio profile the radio uses.
	Note: You can only apply a 2.4G AP radio profile to radio 1. Otherwise, the first radio will not be working.
Radio 1 WDS Profile	This field is available only when the radio is in Root AP or Repeater mode.
	Select the WDS profile the radio uses to connect to a root AP or repeater.
Enable WDS Wireless	This field is available only when the radio is in Root AP or Repeater mode.
Bridging	A Wireless Distribution System is a wireless connection between two or more APs. Enable this to establish wireless links with other APs. See Section 0.2 on page 13 for more information.
	Only the models below support this feature at the time of writing:
	 WAC6103D-I WAC6303D-S WAC6502D-E WAC502D-S WAC6503D-S WAC6552D-S WAC6553D-E WAX610D WAX650S
	Note: You must enable the same wireless security settings on the Zyxel Device and on all wireless clients that you want to associate with it.
Uplink Selection	This field is available only when the radio is in $\mathbf{Repeater}$ mode.
Mode	Select AUIO to have the Zyxel Device automatically use the settings in the applied WDS profile to connect to a root AP or repeater.
	Select Manual to have the Zyxel Device connect to the root AP or repeater with the MAC address specified in the Radio 1 Uplink MAC Address field.
Max Output Power	Enter the maximum output power (between 0 to 30 dBm) of the Zyxel Device in this field. If there is a high density of APs in an area, decrease the output power of the Zyxel Device to reduce interference with other APs.
	Note: Reducing the output power also reduces the Zyxel Device's effective broadcast radius.
MBSSID Settings	

 Table 28
 Configuration > Wireless > AP Management (continued)

IABEL	DESC RIPIIO N
Edit	Click Edit icon (i) to open a screen where you can modify the entry's settings. In some tables you can just click a table entry and edit it directly in the table. For those types of tables small red triangles display for table entries with changes that you have not yet applied.
#	This field shows the index number of the SSID
SSID Profile	This field displays the SSID profile that is associated with the radio profile.
Radio 2 Setting	
Radio 2 Activate	This displays if the Zyxel Device has a second radio.
	Select the check box to enable the Zyxel Device's second radio.
Radio 2 OP Mode	This displays if the Zyxel Device has a second radio. Select the operating mode for radio 2.
	APMode means the radio can receive connections from wireless clients and pass their data traffic through to the Zyxel Device to be managed (or subsequently passed on to an upstream gateway for managing).
	MON Mode means the radio monitors the broadcast area for other APs, then passes their information on to the Zyxel Device where it can be determined if those APs are friendly or rogue. If a radio is set to this mode it cannot receive connections from wireless clients (see Section 1.2.3 on page 15).
	Root AP means the radio acts as an AP and also supports the wireless connections with other APs (in repeater mode) to form a WDS to extend its wireless network.
	Repeater means the radio can establish a wireless connection with other APs (in either root AP or repeater mode) to form a WDS.
Radio 2 Profile	This displays if the Zyxel Device has a second radio. Select the radio profile the radio uses.
	Note: You can only apply a 5G AP radio profile to radio 2. Otherwise, the second radio will not be working.
Radio 2 WDS Profile	This field is available only when the radio is in Root AP or Repeater mode.
	Select the WDS profile the radio uses to connect to a root AP or repeater.
Enable WDS Wireless	This field is available only when the radio is in Root AP or Repeater mode.
Bridging	A Wireless Distribution System is a wireless connection between two or more APs. Enable this to establish wireless links with other APs. See Section 0.2 on page 13 for more information.
	Only the models below support this feature at the time of writing:
	 WAC6103D-I WAC6303D-S WAC6502D-E WAC502D-S WAC6503D-S WAC6552D-S WAC6553D-E WAX610D WAX650S
	Note: You must enable the same wireless security settings on the Zyxel Device and on all wireless clients that you want to associate with it.
Uplink Selection	This field is available only when the radio is in ${f Repeater}$ mode.
Mode	Select AUIO to have the Zyxel Device automatically use the settings in the applied WDS profile to connect to a root AP or repeater.
	Select Manual to have the Zyxel Device connect to the root AP or repeater with the MAC address specified in the Radio 2 Uplink MAC Address field.

Table 28 Configuration > Wireless > AP Management (continued)

NWA/WAC/WAX Series User's Guide

LABEL	DESC RIPTIO N
Max Output Power	Enter the maximum output power (between 0 to 30 dBm) of the Zyxel Device in this field. If there is a high density of APs in an area, decrease the output power of the Zyxel Device to reduce interference with other APs.
	Note: Reducing the output power also reduces the Zyxel Device's effective broadcast radius.
MBSSID Settings	
Edit 💼	Click Edit (i) to open a screen where you can modify the entry's settings. In some tables you can just click a table entry and edit it directly in the table. For those types of tables small red triangles display for table entries with changes that you have not yet applied.
#	This field shows the index number of the SSID
SSID Profile	This field shows the SSID profile that is associated with the radio profile.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 28
 Configuration > Wireless > AP Management (continued)

10.3 Rogue AP

Use this screen to enable **Rogue AP Detection** and import/export a rogue or friendly AP list in a txt file. Click **Configuration > Wire less > Rogue AP** to access this screen.

Rogue APs

A rogue AP is a wireless access point operating in a network's coverage area that is not under the control of the network administrator, and which can potentially open up holes in a network's security.

In the following example, a corporate network's security is compromised by a rogue AP (\mathbf{RG}) set up by an employee at his workstation in order to allow him to connect his notebook computer wirelessly (\mathbf{A}). The company's legitimate wireless network (the dashed ellipse \mathbf{B}) is well-secured, but the rogue AP uses inferior security that is easily broken by an attacker (\mathbf{X}) running readily available encryption-cracking software. In this example, the attacker now has access to the company network, including sensitive data stored on the file server (\mathbf{C}).



Figure 45 Rogue AP Example

Friendly APs

If you have more than one AP in your wireless network, you should also configure a list of "friendly" APs. Friendly APs are wireless access points that you know are not a threat. It is recommended that you export (save) your list of friendly APs often, especially if you have a network with a large number of access points. Exported lists show MAC addresses in txt file format separated by line breaks.

Rogue AP Detection

This feature allows the Zyxel Device to monitor the WiFi signals for other wireless APs (see also Section 1.2.3 on page 15). Detected APs will appear in the **Monitor > Wireless > Detected Device** screen, where the Zyxel Device will label APs with the criteria you select in **Suspected Rogue AP Classification Rule** as a suspected rogue. The APs which you mark as either rogue or friendly APs in the **Monitor > Wireless > Detected Device** screen will appear in the **Wireless > Rogue AP** screen. See Section 1.4 on page 18 to know which models support **Rogue AP Detection**.

Note: Enabling **Rogue AP Detection** might affect the performance of wireless clients associated with the Zyxel Device.

🛱 ever 🖉 🖽 🦄 👘	divers.		
1 Rule-	MAR Address	Description	
l avgus ep	09:00:00 01:23:45	- regione reaction	
egve Af Littimporti	and the second	Brows Brows	1947-14 (1947)
egve Af Utilimportu Helficin: - Salacifi	g/Experting Be auth for heigher Afrikk	(Brows) Specify	-
egve Af Littimporti	g/Experting Be auth for heigher Afrikk	(Brows) Instanting (Equation	-

Figure 46 Configuration > Wireless > Rogue AP (for Zyxel Devices that support Monitor mode)

Figure 47 Configuration > Wireless > Rogue AP (for Zyxel Devices that support Rogue AP Detection)

Requiry Streaty AP List			
togue AP Detection Self	Ing		
🗵 brobie Rogue AP D	efection		
uspected Rogue A* Ch	andfacetion Rule		
🖉 Weal: Security (Ope	en. Mitt MittAvriss)		
V HiddenssiC			
g, SSID Services			
@Add Part There	day.		
1 SSID Konnete			
1 (4:3)			
ageo.Filondiy &P.Ucl			
QANK ZTHE BE			
R DAR-	Mail Address	"Nearing that	
1 Mendle-sp	(AKG): 97 (TD (DC N)		
P Managar	10:40:7.501:33:15	example	
11.1 Page of 1	i i die 20 ⊻ der-		Displaying 1 - 5 of 1
ingso &P list Importing	Expating		
Report Bearing	n put dis Rogas, AP ad		10
tendly AF Uniterpartic	s/Reporting		
Tin Troth [Fright of I]	n path Writinnay of the	distant [mastic] [topotts	a.,
		xpplv teser	

Table 29 Con	figuration > \	Wireless >	Rogue AP
--------------	----------------	------------	----------

IABEL	DESC RIPTIO N
Rogue AP Detection Set	ting
Enable Rogue AP Detection	Select this check box to detect Rogue APs in the network.
Suspected Rogue AP Classification Rule	Select the check boxes (Weak Security (Open, WEP, WPA-PSK), Hidden SSID, SSID Keyword) of the characteristics an AP should have for the Zyxel Device to mark it as a Rogue AP.
Add	Click this to add an SSID Keyword.
Edit	Select an SSID Keyword and click this button to modify it.
Remove	Select an existing SSID keyword and click this button to delete it.
#	This is the SSID Keyword's index number in this list.
SSID Keyword	This field displays the SSID Keyword.
Rogue/Friendly AP List	•
Add	Click this button to add an AP to the list and assign it either friendly or rogue status.
Edit	Select an AP in the list to edit and reassign its status.
Remove	Select an AP in the list to remove.
#	This field is a sequential value, and it is not associated with any interface.
Role	This field indicates whether the selected AP is a rogue-ap or a friendly-ap . To change the AP's role, click the Edit button.
MAC Address	This field indicates the AP's radio MAC address.
Description	This field displays the AP's description. You can modify this by clicking the Edit button.
Rogue/Friendly AP List Importing/Exporting	These controls allow you to export the current list of rogue and friendly APs or import existing lists.
File Path / Browse / Importing	Enter the file name and path of the list you want to import or click the Browse button to locate it. Once the File Path field has been populated, click Importing to bring the list into the Zyxel Device.
	You need to wait a while for the importing process to finish.
Exporting	Click this button to export the current list of either rogue APs or friendly APS.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click $\operatorname{\mathbf{Reset}}$ to return the screen to its last-saved settings.

10.3.1 Add/Edit Rogue/Friendly List

Click Add or select an AP and click the Edit button in the Configuration > Wire less > Rogue AP table to display this screen.

Figure 48 Configuration	ion > wireless > R	ogue AP > Add/Ed	ait Rogue/i
© Edit Rogue/Friendly A	P List	$2 \times$	
MAC:			
Description:		(Optional)	
Role:	Regree AP	Crivenally AP	
	ĺ	Cancel 30	

Figure 48 Configuration > Wireless > Rogue AP > Add/Edit Rogue/Friendly AP List

LABEL	DESC RIPIIO N
MAC	Enter the MAC address of the AP you want to add to the list. A MAC address is a unique hardware identifier in the following hexadecimal format: xx:xx:xx:xx:xx where xx is a hexadecimal number separated by colons.
Description	Enter up to 60 characters for the AP's description. Spaces and underscores are allowed.
Role	Select either Rogue AP or Friendly AP for the AP's role.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to close the window with changes unsaved.

Table 30 Configuration > Wireless > Rogue AP > Add/Edit Rogue/Friendly AP List

10.4 Load Balancing

Use this screen to configure wireless network traffic load balancing between the APs on your network (see Load Balancing on page 125). Click **Configuration > Wire less > Load Balancing** to access this screen.

Figure 49 Configuration > Wireless > Load Balancing

Load Balancing					
Load Italancing Configuration					
🔄 Engble Logd Br	alancing				
Mode:	By Station Number 🛛 🗶				
Max Station Nu	miser: 127 (1-127)				
 Discussociate 	e station when overloaded				
	Apply Reset				

|--|

IABEL	DESC RIPIIO N		
Enable Load	Select this to enable load balancing on the Zyxel Device.		
Balancing	Use this section to configure wireless network traffic load balancing between the managed APs in this group.		
Mode	Select a mode by which load balancing is carried out.		
	Select By Station Number to balance network traffic based on the number of specified stations connected to the Zyxel Device.		
	Select By Tha ffic Level to balance network traffic based on the volume generated by the stations connected to the Zyxel Device.		
	Select By Smart Classmom to balance network traffic based on the number of specified stations connected to the Zyxel Device. The Zyxel Device ignores association request and authentication request packets from any new station when the maximum number of stations is reached.		
	If you select By Station Number or By Traffic Level , once the threshold is crossed (either the maximum station numbers or with network traffic), the Zyxel Device delays association request and authentication request packets from any new station that attempts to make a connection. This allows the station to automatically attempt to connect to another, less burdened AP if one is available.		
Max Station Number	Enter the threshold number of stations at which the Zyxel Device begins load balancing its connections.		
Traffic Level	Select the threshold traffic level at which the Zyxel Device begins load balancing its connections (Iow , Medium , High). The maximum bandwidth allowed for each level is:		
	 Low - 11 Mbps Medium - 23 Mbps High - 35M bps 		
Disassociate station when overloaded	This function is enabled by default and the disassociation priority is always Signal Strength when you set Mode to By Smart Classroom .		
	Select this option to disassociate wireless clients connected to the AP when it becomes overloaded. If you do not enable this option, then the AP simply delays the connection until it can afford the bandwidth it requires, or it transfers the connection to another AP within its broadcast radius.		
	The disassociation priority is determined automatically by the Zyxel Device and is as follows:		
	Idle Timeout - Devices that have been idle the longest will be kicked first. If none of the connected devices are idle, then the priority shifts to Signal Strength.		
	• Sig na 1 Stre ng th - Devices with the weakest signal strength will be kicked first.		
	Note: If you enable this function, you should ensure that there are multiple APs within the broadcast radius that can accept any rejected or kicked wireless clients; otherwise, a wireless client attempting to connect to an overloaded AP will be disassociated permanently and never be allowed to connect.		
Apply	Click Apply to save your changes back to the Zyxel Device.		

10.4.1 Disassociating and Delaying Connections

When your AP becomes overloaded, there are two basic responses it can take. The first one is to "delay" a client connection. This means that the AP withholds the connection until the data transfer throughput is lowered or the client connection is picked up by another AP. If the client is picked up by another AP then the original AP cannot resume the connection.

For example, here the AP has a balanced bandwidth allotment of 6 Mbps. If laptop **R** connects and it pushes the AP over its allotment, say to 7 Mbps, then the AP delays the red laptop's connection until it can afford the bandwidth or the laptop is picked up by a different AP with bandwidth to spare.



The second response your AP can take is to disassociate with clients that are pushing it over its balanced bandwidth allotment.



Connections are cut based on either **idle timeout** or **signal strength**. The Zyxel Device first looks to see which devices have been idle the longest, then starts kicking them in order of highest idle time. If no connections are idle, the next criteria the Zyxel Device analyzes is signal strength. Devices with the weakest signal strength are kicked first.

10.5 DCS

Use this screen to configure dynamic radio channel selection (see Dynamic Channel Selection (DCS) on page 76). Click **Configuration > Wire less > DCS** to access this screen.

gure 52 Configuration > Wileless > DC3
DCS
General Sellings
DCS Now
Apply Resct

Figure 52 Configuration > Wireless > DCS

Each field is described in the following table.

LABEL	DESC RIPTIO N
DCS Now	Click this to have the Zyxel Device scan for and select an available channel immediately.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

Table 32 Configuration > Wireless > DCS

10.6 Technical Reference

The following section contains additional technical information about the features described in this chapter.

Dynamic Channel Selection

When numerous APs broadcast within a given area, they introduce the possibility of heightened radio interference, especially if some or all of them are broadcasting on the same radio channel. If the interference becomes too great, then the network administrator must open his AP configuration options and manually change the channel to one that no other AP is using (or at least a channel that has a lower level of interference) in order to give the connected stations a minimum degree of interference. Dynamic channel selection frees the network administrator from this task by letting the AP do it automatically. The AP can scan the area around it looking for the channel with the least amount of interference.

In the 2.4 GHz spectrum, each channel from 1 to 13 is broken up into discrete 22 MHz segments that are spaced 5 MHz apart. Channel 1 is centered on 2.412 GHz while channel 13 is centered on 2.472 GHz.



Three channels are situated in such a way as to create almost no interference with one another if used exclusively: 1, 6 and 11. When an AP broadcasts on any of these 3 channels, it should not interfere with neighboring APs as long as they are also limited to same trio.





However, some regions require the use of other channels and often use a safety scheme with the following four channels: 1, 4, 7 and 11. While they are situated sufficiently close to both each other and the three so-called "safe" channels (1,6 and 11) that interference becomes inevitable, the severity of it is dependent upon other factors: proximity to the affected AP, signal strength, activity, and so on.

Finally, there is an alternative four channel scheme for ETSI, consisting of channels 1, 5, 9, 13. This offers significantly less overlap that the other one.



Figure 55 An Alternative Four-Channel Deployment

Load Balancing

Because there is a hard upper limit on an AP's wireless bandwidth, load balancing can be crucial in areas crowded with wireless users. Rather than let every user connect and subsequently dilute the

available bandwidth to the point where each connecting device receives a meager trickle, the load balanced AP instead limits the incoming connections as a means to maintain bandwidth integrity.

There are three kinds of wireless load balancing available on the Zyxel Device:

Load balancing by station number limits the number of devices allowed to connect to your AP. If you know exactly how many stations you want to let connect, choose this option.

For example, if your company's graphic design team has their own AP and they have 10 computers, you can load balance for 10. Later, if someone from the sales department visits the graphic design team's offices for a meeting and he tries to access the network, his computer's connection is delayed, giving it the opportunity to connect to a different, neighboring AP. If he still connects to the AP regardless of the delay, then the AP may boot other people who are already connected in order to associate with the new connection.

Load balancing by smart classmom also limits the number of devices allowed to connect to your AP. But any new connections will be just rejected when the AP is overloaded.

Load balancing by traffic level limits the number of connections to the AP based on maximum bandwidth available. If you are uncertain as to the exact number of wireless connections you will have then choose this option. By setting a maximum bandwidth cap, you allow any number of devices to connect as long as their total bandwidth usage does not exceed the configured bandwidth cap associated with this setting. Once the cap is hit, any new connections are rejected or delayed provided that there are other APs in range.

Imagine a coffee shop in a crowded business district that offers free wireless connectivity to its customers. The coffee shop owner can't possibly know how many connections his AP will have at any given moment. As such, he decides to put a limit on the bandwidth that is available to his customers but not on the actual number of connections he allows. This means anyone can connect to his wireless network as long as the AP has the bandwidth to spare. If too many people connect and the AP hits its bandwidth cap then all new connections must basically wait for their turn or get shunted to the nearest identical AP.

C HAPTER 11 Blue to o th

11.1 Overview

Use this screen to configure the iBeacon advertising settings for the Zyxel Device that supports Bluetooth Low Energy (BLE). Bluetooth Low Energy, which is also known as Bluetooth Smart, transmits less data over a shorter distance but consumes less power than classic Bluetooth.

On the WAC5302D-S, you need to attach a supported BLE USB dongle to its USB port to have the AP act as a beacon to broadcast packets. Contact Zyxel customer support if you are not sure whether your BLE USB dongle is compatible with the Zyxel Device.

11.1.1 What You Need To Know

iBeacon is Apple's communication protocol on top of Bluetooth Low Energy wireless technology. Beacons (Bluetooth radio transmitters) or BLE enabled devices broadcast packets to every device around it to announce their presence. Advertising packets contain their iBeacon ID, which consists of the Universally Unique Identifier (UUID), major number, and minor number. These packets also contain a TX (transmit) power measured at a reference point, which is used to approximate a device's distance from the beacon. The UUID can be used to identify a service, a device, a manufacturer or an owner. The 2-byte major number is to identify and distinguish a group, and the 2-byte minor number is to identify and distinguish an individual.

For example, a company can set all its beacons to share the same UUID. The beacons in a particular branch uses the same major number, and each beacon in a branch can have its own minor number.

	COMPANY A		
	BRAN	СНХ	BRANCHY
	BEACON 1	BEACON 2	BEACON 3
UUID	EBAECFAF-DFE0-4039-BE5A-F030EED4303C		
Major	10	10	20
Minor	1	2	1

Developers can create apps that respond to the iBeacon ID that your Zyxel Device broadcasts. An app that is associated with the Zyxel Device's iBeacon ID can measure the proximity of a customer to a beacon. This app can then push messages or trigger prompts and actions based on this information. This allows you to send highly contextual and highly localized advertisements to customers.

130

11.2 Blue to oth Advertising Settings

The Zyxel Device communicates with another BLE enabled device for advertisements. Use this screen to configure up to five beacon IDs to be included in the advertising packet.

To access this screen, click **Configuration > Blue tooth > Advertising Settings**.

Figure 74 Configuration > Bluetooth > Advertising Settings

26d 9 actors 9 • setu	9. D	No:	Minor	
		•/	¥:	
• 😨		Ø)	p	
1. 10			0	
409		9) 8	μ	
5. (梁)		8. C	n -	
1.1. 1999 - 2311	11 100×100 - 101 are		1 is	antra 1 Siat

The following table describes the labels in this screen.

IABEL	DESC RIPTIO N	
Edit	Click this to edit the selected entry.	
Activate	To turn on an entry, select it and click Ac tiva te .	
Inactivate	To turn off an entry, select it and click Inac tiva te .	
#	This field is a sequential value, and it is not associated with a specific entry.	
Status	This field shows whether or not the entry is activated.	
	A yellow bulb signifies that this rule is active. A gray bulb signifies that this rule is not active.	
UUID	This field indicates the UUID to be included in the Bluetooth advertising packets.	
Major	This field indicates the major number to be included in the Bluetooth advertising packets.	
Minor	This field indicates the minor number to be included in the Bluetooth advertising packets.	
Apply	Click Apply to save your changes back to the Zyxel Device.	
Reset	Click Reset to return the screen to its last-saved settings.	

Table 49 Configuration > Bluetooth > Advertising Settings

11.2.1 Edit Advertising Settings

Select an entry in the **Configuration > Blue tooth > Advertising Settings** screen and click the **Edit** icon to open the **Edit Advertising** screen. Use this screen to configure the beacon ID in the Bluetooth advertising packets.

Figure 75 Configuration > Bluetooth > Advertising Settings > Edit

deceloing Sci	leg	
E Actione Ulin		Constra ne U.D.
Majon Misel	2 2	(D-4333) (D-6333)

The following table describes the labels in this screen.
--

Table 50 Configuration > Bluetooth > Advertising Settings > Edit

IABEL	DESC RIPIIO N	
Activate	Select this option to enable the advertising settings.	
UUID	To specify a UUID for the Zyxel Device's beacon ID, enter 32 hexadecimal digits in the range of "A-F", "a-f" and "0-9", split into five groups separated by hyphens (-). The UUID format is as follows: xxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxx (8-4-4-12)	
Generate new UUID	Click this button to have the Zyxel Device generate a new UUID automatically.	
Major	Enter an integer from 0 to 65535 as the major value to identify the group to which the beacon belongs.	
Minor	Enter an integer from 0 to 65535 as the minor value to identify the individual beacon.	
ОК	Click OK to save your changes back to the Zyxel Device.	
Cancel	Click Cancel to exit this screen without saving your changes.	

C HAPTER 11 User

11.1 Overview

This chapter describes how to set up user accounts and user settings for the Zyxel Device.

11.1.1 What You Can Do in this Chapter

- The User screen (see Section 11.2 on page 87) provides a summary of all user accounts.
- The Setting screen (see Section 11.3 on page 89) controls default settings, login settings, lockout settings, and other user settings for the Zyxel Device.

11.1.2 What You Need To Know

The following terms and concepts may help as you read this chapter.

UserAccount

A user account defines the privileges of a user logged into the Zyxel Device. User accounts are used in controlling access to configuration and services in the Zyxel Device.

UserTypes

These are the types of user accounts the Zyxel Device uses.

TYPE	ABILITIES	LOGIN METHOD(S)
Admin Users		
admin	Change Zyxel Device configuration (web, CLI)	WWW, TELNET, SSH, FTP
limited-admin	Look at Zyxel Device configuration (web, CLI)	WWW, TELNET, SSH
	Perform basic diagnostics (CLI)	
Access Users		
User	Used for the embedded RADIUS server and SNMPv3 user access	
	Browse user-mode commands (CLI)	

Table 32 Types of User Accounts

Note: The default **admin** account is always authenticated locally, regardless of the authentication method setting.

11.2 User Summary

The User screen provides a summary of all user accounts. To access this screen click Configuration > Object > User.

Figure 53 Configuration > Object > User

a san Gerrar 🖀 na ha an hai	Pport follower:	
te: Isethiome	Lan Syna	Description
ex.min	ecimie	Administration account.
1 1 10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	A M MARK	Chaptering 1 - 3 of a

The following table describes the labels in this screen.

LABEL	DESC RIPIIO N	
Add	Click this to create a new entry.	
Edit	Double-click an entry or select it and click \mathbf{Edit} to open a screen where you can modify the entry's settings.	
Remove	To remove an entry, select it and click $\operatorname{\mathbf{Remove}}$. The Zyxel Device confirms you want to remove it before doing so.	
Object Reference	Select an entry and click Object Reference to open a screen that shows which settings use the entry.	
#	This field is a sequential value, and it is not associated with a specific user.	
User Name	This field displays the user name of each user.	
User Type	This field displays type of user this account was configured as.	
	 admin - this user can look at and change the configuration of the Zyxel Device limited-admin - this user can look at the configuration of the Zyxel Device but not to change it user - this user has access to the Zyxel Device's services but cannot look at the configuration 	
Description	This field displays the description for each user.	

Table 33 Configuration > Object > User

11.2.1 Add/Edit User

The UserAdd/Edit screen allows you to create a new user account or edit an existing one.

11.2.1.1 Rules for User Names

Enter a user name from 1 to 31 characters.

The user name can only contain the following characters:

- Alphanumeric A-z 0-9 (there is no unicode support)
- _ [underscores]

• - [dashes]

The first character must be alphabetical (A-Z a-z), an underscore (_), or a dash (-). Other limitations on user names are:

- User names are case-sensitive. If you enter a user 'bob' but use 'BOB' when connecting via CIFS or FTP, it will use the account settings used for 'BOB' not 'bob'.
- User names have to be different than user group names.
- Here are the reserved user names:

•	adm	•	admin	•	any	•	bin	•	daemon
•	debug	•	devicehaecived	•	ftp	•	games	•	halt
•	ldap-users	•	lp	•	mail	•	news	•	nobody
•	operator	•	radius-users	•	root	•	shutdown	•	sshd
•	sync	•	uucp	•	zyxel				

To access this screen, go to the User screen, and click Add or Edit.

Care Norma	udnall	
Coor Type:		
hasayosi.	44444	
Becypa	1111	
Decolotion		

Figure 54 Configuration > Object > User > Add/Edit A User

The following table describes the labels in this screen.

Table 34 Configuration > User > User > Add/Edit A User

IABEL	DESC RIPIIO N	
User Name	Type the user name for this user account. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive. User names have to be different than user group names, and some words are reserved.	
User Type	Select what type of user this is. Choices are:	
	• admin - this user can look at and change the configuration of the Zyxel Device	
	 limite d-admin - this user can look at the configuration of the Zyxel Device but not to change it 	
	• user-this is used for embedded RADIUS server and SNMPv3 user access	
Password	Enter the password of this user account. It can consist of 4 - 63 alphanumeric characters.	
Retype	Re-enter the password to make sure you have entered it correctly.	
Description	Enter the description of each user, if any. You can use up to 60 printable ASCII characters. Default descriptions are provided.	

NWA/WAC/WAX Series User's Guide

IABEL	DESC RIPTIO N
Authentication	This field is not available if the user type is user .
Timeout Settings	If you want to set authentication timeout to a value other than the default settings, select Use Manual Settings then fill your preferred values in the fields that follow.
Lease Time	This field is not available if the user type is user .
	Enter the number of minutes this user has to renew the current session before the user is logged out. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited. Admin users renew the session every time the main screen refreshes in the Web Configurator.
Reauthentication Time	This field is not available if the user type is user . Type the number of minutes this user can be logged into the Zyxel Device in one session before the user has to log in again. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited. Unlike Lease Time , the user has no opportunity to renew the session without logging out.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 34 Configuration > User > User > Add/Edit A User (continued)

11.3 Setting

This screen controls default settings, login settings, lockout settings, and other user settings for the Zyxel Device.

To access this screen, login to the Web Configurator, and click Configuration > Object > User > Setting.

2				
han Time and Soffings				
	Lege little		equinerric chart lime	
	1440		90	
h	1410	-	avu -	
	5			
al El Hilanse 🕅 🖉	STR.			Displaying 1. 2 of 2.
er of similitaneous logic Der per old ministration	ns for extraining souther	occourt (I-rati		
22				
el y timi				
eposti -	(j.)	(1999)		
	20	[1.65525.mms.k.		
	for Timeoul Setting: In Sul F. Si Jaka X. [9] er of Smithereousiopo geogen adminutration ge	For Time out Softing: Lease Illine 1 010 In 1 010 1 1 1 1 1 Jack N (e) and er of time tables as logisms for estimicity attac er of time tables as logisms for estimicity attac er of time tables as logisms for estimicity attact er of time tables at a statement of the tables at a statement of tables at a	For Time of Soffrig:	For Time out Softing: Leave Units 140 1990 In 1410 (990) 11 + 41 (size N g) and er of time tableous logours for coincidence as over (rear per domination in 1 (1400) 20 et y time

Figure 55 Configuration > Object > User > Setting

The following table describes the labels in this screen.

IABEL	DESC RIPTIO N
User Default Setting	
Default Authentication Timeout Settings	These authentication timeout settings are used by default when you create a new user account. They also control the settings for any existing user accounts that are set to use the default settings. You can still manually configure any user account's authentication timeout settings.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
#	This field is a sequential value, and it is not associated with a specific entry.
User Type	 These are the kinds of user account the Zyxel Device supports. admin - this user can look at and change the configuration of the Zyxel Device limited-admin - this user can look at the configuration of the Zyxel Device but not to change it user - this is used for embedded RADIUS server and SNMPv3 user access
Lease Time	This is the default lease time in minutes for each type of user account. It defines the number of minutes the user has to renew the current session before the user is logged out. Admin users renew the session every time the main screen refreshes in the Web Configurator.
Reauthentication Time	This is the default reauthentication time in minutes for each type of user account. It defines the number of minutes the user can be logged into the Zyxel Device in one session before having to log in again. Unlike Lease Time , the user has no opportunity to renew the session without logging out.

Table 35 Configuration > Object > User > Setting

IABEL	DESC RIPIIO N
Limit the number of simultaneous logons for administration account	Select this check box if you want to set a limit on the number of simultaneous logins by admin users. If you do not select this, admin users can login as many times as they want at the same time using the same or different IP addresses.
Maximum number per administration account	This field is effective when Limit for a dministration account is checked. Type the maximum number of simultaneous logins by each admin user.
User Lockout Settings	
Enable logon retry limit	Select this check box to set a limit on the number of times each user can login unsuccessfully (for example, wrong password) before the IP address is locked out for a specified amount of time.
Maximum retry count	This field is effective when Enable logon retry limit is checked. Type the maximum number of times each user can login unsuccessfully before the IP address is locked out for the specified lockout period . The number must be between 1 and 99.
Lockout period	This field is effective when Enable logon re try limit is checked. Type the number of minutes the user must wait to try to login again, if logon re try limit is enabled and the maximum re try count is reached. This number must be between 1 and 65,535 (about 45.5 days).
Apply	Click Apply to save the changes.
Reset	Click Reset to return the screen to its last-saved settings.

Table 35 Configuration > Object > User > Setting (continued)

11.3.1 Edit User Authentication Time out Settings

This screen allows you to set the default authentication timeout settings for the selected type of user account. These default authentication timeout settings also control the settings for any existing user accounts that are set to use the default settings. You can still manually configure any user account's authentication timeout settings.

To access this screen, go to the **Configuration > Object > User > Setting** screen, select one of the **Default Authentication Timeout Settings** entry and click the **Edit** icon.

Figure 56 User > Setting > Edit User Authentication Timeout Settings

Edit User Authenticati	on Timeou	ut Settings 👘 🖭
Loss Types	admin	
Leois Ilms	1930	(h. Linki, colorados, fuit, ani colorada
Recuthent parton lime.	17.40	0-1116 minutes, 0 is uniformative
Recufrent parton (the	1740	3-115 minutes, 0 & uni robede
		Standal

The following table describes the labels in this screen.

LABEL	DESC RIPTIO N	
User Type	This read-only field identifies the type of user account for which you are configuring the default settings.	
	 admin - this user can look at and change the configuration of the Zyxel Device. limited-admin - this user can look at the configuration of the Zyxel Device but not to change it. 	
Lease Time	Enter the number of minutes this type of user account has to renew the current session before the user is logged out. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited.	
	Admin users renew the session every time the main screen refreshes in the Web Configurator. Access users can renew the session by clicking the Renew button on their screen. If you allow access users to renew time automatically, the users can select this check box on their screen as well. In this case, the session is automatically renewed before the lease time expires.	
Reauthentication Time	Type the number of minutes this type of user account can be logged into the Zyxel Device in one session before the user has to log in again. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited. Unlike Lease Time , the user has no opportunity to renew the session without logging out.	
OK	Click OK to save your changes back to the Zyxel Device.	
Cancel	Click Cancel to exit this screen without saving your changes.	

C HAPTER 12 AP Pro file

12.1 Overview

This chapter shows you how to configure preset profiles for the Zyxel Device.

12.1.1 What You Can Do in this Chapter

- The Radio screen (Section 12.2 on page 94) creates radio configurations that can be used by the APs.
- The **SSID** screen (Section 12.3 on page 100) configures three different types of profiles for your networked APs.

12.1.2 What You Need To Know

The following terms and concepts may help as you read this chapter.

Wire le ss Pro file s

At the heart of all wireless AP configurations on the Zyxel Device are profiles. A profile represents a group of saved settings that you can use across any number of connected APs. You can set up the following wireless profile types:

- Radio This profile type defines the properties of an AP's radio transmitter. You can have a maximum of 64 radio profiles on the Zyxel Device.
- **SSID** This profile type defines the properties of a single wireless network signal broadcast by an AP. Each radio on a single AP can broadcast up to 8 SSIDs. You can have a maximum of 64 SSID profiles on the Zyxel Device.
- Security This profile type defines the security settings used by a single SSID. It controls the encryption method required for a wireless client to associate itself with the SSID. You can have a maximum of 64 security profiles on the Zyxel Device.
- MAC Filte ring This profile provides an additional layer of security for an SSID, allowing you to block access or allow access to that SSID based on wireless client MAC addresses. If a client's MAC address is on the list, then it is either allowed or denied, depending on how you set up the MAC Filter profile. You can have a maximum of 64 MAC filtering profiles on the Zyxel Device.
- Layer 2 Isolation This profile defines the MAC addresses of the devices that you want to allow the associated wireless clients to have access to when layer-2 isolation is enabled.

SSID

The SSID (Service Set IDentifier) is the name that identifies the Service Set with which a wireless station is associated. Wireless stations associating to the access point (AP) must have the same SSID. In other words, it is the name of the wireless network that clients use to connect to it.

93

WEP

WEP (Wired Equivalent Privacy) encryption scrambles all data packets transmitted between the AP and the wireless stations associated with it in order to keep network communications private. Both the wireless stations and the access points must use the same WEP key for data encryption and decryption.

WPA2

WPA2 (IEEE 802.11i) is a wireless security standard that defines stronger encryption, authentication and key management than WPA. Key differences between WPA2 and WEP are improved data encryption and user authentication.

IEEE 802.1x

The IEEE 802.1x standard outlines enhanced security methods for both the authentication of wireless stations and encryption key management. Authentication is done using an external RADIUS server.

IEEE 802.11k/v Assisted Roaming

IEEE 802.11k is a standard for radio resource management of wireless LANs, which allows clients to request neighbor lists from the connected AP and discover the best available AP when roaming. An 802.11k neighbor list can contain up to six BSSIDs with the highest RCPI (Received Channel Power Indicator) value in both bands (5 GHz and 2.4 GHz, in the ratio of 4:2).

The IEEE 802.11v BSS Transition Management feature lets an AP automatically provide load information of the neighbor APs to clients. It helps the Zyxel Device steer clients to a suitable AP for better performance or load balancing.

12.2 Radio

This screen allows you to create radio profiles for the Zyxel Device. A radio profile is a list of settings that an Zyxel Device can use to configure its radio transmitter(s). To access this screen click **Configuration > Object > AP Profile**.

Note: You can have a maximum of 32 radio profiles on the Zyxel Device.

ġ,	44 2 mil	Provide Q. S. Diale & Printede T	Second Second
•	Sector .	Drušie Nome +	Trequency Rand
		Wit_Rodio_246	2/6
	14	Wit_Rodic_36	36
	<u>.</u>	deta. It	7.4G
	4	deto: 92	36.
π.	1 Marst	1	Districting 1.446

Figure 57 Configuration > Object > AP Profile > Radio

The following table describes the labels in this screen.

LABEL	DESC RIPTIO N					
Add	Click this to add a new radio profile.					
Edit	Click this to edit the selected radio profile.					
Remove	Click this to remove the selected radio profile.					
Activate	To turn on an entry, select it and click Activate.					
Inactivate	To turn off an entry, select it and click Inac tivate .					
Object Reference	Click this to view which other objects are linked to the selected radio profile.					
#	This field is a sequential value, and it is not associated with a specific user.					
Status	This field shows whether or not the entry is activated.					
	A yellow bulb signifies that this rule is active. A gray bulb signifies that this rule is not active.					
Profile Name	This field indicates the name assigned to the radio profile.					
Frequency Band	This field indicates the frequency band which this radio profile is configured to use.					
Apply	Click Apply to save your changes back to the Zyxel Device.					
Reset	Click Reset to return the screen to its last-saved settings.					

Table 37 Configuration > Object > AP Profile > Radio

$12.2.1 \ Add/Edit Radio \ Pro file$

This screen allows you to create a new radio profile or edit an existing one. To access this screen, click the **Add** button or select a radio profile from the list and click the **Edit** button.

			Chapt	er 12	AP Pr	otile			
				- file y	D av al	:- > A	-1-1/5	-1:1	
Figure 58 Configuration	on > Oc	ject 2	> AP PI	one -	> kaa	10 > A	aa/E	air	(Y)
 Add Report From III Factor Anticas and Sellings 									10
Geerandestendingel									
If noware									
Pedle Shreet	-		@.						
6467 66769	las	1000	1						
Charles Market	and a distance	MHO C	7						
Chemical Sciences	18.1357	. 8.4	anna 7			8			
K freed 0010 on Avera									
R. Particip S. Cole 740 August									
Site of promier telephonikeer	idat auro			(m. 1					
	ac (87	2							
ri Pro victor									
# Schedule	1222								
Shart have	52.9	٠ <u>ــــــــــــــــــــــــــــــــــــ</u>	10						
Arri Fire	200 j	VERMAN	R	101227	17	988 3 196	99) 1		
	1.00	te sales.	- 24	Sec. 1	12	Sec. 10			
	1.0	Grader .							
Atamiteter									
Cuttor Histor	19.75	2	100.00						
M Chapter AN 100 Appression									
The second secon									
PROFILE Three books	1242		10.224						
Cappen menol	12		1 and						
ener-	100		The begin	1000					
Destanting of the field			and a						
Marine Nore Threshold	12		1.022.024	646166					
Cardinacian states" (states)			obm 2	1000					
E shey 2 hours (Second and		100							
Starba Tam, Court	a resource and	Concernant of the second	11×100	2					
	1		The second						
Hi wiew 602.11 r /s s stations only	. Paulo								
TI CookerCTS charmainin prise	nce of hadio	5							
W Enddle 902 F a 🛛 🕦									
Methods Settings									
Service designed (Variation	144	Read to	unicipat -	6.2	stic facili	e aut Carle			
Hullcat Cole Vopr	4.6	- 6	11/2	- 1 K	11.24	11 25	1. 4.	- m (4)	
Winter, m WLAR Jole Control (offin				311.63			0.000		
Contraction to the second second second	a a constantino de la	10.00	1.46.22	1896	10.24	10.503	10.4	10.40	
	1.000	100	a construction	attrick.	126922		0.000	Constant of the	
									eresere .
									Gaunt

The following table describes the labels in this screen.

LABEL	DESC RIPIIO N
Hide / Show Advanced Settings	Click this to hide or show the Advanced Settings in this window.
General Settings	
Activate	Select this option to make this profile active.
Profile Name	Enter up to 31 alphanumeric characters to be used as this profile's name. Spaces and underscores are allowed.
802.11 Band	Select whether this radio would use the 2.4 GHz or 5 GHz band.

Table 38	Configuration >	Object > I	A P Profilo >	Padia >	Add/Edit
	Configuration -			ruuio -	AUU/EUI

Table 38	Configuration >	Ohiect >	AP Profile > Radio >	Add/Edit (continued)
	configuration	CDJCCI		

LABEL	DESC RIPIIO N
802.11 Mode	Select how to let wireless clients connect to the AP.
	If 802.11 Band is set to 2.4G:
	 11b/g: allows either IEEE 802.11b or IEEE 802.11g compliant WLAN devices to associate with the Zyxel Device. The Zyxel Device adjusts the transmission rate automatically according to the wireless standard supported by the wireless devices.
	 11n: allows IEEE802.11b, IEEE802.11g and IEEE802.11n compliant WLAN devices to associate with the Zyxel Device.
	 11ax: allows IEEE802.11b, IEEE802.11g, IEEE802.11n, and IEEE802.11ax compliant WLAN devices to associate with the Zyxel Device. If the WLAN device isn't compatible with 802.11ax, the Zyxel Device will communicate with the WLAN device using 802.11n, and so on.
	If 802.11 Band is set to 5G:
	 11a: allows only IEEE 802.11a compliant WLAN devices to associate with the Zyxel Device.
	11n: allows both IEEE802.11n and IEEE802.11a compliant WLAN devices to associate with the Zyxel Device.
	• 11ac : allows IEEE802.11n, IEEE802.11a, and IEEE802.11ac compliant WLAN devices to associate with the Zyxel Device. If the WLAN device isn't compatible with 802.11ac, the Zyxel Device will communicate with the WLAN device using 802.11n, and so on.
	 11ax: allows IEEE802.11n, IEEE802.11a, IEEE802.11ac, and IEEE802.11ax compliant WLAN devices to associate with the Zyxel Device. If the WLAN device isn't compatible with 802.11ax, the Zyxel Device will communicate with the WLAN device using 802.11ac, and so on.
Channel Width	Select the channel bandwidth you want to use for your wireless network.
	Select 20 MHz if you want to lessen radio interference with other wireless devices in your neighborhood.
	Select $20/40~MHz$ to allow the Zyxel Device to choose the channel bandwidth (20 or 40 MHz) that has least interference.
	Select 20/40/80 to allow the Zyxel Device to choose the channel bandwidth (20 or 40 or 80) that has least interference. This option is available only when you select 11ac or 11ax in the 802.11 Mode field.
	Select $20/40/80/160~MHz$ to allow the Zyxel Device to choose the channel bandwidth (20 or 40 or 80 or 160 MHz) that has least interference. This option is available only when you select $11ax$ in the $802.11~Mode$ field.
	Note: If the environment has poor signal-to-noise ratio (SNR), the Zyxel Device will switch to a lower bandwidth.
	Note: 160 MHz is only available in NWA210AX, WAX610D and WAX650S.
Channel Selection	This is the radio channel which the signal will use for broadcasting by this radio profile.
	DCS: Choose Dynamic Channel Selection to have the Zyxel Device choose a radio channel that has least interference.
	• Manual: Choose from the available radio channels in the list. If your Zyxel Device is outdoor type, be sure to choose non-indoors channels.
Enable DCS Client Aware	Select this to have the Zyxel Device switch channels only when there are no clients connected to it. If there is a client connected, the Zyxel Device will not switch channels but generate a log. The Zyxel Device tries to scan and switch channels again at the end of the specified time interval or at the scheduled time.
	If you disable this then the Zyxel Device switches channels immediately regardless of any client connections. In this instance, clients that are connected to the Zyxel Device when it switches channels are dropped.

LABEL	DESC RIPIIO N				
Enable DCS	This field is available when you set Channel Selection to DCS.				
Client Aware	Select this to have the Zyxel Device switch channels only when there are no clients connected to it. If there is a client connected, the Zyxel Device will not switch channels but generate a log. The Zyxel Device tries to scan and switch channels again at the end of the specified time interval or at the scheduled time.				
	If you disable this then the Zyxel Device switches channels immediately regardless of any client connections. In this instance, clients that are connected to the Zyxel Device when it switches channels are dropped.				
2.4 GHz Channel Selection Method	This field is available when you set Channel Selection to DCS .				
2election method	Select how you want to specify the channels the Zyxel Device switches between for 2.4 GHz operation.				
	Select auto to have the Zyxel Device display a 2.4 GHz Channel Deployment field you can use to limit channel switching to 3 or 4 channels.				
	Select $manual$ to select the individual channels the Zyxel Device switches between.				
	Note: The method is automatically set to a uto when no channel is selected or any one of the previously selected channels is not supported.				
Channel ID	This field is available only when you set Channel Selection to DCS and set 2.4 GHz Channel Selection Method to manual .				
	Select the channels that you want the Zyxel Device to use.				
2.4 GHz Channel Deployment	This is available when you set Channel Selection to DCS and the 2.4 GHz Channel Selection Method is set to auto.				
	Select Three-Channel Deployment to limit channel switching to channels 1,6, and 11, the three channels that are sufficiently attenuated to have almost no impact on one another. In other words, this allows you to minimize channel interference by limiting channel-hopping to these three "safe" channels.				
	Select Four Channel Deployment to limit channel switching to four channels. Depending on the country domain, if the only allowable channels are 1-11 then the Zyxel Device uses channels 1, 4, 7, 11 in this configuration; otherwise, the Zyxel Device uses channels 1, 5, 9, 13 in this configuration. Four channel deployment expands your pool of possible channels while keeping the channel interference to a minimum.				
Enable 5 GHz DFS Aware	This field is available only when you select 5G in the 802.11 Band field, set Channel Selection to DCS and set 5 GHz Channel Selection Method to auto .				
	Select this if your APs are operating in an area known to have RADAR devices. This allows the Zyxel Device to downgrade its frequency to below 5 GHz in the event RADAR signal is detected, thus preventing it from interfering with that signal.				
	Enabling this forces the AP to select a non-DFS channel.				
5 GHz Channel Selection Method	Select how you want to specify the channels the Zyxel Device switches between for 5 GHz operation.				
	Select $Auto$ to have the Zyxel Device automatically select the best channel.				
	Select $manual$ to select the individual channels the Zyxel Device switches between.				
	Note: The method is automatically set to a uto when no channel is selected or any one of the previously selected channels is not supported.				
Channel ID	This field is available only when you set Channel Selection to DCS and set 5 GHz Channel Selection Method to manual .				
	Select the channels that you want the Zyxel Device to use.				

Table 38 Configuration > Object > AP Profile > Radio > Add/Edit (continued)

LABEL	DESC RIPIIO N
Time Interval	Select this option to have the Zyxel Device survey the other APs within its broadcast radius at the end of the specified time interval.
DCS Time Interval	This field is available when you set Channel Selection to DCS and select the Time Interval option.
	Enter a number of minutes. This regulates how often the Zyxel Device surveys the other APs within its broadcast radius. If the channel on which it is currently broadcasting suddenly comes into use by another AP, the Zyxel Device will then dynamically select the next available clean channel or a channel with lower interference.
Schedule	Select this option to have the Zyxel Device survey the other APs within its broadcast radius at a specific time on selected days of the week.
Start Time	Specify the time of the day (in 24-hour format) to have the Zyxel Device use DCS to automatically scan and find a less-used channel.
Week Days	Select each day of the week to have the Zyxel Device use DCS to automatically scan and find a less-used channel.
Advanced Settings	
Guard Interval	This field is available only when the channel width is $20/40/Hz$ or $20/40/80MHz$ and the 802.11 Mode is either $11n$ or $11ac$.
	Set the guard interval for this radio profile to either ${f short}$ or ${f long}$.
	The guard interval is the gap introduced between data transmission from users in order to reduce interference. Reducing the interval increases data transfer rates but also increases interference. Increasing the interval reduces data transfer rates but also reduces interference.
Enable A-MPDU	This field is not available when you set $802.11 \ Mode$ to $11a$ or $11b/g$.
Aggregation	Select this to enable A-MPDU aggregation.
	Message Protocol Data Unit (MPDU) aggregation collects Ethernet frames along with their 802.11n headers and wraps them in a 802.11n MAC header. This method is useful for increasing bandwidth throughput in environments that are prone to high error rates.
Enable A-MSDU Aggregation	This field is not available when you set 802.11 Mode to $11a \text{ or } 11b/g$.
	Select this to enable A-MSDU aggregation.
	Mac Service Data Unit (MSDU) aggregation collects Ethernet frames without any of their 802.11n headers and wraps the header-less payload in a single 802.11n MAC header. This method is useful for increasing bandwidth throughput. It is also more efficient than A-MPDU except in environments that are prone to high error rates.
RTS/CTS Threshold	Use RTS/CTS to reduce data collisions on the wireless network if you have wireless clients that are associated with the same AP but out of range of one another. When enabled, a wireless client sends an RTS (Request To Send) and then waits for a CTS (Clear To Send) before it transmits. This stops wireless clients from transmitting packets at the same time (and causing data collisions).
	A wireless client sends an RTS for all packets larger than the number (of bytes) that you enter here. Set the RTS/CTS equal to or higher than the fragmentation threshold to turn RTS/CTS off.
Beacon Interval	When a wirelessly networked device sends a beacon, it includes with it a beacon interval. This specifies the time period before the Zyxel Device sends the beacon again. The interval tells receiving devices on the network how long they can wait in low-power mode before waking up to handle the beacon. A high value helps save current consumption of the access point.
DTIM	Delivery Traffic Indication Message (DTIM) is the time period after which broadcast and multicast packets are transmitted to mobile clients in the Active Power Management mode. A high DTIM value can cause clients to lose connectivity with the network. This value can be set from 1 to 255.

Table 38 Configuration > Object > AP Profile > Radio > Add/Edit (continued)

Table 38	Configuration >	Object >	AP Profile > Radio >	Add/Edit (continued)
----------	-----------------	----------	----------------------	----------------------

LABEL	DESC RIPIIO N
Enable Signal Threshold	Select the check box to use the signal threshold to ensure wireless clients receive good throughput. This allows only wireless clients with a strong signal to connect to the AP.
	Clear the check box to not require wireless clients to have a minimum signal strength to connect to the AP.
Station Signal Threshold	Set a minimum client signal strength. A wireless client is allowed to connect to the AP only when its signal strength is stronger than the specified threshold.
	-20 dBm is the strongest signal you can require and -105 is the weakest.
Disassociate Station Threshold	Set a minimum kick-off signal strength. When a wireless client's signal strength is lower than the specified threshold, the Zyxel Device disconnects the wireless client from the AP.
	-20 dBm is the strongest signal you can require and -105 is the weakest.
Allow Station Connection after Multiple Retries	Select this option to allow a wireless client to try to associate with the AP again after it is disconnected due to weak signal strength.
Station Retry Count	Set the maximum number of times a wireless client can attempt to re-connect to the AP
Allow 802.11n/ ac/ax stations only	Select this option to allow only 802.11 n/ac/ax clients to connect, and reject 802.11a/b/g clients.
Blacklist DFS	This field is available if 802.11 Band is set to 5G and Channel Selection is set to DCS.
channels in presence of radar	Enable this to temporarily blacklist the wireless channels in the Dynamic Frequency Selection (DFS) range whenever a radar signal is detected by the Zyxel Device.
Enable 802.11d	Clear the checkbox to prevent the AP from broadcasting a country code, also called a country Information Element (IE), in beacon frames. This makes the AP incompatible with 802.11d networks and devices.
	802.11d is a WiFi network specification that allows the AP to broadcast a country code to WiFi client. The country code indicates where the AP is located. If WiFi clients are unable to connect to the AP due to an incompatible country code, you should disable 802.11d.
Multicast Settings	
Transmission	Specify how the Zyxel Device handles wireless multicast traffic.
Mode	Select Multic ast to Unic ast to broadcast wireless multicast traffic to all of the wireless clients as unicast traffic. Unicast traffic dynamically changes the data rate based on the application's bandwidth requirements. The retransmit mechanism of unicast traffic provides more reliable transmission of the multicast traffic, although it also produces duplicate packets.
	Select Fixed Multic a st Ra te to send multicast traffic to all wireless clients at a single data rate. You must know the multicast application's bandwidth requirements and set it in the following field.
Multicast Rate(Mbps)	If you set Transmission Mode to Fixed Multicast Rate , select a data rate at which the Zyxel Device transmits multicast packets to wireless clients. For example, to deploy 4 Mbps video, select a fixed multicast rate higher than 4 Mbps.
WLAN Rate Control Setting	Sets the minimum data rate that 2.4Ghz WiFi clients can connect at, in Mbps. At the time of write, allowed values are: 1, 2,5. 5, 6, 9, 11, 12, 18, 24, 36, 48, 54.
	Sets the minimum data rate that 5Ghz WiFi clients can connect at, in Mbps. At the time of write, allowed values are: 6,9, 12, 18, 24, 36, 48, 54.
	Increasing the minimum data rate can reduce network overhead and improve WiFi network performance in high density environments. However, WiFi clients that do not support the minimum data rate will not be able to connect to the AP.
ОК	Click OK to save your changes back to the Zyxel Device.

12.3 SSID

The SSID screens allow you to configure three different types of profiles for your networked APs: an SSID list, which can assign specific SSID configurations to your APs; a security list, which can assign specific encryption methods to the APs when allowing wireless clients to connect to them; and a MAC filter list, which can limit connections to an AP based on wireless clients MAC addresses.

12.3.1 SSID List

This screen allows you to create and manage SSID configurations that can be used by the APs. An SSID, or Service Set IDentifier, is basically the name of the wireless network to which a wireless client can connect. The SSID appears as readable text to any device capable of scanning for wireless frequencies (such as the WiFi adapter in a laptop), and is displayed as the wireless network name when a person makes a connection to it.

To access this screen click Configuration > Object > AP Profile > SSID > SSID List.

Note: You cannot add or remove an SSID profile after running the setup wizard.

Figure 59 Configuration > Object > AP Profile > SSID > SSID List (Default)

550 LINI	Secto	ey an a	AAC THE DR	Loyer-2 isolation	n the		
D Sumn	any:						
CLASS.	1241 11 24	State Barrister	s kindede operation				
	and the second se						
	le sure -	1923	Socially Frohib	1906	MAC Rising	boxe-250401	VL91D
	legure - last	1823 2yxe: 821 A	Security Fredie Confour	New A	Citoble	CRODIE CRODIE	VD401D

551	Dian (3a	contry (Int	MAC Filter Dri	layer-2 letion	ion lini		
йD;	Summary						
2	nd 📲 Operatio	Reflected scale					
1	Final States	2007	County ProBe	sont.	Mike Ellering -	invertiginal d	S VEALEXS
Í.	WI2_230_1	Zyisail	Water Profile.	ANM.	discher	chude	1.10
2	NICSTO,2	Lyoni	West Prof.	AMM	(f) (f) (f)	cluste	ū1
8	K_082_230_3	2900	WESSELMOR.	WMM	concepted.	elegie	1
4	4_CR2_SIN	27.44	Weight Park.	WWW.	10102080	clottile	1.1
5	W2_33D_5	2yout	Wat SEC, Profile.	WMM :	0.0046	clacific -	1
6	0_CR2_314	Lefond .	WE SECTION.	white .	01006	chade	13
У.	W2,330,71	2700	WILSEC Profit	NON	C/ODE	Oberste	1
8	8_UR2_319	21000	WE SEC HUR.	W999	CLOCK .	chepte	11
4	detault -	2yoret 821 A	orient	WMM.	010010	cisciple	19

Figure 60 Configuration > Object > AP Profile > SSID > SSID List (After wizard setup)

The following table describes the labels in this screen.

Table 39 Configuration > Object > AP Profile > SSID > SSID List

IABEL	DESC RIPIIO N			
Add	Click this to add a new SSID profile.			
	This button is not available after you configure the Zyxel Deviceusing the wizard.			
Edit	Click this to edit the selected SSID profile.			
Remove	Click this to remove the selected SSID profile.			
	This button is not available after you configure the Zyxel Deviceusing the wizard.			
Object Reference	Click this to view which other objects are linked to the selected SSID profile (for example, radio profile).			
#	This field is a sequential value, and it is not associated with a specific user.			
Profile Name	This field indicates the name assigned to the SSID profile.			
SSID	This field indicates the SSID name as it appears to wireless clients.			
Security Profile	This field indicates which (if any) security profile is associated with the SSID profile.			
QoS	This field indicates the QoS type associated with the SSID profile.			
MAC Filtering Profile	This field indicates which (if any) MAC filter Profile is associated with the SSID profile.			
Layer-2 Isolation Profile	This field indicates which (if any) layer-2 isolation Profile is associated with the SSID profile.			
VLAN ID	This field indicates the VLAN ID associated with the SSID profile.			

12.3.2 Add/Edit SSID Profile

This screen allows you to create a new SSID profile or edit an existing one. To access this screen, click the **Add** button or select a SSID profile from the list and click the **Edit** button.

Add 5510 Profil	0					
Open La move Obje	v ⁿ					
- ABA ANDAL				4		
NER.		Tear				
Annually Frederic		6/0-1		ł	s O M	
WAAR FRANK (FR	device the transport of the state of the state				0	
$\exp(r^{2} t \ln h h r)$	entile -	- basede			V O	
Que	1	alan .		je je		
Received The	$2\lambda_{1}(0,\infty)$	$\alpha V \alpha J \alpha k$	60			
Description	8	a 👘	mispi 👘	12	20.0 film milled	
Usinia		٥	mopule		Distances 6.0	
VLANTS:		1				
🗖 Hotten 🕸 D						
E state (read	sa mile					
T = 14 (011) (02)	6	999558				
Trata Track	400					
		And				
🐨 Sorrectule Soll.						
in contractor Langua		2 inere	00.00	10	248001196	
Wanday	and a	2 1000	00.02	Ç.	1480 M	
	produced in the local division of	-	00:00	-	24800 1	
Wednesdoy:		X hore	00:00 00	Ç.	2450	
	en altrie	A hore	00.00	to .	2460 11	
The second of		 a suite 	1000	10	24:00 18	
Thursday.	and the second	St. Access	C COLOR L DAG			
Thursday: Molay: Sanuday:	endele endele	anore 🖄	00.00	to	24:00	

Figure 61 Configuration > Object > AP Profile > SSID > SSID List > Add/Edit SSID Profile

The following table describes the labels in this screen.

Table 40	Configuration >	Object > AP Pro	ofile > SSID > SSID List >	Add/Edit SSID Profile
----------	-----------------	-----------------	----------------------------	-----------------------

IABEL	DESC RIPTIO N
Create new Object	Select an object type from the list to create a new one associated with this SSID profile.
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.
SSID	Enter the SSID name for this profile. This is the name visible on the network to wireless clients. Enter up to 32 characters, spaces and underscores are allowed.
Security Profile	Select a security profile from this list to associate with this SSID. If none exist, you can use the Create new Object menu to create one.
	Note: It is highly recommended that you create security profiles for all of your SSIDs to enhance your network security.

IABEL	DESC RIPTIO N			
MAC Filtering Profile	Select a MAC filtering profile from the list to associate with this SSID. If none exist, you can use the Create new Object menu to create one.			
	MAC filtering allows you to limit the wireless clients connecting to your network through a particular SSID by wireless client MAC addresses. Any clients that have MAC addresses not in the MAC filtering profile of allowed addresses are denied connections.			
	The disable setting means no MAC filtering is used.			
Layer-2 Isolation Profile	Select a layer-2 isolation profile from the list to associate with this SSID. If none exist, you can use the Create new Object menu to create one.			
	Layer-2 isolation allows you to prevent wireless clients associated with your Zyxel Device from communicating with other wireless clients, APs, computers or routers in a network.			
	The disable setting means no layer-2 isolation is used.			
QoS	Select a Quality of Service (QoS) access category to associate with this SSID. Access categories minimize the delay of data packets across a wireless network. Certain categories, such as video or voice, are given a higher priority due to the time sensitive nature of their data packets.			
	QoS access categories are as follows:			
	disable: Turns off QoS for this SSID. All data packets are treated equally and not tagged with access categories.			
	WMM : Enables automatic tagging of data packets. The Zyxel Device assigns access categories to the SSID by examining data as it passes through it and making a best guess effort. If something looks like video traffic, for instance, it is tagged as such.			
	WMM_VOICE All wireless traffic to the SSID is tagged as voice data. This is recommended if an SSID is used for activities like placing and receiving VoIP phone calls.			
	WMM_VIDEO : All wireless traffic to the SSID is tagged as video data. This is recommended for activities like video conferencing.			
	WMM_BEST_EFFORT All wireless traffic to the SSID is tagged as "best effort," meaning the data travels the best route it can without displacing higher priority traffic. This is good for activities that do not require the best bandwidth throughput, such as surfing the Internet.			
	WMM_BACKG RO UND : All wireless traffic to the SSID is tagged as low priority or "background traffic", meaning all other access categories take precedence over this one. If traffic from an SSID does not have strict throughput requirements, then this access category is recommended. For example, an SSID that only has network printers connected to it.			
Rate Limiting	•			
Downlink	Define the maximum incoming transmission data rate (either in mbps or kbps) on a per-station basis.			
Uplink	Define the maximum outgoing transmission data rate (either in mbps or kbps) on a per-station basis.			
VLAN ID	Enter a VLAN ID for the Zyxel Device to use to tag traffic originating from this SSID.			
Hidden SSID	Select this if you want to "hide" your SSID from wireless clients. This tells any wireless clients in the vicinity of the AP using this SSID profile not to display its SSID name as a potential connection. Not all wireless clients respect this flag and display it anyway.			
	When a SSID is "hidden" and a wireless client cannot see it, the only way you can connect to the SSID is by manually entering the SSID name in your wireless connection setup screen(s) (these vary by client, client connectivity software, and operating system).			
Enable Intra-BSS Traffic Blocking	Select this option to prevent crossover traffic from within the same SSID on the Zyxel Device.			
Enable U-APSD	Select this option to enable Unscheduled Automatic Power Save Delivery (U-APSD), which is also known as WMM-Power Save. This helps increase battery life for battery-powered wireless clients connected to the Zyxel Device using this SSID profile.			

T I 10				Add/Edit SSID Profile	
	(ODTIGUITATIOD > ODTIGUITAT	ACT > AP PROTIA >	$\sim 1 \times 1 $		ICONTINUEQU
IABEL	DESC RIPIIO N				
-------------------------------	--				
Enable Proxy ARP	The Address Resolution Protocol (ARP) is a protocol for mapping an IP address to a MAC address. An ARP broadcast is sent to all devices in the same Ethernet network to request the MAC address of a target IP address.				
	Select this option to allow the Zyxel Device to answer ARP requests for an IP address on behalf of a client associated with this SSID. This can reduce broadcast traffic and improve network performance.				
802.11k/v Assisted Roaming	Select this option to enable IEEE 802.11k/v assisted roaming on the Zyxel Device. When the connected clients request 802.11k neighbor lists, the Zyxel Device will response with a list of neighbor APs that can be candidates for roaming.				
Schedule SSID	Select this option and set whether the SSID is enabled or disabled on each day of the week. You also need to select the hour and minute (in 24-hour format) to specify the time period of each day during which the SSID is enabled/enabled.				
ОК	Click OK to save your changes back to the Zyxel Device.				
Cancel	Click Cancel to exit this screen without saving your changes.				

Table 40 Configuration > Object > AP Profile > SSID > SSID List > Add/Edit SSID Profile (continued)

12.4 Security List

This screen allows you to manage wireless security configurations that can be used by your SSIDs. Wireless security is implemented strictly between the AP broadcasting the SSID and the stations that are connected to it.

To access this screen click Configuration > Object > AP Profile > SSID > Security List.

Note: You can have a maximum of 32 security profiles on the Zyxel Device.

to dy an	Seconty Urt	MAC HISFLU	Engels Climitation Dat	
curty Summo				
 Prot w Na 	i 📲 famora 🗐 125ja 2014 -	o helwerice.	Security Node	
delocit.	11112		I OLM	

Figure 62 Configuration > Object > AP Profile > SSID > Security List

Table 41 Configuration > Object > AP Profile > SSID > Security List

IABEL	DESC RIPIIO N
Add	Click this to add a new security profile.
Edit	Click this to edit the selected security profile.
Remove	Click this to remove the selected security profile.
Object Reference	Click this to view which other objects are linked to the selected security profile (for example, SSID profile).
#	This field is a sequential value, and it is not associated with a specific user.

Table 41 Configuration > Object > AP Profile > SSID > Security List (continued)

IABEL	DESC RIPIIO N
Profile Name	This field indicates the name assigned to the security profile.
Security Mode	This field indicates this profile's security mode (if any).

12.4.1 Add/Edit Security Profile

This screen allows you to create a new security profile or edit an existing one. To access this screen, click the **Add** button or select a security profile from the list and click the **Edit** button.

Note: These screens' options change based on the Security Mode selected.

Figure 63 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile> Security Mode: none

Ecil Security Profile default		
Lan ockerstart Sortingel		
General Selfings		
no le Same	Shower .	
Scenily Made:	-9°-0	
Automiculos Sellings		
2 briegte		
Reduction Scotles Smooth	i.	(80-50000 Accords 0 stor million)
Ide (mesu)	200	be accessive week
Rodus Settings		
D' Finney Ready Server Asthely	2	
Toothe Server 17 Addition		0
Doolar Server Ports	-	()- (1- (15 M))
The second state of the se	2. Market and a fear that for the fear that fear the fear	
Nocity Server Secret		0
		0
Nocito Server Secreti	-uh-	0
Rocko Server Secreti III Sciencius Rockussicano Acili	rulu Tente	0
Rocks Server Secret III Scrowdae Rock Science Acto E Rimary Address for Jerver Ac	rute terre Activete	
Rocks Lever Leven If Scrowday Rock assessment Acto Dimary Accounting Jerver Ac- Z Secondary Accounting Server	rat. Terte Activele	
Rocks Server Secret E Scoundary Rock assesser Activ E Chinary Accounting Server Activ E Secondary Accounting Server Accounting Server IP Address	rute Terre Activele	0 0 0 0 0 0 0
Rocks Server Secret F Scrowdae Rock Siderer Actor E Primary Addressing Server Actor Secondars Accounting Server Accounting Server IP Address Accounting Server IP Address	aut. Rome Actively	0 0 0 0 0 0 0
Roote Server Secret F Secondas Rodusserson Acto Frimary Accounting Server Ac Secondars Accounting Server Accounting Server P Address Accounting Server Act Accounting Server Act	aut. Rome Actively	0 0 0 0 0 0 0 0
Roote Lever Levert P Scrowdas Rodusioner Acto Dimary Accounting Server Acto Serverdas Accounting Server Accounting Server IP Actions Accounting Server IP Actions Accounting Server IP Accounting Store Secrets E Accounting Intern Bodale	out, forte Schoole	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Rooks Server Secret Considers Rook a Server Activ Rooks Rook a Server Activ Rooks Accounting Server Produces Accounting Interim Rookster Interim Specific Interim	out, forte Schoole	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0

Table 42 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security	
Mode: none	

LABEL	DESC RIPIIO N
General Settings	
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.
Security Mode	Select a security mode from the list: none, enhanced-open, wep, wpa2, wpa2-mix or wpa3.
	enhanced-open uses Opportunistic Wireless Encryption (OWE) which encrypts the wireless connection when possible.
Authentication Settings	
Enterprise	Select this to enable 802.1x secure authentication with a RADIUS server.
ReAuthentication Timer	Enter the interval (in seconds) between authentication requests. Enter a 0 for unlimited time.
Advance	
Note: Click on the Sha	w Advanced Settings button to show the fields describe below.
Idle Timeout	Enter the idle interval (in seconds) that a client can be idle before authentication is discontinued.
Radius Settings	
Primary / Secondary Radius Server Activate	Select this to have the Zyxel Device use the specified RADIUS server.
Radius Server IP Address	Enter the IP address of the RADIUS server to be used for authentication.
Radius Server Port	Enter the port number of the RADIUS server to be used for authentication.
Radius Server Secret	Enter the shared secret password of the RADIUS server to be used for authentication.
Primary / Secondary Accounting Server Activate	Select the check box to enable user accounting through an external authentication server.
Accounting Server IP Address	Enter the IP address of the external accounting server in dotted decimal notation.
Accounting Server Port	Enter the port number of the external accounting server. The default port number is 1813. You need not change this value unless your network administrator instructs you to do so with additional information.
Accounting Share Secret	Enter a password (up to 128 alphanumeric characters) as the key to be shared between the external accounting server and the Zyxel Device. The key must be the same on the external accounting server and your Zyxel Device. The key is not sent over the network.
Accounting Interim Update	This field is available only when you enable user accounting through an external authentication server.
	Select this to have the Zyxel Device send subscriber status updates to the accounting server at the interval you specify.
Interim Update Interval	Specify the time interval for how often the Zyxel Device is to send a subscriber status update to the accounting server.
General Server Settings	
NAS IP Address	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) IP address attribute, enter it here.

Table 42 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Secu	rity Profile> Security
Mode: none (continued)	

LABEL	DESC RIPTIO N
NAS Identifier	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) identifier attribute, enter it here. The NAS identifier is to identify the source of access request. It could be the NAS's fully qualified domain name.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

Figure 64 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile> Security Mode: enhanced- open

Show Advanced Settings General Settings Public Name Analytiches Consistanting	
Nole Name Local Annual State	
Securitation in the second core in the second	
Security Sheep Consistence of Constant	
Authentication Softinge	
2, reactor Moca	
Alvance	
(clentmaster: 2000) (00.00020(control)	
👿 Haragaman hamamawakan 🔮 Sofanat 👔 Yayawa	

The following table describes the labels in this screen.

Table 43 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security Mode: enhanced- open

LABEL	DESC RIPIIO N
General Settings	
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.
Security Mode	Select a security mode from the list: none, enhanced-open, wep, wpa2, wpa2-mix or wpa3. enhanced-open uses Opportunistic Wireless Encryption (OWE) which encrypts the wireless connection when possible.
Authentication Settings	
Transition Mode	Enable this for backwards compatibility. This option is only available if the Security Mode is wpa3 or enhanced-open. This creates two virtual APs (VAPs) with a primary (wpa3 or enhanced-open) and fallback (wpa2 or none) security method.
	If the Security Mode is wpa3, enabling this will force Management Frame Protection to be set to Optional. If this is disabled or if the Security Mode is enhanced-open, Management Frame Protection will be set to Required.

Table 43 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security	
Mode: enhanced- open (continued)	

LABEL	DESC RIPIIO N
Advance	
Note: Click on the Sh	ow Advanced Settings button to show the fields describe below.
Idle Timeout	Enter the idle interval (in seconds) that a client can be idle before authentication is discontinued.
Management Frame Protection	This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes .
	Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11i WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks.
	Select the check box to enable management frame protection (MFP) to add security to 802.11 management frames.
	Select Optional if you do not require the wireless clients to support MFP. Management frames will be encrypted if the clients support MFP.
	Select Required and wireless clients must support MFP in order to join the Zyxel Device's wireless network.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

Figure 65 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security Mode: wep

Edit Security Profile defaul	6	8
Hide Advances Sellings		
General Settings		
Prot e Norse:	Hereite.	
Security Moder	wep.	×
Authentication Settings		
ini Tree poly		
Reference to a	(y	(30: SU00) seconds, 0 h unimited)
Automication Type:	0000	
Key ength	WIT-33	18)
8464: Enter o ASCRETISTOCIE 128 Gr. B. La 1878C (chorac		erochers (10/7, 14-F) Hor exists Ass. (1-4). Signal loss (10 R, 14-F) To gradh Roy (1-4).
I Swy 1	*********	
Cov 2		
(c) Sey 2.		
10 - Sep 4		
<u>al</u> Art <mark>eri</mark> ca		
o e timeo d	2000	(30.50000 schoodk)
Rockus Settings		
Rocke: Settings 191 Princip Rocke, Screen Acci	~~~	
1 1 (C. 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997	~~	
IPI Primery Rodice Server Acti		
IVI Primery Radio Server Activ Radio Server P Actives		0 (*
 Princip Radio Science Activ Radio Science P Active W Radio Science Part 		0 (*
 Princip Rodius Science Actin Radius Accessed P Actine w Radius Server Part Rodius Server Secrets 		0 (*
 Princip Radio Server Activ Radio Server Plant Radio Server Part Radio Server Part Radio Server Server (n, Inclu Activate	0 (*
 Primary Radius Server Activ Radius Server Part Radius Server Part Radius Server Part Radius Server Secrets Baccor serve Radios Server A Trimary Accord stress Server 	n, Inclu Activate	0 (*
 Primary Radius Science Activ Particular Science P Academy Radius Science Cart Radius Science Cart Radius Science Sciences Science Science Sciences Crimary Academy Redox Science Science Cart Academy Redox 	n, Inclu Activate	0 (*

The following table describes the labels in this screen.

Table 44 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security Mode: wep

LABEL	DESC RIPTIO N
General Settings	•
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.
Security Mode	Select a security mode from the list: none, enhanced-open, wep, wpa2, wpa2-mix or wpa3. enhanced-open uses Opportunistic Wireless Encryption (OWE) which encrypts the wireless connection when possible.

Table 44 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile> Security
Mode: wep (continued)

LABEL	DESC RIPTIO N
Authentication Settings	•
Enterprise	Select this to enable 802.1x secure authentication with a RADIUS server.
ReAuthentication Timer	Enter the interval (in seconds) between authentication requests. Enter a 0 for unlimited time.
Authentication Type	Select a WEP authentication method. Choices are Open or Share key.
Key Length	Select the bit-length of the encryption key to be used in WEP connections.
	If you select WEP-64 :
	 Enter 10 hexadecimal digits in the range of "A-F", "a-f" and "0-9" (for example, 0x11AA22BB33) for each Key used.
	or
	 Enter 5 ASCII characters (case sensitive) ranging from "a-z", "A-Z" and "0-9" (for example, MyKey) for each Key used.
	If you select WEP-128 :
	 Enter 26 hexadecimal digits in the range of "A-F", "a-f" and "0-9" (for example, 0x00112233445566778899AABBCC) for each Key used.
	or
	 Enter 13 ASCII characters (case sensitive) ranging from "a-z", "A-Z" and "0-9" (for example, MyKey12345678) for each Key used.
Key 1~4	Based on your Key Length selection, enter the appropriate length hexadecimal or ASCII key.
Advance	
Note: Click on the Sho	w Advanced Settings button to show the fields describe below.
Idle Timeout	Enter the idle interval (in seconds) that a client can be idle before authentication is discontinued.
Radius Settings	
Primary / Secondary Radius Server Activate	Select this to have the Zyxel Device use the specified RADIUS server.
Radius Server IP Address	Enter the IP address of the RADIUS server to be used for authentication.
Radius Server Port	Enter the port number of the RADIUS server to be used for authentication.
Radius Server Secret	Enter the shared secret password of the RADIUS server to be used for authentication.
Primary / Secondary Accounting Server Activate	Select the check box to enable user accounting through an external authentication server.
Accounting Server IP Address	Enter the IP address of the external accounting server in dotted decimal notation.
Accounting Server Port	Enter the port number of the external accounting server. The default port number is 1813. You need not change this value unless your network administrator instructs you to do so with additional information.
Accounting Share Secret	Enter a password (up to 128 alphanumeric characters) as the key to be shared between the external accounting server and the Zyxel Device. The key must be the same on the external accounting server and your Zyxel Device. The key is not sent over the network.
Accounting Interim Update	This field is available only when you enable user accounting through an external authentication server.

Table 44 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security	
Mode: wep (continued)	

IABEL	DESC RIPTIO N
Interim Update Interval	Specify the time interval for how often the Zyxel Device is to send a subscriber status update to the accounting server.
General Server Settings	
NAS IP Address	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) IP address attribute, enter it here.
NAS Identifier	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) identifier attribute, enter it here. The NAS identifier is to identify the source of access request. It could be the NAS's fully qualified domain name.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

Figure 66 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security Mode: wpa2

Edit Security Profile defoult			$\{ i \}$
Rikin Advance Settings			
General Settings			
Profile Nonec	theirs at:		
Security Medic	would	7	
Aalbeelendien lettings			
# Ericionic			
Post-line off-off-of-	6	(90-80000 selected). Ök terkerit	мî,
the Postorial			
Advance			
Cipher Trans	005	1.	
d e fimeral:	300	(20-2000) seizendet .	
Group Key Update Timen	80000	(SI-S000) seconds:	
Pre Notice Dealton	Q.ooc	ar.	
🚊 Monogeroen: Proise Profess	inn		
Rockus Settings			
(a) Trimory Rockus Server Activate			
Rodius Server (* Appress			
Radiu/Server com		🤹 👔 (daada)	
Ruber Server Server	1	•	
III Secondary Poplar Server Anti-	iota.		
Transfer Accounting Server Ac	l-sie		
(b) Socializing Associating Service	hediante.		
Accounting Server P. Adamas	Jamman		
Accounting Server flort:		(j	
Accounting Shore Secret:		<u>a</u>	
bil Accounting intedmit polate			
Interim Underte Interval	10	(* Caal of each a	
CARL PROPERTY AND A CONTRACT OF			
Cenerci Server Setting)			
Ceneral Server Setting) NAS P Accines		(Upfond)	

The following table describes the labels in this screen.

Table 45 Configuration > Object > AP Profile > SSID > Security List > AAdd/Edit Security Profile > Security Mode: wpa2

LABEL	DESC RIPIIO N
General Settings	
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.

LABEL	DESC RIPIIO N
Security Mode	Select a security mode from the list: none, enhanced-open, wep, wpa2, wpa2-mix or wpa3.
	enhanced-open uses Opportunistic Wireless Encryption (OWE) which encrypts the wireless connection when possible.
Authentication Settings	·
Enterprise	Select this to enable 802.1x secure authentication with a RADIUS server.
ReAuthentication Timer	Enter the interval (in seconds) between authentication requests. Enter a 0 for unlimited time.
Personal	This field is available when you select the wpa2 , wpa2-mix or wpa3 security mode.
	Select this option to use a Pre-Shared Key (PSK) with WPA2 encryption or Simultaneous Authentication of Equals (SAE) with WPA3 encryption.
Pre-Shared Key	Enter a pre-shared key of between 8 and 63 case-sensitive ASCII characters (including spaces and symbols) or 64 hexadecimal characters.
Advance	
Note: Click on the Sha	w Advanced Settings button to show the fields describe below.
Cipher Type	Select an encryption cipher type from the list.
- 7	 auto - This automatically chooses the best available cipher based on the cipher in use by the wireless client that is attempting to make a connection.
	• aes - This is the Advanced Encryption Standard encryption method. It is a more recent development over TKIP and considerably more robust. Not all wireless clients may support this.
Idle Timeout	Enter the idle interval (in seconds) that a client can be idle before authentication is discontinued.
Group Key Update Timer	Enter the interval (in seconds) at which the AP updates the group WPA2 encryption key.
Pre-Authentication	Select Enable to allow pre-authentication. Otherwise, select Disable.
Management Frame Protection	This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes .
	Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11i WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks.
	Select the check box to enable management frame protection (MFP) to add security to 802.11 management frames.
	Select Optional if you do not require the wireless clients to support MFP. Management frames will be encrypted if the clients support MFP.
	Select Required and wireless clients must support MFP in order to join the Zyxel Device's wireless network.
Radius Settings	
Primary / Secondary Radius Server Activate	Select this to have the Zyxel Device use the specified RADIUS server.
Radius Server IP Address	Enter the IP address of the RADIUS server to be used for authentication.
Radius Server Port	Enter the port number of the RADIUS server to be used for authentication.

Table 45 Configuration > Object > AP Profile > SSID > Security List > AAdd/Edit Security Profile> Security Mode: wpa2 (continued)

LABEL	DESC RIPIIO N
Radius Server Secret	Enter the shared secret password of the RADIUS server to be used for authentication.
Primary / Secondary Accounting Server Activate	Select the check box to enable user accounting through an external authentication server.
Accounting Server IP Address	Enter the IP address of the external accounting server in dotted decimal notation.
Accounting Server Port	Enter the port number of the external accounting server. The default port number is 1813. You need not change this value unless your network administrator instructs you to do so with additional information.
Accounting Share Secret	Enter a password (up to 128 alphanumeric characters) as the key to be shared between the external accounting server and the Zyxel Device. The key must be the same on the external accounting server and your Zyxel Device. The key is not sent over the network.
Accounting Interim Update	This field is available only when you enable user accounting through an external authentication server.
	Select this to have the Zyxel Device send subscriber status updates to the accounting server at the interval you specify.
Interim Update Interval	Specify the time interval for how often the Zyxel Device is to send a subscriber status update to the accounting server.
General Server Settings	
NAS IP Address	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) IP address attribute, enter it here.
NAS Identifier	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) identifier attribute, enter it here. The NAS identifier is to identify the source of access request. It could be the NAS's fully qualified domain name.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 45 Configuration > Object > AP Profile > SSID > Security List > AAdd/Edit Security Profile > Security Mode: wpa2 (continued)

Figure 67 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile> Security Mode: wpa2-mix

Edit Security Profile defoult		
El Iniciadora de Hinga		
General Settings		
Profile Money	Baltzur	
Successly Medic	14-94 A 1110	
Autoenterolion Settings		
¥ Encrosso		
Provide of allow Torres	(n	(90-80000 seconds, 5k colorals
D Periora		
el Advance		
Dipher June	594 A	1
a e timeaut:	300	(20-2000) seizendet
Group Key topdare Trien	80000	(SI-SUGA) seconds:
	1.0	1. A.
Re Aufrer Bestion Reditor Tethings I minimitatius Server Activate Radius Server IV Activate	Dave.	
Rodius Schings Madus Server Activate Radius Server III Andress Radius Server Part Radius Server Part		0
Rodice Lettings Trimoty Books Server Activate Rodice Server IP Activate Rodice Server Part Pack Server Part Pack Server Fort Pack Server Fort	cie	0
Rodius Ichings Madus Server Activate Radius Server In Activate Radius Server Part Park, Server Part Park, Server Part Secondary Rocka Server Activ Primary According Server Activ	cie I-gic	0
Rodius Inflings Trimoty Kodius Server Activate Rodius Server I'r Archent Rodius Server Fort Rodius Server Fort Rodius Server Fort Secondary Rodius Server Activ Frincis Accounting Server Activ Secondary Accounting Server Activ Secondary Accounting Server Activ	ote Faulo Actione	Q Q (19000) D
Rodius Inflings Trimony Bodius Server Activate Rodius Server IP Activate Rodius Server Part Packs Server Part Packs Server Part Secondary Rocks Server Activ Primony Accounting Server Violation Accounting Server IP Activities	ote I	0 0): sw(3) 0
Rodius Inflings Trimoty Kodius Server Activate Rodius Server I'r Archent Rodius Server Fort Rodius Server Fort Rodius Server Fort Secondary Rodius Server Activ Frincis Accounting Server Activ Secondary Accounting Server Activ Secondary Accounting Server Activ	ote Inglo Actione	Q Q (19000) D
Rodio Iclings	ote Inglo Actione	0 0): sw(3) 0
Rodius Inflings Trimory Bodius Server Activate Rodius Server III Activate Rodius Server Part Pack & Server Part Pack & Server Part Secondary Rodius Server Activ Primary Accounting Server Activ Rodounting Server IV Activeur Accounting Server IV Activeur Accounting Server IV Activeur	ote Inglo Actione	0 0): sw(3) 0
Rective Lettings Trimony Rocks Server Activate Racks Server Fort Racks Server Fort Racks Server Fort Secondary Rocks Server Activ Server daty Accounting Server Act Rocsuming Server Practice Accounting Server Practice Accounting Server Practice Accounting Server Practice (a) Accounting Server Practice (b) Accounting Server Practice (c) Accounting Server (c) Account	cia Ingle	0) (50000) 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Rodios Ictings	cia Ingle	0) (50000) 0 0 0 0 0 0 0 0 0 0 0 0 0 0

The following table describes the labels in this screen.

Table 46 Configuration > Object > AP Profile > SSID > Security List > AAdd/Edit Security Profile> Security Mode: wpa2-mix

LABEL	DESC RIPIIO N
General Settings	
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.

LABEL	DESC RIPTIO N						
Security Mode	Select a security mode from the list: none, enhanced-open, wep, wpa2, wpa2-mix or wpa3.						
	enhanced-open uses Opportunistic Wireless Encryption (OWE) which encrypts the wireless connection when possible.						
Authentication Settings							
Enterprise	Select this to enable 802.1x secure authentication with a RADIUS server.						
ReAuthentication Timer	Enter the interval (in seconds) between authentication requests. Enter a 0 for unlimited time.						
Personal	This field is available when you select the wpa2 , wpa2-mix or wpa3 security mode.						
	Select this option to use a Pre-Shared Key (PSK) with WPA2 encryption or Simultaneous Authentication of Equals (SAE) with WPA3 encryption.						
Pre-Shared Key Enter a pre-shared key of between 8 and 63 case-sensitive ASCII characters (inc spaces and symbols) or 64 hexadecimal characters.							
Advance							
	w Advanced Settings button to show the fields describe below.						
Cipher Type	Select an encryption cipher type from the list.						
	 auto - This automatically chooses the best available cipher based on the cipher in use by the wireless client that is attempting to make a connection. aes - This is the Advanced Encryption Standard encryption method. It is a more recent development over TKIP and considerably more robust. Not all wireless clients may support this. 						
Idle Timeout	Enter the idle interval (in seconds) that a client can be idle before authentication is discontinued.						
Group Key Update Timer	Enter the interval (in seconds) at which the AP updates the group WPA2 encryption key.						
Pre-Authentication	Select Enable to allow pre-authentication. Otherwise, select Disable.						
Radius Settings							
Primary / Secondary Radius Server Activate	Select this to have the Zyxel Device use the specified RADIUS server.						
Radius Server IP Address	Enter the IP address of the RADIUS server to be used for authentication.						
Radius Server Port	Enter the port number of the RADIUS server to be used for authentication.						
Radius Server Secret	Enter the shared secret password of the RADIUS server to be used for authentication.						
Primary / Secondary Accounting Server Activate	Select the check box to enable user accounting through an external authentication server.						
Accounting Server IP Address	Enter the IP address of the external accounting server in dotted decimal notation.						
Accounting Server Port	Enter the port number of the external accounting server. The default port number is 1813. You need not change this value unless your network administrator instructs you to do so with additional information.						
Accounting Share Secret Enter a password (up to 128 alphanumeric characters) as the key to be shared be the external accounting server and the Zyxel Device. The key must be the same o external accounting server and your Zyxel Device. The key is not sent over the net							

Table 46 Configuration > Object > AP Profile > SSID > Security List > AAdd/Edit Security Profile> Security Mode: wpa2-mix (continued)

LABEL	DESC RIPIIO N					
Accounting Interim Update	This field is available only when you enable user accounting through an external authentication server.					
	Select this to have the Zyxel Device send subscriber status updates to the accounting server at the interval you specify.					
Interim Update Interval	Specify the time interval for how often the Zyxel Device is to send a subscriber status update to the accounting server.					
General Server Settings						
NAS IP Address	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) IP address attribute, enter it here.					
NAS Identifier	If the RADIUS server requires the Zyxel Device to provide the NAS (Network Access Server) identifier attribute, enter it here. The NAS identifier is to identify the source of access request. It could be the NAS's fully qualified domain name.					
ОК	Click OK to save your changes back to the Zyxel Device.					
Cancel	Click Cancel to exit this screen without saving your changes.					

Table 46 Configuration > Object > AP Profile > SSID > Security List > AAdd/Edit Security Profile> Security Mode: wpa2-mix (continued)

Figure 68 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile> Security Mode: wpa3

		28) 2
I an extended Setting		
General Settings		
holle Name:	detact	
Social's Pade	1948	ini #A
Automiculus Sellings		
e Blopex		
To Automic Action Terror	15	(80-50030 Journals 0 is or million)
to Personal		
Accorde	1.00.000	
Me franci	300	pe account and
Croko Key Bodote litter	30000	(10x10000/whitehold)
Pre-Ap trentecations	Disclole	*
M. Massagerman Distance Teatractic	w. grinters :	p Province
Radius Settings		
Pirtory Bricky Second Activity		
Deplie Sweet F Address		
Roots Server Port:		011-688780-
Acciestement Secretion		10
III Secondary Sadi II Sama Activ	vir:	
📕 Himory Accounting Server Act	vone	
17 Secondary Accounting Server 2	ted velo	
Access The Second Address		0
Accounting Server Form		avera 0 a-sector
Apploanting Share Secret		
Poppun Englishare Secret 19. Access inglisherin Updala		directory and the second se
	10	(1.1440 mil.ptc.)
 Access in a lefterin Updaty whether specific interval 		
W. Accounting Jelatin Update		

The following table describes the labels in this screen.

Table 47 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security Mode: wpa3

LABEL	DESC RIPIIO N
General Settings	
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.

IABEL	DESC RIPTIO N					
Security Mode	Select a security mode from the list: none, enhanced-open, wep, wpa2, wpa2-mix or wpa3.					
	enhanced-open uses Opportunistic Wireless Encryption (OWE) which encrypts the wireless connection when possible.					
Authentication Settings						
Enterprise	Select this to enable 802.1x secure authentication with a RADIUS server.					
ReAuthentication Timer	Enter the interval (in seconds) between authentication requests. Enter a 0 for unlimited time.					
Personal	This field is available when you select the wpa2 , wpa2-mix or wpa3 security mode.					
	Select this option to use a Pre-Shared Key (PSK) with WPA2 encryption or Simultaneous Authentication of Equals (SAE) with WPA3 encryption.					
Pre-Shared Key	Enter a pre-shared key of between 8 and 63 case-sensitive ASCII characters (including spaces and symbols) or 64 hexadecimal characters.					
Transition Mode	Enable this for backwards compatibility. This option is only available if the Security Mode is wpa3 or enhanced-open . This creates two virtual APs (VAPs) with a primary (wpa3 or enhanced-open) and fallback (wpa2 or none) security method.					
	If the Security Mode is wpa3, enabling this will force Management Frame Protection to be set to Optional. If this is disabled or if the Security Mode is enhanced-open, Management Frame Protection will be set to Required.					
	discontinued.					
Idle Timeout	Enter the idle interval (in seconds) that a client can be idle before authentication is discontinued.					
Group Key Update Timer	Enter the interval (in seconds) at which the AP updates the group WPA2 encryption key					
Pre-Authentication						
Management Frame Protection	Select Enable to allow pre-authentication. Otherwise, select Disable.					
	Select Enable to allow pre-authentication. Otherwise, select Disable. This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes.					
	This field is available only when you select wpa2 in the Security Mode field and set Cipher					
	This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes . Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11 w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11 i WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks.					
	This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes . Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11 WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks. Select the check box to enable management frame protection (MFP) to add security to					
	This field is available only when you select wpa2 in the Security Mode field and set Ciphen Type to aes. Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11 w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11 ii WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks. Select the check box to enable management frame protection (MFP) to add security to 802.11 management frames. Select Optional if you do not require the wireless clients to support MFP. Management					
	This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes. Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11 w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11 ii WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks. Select the check box to enable management frame protection (MFP) to add security to 802.11 management frames. Select Optional if you do not require the wireless clients to support MFP. Management frames will be encrypted if the clients support MFP. Select Require d and wireless clients must support MFP in order to join the Zyxel Device's					
Protection Radius Settings	This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes. Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11 w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11 ii WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks. Select the check box to enable management frame protection (MFP) to add security to 802.11 management frames. Select Optional if you do not require the wireless clients to support MFP. Management frames will be encrypted if the clients support MFP. Select Require d and wireless clients must support MFP in order to join the Zyxel Device's					
Protection Radius Settings Primary / Secondary	This field is available only when you select wpa2 in the Security Mode field and set Cipher Type to aes . Data frames in 802.11 WLANs can be encrypted and authenticated with WEP, WPA or WPA2. But 802.11 management frames, such as beacon/probe response, association request, association response, de-authentication and disassociation are always unauthenticated and unencrypted. IEEE 802.11w Protected Management Frames allows APs to use the existing security mechanisms (encryption and authentication methods defined in IEEE 802.11i WPA/WPA2) to protect management frames. This helps prevent wireless DoS attacks. Select the check box to enable management frame protection (MFP) to add security to 802.11 management frames. Select Optiona1 if you do not require the wireless clients to support MFP. Management frames will be encrypted if the clients support MFP.					

Table 47 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile> Security Mode: wpa3 (continued)

LABEL	DESC RIPTIO N					
Radius Server Secret	Enter the shared secret password of the RADIUS server to be used for authentication.					
Primary / Secondary Accounting Server Activate	Select the check box to enable user accounting through an external authentication server.					
Accounting Server IP Address	Enter the IP address of the external accounting server in dotted decimal notation.					
Accounting Server Port	Enter the port number of the external accounting server. The default port number is 1813. You need not change this value unless your network administrator instructs you to do so with additional information.					
Accounting Share Secret	Enter a password (up to 128 alphanumeric characters) as the key to be shared between the external accounting server and the Zyxel Device. The key must be the same on the external accounting server and your Zyxel Device. The key is not sent over the network.					
Accounting Interim Update	This field is available only when you enable user accounting through an external authentication server.					
	Select this to have the Zyxel Device send subscriber status updates to the accounting server at the interval you specify.					
Interim Update Interval	Specify the time interval for how often the Zyxel Device is to send a subscriber status update to the accounting server.					
General Server Settings						
NAS IP Address If the RADIUS server requires the Zyxel Device to provide the NAS (Network IP address attribute, enter it here.						
NAS Identifier If the RADIUS server requires the Zyxel Device to provide the NAS (Network identifier attribute, enter it here. The NAS identifier is to identify the source request. It could be the NAS's fully qualified domain name.						
ОК	Click OK to save your changes back to the Zyxel Device.					
Cancel	Click Cancel to exit this screen without saving your changes.					

Table 47 Configuration > Object > AP Profile > SSID > Security List > Add/Edit Security Profile > Security
Mode: wpa3 (continued)

12.5 MAC Filter List

This screen allows you to create and manage security configurations that can be used by your SSIDs. To access this screen click Configuration > Object > AP Profile > SSID > MAC Filter List.

Note: You can have a maximum of 32 MAC filtering profiles on the Zyxel Device.

SHULL I	Second y Usi	MAC Filter List	Euryan-Dhoro fon Dat	
C have Lien Su				
87400 Ptm	I Branina Minia	of the way of the		
Sugar Se			File Actor	
I H Rose L	at 1 A A San W	e - one Roma i	100000000000000000000000000000000000000	Y a color facility (c)

The following table describes the labels in this screen.

Table 48 Configuration > Object > AP Profile > SSID > MAC Filter List

LABEL	DESC RIPTIO N					
Add	Click this to add a new MAC filtering profile.					
Edit	Click this to edit the selected MAC filtering profile.					
Remove	Click this to remove the selected MAC filtering profile.					
Object Reference	Click this to view which other objects are linked to the selected MAC filtering profile (for example, SSID profile).					
#	This field is a sequential value, and it is not associated with a specific user.					
Profile Name	This field indicates the name assigned to the MAC filtering profile.					
Filter Action	This field indicates this profile's filter action (if any).					

12.5.1 Add/Edit MAC Filter Profile

This screen allows you to create a new MAC filtering profile or edit an existing one. To access this screen, click the **Add** button or select a MAC filter profile from the list and click the **Edit** button.

Note: Each MAC filtering profile can include a maximum of 512 MAC addresses.

Figure 70	Configuration >	Object >	AP Profile >	SSID > MAC	Filter List > .	Add/Edit MAC	C Filter Profile
-----------	-----------------	----------	--------------	------------	-----------------	--------------	------------------

hdE o blomes			
Inc Actions	GROW	66°2 👸	
Q'Agg 🗐 📖 🔳	Numicing -		
1 MARC +	- 11 12 - Sa	Distription	
M A Avec 1 18	(1) (a) (b) (2) (w) (81)	a. (10010)	 Buildes from dot

The following table describes the labels in this screen.

Table 49 Configuration > Object > AP Profile > SSID > MAC Filter List > Add/Edit MAC Filter Profile

IABEL	DESC RIPIIO N
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.
Filter Action	Select a llow to permit the wireless client with the MAC addresses in this profile to connect to the network through the associated SSID; select $deny$ to block the wireless clients with the specified MAC addresses.
Add	Click this to add a MAC address to the profile's list.
Edit	Click this to edit the selected MAC address in the profile's list.

IABEL	DESC RIPTIO N
Remove	Click this to remove the selected MAC address from the profile's list.
#	This field is a sequential value, and it is not associated with a specific user.
MAC	This field specifies a MAC address associated with this profile. You can click the MAC address to make it editable.
Description	This field displays a description for the MAC address associated with this profile. You can click the description to make it editable. Enter up to 60 characters, spaces and underscores allowed.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 49 Configuration > Object > AP Profile > SSID > MAC Filter List > Add/Edit MAC Filter Profile (continued)

12.6 Layer 2 Isolation List

Layer-2 isolation is used to prevent wireless clients associated with your Zyxel Device from communicating with other wireless clients, APs, computers or routers in a network.

In the following example, layer-2 isolation is enabled on the Zyxel Device to allow a guest wireless client (A) to access the main network router (B). The router provides access to the Internet and the network printer (C) while preventing the client from accessing other computers and servers on the network. The client can communicate with other wireless clients only if Intra-BSS Traffic blocking is disabled.

Note: Intra-BSS Traffic Blocking is activated when you enable layer-2 isolation.



Figure 71 Layer-2 Isolation Application

MAC addresses that are not listed in the layer-2 isolation table are blocked from communicating with the Zyxel Device's wireless clients except for broadcast packets. Layer-2 isolation does not check the traffic between wireless clients that are associated with the same AP. Intra-BSS traffic allows wireless clients associated with the same AP to communicate with each other.

This screen allows you to specify devices you want the users on your wireless networks to access. To access this screen click **Configuration > Object > AP Profile > SSID > Layer 2 Isolation List**.

Concerning of the second	and	1.000000000000		
630.04	security Las	MAC Sterus	Layer-Zisalahan Lot	
yer-2 lapiditor	uni Summary			
DiAce Mad	Remote Bitting	et Bellere (war		
4 Prefie Ha	A18			
Tom 1	1011 - AL 1500 - S	- x x -		Stephen Automates

Figure 72 Configuration > Object > AP Profile > SSID > Layer-2 Isolation List

The following table describes the labels in this screen.

Table 50 Configuration > Object > AP Profile > SSID > Layer-2 Isolation List

IABEL	DESC RIPIIO N
Add	Click this to add a new layer-2 isolation profile.
Edit	Click this to edit the selected layer-2 isolation profile.
Remove	Click this to remove the selected layer-2 isolation profile.
Object Reference	Click this to view which other objects are linked to the selected layer-2 isolation profile (for example, SSID profile).
#	This field is a sequential value, and it is not associated with a specific user.
Profile Name	This field indicates the name assigned to the layer-2 isolation profile.

12.6.1 Add/Edit Layer 2 Isolation Profile

This screen allows you to create a new layer-2 isolation profile or edit an existing one. To access this screen, click the **Add** button or select a layer-2 isolation profile from the list and click the **Edit** button.

- Note: You need to know the MAC address of each wireless client, AP, computer or router that you want to allow to communicate with the Zyxel Device's wireless clients.
- Figure 73 Configuration > Object > AP Profile > SSID > Layer-2 Isolation List > Add/Edit Layer-2 Isolation Profile

100	o bioroca o devices	with inevel	NAC adds	K11 H21:		
Ø e	da gene	W Innis	w.			
4	MAC -			Celerbien		
ίŧ.);	i inge a	A SILVER	5.500 53	e and	No.5cR	lo d'istri

Table 51Configuration > Object > AP Profile > SSID > Layer-2 Isolation List > Add/Edit Layer-2 IsolationProfile

FIOIIIE	
LABEL	DESC RIPIIO N
Profile Name	Enter up to 31 alphanumeric characters for the profile name. This name is only visible in the Web Configurator and is only for management purposes. Spaces and underscores are allowed.
Add	Click this to add a MAC address to the profile's list.
Edit	Click this to edit the selected MAC address in the profile's list.
Remove	Click this to remove the selected MAC address from the profile's list.
#	This field is a sequential value, and it is not associated with a specific user.
MAC	This field specifies a MAC address associated with this profile. You can click the MAC address to make it editable.
Description	This field displays a description for the MAC address associated with this profile. You can click the description to make it editable. Enter up to 60 characters, spaces and underscores allowed.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

C HAPTER 13 MON Profile

13.1 Overview

This screen allows you to set up monitor mode configurations that allow your Zyxel Device to scan for other wireless devices in the vicinity. Once detected, you can use the **Wire less > MON Mode** screen (Section 10.3 on page 79) to classify them as either rogue or friendly.

Not all Zyxel Devices support monitor mode and rogue APs detection.

13.1.1 What You Can Do in this Chapter

The **MON Profile** screen (Section 13.2 on page 117) creates preset monitor mode configurations that can be used by the Zyxel Device.

13.2 MON Profile

This screen allows you to create monitor mode configurations that can be used by the APs. To access this screen, log into the Web Configurator, and click **Configuration > Object > MON Profile**.

Figure 71 Configuration > Object > MON Profile

Acc - Com I	Benote Factors N matters Britser televice	
Malue	rio lle Nome -	
4	detaut	
fega i s	er ()	(Beg) gir (1. 3 of 3

The following table describes the labels in this screen.

LABEL	DESC RIPTIO N
Add	Click this to add a new monitor mode profile.
Edit	Click this to edit the selected monitor mode profile.
Remove	Click this to remove the selected monitor mode profile.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.

Table 50 Configuration > Object > MON Profile

IABEL	DESC RIPTIO N
Object Reference	Click this to view which other objects are linked to the selected monitor mode profile (for example, an AP management profile).
#	This field is a sequential value, and it is not associated with a specific profile.
Status	This field shows whether or not the entry is activated.
Profile Name	This field indicates the name assigned to the monitor profile.

Table 50 Configuration > Object > MON Profile (continued)

$13.2.1 \ Add/Edit\,MON\,Pro\,file$

This screen allows you to create a new monitor mode profile or edit an existing one. To access this screen, click the **Add** button or select and existing monitor mode profile and click the **Edit** button. See Section 1.2.3 on page 15 for more information about MON Mode.

Figure 72 Configuration > Object > MON Profile > Add/Edit MON Profile

Anti-ate		
not e Nome:		
Channel visiel litre;	100 (100+t+ 1000ms)	
Section France Alucies	mana 01 🖉	
l Saan Channel Lot (2)	а оні)	
Chonnel III	207	
1		
2		
3		
4		
5		
٥		
č		
**		
l Soon Channel Ust (5)	SHE)	
Counte ID		
34		
-0		
0		
-11		
1.17		
133		
137		
10 C		

Table 51	Configuration >	Object >	MON Profile >	Add/Edit MON Profile
10010 01	Configuration	00,001		

LABEL	DESC RIPTIO N
Activate	Select this to activate this monitor mode profile.
Profile Name	This field indicates the name assigned to the monitor mode profile.
Channel dwell time	Enter the interval (in milliseconds) before the Zyxel Device switches to another channel for monitoring.
Scan Channel Mode	Select auto to have the Zyxel Device switch to the next sequential channel once the Channel dwell time expires.
	Select manual to set specific channels through which to cycle sequentially when the Channel dwell time expires. Selecting this options makes the Scan Channel List options available.
Set Scan Channel List (2.4 GHz)	Select one or more than one channel to have the Zyxel Device using this profile scan the channel(s) when Scan Channel Mode is set to manual .
	These channels are limited to the 2.4 GHz range (802.11 b/g/n/ax).
Set Scan Channel List (5 GHz)	Select one or more than one channel to have the Zyxel Device using this profile scan the channel(s) when Scan Channel Mode is set to manual .
	These channels are limited to the 5 GHz range (802.11 a/n/ac/ax). Not all Zyxel Devices support both 2.4 GHz and 5 GHz frequency bands.
OK	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

C HAPTER 14 WDS Profile

14.1 Overview

This chapter shows you how to configure WDS (Wireless Distribution System) profiles for the Zyxel Device to form a WDS with other APs.

14.1.1 What You Can Do in this Chapter

The **WDS Profile** screen (Section 14.2 on page 120) creates preset WDS configurations that can be used by the Zyxel Device.

14.2 WDS Profile

This screen allows you to manage and create WDS profiles that can be used by the APs. To access this screen, click **Configuration > Object > WDS Profile**.

gure 73 Configuration > 0		
DS Summary		
Gaze 1/mit Mitemore		
Kristle Norse +	MD6 S83	
l cercet	25/mil_10251	
in in Regels (1) in it Story (r a sta	Department of the t
	(Lopping) (Resetue)	

IABEL	DESC RIPIIO N
Add	Click this to add a new profile.
Edit	Click this to edit the selected profile.
Remove	Click this to remove the selected profile.
#	This field is a sequential value, and it is not associated with a specific profile.
Profile Name	This field indicates the name assigned to the profile.
WDS SSID	This field shows the SSID specified in this WDS profile.

14.2.1 Add/Edit WDS Profile

This screen allows you to create a new WDS profile or edit an existing one. To access this screen, click the **Add** button or select and existing profile and click the **Edit** button.

Figure 74 Configuration > Object > WDS Profile > Add/Edit WDS Profile

Add WDS Profile		<u>×</u> ×
WDS Soffings		
Profile blower: WDS SSID:	0 	
Pre-Shored Key:	12345676	
	OK	Concel

Table 53	Configurati	on > Object > WDS Profile > Add/Edit WDS Profile	

IABEL	DESC RIPTIO N
Profile Name	Enter up to 31 alphanumeric characters for the profile name.
WDS SSID	Enter the SSID with which you want the Zyxel Device to connect to a root AP or repeater to form a WDS.
Pre-Shared Key	Enter a pre-shared key of between 8 and 63 case-sensitive ASCII characters (including spaces and symbols) or 64 hexadecimal characters.
	The key is used to encrypt the traffic between the APs.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

C HAPTER 15 Certific a tes

15.1 Overview

The Zyxel Device can use certificates (also called digital IDs) to authenticate users. Certificates are based on public-private key pairs. A certificate contains the certificate owner's identity and public key. Certificates provide a way to exchange public keys for use in authentication.

15.1.1 What You Can Do in this Chapter

- The My Certificates screens (Section 15.2 on page 125) generate and export self-signed certificates or certification requests and import the Zyxel Device's CA-signed certificates.
- The **Thusted Certificates** screens (Section 15.3 on page 132) save CA certificates and trusted remote host certificates to the Zyxel Device. The Zyxel Device trusts any valid certificate that you have imported as a trusted certificate. It also trusts any valid certificate signed by any of the certificates that you have imported as a trusted certificate.

15.1.2 What You Need to Know

The following terms and concepts may help as you read this chapter.

When using public-key cryptology for authentication, each host has two keys. One key is public and can be made openly available. The other key is private and must be kept secure.

These keys work like a handwritten signature (in fact, certificates are often referred to as "digital signatures"). Only you can write your signature exactly as it should look. When people know what your signature looks like, they can verify whether something was signed by you, or by someone else. In the same way, your private key "writes" your digital signature and your public key allows people to verify whether data was signed by you, or by someone else.

This process works as follows:

- 1 Tim wants to send a message to Jenny. He needs her to be sure that it comes from him, and that the message content has not been altered by anyone else along the way. Tim generates a public key pair (one public key and one private key).
- 2 Tim keeps the private key and makes the public key openly available. This means that anyone who receives a message seeming to come from Tim can read it and verify whether it is really from him or not.
- 3 Tim uses his private key to sign the message and sends it to Jenny.
- 4 Jenny receives the message and uses Tim's public key to verify it. Jenny knows that the message is from Tim, and that although other people may have been able to read the message, no-one can have altered it (because they cannot re-sign the message with Tim's private key).

5 Additionally, Jenny uses her own private key to sign a message and Tim uses Jenny's public key to verify the message.

The Zyxel Device uses certificates based on public-key cryptology to authenticate users attempting to establish a connection, not to encrypt the data that you send after establishing a connection. The method used to secure the data that you send through an established connection depends on the type of connection.

The certification authority uses its private key to sign certificates. Anyone can then use the certification authority's public key to verify the certificates.

A certification path is the hierarchy of certification authority certificates that validate a certificate. The Zyxel Device does not trust a certificate if any certificate on its path has expired or been revoked.

Certification authorities maintain directory servers with databases of valid and revoked certificates. A directory of certificates that have been revoked before the scheduled expiration is called a CRL (Certificate Revocation List). The Zyxel Device can check a peer's certificate against a directory server's list of revoked certificates. The framework of servers, software, procedures and policies that handles keys is called PKI (public-key infrastructure).

Advantages of Certificates

Certificates offer the following benefits.

- The Zyxel Device only has to store the certificates of the certification authorities that you decide to trust, no matter how many devices you need to authenticate.
- Key distribution is simple and very secure since you can freely distribute public keys and you never need to transmit private keys.

Self-signed Certificates

You can have the Zyxel Device act as a certification authority and sign its own certificates.

Factory Default Certificate

The Zyxel Device generates its own unique self-signed certificate when you first turn it on. This certificate is referred to in the GUI as the factory default certificate.

Certificate File Formats

Any certificate that you want to import has to be in one of these file formats:

- Binary X.509: This is an ITU-T recommendation that defines the formats for X.509 certificates.
- PEM (Base-64) encoded X.509: This Privacy Enhanced Mail format uses lowercase letters, uppercase letters and numerals to convert a binary X.509 certificate into a printable form.
- Binary PKCS#7: This is a standard that defines the general syntax for data (including digital signatures) that may be encrypted. A PKCS #7 file is used to transfer a public key certificate. The private key is not included. The Zyxel Device currently allows the importation of a PKS#7 file that contains a single certificate.
- PEM (Base-64) encoded PKCS#7: This Privacy Enhanced Mail (PEM) format uses lowercase letters, uppercase letters and numerals to convert a binary PKCS#7 certificate into a printable form.

• Binary PKCS#12: This is a format for transferring public key and private key certificates. The private key in a PKCS #12 file is within a password-encrypted envelope. The file's password is not connected to your certificate's public or private passwords. Exporting a PKCS #12 file creates this and you must provide it to decrypt the contents when you import the file into the Zyxel Device.

Note: Be careful not to convert a binary file to text during the transfer process. It is easy for this to occur since many programs use text files by default.

15.1.3 Verifying a Certificate

Before you import a trusted certificate into the Zyxel Device, you should verify that you have the correct certificate. You can do this using the certificate's fingerprint. A certificate's fingerprint is a message digest calculated using the MD5 or SHA1 algorithm. The following procedure describes how to check a certificate's fingerprint to verify that you have the actual certificate.

- 1 Browse to where you have the certificate saved on your computer.
- 2 Make sure that the certificate has a ".cer" or ".crt" file name extension.



3 Double-click the certificate's icon to open the Certificate window. Click the Details tab and scroll down to the Thumbprint Algorithm and Thumbprint fields.

Callinate		-
second 15.4 Section of the	6	-
9 m (385).	+1	
Post	Agus	1
Adapti Residenty 2 States Manual Adapt	argia jardat at Later. 200 (2010) ag Dinar Nano Turcost Renowa	5
Sightage .	Derig Signal and Stylf and in	
Accel Street or al	mapping wetty, data wegt	in the second second
C Patert	·	
Estella incontrational canadidate desce	Nillion (m.) Carvet	04.0
		œ

4 Use a secure method to verify that the certificate owner has the same information in the **Thumbprint Algorithm** and **Thumbprint** fields. The secure method may vary based on your situation. Possible examples would be over the telephone or through an HTTPS connection.

15.2 My Certificates

Click **Configuration > Object > Certificate > My Certificates** to open this screen. This is the Zyxel Device's summary list of certificates and certification requests.

Figure 75 Configuration > Object > Certificate > My Certificates

e n	ologe Spor	e in the				
				5.0998 acci.		
ýÖ	erificialen 8	cting				
0,	30 2 m	C Sectory	Constantion			
-	None	7.60	Schipter	$\{4\alpha,\beta^{\alpha}\}^{(1)}$	West from	Vold To:
Υ.,	defo.th	1.5716	Cithners(122.inc	201-0405123.46	2015/08/02 12:02:0	2005-00-20-15-00-2
1.0	I Devid	WE BURN	Q. 15 (24) (A) (K = 1			Stateing 1 + 1 of 1

IABEL	DESC RIPTIO N
PKI Storage Space in Use	This bar displays the percentage of the Zyxel Device's PKI storage space that is currently in use. When the storage space is almost full, you should consider deleting expired or unnecessary certificates before adding more certificates.
Add	Click this to go to the screen where you can have the Zyxel Device generate a certificate or a certification request.
Edit	Double-click an entry or select it and click Edit to open a screen with an in-depth list of information about the certificate.
Remove	The Zyxel Device keeps all of your certificates unless you specifically delete them. Uploading a new firmware or default configuration file does not delete your certificates. To remove an entry, select it and click Remove . The Zyxel Device confirms you want to remove it before doing so. Subsequent certificates move up by one when you take this action.
Object Reference	You cannot delete certificates that any of the Zyxel Device's features are configured to use. Select an entry and click Object Reference to open a screen that shows which settings use the entry.
#	This field displays the certificate index number. The certificates are listed in alphabetical order.
Name	This field displays the name used to identify this certificate. It is recommended that you give each certificate a unique name.
Туре	This field displays what kind of certificate this is.
	REQ represents a certification request and is not yet a valid certificate. Send a certification request to a certification authority, which then issues a certificate. Use the My Certific a te Im port screen to import the certificate and replace the request.
	SELF represents a self-signed certificate.
	CERT represents a certificate issued by a certification authority.
Subject	This field displays identifying information about the certificate's owner, such as CN (Common Name), OU (Organizational Unit or department), O (Organization or company) and C (Country). It is recommended that each certificate have unique subject information.

Table 54 Configuration > Object > Certificate > My Certificates

IABEL	DESC RIPHO N
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as a common name, organizational unit or department, organization or company and country. With self-signed certificates, this is the same information as in the Subject field.
Valid From	This field displays the date that the certificate becomes applicable.
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expired! message if the certificate has expired.
Import	Click Import to open a screen where you can save a certificate to the Zyxel Device.
Refresh	Click Re fre sh to display the current validity status of the certificates.

Table 54 Configuration > Object > Certificate > My Certificates (continued)

15.2.1 Add My Certificates

Click **Configuration > Object > Certificate > My Certificates** and then the **Add** icon to open the **Add My Certificates** screen. Use this screen to have the Zyxel Device create a self-signed certificate, enroll a certificate with a certification authority or generate a certification request.

Figure 76 Configuration > Object > Certificate > My Certificates > Add

Add My Certificates			2 2
Configuration			
Nome		0	
Subject Information			
 Host IP Actoress 			
🛞 Host Damain Nama			
⊕ 5-Mali			
Organizational Units		(Optional)	
Organizations		(Opel terrical)	
Town (City):		(Optional)	
State (Province):		(Optional)	
Country:		(Optional)	
Rey type:	103A-011A-2046	~	
Key Length:	2048	· bit	
Extended Key Usage			
El Server Authentication			
📄 Client Authentication			
Create a self-signed central	ate.		
 B Greaters certification reas 		r leiter menadi en officient	
 Grade a catfloation requi 			
Enrolment fratocal:		broliment protocol (90 ¹⁴	
CAServer Address	and the second sec	•	
CA Cerlificate:	Please selections	S Over Insted CArt	
Request Authentication	4.252/GL/10/10/10/2020		
Keys			
			ancol
Table 55	Configuration >	Object > Certifica	te > My Certificates > Add
----------	-----------------	--------------------	----------------------------

IABEL	DESC RIPTIO N		
Name	Type a name to identify this certificate. You can use up to 31 alphanumeric and ;'~!@#\$%^&()_+[]{}',=- characters.		
Subject Information	Use these fields to record information that identifies the owner of the certificate. You do not have to fill in every field, although you must specify a Host IP Address , Host Domain Name , or E-Mail . The certification authority may add fields (such as a serial number) to the subject information when it issues a certificate. It is recommended that each certificate have unique subject information.		
	Select a radio button to identify the certificate's owner by IP address, domain name or e- mail address. Type the IP address (in dotted decimal notation), domain name or e-mail address in the field provided. The domain name or e-mail address is for identification purposes only and can be any string.		
	A domain name can be up to 255 characters. You can use alphanumeric characters, the hyphen and periods.		
	An e-mail address can be up to 63 characters. You can use alphanumeric characters, the hyphen, the @ symbol, periods and the underscore.		
Organizational Unit	Identify the organizational unit or department to which the certificate owner belongs. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.		
Organization	Identify the company or group to which the certificate owner belongs. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.		
Town (City)	Identify the town or city where the certificate owner is located. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.		
State (Province)	Identify the state or province where the certificate owner is located. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.		
Country	Identify the nation where the certificate owner is located. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.		
Кеу Туре	The Zyxel Device uses the RSA (Rivest, Shamir and Adleman) public-key encryption algorithm. SHA1 (Secure Hash Algorithm) and SHA2 are hash algorithms used to authenticate packet data. SHA2-256 or SHA2-512 are part of the SHA2 set of cryptographic functions and they are considered even more secure than SHA1.		
	Select a key type from RSA-SHA256 and RSA-SHA512 .		
Key Length	Select a number from the drop-down list box to determine how many bits the key should use (1024 to 2048). The longer the key, the more secure it is. A longer key also uses more PKI storage space.		
Extended Key Usage	Select Server Authentic a tion to allow a web server to send clients the certificate to authenticate itself.		
	Select Client Authentication to use the certificate's key to authenticate clients to the secure gateway.		
	These radio buttons deal with how and when the certificate is to be generated.		
Create a self-signed certificate	Select this to have the Zyxel Device generate the certificate and act as the Certification Authority (CA) itself. This way you do not need to apply to a certification authority for certificates.		
Create a certification request and save it locally for later	Select this to have the Zyxel Device generate and store a request for a certificate. Use the My Certificate Edit screen to view the certification request and copy it to send to the certification authority.		
manual enrollment	Copy the certification request from the My Certificate Edit screen and then send it to the certification authority.		

IABEL	DESC RIPTIO N
Create a certification request and enroll for	Select this to have the Zyxel Device generate a request for a certificate and apply to a certification authority for a certificate.
a certificate immediately online	You must have the certification authority's certificate already imported in the Tiruste d Certific a tes screen.
	When you select this option, you must select the certification authority's enrollment protocol and the certification authority's certificate from the drop-down list boxes and enter the certification authority's server address. You also need to fill in the Reference Number and Key if the certification authority requires them.
Enrollment Protocol	This field applies when you select Create a certification request and enroll for a certificate immediately online . Select the certification authority's enrollment protocol from the drop- down list box.
	Simple Certificate Enrollment Protocol (SCEP) is a TCP-based enrollment protocol that was developed by VeriSign and Cisco.
	Certificate Management Protocol (CMP) is a TCP-based enrollment protocol that was developed by the Public Key Infrastructure X.509 working group of the Internet Engineering Task Force (IETF) and is specified in RFC 2510.
CA Server Address	This field applies when you select Create a certification request and enroll for a certificate immediately online . Enter the IP address (or URL) of the certification authority server.
	For a URL, you can use up to 511 of the following characters. a-zA-Z0-9'()+,/:.=?;!*#@\$_%-
CA Certificate	This field applies when you select Create a certification request and enroll for a certificate immediately online . Select the certification authority's certificate from the CA Certificate drop-down list box.
	You must have the certification authority's certificate already imported in the Tirusted Certific a tes screen. Click Trusted CAs to go to the Thusted Certific a tes screen where you can view (and manage) the Zyxel Device's list of certificates of trusted certification authorities.
Request Authentication	When you select Create a certification request and enroll for a certificate immediately online , the certification authority may want you to include a reference number and key to identify you when you send a certification request.
	Fill in both the Reference Number and the Key fields if your certification authority uses the CMP enrollment protocol. Just the Key field displays if your certification authority uses the SCEP enrollment protocol.
	For the reference number, use 0 to 99999999.
	For the key, use up to 31 of the following characters. a-zA-Z0-9; `~!@#\$%^&*()_+\{}':,./ <>=-
ОК	Click OK to begin certificate or certification request generation.
Cancel	Click Cancel to quit and return to the My Certificates screen.

Table 55 Configuration > Object > Certificate > My Certificates > Add (continued)

If you configured the Add My Certific a tes screen to have the Zyxel Device enroll a certificate and the certificate enrollment is not successful, you see a screen with a Return button that takes you back to the Add My Certific a tes screen. Click Return and check your information in the Add My Certific a tes screen. Make sure that the certification authority information is correct and that your Internet connection is working properly if you want the Zyxel Device to enroll a certificate online.

15.2.2 Edit My Certificates

Click Configuration > Object > Certificate > My Certificates and then the Edit icon to open the My Certificate Edit screen. You can use this screen to view in-depth certificate information and change the certificate's name.

Figure 77	Configuration >	Object > Certificate > M	y Certificates > Edit

Cardination Annual Cardination Pair Cardination Pair Contraction Pair Cardination Contraction Cardination Contrac	Edit My Certificates		2
section of an analysis of a section of a sec	Configuration		
/CNERWORDEDUCE CONTRACT CONTRACT STATUS CONTR	Noroni	diminant .	
Balance Sectional Information Type: Self-syneer 8.00° Certificane Yealance Sector Maintaine Sector Maintaine Sector Maintaine Subject: CN4-newsb 125-be_068889/90-6800 News: CN4-newsb 125-be_068889/90-6800 News: CN4-newsb 125-be_068889/90-6800 Signaliane Anal Will R5.05 newspillen Valid Thom: CN4-newsb 125-be_018889/90-6800 Signaliane Anal Will R5.05 newspillen Valid Thom: CN4-newsb 125-be_018889/90-6800 Signaliane Anal Will R5.05 newspillen Valid Thom: Sector Million Valid Thom: Sector Million Valid Thom: newspillen [1000/161] Subject Allementwe Nome: newspillen [1001/161] Nubber Downeric Subject Typewe CA, Pain Lengths Coastrolete1 Nubb Hingerprint: Subject Typewe CA, Pain Lengths Coastrolete1 Nubber Downeric Subject Typewe CA, Pain Lengths Coastrolete1 Nubb Hingerprint: AnAllet Or 1000112/0000 (PM20000000) Stational in PBM (Scase-44) Exceeded Parmet Subject Neword Newspillen (Newspillended Neword)	Continuation Parls		
SetTilacto Information Type: Sulf-Agreed X.509 Confilection Yenden: V0 Setol Number: Signature Subject: CN=nwa5102-co_36855900-860 Nume: CN=nwa5102-co_36855900-860 Signature: CN=nwa5102-co_36855900-860 Signature: CN=nwa5102-co_36855900-860 Signature: CN=nwa5102-co_36855900-860 Signature: V0100120001 GMT VolidTore: 2015 09 02 120021 GMT Sobject Alemetiew Nome: rescinception (100160) Sobject Alemetiew Nome: setSobject Vision (201600) Sobject Alemetiew Nome: setSobject Vision (201600) Sobject Rescince Controle: Sobject Vision (201600) <td>(CN=nwc5123-oc_55651320F</td> <td>680</td> <td></td>	(CN=nwc5123-oc_55651320F	680	
Type: Solf signed X.809 Carlifestic Yeator: V0 Sedal Number: Signature Subject: CN=nwab125-ac_asserd/0-read0 Name: CN=nwab125-ac_asserd/0-read0 Name: CN=nwab125-ac_asserd/0-read0 Signature: Algorithm: Valid Name: Solf of DP 12:00:01 CMP Valid Name: Solf of DP 12:00:01 CMP Valid Ta: Solf of Signature: Kay Encipherment: Data Encipherment: Cerfficient: Sign Solf of Name: Digital Signature: Kay Encipherment: Data Encipherment: Cerfficient: Sign Solf of Name: Digital Signature: Kay Encipherment: Data Encipherment: Cerfficient: Sign Solf of Name: Solf of Type=CA, Path Length: Constraint=1 MD5 Higgerprint: MS Name: California (MSS) (MSS	Reiner		
Vordian: Vol Sedial Number: Signature Subject: CN-model 12-acc_58889900-880 Kours: CN-model 12-acc_58889900-880 Signature Algorithm: Vold Parket Vold Parket: CN-model 12-acc_6889900-880 Signature Algorithm: Vold Parket Vold Parket: 2015 09 09 19:00001 GMT Vold Parket: 2016 09 09:19:00001 GMT Vold Parket: 2019-06:001 GMT Key Ubage: Dig fol Signature: Key Enciptement. Data Enciptement. Certificate Sign Parket: Abstract Vold Parket: Parket: Abstract Vold Parket	ortificate Information		
Sedel Number: Signature Sedel Number: Signature Subject: CN=nwish125-cc_b880500-6800 Nover: CN=nwish125-cc_b880500-6800 Signature: Appletitie: Volid Dram: S015.09_00-15500-01.0441 Volid Dram: S015.09_00-15500-01.0441 Volid Dram: S015.09_00-15500-01.0441 Volid Dram: S025-06-05-15000-01.0441 Volid Dram: S025-06-05-15000-01.0441 Volid Dram: S025-06-05-15000-01.0441 Volid Dram: S025-06-05-15000-01.0441 Volid Dram: socianception (1004-bit) Sobject Attended Wolfser media 123-oc_b8806900-0600 Key Usage: Digital Signature: Key Encipherment: DataBreipherment: Certificete Sign Data Constraint: Subject Type=CA, Path Length Constraint=1 MDb Hingerprint: Apple: Digital Social Micro	Type:	Self signed X.009 Certificate	
Subject: CN=nwd5125-dc_j6885990-680 Num: CN=nwd5125-dc_j6885990-680 Signalore Algorithm: sha1Will-R6A6nangdion Volid France: 2015 0P 0P 12:00:01 GMT New Algorithm: nother cyption (1004 Bft) Subject Alementary Nome: new 01:02-oc_j0889990-9800 Key Budge: Digital Signalore: Key Enciptemment. Data Enciptemment. Certificate Sign Potentiand Kay Dage Digital Signalore: Key Enciptemment. Data Enciptemment. Certificate Sign Potentiand Kay Dage Digital Signalore: Key Enciptemment. Data Enciptemment. Certificate Sign Potentiand Kay Dage Digital Signalore: Key Enciptemment. Data Enciptemment. Certificate Sign Potentiand Kay Dage Digital Signalore: Key Enciptemment. Potentiant: Art Alec C: 12:02:02:02:02:02:02:02:02:02:02:02:02:02	Verden:	N0	
Inversion CN-mixed:125-ec_06885690-6860 Signature: Age/Item che/IWIH66/Encryption Valid Evens 2015 09:00:1200001 GME Valid Evens 2015-09:00:1200001 GME Valid Evens 2015-09:00:1200001 GME Valid Evens 2015-09:00:1200001 GME Valid Evens 2015-09:00:1200001 GME Key Agerffren: nachersyption (1004 bH) Sobject At enviol eventee: newoli 125-ec_08889500680 Key Baage: Dig fol Signation: Key Enciptemment. Data Enciptemment. Certificate Sign Date: Subject Type=CA, Poth Length Constraint=1 M25 Engerprint: 49:0000-060000000000000000000000000000000	Serial Number:	Senature	
Signature Algorithm sha Wilh BSA Broggiton Void Draws 2015 09 00 1200001 GMT Void Draws 2015 09 00 1200001 GMT Void Draws 2026-06-06 1200001 GMT Void Draws 2026-06-06 1200001 GMT Key Algorithm reader cryption (1004-06) Sobject At entrol we Nome: reader Cryption (2004-06) Sobject At entrol we Nome: reader Cryption (2004-06) Sobject Reader Subject Typee CA, Poth Length Constraint=1 M25 Physeprint: 4950-000-0000000000000000000000000000000	Subject:	CN=nwd5122-dd_52284900680	
Volid Prom: 2015 0P 0P 120001 GMT Volid Pro: 2005-06-26 1200021 GMT Key Agorithm: nationaryption [1004 bit] Subject Allemative Name: neobicoryption [1004 bit] Subject Allemative Name: Dig tot Signative: Key Enciptement. Data Enciptement. Conflict to Signative: Key Enciptement. Data Enciptement. Conflict to Signative: Key Enciptement. Data Constraint: Subject Type=CA, Path Length Constraint=1 MD5 Hingerprint: 4000 PD 2500 D0 2500 PM: 4730 PH: 410 PM 2500 PM: 4730 PH: 410 PM 2500 D0 2500 D0 efficient in PEM (Ecce-54) Encoded Parmat Image: Subject Total Action D0 PM 2500 PM: 4200 PM:	lwose:	CN-rwsb125-cc_0888F870-680	
Velid Tv: 2008-06-06 12:00:21 GMT Key Agorithm: nacknowytifien (1008 bit) Subject Allemetiew Nome: nerob 120-oc_06889900680 Key Usage: Dig tot Signaline: Key Enciptement. Data Enciptement. Certificate Sign Extended Coy Usage: Subject Type=CA, Point Length Constraint=1 MD5 hingerprint: 47:00:01:06071800125020440125020440140104000 MD5 hingerprint: 47:00:01:06071800125020440125020440014000 StAT Hingerprint: 47:00:01:06071800125020440125020498-Added/bit 04:00:080209000 StAT Hingerprint: AnsAcco 01:0001:06071800:12:5220798-Added/bit 04:00:080299000 StAT Hingerprint: AnsAcco 01:0001:06071800:12:5220798-Added/bit 04:00:00:080299000 MD5 Control of Lingerprint: AnsAcco 01:0001:06071800:12:5220798-Added/bit 04:00000000000000 MD5 Control of Lingerprint: AnsAcco 01:0001:06071800:12:5220798-Added/bit 04:00000000000000000000000000000000000	Signatore Algorithms	sha1Will-RSABnergalion	
Kev Algorithm: nacknowption (1004 bit) Sobject Alematike None: nexob120-oc_088899006800 Key Usage: Digital Signature, Key Encipherment, Data Breipherment, Certificate Sign Edde, Coertralit: Subject TyperCA, Path Length, Constraint=1 MD5 Higgerprint: 49:000000000000000000000000000000000000	Volid Franc	2015 08 02 12:00:01 GMT	
Subject Alemetive Nome: news/122-sc_0/888*970-880 Key Usage: Dig fol Signal are, Key Encipherment, Data Encipherment, Certificate Sign Data dati Cay Usage: Data Controlat: Subject Type=CA, Path Length Controlate1 MD5 Hingerprint: J0:00000000000000000000000000000000000	Volid Ta:	2028-06-26 12:00:21 GMT	
Key Usage: Digital Signalise: Key Enciptement, Data Enciptement, Certificate Sign Extended Key Usage: Subject Type=CA, Poth Length, Constraint=1 MD5 Engerprint: 300/ect Type=CA, Poth Length, Constraint=1 MD5 Engerprint: 4200/otb/0502044017043020454700045100 SHAT Engerprint: 4200/otb/050204401704020004547000451000000000000000000000000000	Key Algorithm:	radbrarypfian (1024 bH)	
Extended Coy Usage: Subject Type=CA, Polic Length Constraint=1 MD5 Engenprint: 47:000000000000000000000000000000000000	Sobject Alientative Named	nwa6122-ac_0388-670-680	
Date: Constraint: Subject TyperCA, Polit Length: Constraint=1 MD5 Higgeprint: 47x0/01/266000/01/01/2002/01/01/2002/01/01/2002/01/01/2002/01/01/01/2002/01/01/01/01/01/01/01/01/01/01/01/01/01/	Key Barge:	Digital Signal are, Key Breipherment, Data Breipherment, Certificate Sy	je.
MD5 hingerprint:	Extended Key Usage:		
SHAT Hingerprint: At-Abis CriterC1:86/180:12:5207 #8:A6/4/b1:31:70.0.62.59/c/C etilicate in PEW (Sase-34) Encoded Format Image: Control of Control	Rade Constraint	Subject Type=CA, Pote Length Constraint=1	
etilicate in FEW (Sase-34) Encoded Format BEOIN X009 CERTIR CATE ATTIMITER (Update CEUS Velocity) MCAACVERINGS SPURD AQUITAD AVXS/will by TOXOQODD du ATTIMITER (Update CEUS Velocity) MCAACVERINGS WD AQUITAD AVXS/will by TOXOQODD du ATTIMITER (Update CEUS Velocity) MCAACVERINGS WD AQUITAD AVXS/will by TOXOQODD du ATTIMITER (Update CEUS Velocity) MCAACVERINGS WD AQUITAD AVXS/will by TOXOQODD du ATTIMITER (Update CEUS Velocity) MCAACVERINGS WD AQUITAD AVXS/will by TOXOQODD du ATTIMITER (Update CEUS Velocity) MCAACVERINGS WD AQUITAD AVXS/will by TOXOQODD du ATTIMITER (Update CEUS Velocity) MCAACVERINGS WD AQUITAD AVXS/will by TOXOQODD du ATTIMITER (Update CEUS VELOCITY) MCAACVERINGS WD AQUITAD AVXS/WILL BOXOQUE AVXID AVXD/	MD5 hingerprint:	4232/5/07/8320003/44C13/43C27F54780/5/46150E	
EEGIN X009 CERTRICATE MIC teC CASCQAWIDAG EVIDER TAN BOSIGN SPWORAQUE AD AMSAwil Ig/TXXQQDEDIAU ATTI ATTI MER WITK CLUG With WOMEN TAN BOAM DWIN FASM DWIN IGAT WITH TAN MIGHT MAWAY FASCH COLLIG With WOMEN AND SET USAN WITH NEN IG 40 KYSCO TROM TAN MIGHT Col Cascad State Auto AAUC NALEC BOOR SCOTTS of the NEW YOR YOU CORRACT BOOR SCOTTS Col Cascad State Auto AAUC NALEC BOOR SCOTTS of the NEW YOR YOU CORRACT BOOR SCOTTS Col Cascad State Auto AAUC NALEC BOOR SCOTTS of the NEW YOR YOU CORRACT BOOR SCOTTS Col Cascad State Auto AAUC NALEC BOOR SCOTTS of the NEW YOR YOU CORRACT BOOR SCOTTS Col Cascad State Auto AAUC NALEC BOOR SCOTTS of the NEW YOR YOU CORRACT BOOR SCOTTS Col Cascad State Auto AAUC NALEC BOOR SCOTTS of the NEW YOR YOU CORRACT BOOR SCOTTS Col Cascad State Auto AAUC NALEC BOOR SCOTTS of the NEW YOR YOU CORRACT BOOR SCOTTS Col Cascad State Auto AAUC NALEC BOOR SCOTTS OF THE NEW YOR YOU CORRACT BOOR SCOTTS MIC AND YOU CASCAD SCOTTS OF THE NEW YOR YOU CORRECT BOOR SCOTTS OF THE NEW YOR SCOTTS OF THE NE	SHALL Hingerprint:	AnAbb CTD/CT266718012520758A6576H31575/D1625950C	
MIKE BAC CASEOg Awith A gi Evisibili TAN Baskajeki SRWOH A QUITAD AMAS Awili Ightta QQDDDDdu dRELMTREW (Jest KOEUS Mitwol) OLMORA DWDWIN FASMORMURAW (For WRITA 1M) gis MI AwAY Is a MC MID A BAGN VISA MAN POSSY I DWWIN HIGH OPPOSITION CONTRACT CONTRACT AND CARGONIC ANALOG Cost Add Mare A AWAA CON AD CHARGE MORA WITH WINNIN Ight OPPOSITION CONTRACT AND CONTRACT SCORE AND A AWAY AND AD CHARGE WITH WINNIN Ight OPPOSITION CONTRACT AND CONTRACT AND GOVERNUT DIrent 72 of New York Control (1997) I MARK AND AND CONTRACT AND CONT	erhitionste in PEW (Sose-34) En	coded format	
	MIKO BACIC ASTOGANITA BIT VINE HATTI MITHI WI JAEUK OFU GIMENA MI ANA MITA MCIMIDA HABIJIN VISAT CINÇOSI SEDI GILEBALGU MALONIA GUNENU UTO MITA STANDAR MIT ALQUITA MITA STANDAR MIT NU HABITI DI MITA STANDAR MIT NU HABITI DI MITA STANDAR MIT	C[F1MD&vf2vfFA3MD&vf2[F2Fvf3vf4]F2Fvf3vfA4Mjgx MMF35ST1D&vf3vfFMNNNjg+CKrsCTBCN[gwW0C6fv40C5 L2CBCRBgCDDVscHnccgvwwegxcff1BCN[gwW0C6fv40C5 gludusfubUHSF1=Lgw0cW7x95v62vf20C767 d https://dC72f1Dc05Vcfgw0ffmu11Pc7Ay2c52vf20F05 w0ngbM10c6fa4mmAMCArSwgyTDvR050faxw64f1Cbaak cf10112aurem1010101210H182aurem42v682126b8ate66f5	

Table 56	Configuration >	Object > Certificate >	> My Certificates > Edit

LABEL	DESC RIPTIO N	
Name	This field displays the identifying name of this certificate. You can use up to 31 alphanumeric and ;'~ $!@#$ %/&()_+[]{',.=- characters.	
Certification Path	This field displays for a certificate, not a certification request.	
	Click the Re fire sh button to have this read-only text box display the hierarchy of certification authorities that validate the certificate (and the certificate itself).	
	If the issuing certification authority is one that you have imported as a trusted certification authority, it may be the only certification authority in the list (along with the certificate itself). If the certificate is a self-signed certificate, the certificate itself is the only one in the list. The Zyxel Device does not trust the certificate and displays "Not trusted" in this field if any certificate on the path has expired or been revoked.	
Refresh	Click Re fre sh to display the certification path.	
Certificate Information	These read-only fields display detailed information about the certificate.	
Туре	This field displays general information about the certificate. CA-signed means that a Certification Authority signed the certificate. Self-signed means that the certificate's owner signed the certificate (not a certification authority). "X.509" means that this certificate was created and signed according to the ITU-T X.509 recommendation that defines the formats for public-key certificates.	
Version	This field displays the X.509 version number.	
Serial Number	This field displays the certificate's identification number given by the certification authority or generated by the Zyxel Device.	
Subject	This field displays information that identifies the owner of the certificate, such as Common Name (CN), Organizational Unit (OU), Organization (O), State (ST), and Country (C).	
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as Common Name, Organizational Unit, Organization and Country.	
	With self-signed certificates, this is the same as the Subject Name field.	
	"none" displays for a certification request.	
Signature Algorithm	This field displays the type of algorithm that was used to sign the certificate.	
Valid From	This field displays the date that the certificate becomes applicable. "none" displays for a certification request.	
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expired! message if the certificate has expired. "none" displays for a certification request.	
Key Algorithm	This field displays the type of algorithm that was used to generate the certificate's key pair (the Zyxel Device uses RSA encryption) and the length of the key set in bits (1024 bits for example).	
Subject Alternative Name	This field displays the certificate owner's IP address (IP), domain name (DNS) or e-mail address (EMAIL).	
Key Usage	This field displays for what functions the certificate's key can be used. For example, "DigitalSignature" means that the key can be used to sign certificates and "KeyEncipherment" means that the key can be used to encrypt text.	
Extended Key Usage	This field displays for what EKU (Extended Key Usage) functions the certificate's key can be used.	
Basic Constraint	This field displays general information about the certificate. For example, Subject Type=CA means that this is a certification authority's certificate and "Path Length Constraint=1" means that there can only be one certification authority in the certificate's path. This field does not display for a certification request.	

LABEL	DESC RIPTIO N
MD5 Fingerprint	This is the certificate's message digest that the Zyxel Device calculated using the MD5 algorithm.
SHA1 Fingerprint	This is the certificate's message digest that the Zyxel Device calculated using the SHA1 algorithm.
Certificate in PEM (Base-64) Encoded Format	This read-only text box displays the certificate or certification request in Privacy Enhanced Mail (PEM) format. PEM uses lowercase letters, uppercase letters and numerals to convert a binary certificate into a printable form.
	You can copy and paste a certification request into a certification authority's web page, an e-mail that you send to the certification authority or a text editor and save the file on a management computer for later manual enrollment.
	You can copy and paste a certificate into an e-mail to send to friends or colleagues or you can copy and paste a certificate into a text editor and save the file on a management computer for later distribution (via floppy disk for example).
Export Certificate Only	Use this button to save a copy of the certificate without its private key. Click this button and then Save in the File Download screen. The Save As screen opens, browse to the location that you want to use and click Save .
Password	If you want to export the certificate with its private key, create a password and type it here. Make sure you keep this password in a safe place. You will need to use it if you import the certificate to another device.
Export Certificate with Private Key	Use this button to save a copy of the certificate with its private key. Type the certificate's password and click this button. Click Save in the File Download screen. The Save As screen opens, browse to the location that you want to use and click Save .
ОК	Click OK to save your changes back to the Zyxel Device. You can only change the name.
Cancel	Click Cancel to quit and return to the My Certificates screen.

Table 56 Configuration > Object > Certificate > My Certificates > Edit

15.2.3 Import Certificates

Click Configuration > Object > Certificate > My Certificates > Import to open the My Certificate Import screen. Follow the instructions in this screen to save an existing certificate to the Zyxel Device.

Note: You can import a certificate that matches a corresponding certification request that was generated by the Zyxel Device. You can also import a certificate in PKCS#12 format, including the certificate's public and private keys.

The certificate you import replaces the corresponding request in the My Certificates screen.

You must remove any spaces in the certificate's filename before you can import it.

Figure 78 Configuration > Object > Certificate > My Certificates :	> Import
--	----------

Import Co	atticates		1 x
		he certilicate file to be trine following formate	
 Minory 	2,502		
	(cse-6-1) encoded X	509	
 Binwry PEM (I 	nousev Jase 64) encoded Pl	005.37	
	PICCS#19		
- correipon - 2 ₈ 90411, A	ting to the imported	o the successful, ender Inortification must alway the certification reque	dy rold on
Do Poth :	Select a file path		Browne
Pensword:		(FICS#12 only)	
			OK Cancel

IABEL	DESC RIPTIO N
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.
	You cannot import a certificate with the same name as a certificate that is already in the Zyxel Device.
Browse	Click Browse to find the certificate file you want to upload.
Password	This field only applies when you import a binary PKCS#12 format file. Type the file's password that was created when the PKCS #12 file was exported.
OK	Click OK to save the certificate on the Zyxel Device.
Cancel	Click Cancel to quit and return to the My Certificates screen.

15.3 Trusted Certificates

Click **Configuration > Object > Certificate > Trusted Certificates** to open the **Trusted Certificates** screen. This screen displays a summary list of certificates that you have set the Zyxel Device to accept as trusted. The Zyxel Device also accepts any valid certificate signed by a certificate on this list as being trustworthy; thus you do not need to import any certificate that is signed by one of these certificates.

25	rage Space	in Use			
			2.1	Shided.	
11 0	d Certificates	Setting			
4	al E thera	··· Billicert them	liter.		
1		autoret .	Lases -	Velicinica	Vulé lo:
	2970BL-Roost	. CHW. Children C	20-21. U-127 D-234et	eu-v tersteristeries	a C., 2014/05/15/06/18/51 U.,
10	Foge I	et i + el Stev <u>e</u>	• STU		Discover (1.1.1.1

LABEL	DESC RIPTIO N
PKI Storage Space in Use	This bar displays the percentage of the Zyxel Device's PKI storage space that is currently in use. When the storage space is almost full, you should consider deleting expired or unnecessary certificates before adding more certificates.
Edit	Double-click an entry or select it and click Edit to open a screen with an in-depth list of information about the certificate.
Remove	The Zyxel Device keeps all of your certificates unless you specifically delete them. Uploading a new firmware or default configuration file does not delete your certificates. To remove an entry, select it and click Remove . The Zyxel Device confirms you want to remove it before doing so. Subsequent certificates move up by one when you take this action.
Object Reference	You cannot delete certificates that any of the Zyxel Device's features are configured to use. Select an entry and click Object Reference to open a screen that shows which settings use the entry.
#	This field displays the certificate index number. The certificates are listed in alphabetical order.
Name	This field displays the name used to identify this certificate.
Subject	This field displays identifying information about the certificate's owner, such as CN (Common Name), OU (Organizational Unit or department), O (Organization or company) and C (Country). It is recommended that each certificate have unique subject information.
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as a common name, organizational unit or department, organization or company and country. With self-signed certificates, this is the same information as in the Subject field.
Valid From	This field displays the date that the certificate becomes applicable.
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expired! message if the certificate has expired.
Import	Click Import to open a screen where you can save the certificate of a certification authority that you trust, from your computer to the Zyxel Device.
Refresh	Click this button to display the current validity status of the certificates.

Table 58 Configuration > Object > Certificate > Trusted Certificates

15.3.1 Edit Trusted Certificates

Click **Configuration > Object > Certificate > Thusted Certificates** and then a certificate's **Edit** icon to open the **Thusted Certificates Edit** screen. Use this screen to view in-depth information about the certificate, change the certificate's name and set whether or not you want the Zyxel Device to check a certification authority's list of revoked certificates before trusting a certificate issued by the certification

authority.

Sanliguediae Name:	ZysTI-RootCA	
Name	Z/#TI-RootCA	
Contribution Polity		
2C=TW/C=2yx=2CU=VPN	Department/OU=RootC A	
Keirest		
	Mirou Rom Paints and OCSP checking	
E CONServer	a non son na mana ana an ana ang	
URI:		
ID:		
Pastword		
ELIDAT Server		
Adulation	Huri:	
IE:		
Parawards		
Certificate Information		
Ivce	Self-staned X.507 Centricate	
Version:	V0	
Seciol Number:	89 or 22 9 Statistical Hardwards Statistics Statistics and the	
Subject:	CHTM, OriZyxal, OUHVEN Department, OUHReal CA	
Interest	CHW, O-2get, OU-VPN Department, OU-RootCA	
Signature Algorithms	zho 1WithRGAEn cryption	
Volid From:	2018-03-18 05:18:51 (34/1	
Valid Ta:	2014-03-13 03:13:31 GAV	
Key Algorithms	nuEnergation (2048-61)	
Subject Alternative Norm		
Key Usage:		
Extended Key Usage:		
Rode Constraint:	Subject Type=CA, Poth Longth Constraint= 1	
MD5 hingwipdinit:	16943/D5607/C67CD254008HD;EC333#D6757D533#5	
SI M Degerprint	C CUITETO ITE OZNE O 407 CUI ZECO ZI ME CRUE O CUITETO ADO O	
Certificate		
MOSECONTZAQOGO ALUE ENREMEDINGENDANDAN MER MONEMENEN (DHMDANCO AN	NERGENERN Max SY4432 AN 5 give INC 5 wOBACUHADBH SCOMAW SATUE CogwEWINI SYAWEZAYOghIVE AXADD OT FET TH IGLDA23525000 EWH HEN MENWARES MENAWEZAWWINE HINDOW DYCY OR COWLEVED MAX CATUEC gwFW nH2 WwwFzAVBgNVB	
	Shen Qimeson, Daž Soženo, Draggji MACAO Sojo Sibini Qim Nači Savena Jaž Hullof (2016 os 6. Okrali Hullof Avel 1 Horse)	
	l os (Lo	a nia

Table 59 Configuration > Object > Certificate > Trusted Certificates > Edit

IABEL	DESC RIPTIO N
Name	This field displays the identifying name of this certificate. You can change the name. You can use up to 31 alphanumeric and ;'~ $!@#$ %/&()_+[]{',.=- characters.
Certification Path	Click the Re fre sh button to have this read-only text box display the end entity's certificate and a list of certification authority certificates that shows the hierarchy of certification authorities that validate the end entity's certificate. If the issuing certification authority is one that you have imported as a trusted certificate, it may be the only certification authority in the list (along with the end entity's own certificate). The Zyxel Device does not trust the end entity's certificate and displays "Not trusted" in this field if any certificate on the path has expired or been revoked.
Refresh	Click Refiresh to display the certification path.
Enable X.509v3 CRL Distribution Points and OCSP checking	Select this check box to have the Zyxel Device check incoming certificates that are signed by this certificate against a Certificate Revocation List (CRL) or an OCSP server. You also need to configure the OSCP or LDAP server details.
OCSP Server	Select this check box if the directory server uses OCSP (Online Certificate Status Protocol).
URL	Type the protocol, IP address and pathname of the OCSP server.
ID	The Zyxel Device may need to authenticate itself in order to assess the OCSP server. Type the login name (up to 31 ASCII characters) from the entity maintaining the server (usually a certification authority).
Password	Type the password (up to 31 ASCII characters) from the entity maintaining the OCSP server (usually a certification authority).
LDAP Server	Select this check box if the directory server uses LDAP (Lightweight Directory Access Protocol). LDAP is a protocol over TCP that specifies how clients access directories of certificates and lists of revoked certificates.
Address	Type the IP address (in dotted decimal notation) of the directory server.
Port	Use this field to specify the LDAP server port number. You must use the same server port number that the directory server uses. 389 is the default server port number for LDAP.
ID	The Zyxel Device may need to authenticate itself in order to assess the CRL directory server. Type the login name (up to 31 ASCII characters) from the entity maintaining the server (usually a certification authority).
Password	Type the password (up to 31 ASCII characters) from the entity maintaining the CRL directory server (usually a certification authority).
Certificate Information	These read-only fields display detailed information about the certificate.
Туре	This field displays general information about the certificate. CA-signed means that a Certification Authority signed the certificate. Self-signed means that the certificate's owner signed the certificate (not a certification authority). X.509 means that this certificate was created and signed according to the ITU-T X.509 recommendation that defines the formats for public-key certificates.
Version	This field displays the X.509 version number.
Serial Number	This field displays the certificate's identification number given by the certification authority.
Subject	This field displays information that identifies the owner of the certificate, such as Common Name (CN), Organizational Unit (OU), Organization (O) and Country (C).
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as Common Name, Organizational Unit, Organization and Country. With self-signed certificates, this is the same information as in the Subject Name field.
Signature Algorithm	This field displays the type of algorithm that was used to sign the certificate. Some certification authorities use rsa-pkcs1-sha1 (RSA public-private key encryption algorithm and the SHA1 hash algorithm). Other certification authorities may use rsa-pkcs1-md5 (RSA public-private key encryption algorithm and the MD5 hash algorithm).

NWA/WAC/WAX Series User's Guide

LABEL	DESC RIPTIO N
Valid From	This field displays the date that the certificate becomes applicable. The text displays in red and includes a Not Yet Valid! message if the certificate has not yet become applicable.
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expiring! or Expired! message if the certificate is about to expire or has already expired.
Key Algorithm	This field displays the type of algorithm that was used to generate the certificate's key pair (the Zyxel Device uses RSA encryption) and the length of the key set in bits (1024 bits for example).
Subject Alternative Name	This field displays the certificate's owner's IP address (IP), domain name (DNS) or e-mail address (EMAIL).
Key Usage	This field displays for what functions the certificate's key can be used. For example, "DigitalSignature" means that the key can be used to sign certificates and "KeyEncipherment" means that the key can be used to encrypt text.
Basic ConstraintThis field displays general information about the certificate. For example, Subject means that this is a certification authority's certificate and "Path Length Constra means that there can only be one certification authority in the certificate's path.	
MD5 Fingerprint	This is the certificate's message digest that the Zyxel Device calculated using the MD5 algorithm. You can use this value to verify with the certification authority (over the phone for example) that this is actually their certificate.
SHA1 Fingerprint	This is the certificate's message digest that the Zyxel Device calculated using the SHA1 algorithm. You can use this value to verify with the certification authority (over the phone for example) that this is actually their certificate.
Certificate	This read-only text box displays the certificate or certification request in Privacy Enhanced Mail (PEM) format. PEM uses lowercase letters, uppercase letters and numerals to convert a binary certificate into a printable form.
	You can copy and paste the certificate into an e-mail to send to friends or colleagues or you can copy and paste the certificate into a text editor and save the file on a management computer for later distribution (via floppy disk for example).
Export Certificate	Click this button and then Save in the File Download screen. The Save As screen opens, browse to the location that you want to use and click Save .
ОК	Click OK to save your changes back to the Zyxel Device. You can only change the name.
Cancel	Click Cancel to quit and return to the Titusted Certificates screen.

 Table 59 Configuration > Object > Certificate > Trusted Certificates > Edit (continued)

15.3.2 Import Trusted Certificates

Click Configuration > Object > Certificate > Trusted Certificates > Import to open the Import Trusted Certificates screen. Follow the instructions in this screen to save a trusted certificate to the Zyxel Device.

Note: You must remove any spaces from the certificate's filename before you can import the certificate.



Import Inusted Certificates	23
Prove input the File Name	
 Moving X 500 DEM (Deterior) encoded X 500 Binery PCCC47 Moving RCCC47 Moving RCCC47 	
Die Patri : Selectio die peth	(Iburow)
	L OS Constil

Table 60 Configuration > Object > Certificate > Trusted Certificates > Import

LABEL	DESC RIPIIO N
File Path	Type in the location of the file you want to upload in this field or click \mathbf{Browse} to find it.
	You cannot import a certificate with the same name as a certificate that is already in the Zyxel Device.
Browse	Click Browse to find the certificate file you want to upload.
OK	Click OK to save the certificate on the Zyxel Device.
Cancel	Click Cancel to quit and return to the previous screen.

15.4 Technical Reference

The following section contains additional technical information about the features described in this chapter.

OCSP

OCSP (Online Certificate Status Protocol) allows an application or device to check whether a certificate is valid. With OCSP the Zyxel Device checks the status of individual certificates instead of downloading a Certificate Revocation List (CRL). OCSP has two main advantages over a CRL. The first is real-time status information. The second is a reduction in network traffic since the Zyxel Device only gets information on the certificates that it needs to verify, not a huge list. When the Zyxel Device requests certificate status information, the OCSP server returns a "expired", "current" or "unknown" response.

C HAPTER 16 System

16.1 Overview

Use the system screens to configure general Zyxel Device settings.

16.1.1 What You Can Do in this Chapter

- The Host Name screen (Section 16.2 on page 138) configures a unique name for the Zyxel Device in your network.
- The Power Mode screen (Section 17.3 on page 200) configures the Zyxel Device's power settings.
- The Date/Time screen (Section 16.3 on page 139) configures the date and time for the Zyxel Device.
- The **www** screens (Section 16.4 on page 143) configure settings for HTTP or HTTPS access to the Zyxel Device.
- The SSH screen (Section 16.5 on page 151) configures SSH (Secure SHell) for securely accessing the Zyxel Device's command line interface.
- The Te Inet screen (Section 16.6 on page 155) configures Telnet for accessing the Zyxel Device's command line interface.
- The **FIP** screen (Section 16.6 on page 155) specifies FTP server settings. You can upload and download the Zyxel Device's firmware and configuration files using FTP. Please also see Chapter 18 on page 167 for more information about firmware and configuration files.
- The **SNMP** screens (Section 17.9 on page 218) configure the Zyxel Device's SNMP settings, including profiles that define allowed SNMPv3 access.

16.2 Host Name

A host name is the unique name by which a device is known on a network. Click **Configuration > System > Host Name** to open this screen.

Figure 82	Configuration	> System >	Host Name
-----------	---------------	------------	-----------

weiche Nerviel	(Op-tenal)	
yatan destina	Nucleout	
Toroaln Narris	Springert	

IABEL	DESC RIPTIO N
System Name	Choose a descriptive name to identify your Zyxel Device device. This name can be up to 64 alphanumeric characters long. Spaces are not allowed, but dashes (-) underscores (_) and periods (.) are accepted.
System Location	Specify the name of the place where the Zyxel Device is located. You can enter up to 60 alphanumeric and '()' ,:;?! +-*/= #\$%@ characters. Spaces and underscores are allowed. The name should start with a letter.
Domain Name	Enter the domain name (if you know it) here. This name is propagated to DHCP clients connected to interfaces with the DHCP server enabled. This name can be up to 254 alphanumeric characters long. Spaces are not allowed, but dashes "-" are accepted.
Apply	Click \mathbf{Apply} to save your changes back to the Zyxel Device.
Reset	Click $\operatorname{\mathbf{Reset}}$ to return the screen to its last-saved settings.

Table 61 Configuration > System > Host Name

16.3 PowerMode

Use this screen to configure the Zyxel Device's power settings. Click **Configuration > System > Power Mode** to open this screen.

Figure 83	Configuration > System > Power Mode
-----------	-------------------------------------

Fower Mode	
Power Setting	
E Force overni	de the power mode to full power
	ure the power source can provide full power to m Interrupt tissue.
	Apply Reset

Table 62 Con	figuration > System >	Power Mode
--------------	-----------------------	------------

IABEL	DESC RIPTIO N
Force override the power mode to full power	Select this check box if you are using a PoE injector that does not support PoE negotiation. Otherwise, the Zyxel Device cannot draw full power from the power sourcing equipment. Enable this power mode to improve the Zyxel Device's performance in this situation.
	Note: Ensure that the power sourcing equipment can supply enough power to the AP to avoid abnormal system reboots.
	Note: Only enable this if you are using a passive PoE injector that is not IEEE 802.3at/bt compliant but can still provide full power.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

16.4 Date and Time

For effective scheduling and logging, the Zyxel Device system time must be accurate. The Zyxel Device has a software mechanism to set the time manually or get the current time and date from an external server.

To change your Zyxel Device's time based on your local time zone and date, click **Configuration > System > Date/Time**. The screen displays as shown. You can manually set the Zyxel Device's time and date or have the Zyxel Device get the date and time from a time server.

Figure 84 Configuration > System > Date/Time

Date/Time	
Current Time and Date	
Current Time:	202206 (540-020)
Current Done:	2019-07-30
Time and Date Setup	
Manual	
New York (Marrina)	
new thits (www.enn.cidi):	
Get born Time Server	
Time Server Address*:	Oppolintp.org
"Optimiz: There is o pre-de	med MP Ime server bit.
Time Jone Setup	
Tene Zonal	(GA/E0000); Gesenwich Maan Time : Dublin, Edinburgh, =
Enable Daytight Saving	
Distants.	HE A HARD AND AND AND AND
Street (Associ	the product of the second of the state
Client	Transfer allows
	Apply Reset

The following table describes the labels in this screen.

Table 63	Configuration >	System >	Date/Time
----------	-----------------	----------	-----------

LABEL	DESC RIPIIO N		
Current Time and Date			
Current Time This field displays the present time of your Zyxel Device.			
Current Date This field displays the present date of your Zyxel Device.			
Time and Date Setup			
Manual	Select this radio button to enter the time and date manually. If you configure a new time and date, time zone and daylight saving at the same time, the time zone and daylight saving will affect the new time and date you entered. When you enter the time settings manually, the Zyxel Device uses the new setting once you click Apply .		
New Time (hh:mm:ss)	This field displays the last updated time from the time server or the last time configured manually. When you set Time and Date Setup to Manual , enter the new time in this field and then click Apply .		

NWA/WAC/WAX Series User's Guide

IABEL	DESC RIPTIO N
New Date (yyyy-mm-dd)	This field displays the last updated date from the time server or the last date configured manually.
(),),) 22)	When you set Time and Date Setup to Manual , enter the new date in this field and then click Apply .
Get from Time Server	Select this radio button to have the Zyxel Device get the time and date from the time server you specify below. The Zyxel Device requests time and date settings from the time server under the following circumstances.
	 When the Zyxel Device starts up. When you click Apply or Sync. Now in this screen. 24-hour intervals after starting up.
Time Server Address	Enter the IP address or URL of your time server. Check with your ISP/network administrator if you are unsure of this information.
Sync. Now	Click this button to have the Zyxel Device get the time and date from a time server (see the Time ServerAddress field). This also saves your changes (except the daylight saving settings).
Time Zone Setup	
Time Zone	Choose the time zone of your location. This will set the time difference between your time zone and Greenwich Mean Time (GMT).
Enable Daylight Saving	Daylight saving is a period from late spring to fall when many countries set their clocks ahead of normal local time by one hour to give more daytime light in the evening.
	Select this option if you use Daylight Saving Time.
Start Date	Configure the day and time when Daylight Saving Time starts if you selected Enable Daylight Saving . The at field uses the 24 hour format. Here are a couple of examples:
	Daylight Saving Time starts in most parts of the United States on the second Sunday of March. Each time zone in the United States starts using Daylight Saving Time at 2 A.M. local time. So in the United States you would select Second , Sunday , March and type 2 in the at field.
	Daylight Saving Time starts in the European Union on the last Sunday of March. All of the time zones in the European Union start using Daylight Saving Time at the same moment (1 A.M. GMT or UTC). So in the European Union you would select Last , Sunday , March . The time you type in the at field depends on your time zone. In Germany for instance, you would type 2 because Germany's time zone is one hour ahead of GMT or UTC (GMT+1).
End Date	Configure the day and time when Daylight Saving Time ends if you selected Enable Daylight Saving . The at field uses the 24 hour format. Here are a couple of examples:
	Daylight Saving Time ends in the United States on the first Sunday of November. Each time zone in the United States stops using Daylight Saving Time at 2 A.M. local time. So in the United States you would select First , Sunday , November and type 2 in the at field.
	Daylight Saving Time ends in the European Union on the last Sunday of October. All of the time zones in the European Union stop using Daylight Saving Time at the same moment (1 A.M. GMT or UTC). So in the European Union you would select Last , Sunday , October . The time you type in the at field depends on your time zone. In Germany for instance, you would type 2 because Germany's time zone is one hour ahead of GMT or UTC (GMT+1).
Offset	Specify how much the clock changes when daylight saving begins and ends.
	Enter a number from 1 to 5.5 (by 0.5 increments).
	For example, if you set this field to 3.5, a log occurred at 6 P.M. in local official time will appear as if it had occurred at 10:30 P.M.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

Table 63 Configuration > System > Date/Time (continued)

16.4.1 Pre-defined NTP Time Servers List

When you turn on the Zyxel Device for the first time, the date and time start at 2003-01-01 00:00:00. The Zyxel Device then attempts to synchronize with one of the following pre-defined list of Network Time Protocol (NTP) time servers.

The Zyxel Device continues to use the following pre-defined list of NTP time servers if you do not specify a time server or it cannot synchronize with the time server you specified.

Table 64Default Time Servers0.pool.ntp.org1.pool.ntp.org2.pool.ntp.org

When the Zyxel Device uses the pre-defined list of NTP time servers, it randomly selects one server and tries to synchronize with it. If the synchronization fails, then the Zyxel Device goes through the rest of the list in order from the first one tried until either it is successful or all the pre-defined NTP time servers have been tried.

16.4.2 Time Server Synchronization

Click the Sync. Now button to get the time and date from the time server you specified in the Time Server Address field.

When the Loading message appears, you may have to wait up to one minute.



The **Current Time** and **Current Date** fields will display the appropriate settings if the synchronization is successful.

If the synchronization was not successful, a log displays in the View Log screen. Try re-configuring the Date/Time screen.

To manually set the Zyxel Device date and time:

- 1 Click System > Date / Time.
- 2 Select Manual Under Time and Date Setup.
- 3 Enter the Zyxel Device's time in the New Time field.
- 4 Enter the Zyxel Device's date in the New Date field.
- 5 Under Time Zone Setup, select your Time Zone from the list.
- 6 As an option you can select the **Enable Daylight Saving** check box to adjust the Zyxel Device clock for daylight savings.
- 7 Click Apply.

To get the Zyxel Device date and time from a time server:

- 1 Click System > Date / Time.
- 2 Select Get from Time Server Under Time and Date Setup.
- 3 Under Time Zone Setup, select your Time Zone from the list.
- 4 Under Time and Date Setup, enter a Time Server Address.
- 5 Click Apply.

16.5 WWW Overview

The following figure shows secure and insecure management of the Zyxel Device coming in from the WAN. HTTPS and SSH access are secure. HTTP, Telnet, and FTP management access are not secure.



Figure 86 Secure and Insecure Service Access From the WAN

16.5.1 Service Access Limitations

A service cannot be used to access the Zyxel Device when you have disabled that service in the corresponding screen.

16.5.2 System Timeout

There is a lease timeout for administrators. The Zyxel Device automatically logs you out if the management session remains idle for longer than this timeout period. The management session does not time out when a statistics screen is polling.

Each user is also forced to log in the Zyxel Device for authentication again when the reauthentication time expires.

You can change the timeout settings in the User screens.

16.5.3 HTIPS

You can set the Zyxel Device to use HTTP or HTTPS (HTTPS adds security) for Web Configurator sessions.

HTTPS (HyperText Transfer Protocol over Secure Socket Layer, or HTTP over SSL) is a web protocol that encrypts and decrypts web pages. Secure Socket Layer (SSL) is an application-level protocol that enables secure transactions of data by ensuring confidentiality (an unauthorized party cannot read the transferred data), authentication (one party can identify the other party) and data integrity (you know if data has been changed).

It relies upon certificates, public keys, and private keys (see Chapter 15 on page 122 for more information).

HTTPS on the Zyxel Device is used so that you can securely access the Zyxel Device using the Web Configurator. The SSL protocol specifies that the HTTPS server (the Zyxel Device) must always authenticate itself to the HTTPS client (the computer which requests the HTTPS connection with the Zyxel Device), whereas the HTTPS client only should authenticate itself when the HTTPS server requires it to do so (select **Authenticate Client Certificates** in the **WWW** screen). **Authenticate Client Certificates** is optional and if selected means the HTTPS client must send the Zyxel Device a certificate. You must apply for a certificate for the browser from a CA that is a trusted CA on the Zyxel Device.

Please refer to the following figure.

- 1 HTTPS connection requests from an SSL-aware web browser go to port 443 (by default) on the Zyxel Device's web server.
- 2 HTTP connection requests from a web browser go to port 80 (by default) on the Zyxel Device's web server.



Figure 87 HTTP/HTTPS Implementation

Note: If you disable **HTTP** in the **WWW** screen, then the Zyxel Device blocks all HTTP connection attempts.

16.5.4 Configuring WWW Service Control

Click **Configuration > System > WWW** to open the **WWW** screen. Use this screen to specify HTTP or HTTPS settings.

Figure 88	Configuration	> System > WWW :	> Service Control
-----------	---------------	------------------	-------------------

Sentce Control	,		
HTTPS.			
W Bubb Scientificati	122		
📄 Authenflöhre Cilen/	Celtificates (Son Trav	er a cong	
Server Certificate:	derout		
III Foreford UTP to UTP	8 94.0 mpc		
HTTP			
vi Tontik			
Server Ports	00		
		App ()	

LABEL	DESC RIPTIO N
HTTPS	
Enable Select the check box to allow or disallow the computer with the IP address that is the IP address (es) in the Service Control table to access the Zyxel Device Web Configurator using secure HTTPs connections.	
Server Port	The HTTPS server listens on port 443 by default. If you change the HTTPS server port to a different number on the Zyxel Device, for example 8443, then you must notify people who need to access the Zyxel Device Web Configurator to use "https://Zyxel Device IP Address:8443" as the URL.
Authenticate Client Certificates	Select Authenticate Client Certificates (optional) to require the SSL client to authenticate itself to the Zyxel Device by sending the Zyxel Device a certificate. To do that the SSL client must have a CA-signed certificate from a CA that has been imported as a trusted CA on the Zyxel Device.
Server Certificate	Select a certificate the HTTPS server (the Zyxel Device) uses to authenticate itself to the HTTPS client. You must have certificates already configured in the My Certific a tes screen.
Redirect HTTP to HTTPS	To allow only secure Web Configurator access, select this to redirect all HTTP connection requests to the HTTPS server.
HTTP	
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the Zyxel Device Web Configurator using HTTP connections.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service to access the Zyxel Device.
Apply	Click $Apply$ to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

Table 65 Configuration > System > WWW > Service Control

16.5.5 HTIPS Example

If you have not changed the default HTTPS port on the Zyxel Device, then in your browser enter "https:// Zyxel Device IP Address/" as the web site address where "Zyxel Device IP Address" is the IP address or domain name of the Zyxel Device you wish to access.

16.5.5.1 Google Chrome Warning Messages

When you attempt to access the Zyxel Device HTTPS server, you will see the error message shown in the following screen.

```
Figure 89 Security Alert Dialog Box (Google Chrome)
```



Select Advanced > Proceed to 192.168.1.2 (unsafe) to proceed to the Web Configurator login screen.

16.5.5.2 Mozilla Fire fox Warning Messages

When you attempt to access the Zyxel Device HTTPS server, a Warning screen appears as shown in the following screen. Click **Leam More...** if you want to verify more information about the certificate from the Zyxel Device.

Click Advanced > Accept the Risk and Continue.



Warning: Potential Security Risk Ahead
Firefox detected a potential security threat and did not continue to 192.158.1.2. If you visit this site, attackers could try to steal information like your passwords, emails, or oradit card details.
Learn more
tio Badi (Recommended) Advanced
Websites prove their identity via certificates. Firefor does not trust this site because it uses a certificate that is not valid for 1921681.2. The certificate is only valid for . Error code: MOZELIA_PROX_ERROR_SELF_SIONED_CERT View Certificate
Go Back (Recommended) Accept the Rek and Costinue

16.5.5.3 Avoiding Browser Warning Messages

Here are the main reasons your browser displays warnings about the Zyxel Device's HTTPS server certificate and what you can do to avoid seeing the warnings:

- The issuing certificate authority of the Zyxel Device's HTTPS server certificate is not one of the browser's trusted certificate authorities. The issuing certificate authority of the Zyxel Device's factory default certificate is the Zyxel Device itself since the certificate is a self-signed certificate.
- For the browser to trust a self-signed certificate, import the self-signed certificate into your operating system as a trusted certificate.
- To have the browser trust the certificates issued by a certificate authority, import the certificate authority's certificate into your operating system as a trusted certificate. Refer to Appendix A on page 206 for details.

16.5.5.4 Enrolling and Importing SSLC lient Certificates

The SSL client needs a certificate if Authenticate Client Certificates is selected on the Zyxel Device.

You must have imported at least one trusted CA to the Zyxel Device in order for the **Authenticate Client Certificates** to be active (see the Certificates chapter for details).

Apply for a certificate from a Certification Authority (CA) that is trusted by the Zyxel Device (see the Zyxel Device's **Thusted Certificates** Web Configurator screen).

Figure 91 Trusted Certificates

Shor	rage Space In	r Úcie :			
			23/24.1892.		
te d	Certification 2	etting			
(Tay	d Electro	Ballostiterente			
	Norre +	subject	Loger .	Weldmen.	Vule for
	2998D-Reiot	CHW. CHENEL OUW	U-IW D-Synet OU-Y	2013-05-13 00:12/01 01	2014/05/75/06/18/51 U
	Face 5 ct	1 +			Distance 1 1 and 1
1. 4					

The CA sends you a package containing the CA's trusted certificate(s), your personal certificate(s) and a password to install the personal certificate(s).

16.5.5.5 Installing a Personal Certificate

You need a password in advance. The CA may issue the password or you may have to specify it during the enrollment. Double-click the personal certificate given to you by the CA to produce a screen similar to the one shown next.

1 Click **Next** to begin the wizard.



2 The file name and path of the certificate you double-clicked should automatically appear in the File name text box. Click **Browse** if you wish to import a different certificate.

		Browner
oter Moretherus	a cartification care has strengt in a smalle Mar.	in the following formation
		In the reserving formatio
Personal Inform	Rice Exchange- PRCS #12 (PFE, P12)	
Cryptographic M	essage Syntax Mandard-PRCS #7 Certific	cates (1978)
Bernard Small	nd Certificate Store (.351)	1101010101
THE COST COST OF	and response includes reported of Fight (A	

3 Enter the password given to you by the CA.

Certificate Import Woard			×
Passesied To maintain security, the private tay use	potested with a	passord.	
Type the password for the private key. Enumeral			
Bodde strang private key protects prompted every trace the private k nockstow if you enable this optic	ey is used to an		
F gark the private key as exportable	-		
	(\$et	ljøzt >	Canesi

4 Have the wizard determine where the certificate should be saved on your computer or select Place all certificates in the following store and choose a different location.

 Automatically send the ce Figure all net ficates in the literation 	ethoite store based on the type of certificate
Carrier and a service of the service	conserve of second
	1. Down

5 Click Finish to complete the wizard and begin the import process.

Certificate Import Woard		Certificate Import	IX.
E	Wizard You have successfully range woard.	fected the Carthicade Deport	
	You have specified the follo	who settings:	
	Certificate Store Selected Control File Name	Automatically determined by t proc brithrometry_2000-109/01121/pr	
	4		
	(jei	Fissh Careed	Ð,

6 You should see the following screen when the certificate is correctly installed on your computer.



16.5.5.6 Using a Certificate When Accessing the Zyxel Device

To access the Zyxel Device via HTTPS:

1 Enter 'https://Zyxel Device IP Address/' in your browser's web address field.



2 When Authentic ate Client Certific ates is selected on the Zyxel Device, the following screen asks you to select a personal certificate to send to the Zyxel Device. This screen displays even if you only have a single certificate as in the example.

lines Authoritication		<u>11×</u>
Identification		
The Web a	te you want to view requests identific tertificate to use when correlating	adapte.
	care to the meriod record	
lettr		
	Moss Mrs. View	Certificate.
	OK I	Tarcel
		Hotosin Politica

3 You next see the Web Configurator login screen.

16.6 SSH

You can use SSH (Secure SHell) to securely access the Zyxel Device's command line interface.

SSH is a secure communication protocol that combines authentication and data encryption to provide secure encrypted communication between two hosts over an unsecured network. In the following figure, computer B on the Internet uses SSH to securely connect to the Zyxel Device (A) for a management session.

Figure 92 SSH Communication Over the WAN Example



16.6.1 How SSH Works

The following figure is an example of how a secure connection is established between two remote hosts using SSH v1.





1 Host Identification

The SSH client sends a connection request to the SSH server. The server identifies itself with a host key. The client encrypts a randomly generated session key with the host key and server key and sends the result back to the server.

The client automatically saves any new server public keys. In subsequent connections, the server public key is checked against the saved version on the client computer.

2 Encryption Method

Once the identification is verified, both the client and server must agree on the type of encryption method to use.

3 Authentication and Data Transmission

After the identification is verified and data encryption activated, a secure tunnel is established between the client and the server. The client then sends its authentication information (user name and password) to the server to log in to the server.

16.6.2 SSH Implementation on the Zyxel Device

Your Zyxel Device supports SSH versions 1 and 2 using RSA authentication and four encryption methods (AES, 3DES, Archfour, and Blowfish). The SSH server is implemented on the Zyxel Device for management using port 22 (by default).

16.6.3 Requirements for Using SSH

You must install an SSH client program on a client computer (Windows or Linux operating system) that is used to connect to the Zyxel Device over SSH.

16.6.4 Configuring SSH

Click **Configuration > System > SSH** to open the following screen. Use this screen to configure your Zyxel Device's Secure Shell settings.

Note: It is recommended that you disable Telnet and FTP when you configure SSH for secure connections.

		-
SSH		
General Settings		
Enable		
Vesion 1		
Server Port:	22	
Server Certificate:	default	w
		Apply Reset

LABEL	DESC RIPIIO N
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the Zyxel Device CLI using this service.
	Note: The Zyxel Device uses only SSH version 2 protocol.
Version 1	Select the check box to have the Zyxel Device use both SSH version 1 and version 2 protocols. If you clear the check box, the Zyxel Device uses only SSH version 2 protocol.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.
Server Certificate	Select the certificate whose corresponding private key is to be used to identify the Zyxel Device for SSH connections. You must have certificates already configured in the My Certificates screen.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

Table 66 Configuration > System > SSH

16.6.5 Examples of Secure Telnet Using SSH

This section shows two examples using a command interface and a graphical interface SSH client program to remotely access the Zyxel Device. The configuration and connection steps are similar for most SSH client programs. Refer to your SSH client program user's guide.

16.6.5.1 Example 1: Microsoft Windows

This section describes how to access the Zyxel Device using the Secure Shell Client program.

- 1 Launch the SSH client and specify the connection information (IP address, port number) for the Zyxel Device.
- **2** Configure the SSH client to accept connection using SSH version 2.
- 3 A window displays prompting you to store the host key in you computer. Click Yes to continue.
 - Figure 95 SSH Example 1: Store Host Key

Sile Seat	a v, Novina		
100	Unknown Bast key		
100	terni depatrister Lidore Pertechyd	i 2 juni 2 juni ego ereder bel reden soret eredere e (ered	azi un deș ol e nem le se
	The year that is access the	chatting)	
	Americas	(Sence

Enter the password to log in to the Zyxel Device. The CLI screen displays next.

16.6.5.2 Example 2: Linux

This section describes how to access the Zyxel Device using the OpenSSH client program that comes with most Linux distributions.

1 Test whether the SSH service is available on the Zyxel Device.

Enter "telnet 192.168.1.2 22" at a terminal prompt and press [ENTER]. The computer attempts to connect to port 22 on the Zyxel Device (using the default IP address of 192.168.1.2).

A message displays indicating the SSH protocol version supported by the Zyxel Device.

Figure 96 SSH Example 2: Test

```
$ telnet 192.168.1.2 22
Trying 192.168.1.2...
Connected to 192.168.1.2.
Escape character is '^]'.
SSH-1.5-1.0.0
```

2 Enter "ssh -2 192.168.1.2". This command forces your computer to connect to the Zyxel Device using SSH version 1. If this is the first time you are connecting to the Zyxel Device using SSH, a message displays prompting you to save the host information of the Zyxel Device. Type "yes" and press [ENTER].

Then enter the password to log in to the Zyxel Device.

Figure 97 SSH Example 2: Log in

```
$ ssh -2 192.168.1.2
The authenticity of host '192.168.1.2 (192.168.1.2)' can't be established.
RSA1 key fingerprint is 21:6c:07:25:7e:f4:75:80:ec:af:bd:d4:3d:80:53:d1.
Are you sure you want to continue connecting (yes/no)? yes
Warning: Permanently added '192.168.1.2' (RSA1) to the list of known hosts.
Administrator@192.168.1.2's password:
```

3 The CLI screen displays next.

16.7 Telnet

You can use Telnet to access the Zyxel Device's command line interface. Click **Configuration > System > TELNET** to configure your Zyxel Device for remote Telnet access. Use this screen to enable or disable Telnet and set the server port number.

Figure 98 Configuration > System > TELNET

TELNET	
General Settings	
🗈 bhable	
Server Port:	23
	A opty [] Notat

LABEL	DESC RIPIIO N
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the Zyxel Device CLI using this service.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.

Table 67 Configuration > System > TELNET

Table 67 Configuration > System > TELNET (continued)

LABEL	DESC RIPIIO N
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

16.8 FIP

You can upload and download the Zyxel Device's firmware and configuration files using FTP. To use this feature, your computer must have an FTP client. See Chapter 18 on page 167 for more information about firmware and configuration files.

To change your Zyxel Device's FTP settings, click **Configuration > System > FIP** tab. The screen appears as shown. Use this screen to specify FTP settings.

Figure 99 Configuration > System > FTP

FIR	
General Setting:	
🗟 brable	
III TLS required	
Server Port:	\$1
Server Certificate:	defa.tr 🕑
	Apply

IABEL	DESC RIPIIO N	
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the Zyxel Device using this service.	
TLS required	Select the check box to use FTP over TLS (Transport Layer Security) to encrypt communication.	
	This implements TLS as a security mechanism to secure FTP clients and/or servers.	
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.	
Server Certificate	Select the certificate whose corresponding private key is to be used to identify the Zyxel Device for FTP connections. You must have certificates already configured in the My Certificates screen.	
Apply	Click Apply to save your changes back to the Zyxel Device.	
Reset	Click Reset to return the screen to its last-saved settings.	

Table 68 Configuration > System > FTP

16.9 SNMP

Simple Network Management Protocol is a protocol used for exchanging management information between network devices. Your Zyxel Device supports SNMP agent functionality, which allows a manager station to manage and monitor the Zyxel Device through the network. The Zyxel Device supports SNMP version one (SNMPv1), version two (SNMPv2c), and version three (SNMPv3). The next figure illustrates an SNMP management operation.





An SNMP managed network consists of two main types of component: agents and a manager.

An agent is a management software module that resides in a managed device (the Zyxel Device). An agent translates the local management information from the managed device into a form compatible with SNMP. The manager is the console through which network administrators perform network management functions. It executes applications that control and monitor managed devices.

The managed devices contain object variables/managed objects that define each piece of information to be collected about a device. Examples of variables include such as number of packets received, node port status etc. A Management Information Base (MIB) is a collection of managed objects. SNMP allows a manager and agents to communicate for the purpose of accessing these objects.

SNMP itself is a simple request/response protocol based on the manager/agent model. The manager issues a request and the agent returns responses using the following protocol operations:

- Get Allows the manager to retrieve an object variable from the agent.
- GetNext Allows the manager to retrieve the next object variable from a table or list within an agent. In SNMPv1, when a manager wants to retrieve all elements of a table from an agent, it initiates a Get operation, followed by a series of GetNext operations.
- Set Allows the manager to set values for object variables within an agent.

• Trap - Used by the agent to inform the manager of some events.

16.9.1 Supported MIBs

The Zyxel Device supports MIB II that is defined in RFC-1213 and RFC-1215. The Zyxel Device also supports private MIBs (ZYXEL-ES-CAPWAP.MIB, ZYXEL-ES-COMMON.MIB, ZYXEL-ES-ZyXELAPMgmt.MIB, ZYXEL-ES-PROWLAN.MIB, ZYXEL-ES-RFMGMT.MIB, ZYXEL-ES-SMI.MIB, and ZYXEL-ES-WIRELESS.MIB) to collect information about CPU and memory usage and VPN total throughput. The focus of the MIBs is to let administrators collect statistical data and monitor status and performance. You can download the Zyxel Device's MIBs from www.zyxel.com.

16.9.2 SNMP Traps

The Zyxel Device will send traps to the SNMP manager when any one of the following events occurs.

OBJECTIABEL	OBJECTID	DESC RIPTIO N
linkDown	1.3.6.1.6.3.1.1.5.3	This trap is sent when the Ethernet link is down.
linkUp	1.3.6.1.6.3.1.1.5.4	This trap is sent when the Ethernet link is up.
authenticationFailure	1.3.6.1.6.3.1.1.5.5	This trap is sent when an SNMP request comes from non- authenticated hosts.

Table 69 SNMP Traps

16.9.3 Configuring SNMP

To change your Zyxel Device's SNMP settings, click **Configuration > System > SNMP** tab. The screen appears as shown. Use this screen to configure your SNMP settings. You can also configure user profiles that define allowed SNMPv3 access.

Pl Brock					
Server from	181				
isup:					
Contributility	puble	(Opelation)			
Central and		(Optional)			
 If Top Verdew? 	cont.				
🔄 \$100Pv2e					
Gel Comuscillat	polote				
Self-Commonlys	prod/				
MI SNMPKI					
13 Acre 17 and 18 frame	199				
1 Constraint	out wheaton		Strong	Statigs.	
11 1 Not 1 10 1 1	1 See - 1000				An additional a

Figure 101 Configuration > System > SNMP

IABEL	DESC RIPIIO N		
Enable	Select the check box to allow or disallow users to access the Zyxel Device using SNMP.		
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.		
Trap			
Community	Type the trap community, which is the password sent with each trap to the SNMP manager. The default is public and allows all requests.		
Destination	Type the IP address of the station to send your SNMP traps to.		
Trap Wireless Event	Select this to have the Zyxel Device send a trap to the SNMP manager when a wireless client is connected to or disconnected from the Zyxel Device.		
SNMPv2c	Select this to allow SNMP managers using SNMPv2c to access the Zyxel Device.		
Get Community	nmunity Enter the Get Community , which is the password for the incoming Get and GetNext requirement from the management station. The default is public and allows all requests.		
Set Community	nmunity Enter the Set community , which is the password for incoming Set requests from the management station. The default is private and allows all requests.		
SNMPv3	Select this to allow SNMP managers using SNMPv3 to access the Zyxel Device.		
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.		
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.		
Remove	To remove an entry, select it and click $\operatorname{\mathbf{Remove}}$. The Zyxel Device confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.		
#	This the index number of an SNMPv3 user profile.		
User Name	This is the name of the user for which this SNMPv3 user profile is configured.		
Authentication	This field displays the type of authentication the SNMPv3 user must use to connect to the Zyxel Device using this SNMPv3 user profile.		
Privacy	This field displays the type of encryption the SNMPv3 user must use to connect to the Zyxel Device using this SNMPv3 user profile.		
Privilege	This field displays whether the SNMPv3 user can have read-only or read and write access to the Zyxel Device using this SNMPv3 user profile.		
Apply	Click Apply to save your changes back to the Zyxel Device.		
Reset	Click \mathbf{Reset} to return the screen to its last-saved settings.		

Table 70 Configuration > System > SNMP

16.9.4 Adding or Editing an SNMPv3 User Profile

This screen allows you to add or edit an SNMPv3 user profile. To access this screen, click the **Configuration > System > SNMP** screen's **Add** button or select a SNMPv3 user profile from the list and click the **Edit** button.

auration > System >	SNMP > Add
(guration > System >

Nor Vielme	o.ámin	1
vehenication:	6ms	*
Mvacys	NCME	<u>(1</u>
the second	Keyconthe	12

Table 71	Configuration > S	ystem > SNMP
		,

LABEL	DESC RIPTIO N
User Name	Select the user name of the user account for which this SNMPv3 user profile is configured.
Authentication	Select the type of authentication the SNMPv3 user must use to connect to the Zyxel Device using this SNMPv3 user profile.
	Select $\mathbf{MD5}$ to require the SNMPv3 user's password be encrypted by MD5 for authentication.
	Select SHA to require the SNMPv3 user's password be encrypted by SHA for authentication.
Privacy	Select the type of encryption the SNMPv3 user must use to connect to the Zyxel Device using this SNMPv3 user profile.
	Select NONE to not encrypt the SNMPv3 communications.
	Select DES to use DES to encrypt the SNMPv3 communications.
	Select AES to use AES to encrypt the SNMPv3 communications.
Privilege	Select whether the SNMPv3 user can have read-only or read and write access to the Zyxel Device using this SNMPv3 user profile.
ОК	Click OK to save your changes back to the Zyxel Device.
Cancel	Click Cancel to exit this screen without saving your changes.

C HAPTER 17 Log and Report

17.1 Overview

Use the system screens to configure daily reporting and log settings.

17.1.1 What You Can Do In this Chapter

- The Email Daily Report screen (Section 18.2 on page 222) configures how and where to send daily reports and what reports to send.
- The Log Setting screens (Section 17.2 on page 157) specify which logs are e-mailed, where they are e-mailed, and how often they are e-mailed.

17.2 Email Daily Report

Use this screen to start or stop data collection and view various statistics about traffic passing through your Zyxel Device.

Note: Data collection may decrease the Zyxel Device's traffic throughput rate.

Click **Configuration > Log & Report > Email Daily Report** to display the following screen. Configure this screen to have the Zyxel Device e-mail you system statistics every day.

Figure 98	Configuration	> Log & Report >	Email Daily Report

Errori Daity Seport			
General Settings			
ej Encloio Emai Dolly Rep	n-		
Recail Sollings			
MakSerich	1	Obligging SMIP Server Name or IP Address	
\$31/715 Proceptions	1965	in .	
Mail Server Ports	25	ji sasta parimat	
(Acti Subject:			
團 Appendixes on the	194		
📷 Appendiants time.			
(Aat from)	Lawaren	🥡 T.S. Adister)	
Moll to:		(ktrist (Addrew)	
		(small-Address)	
		(Frinklik delawa)	
		(Erral Address)	
	8	jorral Addren)	
SVP Autrentication			
Law Homen			
P (b C and a l l)			
helding at how			
behedule			
this for which graper t	<u>u</u> , 74	un) <u>oo (</u> imihuyee)	
Topost Pores			
water from the keys			
(P) CPU 0.026			
M Venary Brase			
🗑 Peri Uroge			
Wildless Report			
III Station Cost (
🔝 TOPI Proteins			
🗐 Kerel oppiners offerse	engisportuce	aven of p	
[Develop. Countres.]			
		dama [] Beat	
		Terraneous I Increased	

Table 67 Configuration > Log & Report > Email Daily Report

LABEL	DESC RIPTIO N
Enable Email Daily Report	Select this to send reports by e-mail every day.
Mail Server	Type the name or IP address of the outgoing SMTP server.

NWA/WAC/WAX Series User's Guide
LABEL	DESC RIPTIO N
SSL/TLS Encryption	Select SSL/ TLS to use Secure Sockets Layer (SSL) or Transport Layer Security (TLS) if you want encrypted communications between the mail server and the Zyxel Device.
	Select STARTIES to upgrade a plain text connection to a secure connection using SSL/TLS.
	Select \mathbf{N}_{0} to not encrypt the communications.
Mail Server Port	Enter the same port number here as is on the mail server for mail traffic.
Mail Subject	Type the subject line for the outgoing e-mail. Select Append system name to add the Zyxel Device's system name to the subject. Select Append date time to add the Zyxel Device's system date and time to the subject.
Mail From	Type the e-mail address from which the outgoing e-mail is delivered. This address is used in replies.
Mail To	Type the e-mail address (or addresses) to which the outgoing e-mail is delivered.
SMTP Authentication	Select this check box if it is necessary to provide a user name and password to the SMTP server.
User Name	This box is effective when you select the SMTP Authentic ation check box. Type the user name to provide to the SMTP server when the log is e-mailed.
Password	This box is effective when you select the SMTP Authentic a tion check box. Type the password to provide to the SMTP server when the log is e-mailed.
Send Report Now	Click this button to have the Zyxel Device send the daily e-mail report immediately.
Time for sending report	Select the time of day (hours and minutes) when the log is e-mailed. Use 24-hour notation.
Report Items	Select the information to include in the report. Select Reset counters after sending report successfully if you only want to see statistics for a 24 hour period.
Reset All Counters	Click this to discard all report data and start all of the counters over at zero.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click $\operatorname{\mathbf{Re}}\operatorname{\mathbf{se}}\operatorname{\mathbf{t}}$ to return the screen to its last-saved settings.

Table 67 Configuration > Log & Report > Email Daily Report (continued)

17.3 Log Setting

These screens control log messages and alerts. A log message stores the information for viewing (for example, in the **Monitor > View Log** screen) or regular e-mailing later, and an alert is e-mailed immediately. Usually, alerts are used for events that require more serious attention, such as system errors and attacks.

The Zyxel Device provides a system log and supports e-mail profiles and remote syslog servers. The system log is available on the **View Log** screen, the e-mail profiles are used to mail log messages to the specified destinations, and the other four logs are stored on specified syslog servers.

The **Log Setting** tab also controls what information is saved in each log. For the system log, you can also specify which log messages are e-mailed, where they are e-mailed, and how often they are e-mailed.

For alerts, the Log Setting screen controls which events generate alerts and where alerts are e-mailed.

The **Log Setting** screen provides a summary of all the settings. You can use the **Edit Log Setting** screen to maintain the detailed settings (such as log categories, e-mail addresses, server names, etc.) for any log. Alternatively, if you want to edit what events is included in each log, you can also use the **Active Log**

 $\operatorname{Summ} a\, ry$ screen to edit this information for all logs at the same time.

17.3.1 Log Setting Screen

To access this screen, click Configuration > Log & Report > Log Setting.

16	a 7 m	tycle Phoetycle			
1 ¹	Store.	Same	Jog Formal	aneras.	
	Statements of the local division of the	ladou tvo	inkesit.	E-mail Jurney 1 (400.5 cover pact Server Port, 75 10.00.5 brock all server tract 5 deject server for an term traction with representations Store you Devec forem Server Store to Server Store to Server Store to Server Store to Server Store to Server Store to Server Store to	
10	2	Sistem Log	okead	E multi Storven 2 Noti Storven Utali Storven Porti 35 Novi ILS brody pifant no Noti Subjeren opported system Hamer vez opported system Hamer vez Storve Forms Dana Libb for Storve Abit for Storve July Store for	
8	9	Remete here a	V RPT/Syskup	Šerva Addresa Log Socilep Local S	
4	8	Romoto Soever P	vernasslog	Secon Address Teg Teglity Local 1	
5	3	Percole Server 2	vorh/stysleg	Server Addisor: Tog roothy Lodd I	
6	<u>R</u>	Memote Server 3	val*1/Svelog	Servet Address: Log Rocities Everal	

Figure 99 Configuration > Log & Report > Log Setting

The following table describes the labels in this screen.

IABEL	DESC RIPIIO N
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Activate	To turn on an entry, select it and click Ac tiva te.
Inactivate	To turn off an entry, select it and click Inactivate .
#	This field is a sequential value, and it is not associated with a specific log.
Status	This field shows whether the log is active or not.
Name	This field displays the name of the log (system log or one of the remote servers).

Table 68 Configuration > Log & Report > Log Setting

	o " "			<i>,</i> ,, ,,
Table 68	Configuration >	> Loa & Report >	> Loa Settina	(continued)
10010 00	Gormgoranori	Loganopon	Logooning	

IABEL	DESC RIPTIO N
Log Format	This field displays the format of the log.
	Intemal - system log; you can view the log on the View Log tab.
	VRPT Syslog - Zyxel's Vantage Report, syslog-compatible format.
	CEF/Syslog - Common Event Format, syslog-compatible format.
Summary	This field is a summary of the settings for each log.
Active Log Summary	Click this button to open the Active Log Summary screen.
Apply	Click this button to save your changes (activate and deactivate logs) and make them take effect.

17.3.2 Edit System Log Settings

This screen controls the detailed settings for each log in the system log (which includes the e-mail profiles). Select a system log entry in the **Log Setting** screen and click the **Edit** icon.

42 2	Sing dag sive a s Sing dag sive a s Sing dag sing sing si	ur a na ú <mark>l</mark> Addres,
- 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	izi.	in Faith Cir <mark>le</mark> (alabera)
- 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	- decad	
12	to serve of provide	
	2- Chail Problems of	
), Vul Adden ; 3 (Vul Adden ; Aret 1 al		
Sec.	A.	
660		
-		525 505 6 60 6 7
	No year y William	ser provident and the series of the series o
1.1		en en anter and
i la la b ase desert		
Sector Sector Sector		and large a
468	4.5	14
		er. 14
19.8.0	18 D	10 M
10.06.05	14.14	16.0
10 M II	18 III	90.01
1000	清洁.	使用
0.24	8.0	9.10
		1999 (d)
		2 D
		- ::::::::::::::::::::::::::::::::::::
	5 T T T T T T T T T T T T T T T T T T T	8.0
		10
	10000	100 March 100 Ma
	A 10	10
A4.04.02	199.00	(Brid
0.00	M.D.	- Milli
10.00	18.11	1.000
	18 D	10.0
1.000	(A) (4	10.00
0.01	18 III	10.00
	23	12.5
1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	100 March 100 Ma	8.0
		10 Million (1990)
	107 CT	2 D
	AC 11	- 現長
ALC NORT		Lighter and a
6	(D) SZ Overetti,	
16)	(15, 53° Svensta)	

Figure 100 Configuration > Log & Report > Log Setting > Edit System Log Setting

The following table describes the labels in this screen.

Table 69 Configuration > Log & Report > Log Setting > Edit System Log Setting

IABEL	DESC RIPTIO N
E-Mail Server 1/2	
Active	Select this to send log messages and alerts according to the information in this section. You specify what kinds of log messages are included in log information and what kinds of log messages are included in alerts in the Active Log and Alert section.
Mail Server	Type the name or IP address of the outgoing SMTP server.
SSL/TLS Encryption	Select SSL/ TLS to use Secure Sockets Layer (SSL) or Transport Layer Security (TLS) if you want encrypted communications between the mail server and the Zyxel Device.
	Select STARTILS to upgrade a plain text connection to a secure connection using SSL/TLS.
	Select \mathbf{No} to not encrypt the communications.
Mail Server Port	Enter the same port number here as is on the mail server for mail traffic.
Mail Subject	Type the subject line for the outgoing e-mail. Select Append system name to add the Zyxel Device's system name to the subject. Select Append date time to add the Zyxel Device's system date and time to the subject.
Send From	Type the e-mail address from which the outgoing e-mail is delivered. This address is used in replies.
Send Log To	Type the e-mail address to which the outgoing e-mail is delivered.
Send Alerts To	Type the e-mail address to which alerts are delivered.
Sending Log	Select how often log information is e-mailed. Choices are: When Full, Hourly and When Full, Daily and When Full, and Weekly and When Full.
Day for Sending Log	This field is available if the log is e-mailed weekly. Select the day of the week the log is e-mailed.
Time for Sending Log	This field is available if the log is e-mailed weekly or daily. Select the time of day (hours and minutes) when the log is e-mailed. Use 24-hour notation.
SMTP Authentication	Select this check box if it is necessary to provide a user name and password to the SMTP server.
User Name	This box is effective when you select the SMTP Authentic ation check box. Type the user name to provide to the SMTP server when the log is e-mailed.
Password	This box is effective when you select the SMTP Authentication check box. Type the password to provide to the SMTP server when the log is e-mailed.
Active Log and Alert	
System log	Use the System Log drop-down list to change the log settings for all of the log categories.
	disable all logs (red X) - do not log any information for any category for the system log or e-mail any logs to e-mail server 1 or 2.
	enable nomallogs (green check mark) - create log messages and alerts for all categories for the system log. If e-mail server 1 or 2 also has normal logs enabled, the Zyxel Device will e-mail logs to them.
	enable nomal logs and debug logs (yellow check mark) - create log messages, alerts, and debugging information for all categories. The Zyxel Device does not e-mail debugging information, even if this setting is selected.

IABEL	DESC RIPIIO N
E-mail Server 1	Use the E-Mail Server 1 drop-down list to change the settings for e-mailing logs to e-mail server 1 for all log categories.
	Using the System Log drop-down list to disable all logs overrides your e-mail server 1 settings.
	enable nomallogs (green check mark) - e-mail log messages for all categories to e-mail server 1.
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 1.
E-mail Server 2	Use the E-Mail Server2 drop-down list to change the settings for e-mailing logs to e-mail server 2 for all log categories.
	Using the System Log drop-down list to disable all logs overrides your e-mail server 2 settings.
	enable nomallogs (green check mark) - e-mail log messages for all categories to e-mail server 2.
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 2.
#	This field is a sequential value, and it is not associated with a specific address.
Log Category	This field displays each category of messages. It is the same value used in the Display and Category fields in the View Log tab. The Default category includes debugging messages generated by open source software.
System log	Select which events you want to log by Log Category. There are three choices:
	disable all logs (red X) - do not log any information from this category
	enable nomallogs (green check mark) - create log messages and alerts from this category
	enable nomallogs and debug logs (yellow check mark) - create log messages, alerts, and debugging information from this category; the Zyxel Device does not e-mail debugging information, however, even if this setting is selected.
E-mail Server 1	Select whether each category of events should be included in the log messages when it is e-mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E Mail Server 1. The Zyxel Device does not e-mail debugging information, even if it is recorded in the System log.
E-mail Server 2	Select whether each category of events should be included in log messages when it is e- mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E-Mail Server2 . The Zyxel Device does not e-mail debugging information, even if it is recorded in the System log .
Log Consolidation	
Active	Select this to activate log consolidation. Log consolidation aggregates multiple log messages that arrive within the specified $\log C \cos i da t \cos n t$. In the $\operatorname{View} \log t$ ab, the text "[count=x]", where x is the number of original log messages, is appended at the end of the $\operatorname{Message}$ field, when multiple log messages were aggregated.
Log Consolidation Interval	Type how often, in seconds, to consolidate log information. If the same log message appears multiple times, it is aggregated into one log message with the text "[count= x]", where x is the number of original log messages, appended at the end of the Message field.
ОК	Click this to save your changes and return to the previous screen.
ÖR	

 Table 69
 Configuration > Log & Report > Log Setting > Edit System Log Setting (continued)

17.3.3 Edit Remote Server

This screen controls the settings for each log in the remote server (syslog). Select a remote server entry in the Log Setting screen and click the Edit icon.

Figure 101 Configuration > Log & Report > Log Setting > Edit Remote Server

Notive					
Logilamon		Voltovskak, Mi			
	erver Abdress	and the second s	(Server Name or IP Address)		
11	og foclity:	Local i 👘			
ety	elog				
Đ:	Selection				
	Log Caregory		Seeuka		
ð			8 88		
	Account		K (0,0		
	Noetcolh		# @ D		
	hullhin Service		8 5 0		
1	ConnectWhy Check		* © 0		
	Doily Report	8 0 0			
	Delout	8 0 0			
	Device HA	800			
8	Dynamic Requercy Selection		* 0 0		
1	DHCP	# 42 11			
9.)	He Monoger	8 62 53			
4	tone Authention		* 0 0		
25			* 0 0		
3.	Interface Statistics		8.0.0		
н.	PO		800		
5			800		
6	System Monitoring		800		
7.	0.073377543		*00		
8	Low		* 0 0 * 0 0		
			8 0 D		
	With Are frond Select	0 Miana	Dissinying 1 - 30 st 31		

The following table describes the labels in this screen.

Table 70 Configuration > Log & Report > Log Setting > Edit Remote Server

LABEL	DESC RIPIIO N
Log Settings for Re	mote Server
Active	Select this check box to send log information according to the information in this section. You specify what kinds of messages are included in log information in the Active Log section.
Log Format	This field displays the format of the log information. It is read-only.
	VRPT Syslog - Zyxel's Vantage Report, syslog-compatible format.
	$\operatorname{\mathbf{CEF}}\operatorname{\mathbf{Syslog}}$ - Common Event Format, syslog-compatible format.
Server Address	Type the server name or the IP address of the syslog server to which to send log information.
Log Facility	Select a log facility. The log facility allows you to log the messages to different files in the syslog server. Please see the documentation for your syslog program for more information.
Active Log	
Selection	Use the $\operatorname{Selection}$ drop-down list to change the log settings for all of the log categories.
	$disable \ all \log s$ (red X) - do not send the remote server logs for any log category.
	enable nomallogs (green check mark) - send the remote server log messages and alerts for all log categories.
	enable nomallogs and debug logs (yellow check mark) - send the remote server log messages, alerts, and debugging information for all log categories.
#	This field is a sequential value, and it is not associated with a specific address.
Log Category	This field displays each category of messages. It is the same value used in the Display and Category fields in the View Log tab. The Default category includes debugging messages generated by open source software.
Selection	Select what information you want to log from each Log Category (except All Logs; see below). Choices are:
	$disable \ all \log s$ (red X) - do not log any information from this category
	enable nomallogs (green checkmark) - log regular information and alerts from this category
	enable nomallogs and debug logs (yellow check mark) - log regular information, alerts, and debugging information from this category
ОК	Click this to save your changes and return to the previous screen.
Cancel	Click this to return to the previous screen without saving your changes.

17.3.4 Active Log Summary

This screen allows you to view and to edit what information is included in the system log, e-mail profiles, and remote servers at the same time. It does not let you change other log settings (for example, where and how often log information is e-mailed or remote server names). To access this screen, go to the **Log Setting** screen, and click the **Active Log Summary** button.

	n Colea	Astonios Seco	THE REAL PROPERTY.	nich anna anna		Contract of the Contract of th	STREET, SALARY	e Server (11) 🕌 Ber 9 Domoto Soeve Ser an 19 66 Se
	locuel .	0.40	8 H	N H	800	100	· # 2.0	R 40 0
E4	ue levil	0.8.0	16 M	80.00	50.0	800	100	* 0 0
8 B.	divises	故事中	R U	20 UL	000	800	0.00	0.0 0
4 C	un de la la	0.8.0	(et 11)	10 CH	0.0.6	50.0	+00	* 0 D
5 E-	dy Repair	$0 \approx 0$	82 H	18 H.	800	10 O O	6.2.0	W-0-0
6 (D-	dial.	0.019	10 M	91 (u	9:00.0	800		* 0 0
E ₂	dece HA	0.00	R U	10 UL	0.00	8.0.0	00.0	
	nan y F	0.8.0	(et 111	10	8.0.0	6.0.0	+00	# 0 D
6 E4	-02	$0 \gg 0$	名 田	18 E	* 0 0	- 新設会	6.2.0	9.0.9
10 F	v Managér	0.8.0	12 (d)	8 W.	800	800	# 9-0	800
11. F.	ing bahara	包裹 使	12 U	8 W.		80.0	0.0.0	+ 0.0
12 6	leiture -	0.8.0	18 UK	10 UL	7.00	# 0.0	+0.0	* * *
13 1	leilige &	海棠等	10.00	10.00	* 0 0	- 新設会	6.2.0	W-0-0
14 PP	1	0.8.0	N 10	81 W.	9 (D (D	8.0.0	# 2-0	* 0 0
15 89	Je-	前(新市)	92 U	10 M	8.00	8.0.0	0.00	0.0.0
16 . 29	de Mari	994	12.12	12.52	8.0.0	6.0.0	+0.0	# @ D
12 50	uffic bog	0.214	10.10	10.00	800	6.0.0	6.2.0	* 0 U
18 10.	à	0.8.0	14 M	81 UL -	B (0, 0	80.0		800
14. L W	heleo LAH.	0.8.0	10 M	10 U.	8.0.0	80.0	0.0.0	+ 0.0
20 SW	UNH Egen	0.8.0	14 U	10 U	8.0.0	50.0	+00	10 O D
14.4	2000 1 1 m 2	I In Million 2	(internal					Soleno, Contaile

This screen provides a different view and a different way of indicating which messages are included in each log and each alert. (The **Default** category includes debugging messages generated by open source software.)

The following table describes the fields in this screen.

Table 71	Configuration >	t N L a a Sattina N	Active Log Summary
			ACTIVE LOG SUTTITION

LABEL	DESC RIPIIO N
Active Log Summary	If the Zyxel Device is set to controller mode, the AC section controls logs generated by the controller and the AP section controls logs generated by the managed APs.
System log	Use the System Log drop-down list to change the log settings for all of the log categories.
	disable all logs (red X) - do not log any information for any category for the system log or e-mail any logs to e-mail server 1 or 2.
	enable normallogs (green check mark) - create log messages and alerts for all categories for the system log. If e-mail server 1 or 2 also has normal logs enabled, the Zyxel Device will e-mail logs to them.
	enable normallogs and debug logs (yellow check mark) - create log messages, alerts, and debugging information for all categories. The Zyxel Device does not e-mail debugging information, even if this setting is selected.

IABEL	DESC RIPIIO N				
E-mail Server 1	Use the E-Mail Server1 drop-down list to change the settings for e-mailing logs to e-mail server 1 for all log categories.				
	Using the System Iog drop-down list to disable all logs overrides your e-mail server 1 settings.				
	enable nomallogs (green check mark) - e-mail log messages for all categories to e-mail server 1.				
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 1.				
E-mail Server 2	Use the E-Mail Server2 drop-down list to change the settings for e-mailing logs to e-mail server 2 for all log categories.				
	Using the \mathbf{System} Iog drop-down list to disable all logs overrides your e-mail server 2 settings.				
	enable nomallogs (green check mark) - e-mail log messages for all categories to e-mail server 2.				
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 2.				
Remote Server 1~4	For each remote server, use the $\mathbf{Se} \mathbf{le} \mathbf{c} \mathbf{tio} \mathbf{n}$ drop-down list to change the log settings for all of the log categories.				
	disable all $\log s$ (red X) - do not send the remote server logs for any log category.				
	enable normallogs (green check mark) - send the remote server log messages and alerts for all log categories.				
	enable nomallogs and debug logs (yellow check mark) - send the remote server log messages, alerts, and debugging information for all log categories.				
#	This field is a sequential value, and it is not associated with a specific address.				
Log Category	This field displays each category of messages. It is the same value used in the Display and Category fields in the View Log tab. The Default category includes debugging messages generated by open source software.				
System log	Select which events you want to log by Log Category. There are three choices:				
	disable all logs (red X) - do not log any information from this category				
	enable nomallogs (green checkmark) - create log messages and alerts from this category				
	enable nomallogs and debug logs (yellow check mark) - create log messages, alerts, and debugging information from this category; the Zyxel Device does not e-mail debugging information, however, even if this setting is selected.				
E-mail Server 1 E- mail	Select whether each category of events should be included in the log messages when it is e- mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E Mail Server 1. The Zyxel Device does not e-mail debugging information, even if it is recorded in the System log.				
E-mail Server 2 E- mail	Select whether each category of events should be included in log messages when it is e- mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E Mail Server 2. The Zyxel Device does not e-mail debugging information, even if it is recorded in the System log.				
Remote Server 1~4 Syslog	For each remote server, select what information you want to log from each Log Category (except All Logs; see below). Choices are:				
	disable all logs (red X) - do not log any information from this category				
	enable nomallogs (green checkmark) - log regular information and alerts from this category				
	enable nomal logs and debug logs (yellow check mark) - log regular information, alerts, and debugging information from this category				
OK	Click this to save your changes and return to the previous screen.				
Cancel	Click this to return to the previous screen without saving your changes.				

Table 71	Configuration 2	> Log & Report :	> Loa Settina > /	Active Log Summary	(continued)
	Configuration 2	LOG & ROPOIL	- LOG Johning - /	ACTIVE LOG SOTTITION	(commoca)

CHAPTER 18 File Manager

18.1 Overview

Configuration files define the Zyxel Device's settings. Shell scripts are files of commands that you can store on the Zyxel Device and run when you need them. You can apply a configuration file or run a shell script without the Zyxel Device restarting. You can store multiple configuration files and shell script files on the Zyxel Device. You can edit configuration files or shell scripts in a text editor and upload them to the Zyxel Device. Configuration files use a .conf extension and shell scripts use a .zysh extension.

18.1.1 What You Can Do in this Chapter

- The **Configuration File** screen (Section 18.2 on page 168) stores and names configuration files. You can also download and upload configuration files.
- The Firmware Package screen (Section 18.3 on page 173) checks your current firmware version and uploads firmware to the Zyxel Device.
- The Shell Script screen (Section 18.4 on page 175) stores, names, downloads, uploads and runs shell script files.

18.1.2 What you Need to Know

The following terms and concepts may help as you read this chapter.

Configuration Files and Shell Scripts

When you apply a configuration file, the Zyxel Device uses the factory default settings for any features that the configuration file does not include. When you run a shell script, the Zyxel Device only applies the commands that it contains. Other settings do not change.

These files have the same syntax, which is also identical to the way you run CLI commands manually. An example is shown below.

Figure 102 Configuration File / Shell Script: Example

```
# enter configuration mode
configure terminal
# change administrator password
username admin password 4321 user-type admin
#configure default radio profile, change 2GHz channel to 11 & Tx output
power # to 50%
wlan-radio-profile default
2g-channel 11
output-power 50%
exit
write
```

While configuration files and shell scripts have the same syntax, the Zyxel Device applies configuration files differently than it runs shell scripts. This is explained below.

Configuration Files (.conf)	Shell Scripts (.zysh)	
 Resets to default configuration. Goes into CLI Configuration mode. Runs the commands in the configuration file. 	 Goes into CLI Privilege mode. Runs the commands in the shell script. 	

Table 71 Configuration Files and Shell Scripts in the Zyxel Device

You have to run the aforementioned example as a shell script because the first command is run in **Privilege** mode. If you remove the first command, you have to run the example as a configuration file because the rest of the commands are executed in **Configuration** mode.

Comments in Configuration Files or Shell Scripts

In a configuration file or shell script, use "#" or "!" as the first character of a command line to have the Zyxel Device treat the line as a comment.

Your configuration files or shell scripts can use "exit" or a command line consisting of a single "!" to have the Zyxel Device exit sub command mode.

Note: "exit" or "!" must follow sub commands if it is to make the Zyxel Device exit sub command mode.

In the following example lines 1 and 2 are comments. Line 7 exits sub command mode.

```
! this is from Joe
# on 2010/12/05
wlan-ssid-profile default
ssid Joe-AP
qos wmm
security default
'
```

Errors in Configuration Files or Shell Scripts

When you apply a configuration file or run a shell script, the Zyxel Device processes the file line-by-line. The Zyxel Device checks the first line and applies the line if no errors are detected. Then it continues with the next line. If the Zyxel Device finds an error, it stops applying the configuration file or shell script and generates a log.

You can change the way a configuration file or shell script is applied. Include setenv stop-on-error off in the configuration file or shell script. The Zyxel Device ignores any errors in the configuration file or shell script and applies all of the valid commands. The Zyxel Device still generates a log for any errors.

18.2 Configuration File

Click **Maintenance > File Manager > Configuration File** to open this screen. Use the **Configuration File** screen to store, run, and name configuration files. You can also download configuration files from the Zyxel Device to your computer and upload configuration files from your computer to the Zyxel Device.

Once your Zyxel Device is configured and functioning properly, it is highly recommended that you back up your configuration file before making further configuration changes. The backup configuration file will be useful in case you need to return to your previous settings.

Configuration File Flow at Restart

rightp-config-bad.conf

intgood detaut cont

putobackup-4.00.conf.

which gitte

Popel 1 of a si Show w / flame.

Incolation F

Upload Configuration File

2003

cictwici.

- If there is not a **startup-config.conf** when you restart the Zyxel Device (whether through a management interface or by physically turning the power off and back on), the Zyxel Device uses the **system-default.conf** configuration file with the Zyxel Device's default settings.
- If there is a **startup-config.conf**, the Zyxel Device checks it for errors and applies it. If there are no errors, the Zyxel Device uses it and copies it to the **lastgood.conf** configuration file as a back up file. If there is an error, the Zyxel Device generates a log and copies the **startup-config.conf** configuration file to the **startup-config-bad.conf** configuration file and tries the existing **lastgood.conf** configuration file. If there isn't a **lastgood.conf** configuration file or it also has an error, the Zyxel Device applies the **system-default.conf** configuration file.
- You can change the way the **startup**-**config.conf** file is applied. Include the setenv-startup stopon-error off command. The Zyxel Device ignores any errors in the **startup**-**config.conf** file and applies all of the valid commands. The Zyxel Device still generates a log for any errors.

2019-07-29 14:12:39

2019-07-27 14:13:20

2019-07-29-13:58:54

2019-07-29 14:14:30 2019-07-29 14:11:38

Thinking 1-7 of 7

Configuration File	Himwas Factory		
onliguration files	- Booming Bo		
N	100	Str	(miWooRec
1 statup-config.	sont	4267	2019-07-29 14:35:42
the second	cont	3955	2019-07-29 14:11:39

3876

3765.

4267

3876

To upload a cottinguration file, browse to the location of the file (conf) and then click Upload.

52

Figure 103	Maintenance > File	Manager >	Configuration File
------------	--------------------	-----------	--------------------

Do not turn off the Zyxel Device while configuration file upload is in progress.

The following table describes the labels in this screen.

IABEL	DESC RIPTIO N				
Rename	Use this button to change the label of a configuration file on the Zyxel Device. You can only rename manually saved configuration files. You cannot rename the lastgood.conf , system- default.conf and startup-config.conf files.				
	You cannot rename a configuration file to the name of another configuration file in the Zyxel Device.				
	Click a configuration file's row to select it and click Rename to open the Rename File screen.				
	D Rendme EB				
	Several Sol - subblees g 1.0 cont Toget Le:				
	Law OK-Law (Concell)				
	Specify the new name for the configuration file. Use up to 25 characters (including a-zA-Z0- 9 ;'~!@# $\%^{0}=1$.				
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.				
Remove	Click a configuration file's row to select it and click Remove to delete it from the Zyxel Device. You can only delete manually saved configuration files. You cannot delete the system - de fault.conf , startup-config.conf and lastgood.conf files.				
	A pop-up window asks you to confirm that you want to delete the configuration file. Click OK to delete the configuration file or click Cancel to close the screen without deleting the configuration file.				
Download	Click a configuration file's row to select it and click Download to save the configuration to your computer.				
Сору	Use this button to save a duplicate of a configuration file on the Zyxel Device.				
	Click a configuration file's row to select it and click Copy to open the Copy File screen.				
	后 Copy file 臣国				
	Source Reg Utamia config cont Tarpet Le:				
	(
	Specify a name for the duplicate configuration file. Use up to 25 characters (including a-zA-Z0- 9 ;'~!@#\$%^&()_+[]{',=-}.				
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.				

	intenance > File Manager > Configuration File (continued)				
IABEL	DESC RIPIIO N				
Apply	Use this button to have the Zyxel Device use a specific configuration file.				
	Click a configuration file's row to select it and click Apply to have the Zyxel Device use that configuration file. The Zyxel Device does not have to restart in order to use a different configuration file, although you will need to wait for a few minutes while the system reconfigures.				
	The following screen gives you options for what the Zyxel Device is to do if it encounters an error in the configuration file.				
	Apply Configuration Rie (학교)				
	Apply Configuration No				
	File Name: system-default.cont				
	If applying the configuration file encounters an error:				
	() Immediately stop oppiding the configuration file				
	Immediately stop applying the configuration file and rollback to the previous configuration				
	 Ignore error and initial applying the configuration file 				
	 Ignore even and light applying the configuration file and the notice to the previous configuration 				
	OK Concel				
	Immediately stop applying the configuration file - this is not recommended because it would leave the rest of the configuration blank. If the interfaces were not configured before the first error, the console port may be the only way to access the Zyxel Device.				
	Immediately stop applying the configuration file and roll back to the previous configuration - this gets the Zyxel Device started with a fully valid configuration file as quickly as possible.				
	Ignore errors and finish applying the configuration file - this applies the valid parts of the configuration file and generates error logs for all of the configuration file's errors. This lets the Zyxel Device apply most of your configuration and you can refer to the logs for what to fix.				
	Ig nore errors and finish applying the configuration file and then roll back to the previous configuration - this applies the valid parts of the configuration file, generates error logs for all of the configuration file's errors, and starts the Zyxel Device with a fully valid configuration file.				
	Click OK to have the Zyxel Device start applying the configuration file or click Cancel to close the screen.				
#	This column displays the number for each configuration file entry. This field is a sequential value and it is not associated with a specific address. The total number of configuration files that yo can save depends on the sizes of the configuration files and the available flash storage space				
File Name	This column displays the label that identifies a configuration file.				
	You cannot delete the following configuration files or change their file names.				
	The system-default.conf file contains the Zyxel Device's default settings. Select this file and click Apply to reset all of the Zyxel Device settings to the factory defaults. This configuration file is included when you upload a firmware package.				
	The startup-config.conf file is the configuration file that the Zyxel Device is currently using. If yo make and save changes during your management session, the changes are applied to this configuration file. The Zyxel Device applies configuration changes made in the Web Configurator to the configuration file when you click Apply or OK . It applies configuration changes made via commands when you use the write command.				

abla 72 Maint

apply lastgood.conf to return to a valid configuration.

This column displays the size (in KB) of a configuration file.

Size

The lastgood.conf is the most recently used (valid) configuration file that was saved when the Zyxel Device last restarted. If you upload and apply a configuration file with an error, you can

IABEL	DESC RIPTIO N
Last Modified	This column displays the date and time that the individual configuration files were last changed or saved.
Upload Configuration File	The bottom part of the screen allows you to upload a new or previously saved configuration file from your computer to your Zyxel Device.
	You cannot upload a configuration file named system-default.conf Or lastgood.conf .
	If you upload startup-config.conf , it will replace the current configuration and immediately apply the new settings.
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.
Browse	Click Browse to find the .conf file you want to upload. The configuration file must use a ".conf" filename extension. You will receive an error message if you try to upload a fie of a different format. Remember that you must decompress compressed (.zip) files before you can upload them.
Upload	Click Upload to begin the upload process. This process may take up to two minutes.

 Table 72
 Maintenance > File Manager > Configuration File (continued)

18.2.1 Example of Configuration File Download Using FIP

The following example gets a configuration file named startup-config.conf from the Zyxel Device and saves it on the computer.

- 1 Connect your computer to the Zyxel Device.
- 2 The FTP server IP address of the Zyxel Device in standalone mode is 192.168.1.2, so set your computer to use a static IP address from 192.168.1.3 ~192.168.1.254.
- 3 Use an FTP client on your computer to connect to the Zyxel Device. For example, in the Windows command prompt, type ftp 192.168.1.2. Keep the console session connected in order to see when the firmware recovery finishes.
- 4 Enter your user name when prompted.
- 5 Enter your password as requested.
- 6 Use "cd" to change to the directory that contains the files you want to download.
- 7 Use "dir" or "Is" if you need to display a list of the files in the directory.
- 8 Use "get" to download files. Transfer the configuration file on the Zyxel Device to your computer. Type get followed by the name of the configuration file. This examples uses get startup-config.conf.

```
C:\>ftp 192.168.1.2
Connected to 192.168.1.2.
220----- Welcome to Pure-FTPd [privsep] [TLS] ------
220-You are user number 1 of 5 allowed.
220-Local time is now 21:28. Server port: 21.
220-This is a private system - No anonymous login
220 You will be disconnected after 600 minutes of inactivity.
User (192.168.1.2: (none)): admin
331 User admin OK. Password required
Password:
230 OK. Current restricted directory is /
ftp> cd conf
250 OK. Current directory is /conf
ftp> ls
200 PORT command successful
150 Connecting to port 5001
lastgood.conf
startup-config.conf
system-default.conf
226 3 matches total
ftp: 57 bytes received in 0.33Seconds 0.17Kbytes/sec.
ftp> get startup-config.conf
200 PORT command successful
150 Connecting to port 5002
226-File successfully transferred
226 0.002 seconds (measured here), 1.66 Mbytes per second
ftp: 2928 bytes received in 0.02Seconds 183.00Kbytes/sec.
ftp>
```

- 9 Wait for the file transfer to complete.
- **10** Enter "quit" to exit the ftp prompt.

18.3 Firmware Package

Click Maintenance > File Manager > Fimware Package to open this screen. Use the Fimware Package screen to check your current firmware version and upload firmware to the Zyxel Device.

Note: The Web Configurator is the recommended method for uploading firmware. You only need to use the command line interface if you need to recover the firmware. See the CLI Reference Guide for how to determine if you need to recover the firmware and how to recover it.

Find the firmware package at www.zyxel.com in a file that (usually) uses a .bin extension.

The firm ware update can take up to five minutes. Do not turn off or reset the Zyxel Device while the firm ware update is in progress!

Fig ure	104	Maintenance >	> File N	/anager >	Firmware	Package
---------	-----	---------------	----------	-----------	----------	---------

Configuration Fil	e Firmware Package Shell Script
Version	
Boot Module:	VI.8
Current Version:	V6.00(ABM.3)b1
Released Date:	2019-09-19 03:19:47
Upload Rie	
To upload firmw	are, brawse to the location of the file (*Join) and then click Upload.
File:	Select a file Browse Upload

The following table describes the labels in this screen.

LABEL	DESC RIPIIO N	
Boot Module	This is the version of the boot module that is currently on the Zyxel Device.	
Current Version	This is the firmware version and the date created.	
Released Date	This is the date that the version of the firmware was created.	
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.	
Browse	Click Browse to find the .bin file you want to upload. Remember that you must decompress compressed (.zip) files before you can upload them.	
Upload	Click Upload to begin the upload process. This process may take up to two minutes.	

After you see the **Firm ware Upload in Process** screen, wait two minutes before logging into the Zyxel Device again.

Note: The Zyxel Device automatically reboots after a successful upload.

The Zyxel Device automatically restarts causing a temporary network disconnect. In some operating systems, you may see the following icon on your desktop.

Figure 105 Network Temporarily Disconnected



After five minutes, log in again and check your new firmware version in the **Dashboard** screen.

18.3.1 Example of Firmware Upload Using FIP

This procedure requires the Zyxel Device's firmware. Download the firmware package from www.zyxel.com and unzip it. The firmware file uses a .bin extension, for example, "600ABFH0C0.bin". Do the following after you have obtained the firmware file.

- 1 Connect your computer to the Zyxel Device.
- 2 The FTP server IP address of the Zyxel Device in standalone mode is 192.168.1.2, so set your computer to use a static IP address from 192.168.1.3 ~192.168.1.254.

- 3 Use an FTP client on your computer to connect to the Zyxel Device. For example, in the Windows command prompt, type ftp 192.168.1.2. Keep the console session connected in order to see when the firmware recovery finishes.
- 4 Enter your user name when prompted.
- 5 Enter your password as requested.
- 6 Enter "hash" for FTP to print a `#' character for every 1024 bytes of data you upload so that you can watch the file transfer progress.
- 7 Enter "bin" to set the transfer mode to binary.
- 8 Transfer the firmware file from your computer to the Zyxel Device. Type put followed by the path and name of the firmware file. This examples uses put C:\ftproot\Zyxel Device_FW\600ABFH0C0.bin.

```
C:\>ftp 192.168.1.2
Connected to 192.168.1.2.
220----- Welcome to Pure-FTPd [privsep] [TLS] ------
220-You are user number 1 of 5 allowed.
220-Local time is now 21:28. Server port: 21.
220-This is a private system - No anonymous login
220 You will be disconnected after 600 minutes of inactivity.
User (192.168.1.2: (none)): admin
331 User admin OK. Password required
Password:
230 OK. Current restricted directory is /
ftp> hash
Hash mark printing On ftp: (2048 bytes/hash mark) .
ftp> bin
200 TYPE is now 8-bit binary
ftp> put C:\ftproot\Zyxel Device FW\600ABFH0C0.bin
```

- 9 Wait for the file transfer to complete.
- **10** Enter "quit" to exit the ftp prompt.

18.4 Shell Script

Use shell script files to have the Zyxel Device use commands that you specify. Use a text editor to create the shell script files. They must use a ".zysh" filename extension.

Click **Maintenance > File Manager > Shell Script** to open this screen. Use the **Shell Script** screen to store, name, download, upload and run shell script files. You can store multiple shell script files on the Zyxel Device at the same time.

Note: You should include write commands in your scripts. If you do not use the write command, the changes will be lost when the Zyxel Device restarts. You could use multiple write commands in a long script.

Figure 106 Maintenance > File Manager > Shell Script

and the second second		ware Rockoge	Shell Script	L	
heli Scrip	nta				
China	mini 🗮 Harmone 🖽 O	www.itt 🔤 Carri	12,3909		
1 TH	i Nation		Sim	inter Modifiert	
Statement of the local division of the	Hoten Foge 1. of the P	i Show W 🖻 h		ani Mosteri	No data to display
10.10	Prige III of 1.1 P. P		ma		No data to display
uplood Sr	Page 1 of 1 P 1				No data lo diplay

Each field is described in the following table.

Table 74 Maintenance > File Manager > Shell Script

IABEL	DESC RIPTIO N			
Rename	Use this button to change the label of a shell script file on the Zyxel Device.			
	You cannot rename a shell script to the name of another shell script in the Zyxel Device.			
	Click a shell script's row to select it and click Rename to open the Rename File screen.			
	Specify the new name for the shell script file. Use up to 25 characters (including a-zA-Z0- 9 ;'~!@#\$%^&()_+[]{}',=-).			
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.			
Remove	Click a shell script file's row to select it and click Delete to delete the shell script file from the Zyxel Device.			
	A pop-up window asks you to confirm that you want to delete the shell script file. Click OK to delete the shell script file or click Cancel to close the screen without deleting the shell script file.			
Download	Click a shell script file's row to select it and click Download to save the configuration to your computer.			
Сору	Use this button to save a duplicate of a shell script file on the Zyxel Device.			
	Click a shell script file's row to select it and click Copy to open the Copy File screen.			
	Specify a name for the duplicate file. Use up to 25 characters (including a-zA-Z0-9;'~!@#\$%^&()_+[]{}',.=-).			
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.			
Apply	Use this button to have the Zyxel Device use a specific shell script file.			
	Click a shell script file's row to select it and click Apply to have the Zyxel Device use that shell script file. You may need to wait awhile for the Zyxel Device to finish applying the commands.			
#	This column displays the number for each shell script file entry.			
File Name	This column displays the label that identifies a shell script file.			
Size	This column displays the size (in KB) of a shell script file.			
Last Modified	This column displays the date and time that the individual shell script files were last changed or saved.			
Upload Shell Script	The bottom part of the screen allows you to upload a new or previously saved shell script file from your computer to your Zyxel Device.			
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.			

Table 74 Maintenance > File Manager > Shell Script (continued)

LABEL	DESC RIPIIO N
Browse	Click Browse to find the .zysh file you want to upload.
Upload	Click $Up \log d$ to begin the upload process. This process may take up to several minutes.

C HAPTER 19 Diagnostics

19.1 Overview

Use the diagnostics screen for troubleshooting.

19.1.1 What You Can Do in this Chapter

The **Diagnostics** screen (Section 19.2 on page 178) generates a file containing the Zyxel Device's configuration and diagnostic information if you need to provide it to customer support during troubleshooting.

19.2 Diagnostics

This screen provides an easy way for you to generate a file containing the Zyxel Device's configuration and diagnostic information. You may need to generate this file and send it to customer support during troubleshooting. All categories of settings and shell script files stored on the Zyxel Device will be included in the diagnostic file.

Click Maintenance > Diagnostics > Diagnostics to open the Diagnostics screen. Click Collect Now to have the Zyxel Device create a new diagnostic file.

Figure 107 Maintenance > Diagnostics> Diagnostics



The **Debug Information Center** screen then displays showing whether the collection is in progress, was successful, or has failed. When the data collection is done, click **Download** to save the most recent diagnostic file to a computer.

Figure 108 Maintenance > Diagnostics: Debug Information Collector



19.3 Remote Capture

Use this screen to capture network traffic going through the Zyxel Device connected to the Zyxel gateway or ZyWALL, and output the captured packets to a packet analyzer (also known as network or protocol analyzer) such as Wireshark.

Click Maintenance > Diagnostics > Remote Capture to open the Remote Capture screen.

Figure 109	Maintenance > Diagnostics> Remote Capture
------------	---

Diagnostias	Remote Capture	k
Remote Copture		
Server Port:	2002	
Wireless Monitor I	ntertace Support	
		Stort Stop

The following table describes the labels in this screen.

IABEL	DESC RIPTIO N
Server Port	Enter the number of the server port you want the packet analyzer to connect to in order to capture traffic going through the Zyxel Device. The default port number is 2002.
Start	Click this button to allow the packet analyzer to start capturing traffic going through the Zyxel Device.
Stop	Click this button to stop the packet analyzer from capturing traffic going through the Zyxel Device.

C HAPTER 20 LEDs

20.1 Overview

The LEDs of your Zyxel Device can be controlled such that they stay lit (ON) or OFF after the Zyxel Device is ready. There are two features that control the LEDs of your Zyxel Device - Locator and Suppression (see Section 1.4 on page 18).

20.1.1 What You Can Do in this Chapter

- The **Suppression** screen (Section 20.2 on page 180) allows you to set how you want the LEDs to behave after the Zyxel Device is ready.
- The Locator screen (Section 20.3 on page 181) allows users to see the actual location of the Zyxel Device between several devices in the network.

20.2 Suppression Screen

The LED Suppression feature allows you to control how the LEDs of your Zyxel Device behave after it's ready. The default LED suppression setting of your AP is different depending on your Zyxel Device model.

You can go to the **Maintenance** > **IEDs** > **Suppression** screen to see the default LED behavior and change the LED suppression setting. After you make changes in the suppression screen, it will be stored as the default when the Zyxel Device is restarted. See (Section 3.2 on page 36) for information on default values for different models.

Note: When the ZyXEL Device is booting or performing firmware upgrade, the LEDs will light up regardless of the setting in LED suppression.

To access this screen, click Maintenance > LEDs > Suppression.

180

Figure 110	Maintenance >	LEDs > Su	ppression
------------	---------------	-----------	-----------

Suppression	Locater
Configuration	
T Suppletion	36 ·
Q Note:	
L.Cesterio 2. Sectorio	e the experienciwhen 250 suppression Hodels On. An onling Hintograd Dograde Androg a mach wass ant apply the control TT.
	Apply Devet

The following table describes fields in the above screen.

LABEL	DESC RIPIIO N	
Suppression On	If the $\mathbf{Suppression} \ \mathbf{On}$ check box is checked, the LEDs of your Zyxel Device will turn off after it's ready.	
	If the check box is unchecked, the LEDs will stay lit after the Zyxel Device is ready.	
Apply	Click Apply to save your changes back to the Zyxel Device.	
Reset	Click Reset to return the screen to its last-saved settings.	

20.3 LocatorScreen

The Locator feature identifies the location of your Zyxel Device among several devices in the network. You can run this feature and set a timer in this screen.

To run the locator feature, enter a number of minutes and click **Tum On** button to have the Zyxel Device find its location. The Locator LED will start to blink for the number of minutes set in the **Locator** screen. The default setting is 10 minutes. While the locator is running, the turn on button will gray out and return after it's finished. If you make changes to the time default setting, it will be stored as the default when the Zyxel Device restarts.

Note: The Locator feature is not affected by the Suppression setting.

To access this screen, click Maintenance > LEDs > Locator.

s > Locator

Rugen unen Locarer	
Configuration	
dam deg (dam det.) Ar tom of Loting Ath After 10	(1-83 minores)
00000000000000000000000000000000000000	
	Appen (Invitatio

The following table describes fields in the above screen.

Table // Maintenance > LED > Locator	
IABEL	DESC RIPIIO N
Turn On Turn Off	Click Tum On button to activate the locator. The Locator function will show the actual location of the Zyxel Device between several devices in the network.
	Otherwise, click Tum Off to disable the locator feature.
Automatically Extinguish After	Enter a time interval between 1 and 60 minutes to stop the locator LED from blinking. Default is 10 minutes.
Apply	Click Apply to save changes in this screen.
Refresh	Click $\operatorname{Re}\operatorname{fre}\operatorname{sh}$ to update the information in this screen.

Table 77 Maintenance > LED > Locator

C HAPTER 22 Antenna Switch

22.1 Overview

Use this screen to adjust coverage depending on the orientation of the antenna.

22.1.1 What You Need To Know

Positioning the antennas properly increases the range and coverage area of a wireless LAN.

On the Zyxel Device that comes with internal antennas and also has an antenna switch, you can adjust coverage depending on the antenna orientation for the Zyxel Device radios using the web configurator, the command line interface (CLI) or a physical switch. Check Section 1.4 on page 18 to see if your Zyxel Device has an antenna switch.



- Note: With the physical antenna switch, you apply the same antenna orientation settings to both radios. You can set the radios to have different settings while using the Web Configurator or the command line interface.
- Note: The antenna switch in the Web Configurator has priority over the physical antenna switch after you **Enable Software Control** in the **Maintenance > Antenna** screen. By default, software control is disabled.

22.2 Antenna Switch Screen

To access this screen, click **Maintenance** > **Antenna**.

The screen varies depending on whether the Zyxel Device has a physical antenna switch or allows you to change antenna orientation settings on a per-radio basis or on a per-AP basis.

250

Figure 152 Maintenance > Antenna > Antenna Switch (Per Rad	lio)
--	------

Anienna Swiich			
Configeration			
🝸 brable dollara	e Control		
ivy of	et Mat	il Critig	
www.o21	w wet	0.4y	
		Apply East	



Antenna Switch	
Configuration	
 Wall C 	Colling
	Apply Reset

If the Zyxel Device has a physical antenna switch, select the **Enable Software Control** option to use the Web Configurator to adjust coverage depending on each radio's antenna orientation for better coverage.

Select **Wall** if you mount the Zyxel Device to a wall. Select **Ceiling** if the Zyxel Device is mounted on a ceiling. You can switch from **Wall** to **Ceiling** if there are still wireless dead zones, and vice versa.

Click Apply to save your changes or click Reset to return the screen to its last-saved settings.

CHAPTER 21 Reboot

21.1 Overview

Use this screen to restart the Zyxel Device.

21.1.1 What You Need To Know

If you applied changes in the Web Configurator, these were saved automatically and do not change when you reboot. If you made changes in the CLI, however, you have to use the write command to save the configuration before you reboot. Otherwise, the changes are lost when you reboot.

Reboot is different to reset; reset returns the Zyxel Device to its default configuration.

21.2 Reboot

This screen allows remote users can restart the Zyxel Device. To access this screen, click Maintenance > Reboot.

Figure 112 Maintenance > Reboot

bool	
Clock 1-2 Rebusilibuilib Sectoral concess two	 Excepted the cover, Reactive and a low innotes and the topic stream approximiting the legitherem the Lipplanes of the device is your way browse.

Click the **Reboot** button to restart the Zyxel Device. Wait a few minutes until the login screen appears. If the login screen does not appear, type the IP address of the Zyxel Device in your Web browser.

You can also use the CLI command reboot to restart the Zyxel Device.

183

C HAPTER 22 Shutdown

22.1 Overview

Use this screen to shut down the Zyxel Device.

Always use Maintenance > Shutdown > Shutdown or the shutdown command before you tum off the Zyxel Device or remove the power. Not doing so can cause the firmware to become corrupt.

22.1.1 What You Need To Know

Shutdown writes all cached data to the local storage and stops the system processes. Shutdown is different to reset; reset returns the Zyxel Device to its default configuration.

22.2 Shutdown

To access this screen, click **Maintenance** > **Shutdown**.

Figure 113	Maintenance	> Shutdown
------------	-------------	------------

Stuidown	
Sharkhawn	
fills the Weissland Butten In the Lawy the Sevien	
	Anutdown.

Click the **Shutdown** button to shut down the Zyxel Device. Wait for the Zyxel Device to shut down before you manually turn off or remove the power. It does not turn off the power.

You can also use the CLI command shutdown to shut down the Zyxel Device.

PART II Local Configuration in Cloud Mode

CHAPTER 23 Cloud Mode

23.1 Overview

The Zyxel Device is managed and provisioned automatically by the *NCC (Ne bula Control Center*) when it is connected to the Internet and has been registered in the NCC. If you need to change the Zyxel Device's VLAN setting or manually set its IP address, access its simplified web configurator (see Chapter 4 on page 30). You can check the NCC's Access Point > Monitor > Access Points screen or the connected gateway for the Zyxel Device's current LAN IP address. Alternatively, disconnect the gateway or disable its DHCP server function and use the Zyxel Device's default static LAN IP address (192.168.1.2).





23.2 Cloud Mode Web Configurator Screens

When your Zyxel Device is managed through NCC, you can access only the following screens through the Web Configurator:

- Dashboard
- Configuration > Network > IP Setting
- Configuration > Network > VIAN

- Maintenance > Shell Script
- Maintenance > Diagnostics
- Maintenance > Log

These screens also have fewer options than those in standalone Zyxel Devices. The rest of the Zyxel Device's features must be configured through the NCC.

23.3 Dashboard

This screen displays general AP information, and client information in widgets that you can re-arrange to suit your needs. You can also collapse, refresh, and close individual widgets.



EAP adorealises	(TE
WAC Address:	SO SEPTOT ELEO
Seriel Marchen	\$1921590000P
PHONE MADRIE	WARSI CO
2.4C Chantel Internet of	Chernel b CH 17 Nominit power b 20 cdH
10 Channel Internations	Channel & CLORE (CARPER) Training Actives 3, 20 (2011)
Blocker	This area part to deer to consider to be a loss in the set. P. Address: 123,165,1,10
mernett	This occase point is connected to the internet.
skih in Compativity Status	Fieldcows point biscows of y consected to the Netslet
Nebula Control Center Activations club	his dependent in the state of the Metada.
Use Provy to Append	05

The following table describes the labels in this screen.

LABEL	DESC RIPTIO N				
AP Information	Information				
MAC Address	This field displays the MAC address of the Zyxel Device.				
Serial Number	This field displays the serial number of the Zyxel Device.				
Product Model	This field displays the model name of the Zyxel Device.				
2.4G Channel Information	This field displays the channel number the Zyxel Device is using and its output power in the 2.4 GHz spectrum. This shows Not a c tiva te d if the wireless LAN is disabled.				
5G Channel Information	This field displays the channel number the Zyxel Device is using and its output power in the 5 GHz spectrum. This shows Not a c tiva te d if the wireless LAN is disabled.				
Ethernet	This field displays whether the Zyxel Device's Ethernet port is connected and the IP address of the gateway to which the Zyxel Device is connected.				
Internet	This field displays whether the Zyxel Device is connecting to the Internet.				
Nebula Connectivity Status	This field displays whether the Zyxel Device can connect to the Zyxel Nebula Control Center (NCC).				

Table 78 Dashboard

Table 78 Dashboard (continued)

LABEL	DESC RIPTIO N			
Nebula Control Center Activation Status	This field displays whether the Zyxel Device has been registered and can be managed by the NCC.			
Use Proxy to Access NCC	This displays whether the NAP uses a proxy server to access the NCC (Nebula Control Center).			

If the Zyxel Device cannot connect to the Internet or to NCC, an error message is shown on this screen, as in the following.

1	AF Information	
	MAC Address:	02.11.49.00.00.01
	Serial Number:	234131340 80-009-01 1001 AA
	Product Model:	WAX510D
	2.4G Channel Information:	Channel & CH 6 / Trahumit power's 25 dBm
	SG Channel Information:	Channel's CH 36/40/44/45 / Transnill power's 25 dBm
	Ethomat;	The account point is helping to just a network or finance wanting Ethernal conversions
	Internet:	This access point is connected to the internet.
	Neoula Connectivity Status	We states point a toractive shear to the resource (the section in the second). (the section is said to be the second seco
	Nebula Control Center Activation Status;	This access point has not been registered to the Nebula.
	Use Proxy to Access NGC:	no

C HAPTER 24 Network

24.1 Overview

This chapter describes how you can configure the management IP address and VLAN settings of your Zyxel Device in cloud mode.

See Section 9.1 on page 69 for information about IP addresses.

Note: Make sure your VLAN settings allow the Zyxel Device to connect to the Internet so you could manage it with NCC.

24.1.1 What You Can Do in this Chapter

- The IP Setting screen (Section 24.2 on page 189) configures the Zyxel Device's LAN IP address.
- The VIAN screen (Section 24.3 on page 191) configures the Zyxel Device's VLAN settings.

24.2 IP Setting

Use this screen to configure the IP address for your Zyxel Device. To access this screen, click **Configuration > Network > IP Setting**.

IP Setting VIXII		
IP Address Assignment		
 Get Automatically 		
 Use Rived IP Activities 		
P Addreas	192,568.1.1	
Submet Mask:	255.255.252.9	
Goleway:	192.168.1.5	(Opficeal)
DNS Server IP Address.	192,168.1,11	(Optional)
Use Proxy to Access NCC		
Hosy larver-		
Prove Parts		1 (7 - KSSDD
23 Autoenticulus		
Over Science		
Printerett		
		Apply Reset

Figure 116 Configuration > Network > IP Setting

Each field is described in the following table.

Table 79	Configuration > Network > IP Setting
----------	--------------------------------------

LABEL	DESC RIPIIO N				
IP Address Assignment					
Get Automatically	Select this to make the interface a DHCP client and automatically get the IP address, subnet mask, and gateway address from a DHCP server.				
Use Fixed IP Address	Select this if you want to specify the IP address, subnet mask, and gateway manually.				
IP Address	Enter the IP address for this interface.				
Subnet Mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.				
Gateway	Enter the IP address of the gateway. The Zyxel Device sends packets to the gateway when it does not know how to route the packet to its destination. The gateway should be on the same network as the interface.				
DNS Server IP Address	Enter the IP address of the DNS server.				
Use Proxy to Access Internet	If the Zyxel Device is behind a proxy server, you need to select this option and configure the proxy server settings so that the Zyxel Device can access the NCC through the proxy server.				
Proxy Server	Enter the IP address of the proxy server.				
Proxy Port	Enter service port number used by the proxy server.				
Authentication	Select this option if the proxy server requires authentication before it grants access to the Internet.				
User Name	Enter your proxy user name.				
Password	Enter your proxy password.				
Apply	Click Apply to save your changes back to the Zyxel Device.				
Reset	Click Reset to return the screen to its last-saved settings.				
24.3 VIAN

This section discusses how to configure the Zyxel Device's VLAN settings. See Section 9.3 on page 71 for more information about VLAN.

Use this screen to configure the VLAN settings for your Zyxel Device. To access this screen, click **Configuration > Network > VIAN**.

Figure 117 Configuration > Network > VLAN

IP Setting	VLAN
VLAN Settings	
Management	VLANID: 1 (1-4094)
Untragged	© Tagged
	Apply Reset

Each field is described in the following table.

LABEL	DESC RIPIIO N
VLAN Settings	
Management VLAN ID	Enter a VLAN ID for the Zyxel Device.
Untagged/ Tagged	Set whether the Zyxel Device adds the VLAN ID to outbound traffic transmitted through its Ethernet port.
Apply	Click Apply to save your changes back to the Zyxel Device.
Reset	Click Reset to return the screen to its last-saved settings.

Table 80 Configuration > Network > VLAN

C HAPTER 25 Maintenance

25.1 Overview

When the Zyxel Device is set to work in cloud mode, the **Maintenance** screens let you mange shell script files on the Zyxel Device, generate a diagnostic file, or view log messages.

See Chapter 18 on page 167 for information about shell scripts.

25.1.1 What You Can Do in this Chapter

- The Shell Script screen (Section 25.2 on page 192) stores, names, downloads, and uploads shell script files.
- The **Diagnostics** screen (Section 25.3 on page 193) generates a file containing the Zyxel Device's configuration and diagnostic information if you need to provide it to customer support during troubleshooting.
- The Log > View Log screen (Section 25.4 on page 194) displays the Zyxel Device's current log messages when it is disconnected from the NCC.

25.2 Shell Script

Use shell script files to have the Zyxel Device use commands that you specify. Use a text editor to create the shell script files. They must use a ".zysh" filename extension.

Click **Maintenance > Shell Script** to open this screen. Use the **Shell Script** screen to store, name, download, and upload shell script files. You can store multiple shell script files on the Zyxel Device at the same time.

Clinica	nie 🖀 Herzone 🔠 Otwardie	nt 🔁 Cone		
r The	histon	Tim	ioni Mostina	
10 m CT	toge Elliot 11 P. H Sho	w. W. L. M. Barra		No data to displa
	eli Script			
nioad Sh	eli Script			

Figure 118 Maintenance > Shell Script

192

Each field is described in the following table.

IABEL	DESC RIPTIO N
Rename	Use this button to change the label of a shell script file on the Zyxel Device.
	You cannot rename a shell script to the name of another shell script in the Zyxel Device.
	Click a shell script's row to select it and click Rename to open the Rename File screen.
	Specify the new name for the shell script file. Use up to 25 characters (including a-zA-Z0-9;'~!@#\$%^&()_+[]{}',.=-).
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.
Remove	Click a shell script file's row to select it and click Delete to delete the shell script file from the Zyxel Device.
	A pop-up window asks you to confirm that you want to delete the shell script file. Click OK to delete the shell script file or click Cancel to close the screen without deleting the shell script file.
Download	Click a shell script file's row to select it and click Download to save the configuration to your computer.
Сору	Use this button to save a duplicate of a shell script file on the Zyxel Device.
	Click a shell script file's row to select it and click Copy to open the Copy File screen.
	Specify a name for the duplicate file. Use up to 25 characters (including a-zA-Z0-9;'~!@#\$%^&()_+[]{}',.=-).
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.
#	This column displays the number for each shell script file entry.
File Name	This column displays the label that identifies a shell script file.
Size	This column displays the size (in KB) of a shell script file.
Last Modified	This column displays the date and time that the individual shell script files were last changed or saved.
Upload Shell Script	The bottom part of the screen allows you to upload a new or previously saved shell script file from your computer to your Zyxel Device.
File	Type in the location of the file you want to upload in this field or click Browse to find it.
Browse	Click Browse to find the .zysh file you want to upload.
Upload	Click Upload to begin the upload process. This process may take up to several minutes.

Table 81 Maintenance > Shell Script

25.3 Diagnostics

This screen provides an easy way for you to generate a file containing the Zyxel Device's configuration and diagnostic information. You may need to generate this file and send it to customer support during troubleshooting. All categories of settings and shell script files stored on the Zyxel Device will be included in the diagnostic file.

Click **Maintenance > Diagnostics** to open the **Diagnostics** screen. Click **Collect Now** to have the Zyxel Device create a new diagnostic file.



Disgonities	Frende Caphae
Disgunde infer	rentas Collector
Press Fr. Lands	es l'Mere les les des les soites l'aspess les mis consisers les mars a l'anti-fire baix et land cols
	College Have

The **Debug Information Center** screen then displays showing whether the collection is in progress, was successful, or has failed. When the data collection is done, click **Download** to save the most recent diagnostic file to a computer.

Figure 120 Maintenance > Diagnostics: Debug Information Collector



25.4 View Log

The NCC periodically gathers log files from the devices being managed by it. Before the NCC pulls logs from the Zyxel Device or when the Zyxel Device is disconnected from the NCC, you can use this screen to view its current log messages. To access this screen, click **Maintenance > Log**.

Note: When a log reaches the maximum number of log messages, new log messages automatically overwrite existing log messages, starting with the oldest existing log message first.

Events that generate an alert (as well as a log message) display in red. Regular logs display in black. Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

Figure 121 Maintenance >	> Log	> \	/iew	Log
--------------------------	-------	-----	------	-----

log - be ecolated?	viten i 😵 access p	ntis net confecto	(10 methe suis :		
(Opplay	spicial	l a	Profile:	4.7	1×
zource Addiess	- 100		Electionaria n Alda lossi		
Shurde Interforme	0.0	in.	Certification stedare	40.2	
na cite.	177		Repaine.		
Secret					
@Retrette # Clean	taj				
STATE.	P stelowy	Part of the second	(Jane	Texturnie)	térie,
30 (2020) I de Orige	pCarlot	see the file of the	h Received	meth. Cali	00042
We Alexandra	ona Sar Sath	-control			
67 2012-11-27 05.24	. La las mint	ekeré (ANE comiço)	ned - conceptuary with a la		
88 10-14-14-20 Mar.		1005	IN THE REAL PLAN		
rus - Mark III America	pin the field D	A scentrul			
The second reading the					

The following table describes the labels in this screen.

|--|

LABEL	DESC RIPTIO N	
Show Filter / Hide	Click this button to show or hide the filter settings.	
Filter	lf the filter settings are hidden, the Display , Email Log Now , Refresh , and Clear Log fields are available.	
	If the filter settings are shown, the Display , Priority , Source Address , Destination Address , Source Interface, Destination Interface, Protocol, Keyword, and Search fields are available.	
Display	Select the category of log message(s) you want to view. You can also view All Logs at one time, or you can view the Debug Log .	
Priority	This displays when you show the filter. Select the priority of log messages to display. The log displays the log messages with this priority or higher. Choices are: any , emerg , alert , crit , emor , wam , notice , and info , from highest priority to lowest priority. This field is read-only if the Category is Debug Log .	
Source Address	This displays when you show the filter. Type the source IP address of the incoming packet that generated the log message. Do not include the port in this filter.	
Destination Address	This displays when you show the filter. Type the IP address of the destination of the incoming packet when the log message was generated. Do not include the port in this filter.	
Source Interface	This displays when you show the filter. Select the source interface of the packet that generated the log message.	
Destination Interface	This displays when you show the filter. Select the destination interface of the packet that generated the log message.	
Protocol	This displays when you show the filter. Select a service protocol whose log messages you would like to see.	
Keyword	This displays when you show the filter. Type a keyword to look for in the Message , Source , Destination and Note fields. If a match is found in any field, the log message is displayed. Y can use up to 63 alphanumeric characters and the underscore, as well as punctuation mot ()', ;;?! +-*/= #\$% @; the period, double quotes, and brackets are not allowed.	
Search	This displays when you show the filter. Click this button to update the log using the current filter settings.	

IABEL	DESC RIPIIO N
Refresh	Click this to update the list of logs.
Clear Log	Click this button to clear the whole log, regardless of what is currently displayed on the screen.
#	This field is a sequential value, and it is not associated with a specific log message.
Time	This field displays the time the log message was recorded.
Priority	This field displays the priority of the log message. It has the same range of values as the Priority field above.
Category	This field displays the log that generated the log message. It is the same value used in the Display and (other) Category fields.
Message	This field displays the reason the log message was generated. The text "[count= x]", where x is a number, appears at the end of the Message field if log consolidation is turned on and multiple entries were aggregated to generate into this one.
Source	This field displays the source IP address and the port number in the event that generated the log message.
Source Interface	This field displays the source interface of the packet that generated the log message.
Destination	This field displays the destination IP address and the port number of the event that generated the log message.
Destination Interface	This field displays the destination interface of the packet that generated the log message.
Protocol	This field displays the service protocol in the event that generated the log message.
Note	This field displays any additional information about the log message.

 Table 82
 Maintenance > Log > View Log (continued)

PART III Appendices and Trouble shooting

C HAPTER 26 Trouble shooting

26.1 Overview

This chapter offers some suggestions to solve problems you might encounter. The potential problems are divided into the following categories.

- Power, Hardware Connections, and LED
- Zyxel Device Management, Access, and Login
- Internet Access
- WiFi Network
- Resetting the Zyxel Device

26.2 Power, Hardware Connections, and LED

The Zyxel Device does not turn on. The LED is not on.

- 1 Make sure you are using the power adapter included with the Zyxel Device or a PoE power injector/ switch.
- 2 Make sure the power adapter or PoE power injector/switch is connected to the Zyxel Device and plugged in to an appropriate power source. Make sure the power source is turned on.
- 3 Disconnect and re-connect the power adapter or PoE power injector/switch.
- 4 Inspect your cables for damage. Contact the vendor to replace any damaged cables.
- 5 If none of these steps work, you may have faulty hardware and should contact your Zyxel Device vendor.

The LED does not behave as expected.

- 1 Make sure you understand the normal behavior of the LED. See Section 3.2 on page 36.
- 2 Check the hardware connections. See the Quick Start Guide.
- 3 Inspect your cables for damage. Contact the vendor to replace any damaged cables.

- 4 Disconnect and re-connect the power adapter or PoE power injector to the Zyxel Device.
- 5 If the problem continues, contact the vendor.

26.3 Zyxel Device Management, Access, and Login

I forgot the IP address for the Zyxel Device.

- 1 The default in-band IP address in standalone mode is http://DHCP-assigned IP (when connecting to a DHCP server) or 192.168.1.2.
- 2 If you changed the IP address and have forgotten it, you have to reset the Zyxel Device to its factory defaults. See Section 26.6 on page 205.
- 3 If your Zyxel Device is a DHCP client, you can find your IP address from the DHCP server. This information is only available from the DHCP server which allocates IP addresses on your network. Find this information directly from the DHCP server or contact your system administrator for more information.
- 4 If the NCC has managed the Zyxel Device, you can also check the NCC's **AP > Monitor > Access Point** screen for the Zyxel Device's current LAN IP address.

I cannot see or access the Login screen in the Web Configurator.

- 1 Make sure you are using the correct IP address.
 - The default IP address (in standalone mode) is 192.168.1.2.
 - If you changed the IP address, use the new IP address.
 - If you changed the IP address and have forgotten it, see the troubleshooting suggestions for I forgot the IP address for the Zyxel Device.
- 2 Check the hardware connections, and make sure the LED is behaving as expected. See the Quick Start Guide and Section 3.2 on page 36.
- 3 Make sure your Internet browser does not block pop-up windows and has JavaScripts and Java enabled.
- 4 Make sure your computer is in the same subnet as the Zyxel Device. (If you know that there are routers between your computer and the Zyxel Device, skip this step.)
 - If there is a DHCP server on your network, make sure your computer is using a dynamic IP address.
 - If there is no DHCP server on your network, make sure your computer's IP address is in the same subnet as the Zyxel Device.
- 5 Reset the Zyxel Device to its factory defaults, and try to access the Zyxel Device with the default IP address. See Section 26.6 on page 205.

6 If the problem continues, contact the network administrator or vendor, or try one of the advanced suggestions.

Advanced Suggestions

- Try to access the Zyxel Device using another service, such as Telnet. If you can access the Zyxel Device, check the remote management settings to find out why the Zyxel Device does not respond to HTTP.
- If your computer is connected wirelessly, use a computer that is connected to a IAN/EIHERNET port.

I forgot the password.

- 1 The default password is 1234. If the Zyxel Device is connected to the NCC and registered, check the NCC for the password.
- 2 If this does not work, you have to reset the Zyxel Device to its factory defaults. See Section 26.6 on page 205.

I can see the Login screen, but I cannot log in to the Zyxel Device.

- 1 Make sure you have entered the user name and password correctly. The default password is **1234**. This fields are case-sensitive, so make sure [Caps Lock] is not on.
- 2 You cannot log in to the Web Configurator while someone is using Telnet to access the Zyxel Device. Log out of the Zyxel Device in the other session, or ask the person who is logged in to log out.
- 3 Disconnect and re-connect the power adapter or PoE power injector to the Zyxel Device.
- 4 If this does not work, you have to reset the Zyxel Device to its factory defaults. See Section 26.6 on page 205.

I cannot use FTP to upload / download the configuration file. / I cannot use FTP to upload new firmware.

See the troubleshooting suggestions for I cannot see or access the Login screen in the Web Configurator. Ignore the suggestions about your browser.

I cannot access the Zyxel Device directly anymore after switching to NCC management.

• Check the Zyxel Device IP address and login credentials using the NCC and use them to access the Zyxel Device. Note that the built-in Web Configurator will have limited functionality when managed through NCC.

I enabled NCC Discovery, but the Zyxel Device is still in standalone mode.

Make sure your Zyxel Device is registered to the NCC.

The Zyxel Device is already registered with NCC, but it is still in standalone mode; it cannot connect to the NCC.

- 1 Make sure that NCC Discovery is enabled (see Section 9.4 on page 73).
- 2 Check your network's firewall/security settings. Make sure the following TCP ports are allowed: 443, 4335, and 6667.
- **3** Make sure your Zyxel Device can access the Internet.
- 4 Check your network's VLAN settings (see Section 9.3 on page 71). You may have to change the Management VLAN settings of the Zyxel Device to allow it to connect to the Internet and access the NCC.
 - Note: Changing the management VLAN and IP address settings on the Zyxel Device also pushes these changes to the NCC. Do this only if your device cannot otherwise connect to the NCC.
- 5 Make sure your Zyxel Device does not have to go through network authentication such as a captive portal, If your network uses a captive portal, the network administrator may have to create a new VLAN without this requirement. Change your Zyxel Device's management VLAN settings as necessary.

I want to switch from NCC to AC management, but I could not find the **AC Discovery** menu in the Zyxel Device Web Configurator.

- 1 Unregister the Zyxel Device from the NCC.
- 2 Reset your Zyxel Device to the factory defaults.
- 3 Make sure that your Zyxel Device is in the same subnet as the AC, and enable AC Discovery in Configuration > Network > AC Discovery.

The Zyxel Device cannot discover the AC.

- 1 Make sure your Zyxel Device is not registered to NCC.
- 2 Enable AC Discovery in Configuration > Network > AC Discovery.

- 3 Make sure that the Zyxel Device and the AC are both in the same subnet.
- 4 If you have to set them up in different subnets, see AC management and IP Subnets on page 104.

I accidentally pressed the Nebula button in the NXC's Web Configurator. How do I undo it?

- 1 If the Zyxel Device is not registered with the NCC, register it first.
- 2 Unregister the Zyxel Device from the NCC.
- **3** Reset the Zyxel Device to the factory defaults.

Some features I set using the NCC do not work as expected.

- 1 Make sure your Zyxel Device can access the Internet.
- 2 Check your network's firewall/security settings. Make sure the following ports are allowed:
 - TCP: 443, 4335, and 6667
 - UDP: 123
- **3** After changing your Zyxel Device settings using the NCC, wait 1-2 minutes for the changes to take effect.

I can only see newer logs. Older logs are missing.

When a log reaches the maximum number of log messages (see Section 1.4 on page 18), new log messages automatically overwrite the oldest log messages.

The commands in my configuration file or shell script are not working properly.

- In a configuration file or shell script, use "#" or "!" as the first character of a command line to have the Zyxel Device treat the line as a comment.
- Your configuration files or shell scripts can use "exit" or a command line consisting of a single "!" to have the Zyxel Device exit sub command mode.
- Include write commands in your scripts. Otherwise the changes will be lost when the Zyxel Device restarts. You could use multiple write commands in a long script.

Note: "exit" or "!" must follow sub commands if it is to make the Zyxel Device exit sub command mode.

I cannot upload the firmware uploaded using FTP.

The Web Configurator is the recommended method for uploading firmware in standalone mode. For managed Zyxel Devices, using the NCC or AC is recommended. You only need to use FTP if you need to recover the firmware. See the CLI Reference Guide for how to determine if you need to recover the firmware and how to recover it.

26.4 Internet Access

Clients cannot access the Internet through the Zyxel Device.

- 1 Check the Zyxel Device's hardware connections, and make sure the LEDs are behaving as expected (refer to Section 3.2 on page 36). See the Quick Start Guide and Section 26.2 on page 198.
- 2 Make sure the Zyxel Device is connected to a broadband modem or router with Internet access and your computer is set to obtain an dynamic IP address.
- 3 If clients are trying to access the Internet wirelessly, make sure the wireless settings on the wireless clients are the same as the settings on the Zyxel Device.
- 4 Disconnect all the cables from your Zyxel Device, and follow the directions in the Quick Start Guide again.
- **5** Reboot the client and reconnect to the Zyxel Device.
- 6 If the problem continues, contact your ISP.

The Internet connection is slow or intermittent.

- 1 There might be a lot of traffic on the network. Look at the LEDs, and check Section 3.2 on page 36. If the Zyxel Device is sending or receiving a lot of information, try closing some programs that use the Internet, especially peer-to-peer applications.
- 2 Check the signal strength using the NCC, AC, Zyxel Device Web Configurator, or the client device itself. If the signal is weak, try moving the client closer to the Zyxel Device (if possible), and look around to see if there are any devices that might be interfering with the wireless network (microwaves, other wireless networks, and so on).
- **3** Reboot the Zyxel Device using the Web Configurator/CLI or the NCC or AC.
- 4 Check the settings for QoS. If it is disabled, activate it. When enabled, raise or lower the priority for some applications.

5 If the problem continues, contact the network administrator or vendor.

26.5 WiFi Network

I cannot access the Zyxel Device or ping any computer from the WLAN.

- 1 Make sure the wireless LAN (wireless radio) is enabled on the Zyxel Device.
- 2 Make sure the radio or at least one of the Zyxel Device's radios is operating in AP mode.
- 3 Make sure the wireless adapter (installed on your computer) is working properly.
- 4 Make sure the wireless adapter (installed on your computer) is IEEE 802.11 compatible and supports the same wireless standard as the Zyxel Device's active radio.
- 5 Make sure your computer (with a wireless adapter installed) is within the transmission range of the Zyxel Device.
- 6 Check that both the Zyxel Device and your computer are using the same wireless and wireless security settings.

Hackers have accessed my WEP-encrypted wireless LAN.

WEP is extremely insecure. Its encryption can be broken by an attacker, using widely-available software. It is strongly recommended that you use a more effective security mechanism. Use the strongest security mechanism that all the wireless devices in your network support. WPA2 or WPA2-PSK is recommended.

The wireless security is not following the re-authentication timer setting I specified.

If a RADIUS server authenticates wireless stations, the re-authentication timer on the RADIUS server has priority. Change the RADIUS server's configuration if you need to use a different re-authentication timer setting.

I cannot import a certificate into the Zyxel Device.

- 1 For My Certificates, you can import a certificate that matches a corresponding certification request that was generated by the Zyxel Device. You can also import a certificate in PKCS#12 format, including the certificate's public and private keys.
- 2 You must remove any spaces from the certificate's filename before you can import the certificate.

- 3 Any certificate that you want to import has to be in one of these file formats:
 - Binary X.509: This is an ITU-T recommendation that defines the formats for X.509 certificates.
 - PEM (Base-64) encoded X.509: This Privacy Enhanced Mail format uses lowercase letters, uppercase letters and numerals to convert a binary X.509 certificate into a printable form.
 - Binary PKCS#7: This is a standard that defines the general syntax for data (including digital signatures) that may be encrypted. A PKCS #7 file is used to transfer a public key certificate. The private key is not included. The Zyxel Device currently allows the importation of a PKS#7 file that contains a single certificate.
 - PEM (Base-64) encoded PKCS#7: This Privacy Enhanced Mail (PEM) format uses lowercase letters, uppercase letters and numerals to convert a binary PKCS#7 certificate into a printable form.
 - Binary PKCS#12: This is a format for transferring public key and private key certificates. The private key in a PKCS #12 file is within a password-encrypted envelope. The file's password is not connected to your certificate's public or private passwords. Exporting a PKCS #12 file creates this and you must provide it to decrypt the contents when you import the file into the Zyxel Device.

Note: Be careful not to convert a binary file to text during the transfer process. It is easy for this to occur since many programs use text files by default.

Wireless clients are not being load balanced among my Zyxel Devices.

- Make sure that all the Zyxel Devices used by the wireless clients in question share the same SSID, security, and radio settings.
- Make sure that all the Zyxel Devices are in the same broadcast domain.
- Make sure that the wireless clients are in range of the other Zyxel Devices; if they are only in range of a single Zyxel Device, then load balancing may not be as effective.

In the **Monitor > Wire less > AP Inform a tion > Ra dio List** screen, there is no load balancing indicator associated with any Zyxel Devices assigned to the load balancing task.

- Check that the AP profile which contains the load balancing settings is correctly assigned to the Zyxel Devices in question.
- The load balancing task may have been terminated because further load balancing on the Zyxel Devices in question is no longer required.

26.6 Resetting the Zyxel Device

If you cannot access the Zyxel Device by any method, try restarting it by turning the power off and then on again. If you still cannot access the Zyxel Device by any method or you forget the administrator password(s), you can reset the Zyxel Device to its factory-default settings. Any configuration files or shell scripts that you saved on the Zyxel Device should still be available afterwards.

Use the following procedure to reset the Zyxel Device to its factory-default settings. This overwrites the settings in the startup-config.conf file with the settings in the system-default.conf file.

205

Note: This procedure removes the current configuration.

- 1 Make sure the Power LED is on and not blinking.
- 2 Press the **RESET** button and hold it until the Power LED begins to blink. (This usually takes about ten seconds.)
- **3** Release the **RESET** button, and wait for the Zyxel Device to restart.

You should be able to access the Zyxel Device in standalone mode using the default settings.

26.7 Getting More Trouble shooting Help

Search for support information for your model at www.zyxel.com for more troubleshooting suggestions.



A PPENDIX A Importing Certificates

This appendix shows you how to import public key certificates into your web browser.

Public key certificates are used by web browsers to ensure that a secure web site is legitimate. When a certificate authority such as VeriSign, Comodo, or Network Solutions, to name a few, receives a certificate request from a website operator, they confirm that the web domain and contact information in the request match those on public record with a domain name registrar. If they match, then the certificate is issued to the website operator, who then places it on the site to be issued to all visiting web browsers to let them know that the site is legitimate.

Many Zyxel products, such as the Zyxel Device, issue their own public key certificates. These can be used by web browsers on a LAN or WAN to verify that they are in fact connecting to the legitimate device and not one masquerading as it. However, because the certificates were not issued by one of the several organizations officially recognized by the most common web browsers, you will need to import the Zyxel-created certificate into your web browser and flag that certificate as a trusted authority.

Note: You can see if you are browsing on a secure website if the URL in your web browser's address bar begins with https:// or there is a sealed padlock icon (🔒) somewhere in the main browser window (not all browsers show the padlock in the same location).

Google Chrome

The following example uses Google Chrome on Windows 7. You first have to store the certificate in your computer and then install it as a Trusted Root CA, as shown in the following tutorials.

Export a Certificate

1 If your device's Web Configurator is set to use SSL certification, then upon browsing with it for the first time, you are presented with a certification error.



2 Click Advanced > Proceed to x.x.x.x (unsafe).



3 In the Address Bar, Click Not Secure > Certificate (Invalid).



4 In the Certificate dialog box, click Details > Copy to File.

Nom CAD-	*	
Reid	Value	1
Wetaba	1/2	1
Serial number	96 24 bf 08	E.
Signature elgorithm	inu 385A	
Signature hash algorithm	shat	
lamor	ung63_5550° 3°5032A	
Valid from	Monday, October 19, 2015 St	
Valid to	Thursday, October 18, 3025 5	
Section 1	14191 REPAIR 1997/17/2	0
6	Fall Properties	

5 In the Certificate Export Wizard, Click Next.

Certificate Export Waard	
	Welcome to the Certificate Export Wizard This waard helps you coop certificates, certificate trust is and certificate revocation but from a certificate trust store to your dok. A certificate, which is second by a certification activative, a n confirmation of your dentity and contents information used to protect data or to establish second intervert contections. A certificate store is the system area where certificates are lapt. To continue, click Next:
	e tuck (Rest >) Ganad

NWA/WAC/WAX Series User's Guide

6 Select the format and settings you want to use and then click Next.

Certificate Export Wizerd	
Expert File Format Cartificates can be exported in a variance of Re formats.	
Select the formet you want to user	
CER encoded binary X.539 (, CER)	
(2) Base-64 encoded X.509 (LER.)	
O votographic Message Syntax Standard - PKCS #7 Certificates (.P78) Dictude all certificates in the certification path if possible	
Personal Information Exchange - PRCS #12 (JPFN) Declarle all cartification in the cartification path if possible	
Deleter the private tay if the export is accounted Depart all extended properties	
Microsoft Serviced Certificate Stare C567)	
Learn more about <u>on tificate. Re formate</u>	
< Sack Next > Car	

7 Type a filename and specify a folder to save the certificate in. Click Next.

those Export Woord	
No to Export Specify the name of the Ne	you want to export
Pie rens:	
Dricert.cer	Bronsen
	< Sad Next > Cancel

8 In the Completing the Certificate Export Wizard screen, click Finish.

	Completing the Certificate Export Wizard You have successfully completed the Certificate Expo
8	No have specified the following settings: Depart Name No Didude all certificates in the certification path No File Parmet DER

9 Finally, click **OK** when presented with the successful certificate export message.



Import a Certificate

After storing the certificate in your computer (see Export a Certificate), you need to install it as a trusted root certification authority using the following steps:

1 Open your web browser, click the menu icon, and click Settings.



2 Scroll down and click Advanced to expand the menu. Under Privacy and security, click Manage certificates.

iwawa and an aurity	
way and any interesting	
Byte and buogle veryies a	
More settings that while to prove streaming and data to be too	200
A low Corona a gran	1.00
By botting the off, you can again to Geogle area. We Creat websits again got to Stanto	
sond a "under unde" requires a thread burst long to the	())))) ()))))
A low all an to check If you have payment methods as well	
Protoci pages for tester browing and averability	-
Were bodyley to remember your preferences, even if you don't net those begins	
Managa cardinalar	E
Manage 1 TTP0/221 cardinarys at 2 sellings	- 7.県:
Contact will trac	10
Contract which information responses can use use which extract they can whom you	50
Case intervang sala	78
Charling cooking eveng and man	

3 In the Certificates pop-up screen, click Trusted Root Certification Authorities. Click Import to start the Certificate Import Wizard.

Issued To	Issued By	Expirate	Friendly Name	2
AffirmTrust Comme Baltimorie CyberTru Certum CA Cartum Trusted Na Class 3 Public Prime ICDMODIO RSA Cart Copyright (c) 1997	AddThart Externel CA AffiniTrust Convienced Beltimore DyberTrust Certuri CA Cartaen Trusted Network Close 3 Public Primery COMODO RSA Carthfe Copyright (c) 3997 Mi DigiCert Assured ID R	12/31/2030 5/11/2025 6/11/2027 12/31/2029 8/2/2028 1/19/2038 12/31/1999	AffeniTrust Com DigiCert Behavor Certum Certum Trusted VerSign Oess 3 Sectigo (formet Moresoft Timest	
opertuin) Econ	Return		Adva	10

4 Click Next when the wizard pops up, and then on the following screen click Browse.



5 Select the certificate file you want to import and click **Open**.

Species + 11. Species	100 C				- 10 × 10 4
Git in		Line reaction in 2.0	1965	(1994-)	
1	1.	4 70	1.		
	11.	10.00	******		
1 ····		11 - 12 11 - 12	1.2		
a 45	1	·····································	1.5		
5 26 L	1	14, 460 25, 38 35, 69	1		
-	N N	A	1		
Second .	in the sector	-) di du chi e	Contraction of the local distribution of the	- (1890 Gen	

6 Click Next.

ertificate Import Wicard		
Certificate Store Certificate stores are system areas w	vhere certificates are kep	6
Windows can automotically select a o the cartificate.	ethatestore, or you a	in specify a location for
C Automatically select the certific	tate store based on the t	vpe of certificate
Place all certificates in the follo	wing store	
Certificate stores		
Trusted Root Certification Au	ifnantee	Drovies
Least nore about <u>an triate allocat</u>		
		Next >

7 Confirm the settings displayed and click Finish.

<u>_</u>	Completing the Certi Wizard The certificate will be reported of You have specified the following of	lar you dide Prisite.
	Den blocke Steen Delected for un Conternt Pile Pileme	Trusted Root Certifica Certificate Drijoert_test.cer
	* 0	<u> </u>
	< Bod	Fireth Cancel

8 If presented with a security warning, click Yes.

Ormog		X
You are about to install a catificata from a claiming to represent:	certification aut	unity (CA)
05, "**** (d) ***		
⁺ ν 9′ Ψ −? 2 ⁺ . You should confirm	n ita lorigin by co	intecting
Thumbprint (shall): (" r 19 0. 8 . 5.35	2 N 16 8	1781
cartificate issued by this CA. Installing a car	tificate with en s	nconfirmed
Do you want to install this cartificate?		
		10
	claiming to represent $US_{k} \stackrel{(r, q)}{\longrightarrow} \frac{(q)}{2} \stackrel{(r, q)}{\longrightarrow} \frac{(q)}{2}$ Windows connot validate that the certification $\frac{(q)}{2} \stackrel{(q)}{\longrightarrow} \frac{(q)}{2} \stackrel{(r, q)}{\longrightarrow} \frac{(q)}{2} \stackrel$	$\begin{split} & U_{h_{k_{1}}} \stackrel{\text{def}}{=} x_{2k_{1}} \stackrel{\text{def}}{$

9 Finally, click **OK** when you are notified of the successful import.



Install a Stand-Alone Certificate File

Rather than installing a public key certificate using web browser settings, you can install a stand-alone certificate file if one has been issued to you.

1 Double-click the public key certificate file.



2 Click Install Certific a te.

-	ertificate Information
install th	Root certificate is not trusted. To enable trust, as certificate in the Trusted Root Certification ies store.
199	ied tox 19290_3939F3FEX324
Iss	red Byr 1.5990_5989F3PED33A
	d from 10/ 19/ 2015 to 10/ 16/ 2025

3 Click Next on the first wizard screen, click Place all certificates in the following store, and click Browse.

Centricate Import Wisard	100
Certificate Store Certificate stores are system areas ofers certificates are legit.	
Windows can automatically select a certificate store, or you can specify a location f the certificate	iar
C Automatorally select the pertificate store based on the type of certificate	
@ Recs al cartificates in the following store	
Cartificate store:	
Learn more about certificate stores	
Sector Sector	ascei

4 Select Trusted Root Certificate Authorities > OK, and then click Next.

Rersonal Rerson

5 Confirm the information shown on the final wizard screen and click Finish.

Certificate Import Waard			
	Completing the Certificate Import Wizard The certificate will be imported after you clid. Prish.		
	Content	Environment Certifics Certificate	
	·	a a	
		Presh Cancel	

6 If presented with a security warning, click Yes.

www.g		đ
You are about to install a cartificate fr claiming to represent:	rom a certification auth	sariky (CA)
$= E_{ij} \omega^{ij} + e^{-i\omega^{ij}} + e^{-i\omega^{ij}}$		
5. N. T. You should a	confirm its polyin by co	intacting
Thumbprint (shalls for short and	AL 8 E 9 1	0.05 (1.4)
certificate issued by this CA Installing	g a certificate with an u	nconfirmed
$D_{D,\gamma} \omega_{0}$ want to install this certificated		
	claiming to represent. $\frac{1}{2} \left[\Gamma_{-1} \frac{1}{2} + e^{-i\Gamma_{-1}} + \frac{1}{2} \right]$ Windows connot will date that the cerit $S_{-1} = 3$ and $T_{-1} = V_{-1}^{-1}$ You should a $\frac{1}{2} \frac{1}{2} \frac{1}{2} + \frac{1}{2} \frac{1}{2} + \frac{1}{2} 1$	$\label{eq:constraint} \begin{array}{l} 1 & \Gamma \rightarrow 2^{n-1} & = -2^n \\ \\ \mbox{Windows cannot will date that the certificate is actually from 1 & S_{n} = 3 & T \rightarrow 3^n$. You should confirm its origin by contact, $S_{n} = E^{n-1} + E^$

7 Finally, click **OK** when you are notified of the successful import.



Remove a Certificate in Google Chrome

This section shows you how to remove a public key certificate in Google Chrome on Windows 7.

1 Open your web browser, click the menu icon, and click Settings.



2 Scroll down and click Advanced to expand the menu. Under Privacy and security, click Manage certificates.

areanrad -	
anay and an analy	
By it: and people converts	¥2
More settings that while to prevery security and data to bestion	
Allow Concerning in an	1100
By barring the off, you can again to Beogle areas line threat we reac again goe to Standay	-0
sond a funder under oppertie they are brokening to the	()))))))))))))))))))))))))))))))))))))
A tow all a fix check if you have payment methods caned	
Privileus pages for pasteriar strating and avarching	
User boddey is rener the your preleve over even if you don't net those arges	
Manage conduction	
Manage 1 TTP-0155, participant and adding a	
Gartapiaviinga	17
Control which information reportion can assume which contains they can whole you	200
Case interangents	22
Char history cook his evene, and mea	

3 In the Certificates pop-up screen, click Thusted Root Certification Authorities.

Issuest To	Issued By	Siprata	Friendly Name	1
AddTrust Externel AffirmTrust Conne Baltimore CyberThu Certum CA Contum Trusted Ne Concord Public Prints COMODIO RSA Cart Copyright (c) 1997 DigiGert Assured ID	AffiniTrust Convierceal Baltimarie OptierTrust Certum CA Cartum Trusted Network Class 3 Public Primary COMODIO RSA Carthfic Copyright (c) 3997 Mi	12/51/2030 5/13/2025 6/11/2027 12/31/2029 8/2/2028 1/16/2038 12/51/1999	Certum Certum Trusted Verlägt Oass 3 Sectigs (farmer Morosoft Timest	
legart			Adve	tcinit

NWA/WAC/WAX Series User's Guide

- 4 Select the certificate you want to remove and click **Remove**.
- 5 Click Yes when you see the following warning message.



6 Confirm the details displayed in the warning message and click Yes.



Fire fo x

The following example uses Mozilla Firefox on Windows 7. You first have to store the certificate in your computer and then install it as a Trusted Root CA, as shown in the following tutorials.

Export a Certificate

1 If your device's Web Configurator is set to use SSL certification, then the first time you browse to it you are presented with a certification error. Click **Advanced**.



2 Click View Certific a te.

Warning: Potential Security Risk Ahead
Electors detected a potential security threat and did not continue to 192.1601.2. If you visit this site, attackers could by to steal information like your passwords, emails, or credit card details.
Learn more .
Go Back (Recommender) Advanced
We below prove them identify we carbinates threfox does not host the site because it uses a carbinate that is not valid for 192.108.1.2. The certificate is only valid for . Bron code: MOZILLA PICK BERGE SELF BIGNED CERT View Certificate
Go Back (Recommended) Accept the Kek and Continue

3 Click Details > Export.

Company (1949-1) Les tills alle Blassen its Les gifts (SREDIT ITECE) A	
Largeto Stationo modela.	Ť.
Latitute sets	
	2
4 Con Scott	÷.
Vertice	1
Gef () Humber	
Can Sciencification Algorithm	
- Velak,	
Finit Salar	

4 Type a filename and click Save.

🖓 la trace					in Twa Tikas Sid ga Sa
ipini - Nortalia		7766 1176			371.7
<u>×</u> 10,		10decostiled -	l'ar	See	
_					
3					
3					
_					
2					
16					
filenens Million	1210010				
Secretigen 200 stade	NCT 3114				

Import a Certificate

After storing the certificate in your computer, you need to import it in trusted root certification authorities using the following steps:

1 Open Fire fox and click Tools > Options.


2 In the Options page, click Privacy & Security, scroll to the bottom of the page, and then click View Certificates.



3 In the Certificate Manager, Click Authorities > Import.

die Hillende Blassey e	×
You California People Salvas Autoritas	
You have confidence on the that identity these settificates autorities	
Community Manager Section (1996)	λ.
UCA Global California Ballin Object Taken	
UCALISTENEES Verlanden Koel. Builtin Object Taken	
w un acto de la ciela	
Cortem Keel CA Built in Object Taken	
≥ unitete i echiekejike s2	
Certain Interest Network (24) Bellin Object Texer	
Certain Interest Network C4.2 Builtin Object Takey	
Mikaning Jaw	-
View Kolders append Kopend Address Iward	
	•

- 🌜 Senat Tile anna an Chrant Lana, so import -101 and a summer that the field of a Fp Trick constraints \mathcal{D} Cognition in Gen Volcan E 0 審 250 Nice Data race Set 3,95 100 10 Υ.P 20 15 10 ${\rm T}^{\rm c}_{\rm c}$ 1.0 664 25 Q, - $\{i\}$ 20 11 14 14.0 k ÷. \mathcal{H} 4h×. 641 Т. 25 ÷. Ħ. 886 \mathbf{M}_{i} 24 23. 20 300ist. \overline{M}_{2} a. эċ, ł 26 110J. ¢Φ. ${\mathcal U}_{i}$ 2 54 122 6 10. ΞŔ., 14 136 815 朣 $\overline{27}$ 10 144 10 10 10 144 10 1911 $M_{\rm e}^{\rm c}$ 1 ii. ю. 18 14. ÷. 21 ${\mathcal D}_{i,i}$ $C(\mathbf{R})$ ū. i. 141 -18 a, 93 2/2la. ie. a samari 4/DUDGELAR 741 Secure/ Cardiness 2.00 lis neres. 362236 Configure Disc. βz -See. Course
- 4 Use the Select File dialog box to locate the certificate and then click Open.

5 Select Thust this CA to identify websites and click OK.

Downloading Certificate	-
You have been asked to trust a new Certificate Authority (CA).	
Do you want to trust "usg00_30886F3FED32A" for the following purposes?	
Trust this CA to identify websites.	
Trust this CA to identify small users.	
Before trusting this CA for any purpose, you should examine its certificate and its policy and procedures (if evailable).	
CK Car	8

Removing a Certificate in Firefox

This section shows you how to remove a public key certificate in Firefox.

1 Open Fire fox and click Tools > Options.



2 In the Options page, click Privacy & Security, scroll to the bottom of the page, and then click View Certificates.



3 In the Certificate Manager, click Authorities and select the certificate you want to remove. Click Delete or Distrust.

×	ra Managar	CartRic		
)	erver Automae	arte (cales -ce	White Certif
	y these certificate authoritiks	met kland	fiszbes on filo	You have certi
2	Security Device		atte.	Cert Robel N
	Builtin Object Token	ek CA	fusical Netwo	Century T
	Builtin Ölgent Token	ick CA.2	fusted Netwo	Centurn T
			BR3FEDROA	× 15060_588
	Software Security Design		ser en cara esta esta esta esta esta esta esta est	uuq60_58
	 Builth Object Town Builth Object Town Builth Object Town 	Niner O	Clas I Poble Clas 2 Poble	Vaision
ote or EVaruat .	en., Egents, Dela	i kos	Edit Trust	ylen.,
1	6., Buith Sheat Toos 1., Buith Sheat Toos	Printery O Printery O	Chas I Public Chas 2 Public Chas 3 Public	Verbion Verbion Verbion

4 In the following dialog box, click **OK**.

Delete or Distruct CA Certificates	
You have requested to delete these CA cartificates. For by which has the same effect. Are you sure you want to delete	
NUMBER OF THE OWNER OF THE OWNER	
If you delete or distruct a cartificate authority (CA) certific	ate this application will no longer trust any
certificates insued by that CA	· · · · · · · · · · · · · · · · · · ·
	OK Cancel

5 The next time you go to the web site that issued the public key certificate you just removed, a certification error appears.

A PPENDIX B IPv6

O ve rvie w

IPv6 (Internet Protocol version 6), is designed to enhance IP address size and features. The increase in IPv6 address size to 128 bits (from the 32-bit IPv4 address) allows up to 3.4×10^{38} IP addresses.

IPv6 Addressing

The 128-bit IPv6 address is written as eight 16-bit hexadecimal blocks separated by colons (:). This is an example IPv6 address 2001:0db8:1a2b:0015:0000:1a2f:0000.

IPv6 addresses can be abbreviated in two ways:

- Leading zeros in a block can be omitted. So 2001:0db8:1a2b:0015:0000:0000:1a2f:0000 can be written as 2001:db8:1a2b:15:0:0:1a2f:0.
- Any number of consecutive blocks of zeros can be replaced by a double colon. A double colon can only appear once in an IPv6 address. So 2001:0db8:0000:0000:1a2f:0000:0000:0015 can be written as 2001:0db8::1a2f:0000:0000:0015, 2001:0db8:0000:0000:1a2f::0015, 2001:db8::1a2f:0:0:15 or 2001:db8:0:0:1a2f::15.

Prefix and Prefix Length

Similar to an IPv4 subnet mask, IPv6 uses an address prefix to represent the network address. An IPv6 prefix length specifies how many most significant bits (start from the left) in the address compose the network address. The prefix length is written as "/x" where x is a number. For example,

2001:db8:1a2b:15::1a2f:0/32

means that the first 32 bits (2001:db8) is the subnet prefix.

Link-local Address

A link-local address uniquely identifies a device on the local network (the LAN). It is similar to a "private IP address" in IPv4. You can have the same link-local address on multiple interfaces on a device. A link-local unicast address has a predefined prefix of fe80::/10. The link-local unicast address format is as follows.

Table 83 Link-local Unicast Address Format

1111 1110 10	0	Interface ID
10 bits	54 bits	64 bits

GlobalAddress

A global address uniquely identifies a device on the Internet. It is similar to a "public IP address" in IPv4. A global unicast address starts with a 2 or 3.

Unspecified Address

An unspecified address (0:0:0:0:0:0:0:0:0 or ::) is used as the source address when a device does not have its own address. It is similar to "0.0.0.0" in IPv4.

Loopback Address

A loopback address (0:0:0:0:0:0:0:0:1 or ::1) allows a host to send packets to itself. It is similar to "127.0.0.1" in IPv4.

Multic a st Addre ss

In IPv6, multicast addresses provide the same functionality as IPv4 broadcast addresses. Broadcasting is not supported in IPv6. A multicast address allows a host to send packets to all hosts in a multicast group.

Multicast scope allows you to determine the size of the multicast group. A multicast address has a predefined prefix of ff00::/8. The following table describes some of the predefined multicast addresses.

Table 84 Predefined Multicast Address

MULTICASTADDRESS	DESC RIPHO N
FF01:0:0:0:0:0:0:1	All hosts on a local node.
FF01:0:0:0:0:0:0:2	All routers on a local node.
FF02:0:0:0:0:0:0:1	All hosts on a local connected link.
FF02:0:0:0:0:0:0:2	All routers on a local connected link.
FF05:0:0:0:0:0:0:2	All routers on a local site.
FF05:0:0:0:0:1:3	All DHCP severs on a local site.

The following table describes the multicast addresses which are reserved and can not be assigned to a multicast group.

	1000000	
MULTICA	STADDRE	SS
FF00:0:0	:0:0:0:0	: 0
FF01:0:0	:0:0:0:0	: 0
FF02:0:0	:0:0:0:0	: 0
FF03:0:0	:0:0:0:0	: 0
FF04:0:0	:0:0:0:0	: 0
FF05:0:0	:0:0:0:0	: 0
FF06:0:0	:0:0:0:0	: 0
FF07:0:0	:0:0:0:0	: 0
FF08:0:0	:0:0:0:0	: 0
FF09:0:0	:0:0:0:0	: 0
FF0A:0:0	:0:0:0:0	: 0
FF0B:0:0	:0:0:0:0	: 0
FF0C:0:0	:0:0:0:0	: 0
FF0D:0:0	:0:0:0:0	: 0
FF0E:0:0	:0:0:0:0	: 0
FF0F:0:0	:0:0:0:0	: 0

${\bf Subnet\,Ma\,sking}$

Interface ID

In IPv6, an interface ID is a 64-bit identifier. It identifies a physical interface (for example, an Ethernet port) or a virtual interface (for example, the management IP address for a VLAN). One interface should have a unique interface ID.

EUI-64

The EUI-64 (Extended Unique Identifier) defined by the IEEE (Institute of Electrical and Electronics Engineers) is an interface ID format designed to adapt with IPv6. It is derived from the 48-bit (6-byte) Ethernet MAC address as shown next. EUI-64 inserts the hex digits fffe between the third and fourth bytes of the MAC address and complements the seventh bit of the first byte of the MAC address. See the following example.

Table 86

MAC		(00	:	13	:	49	:	12	:	34	:	56		
Table 87															
EUI-64	02	: 1	13	:	49	:	FF	:	FE	:	12	:	34	•	56

State less Autoconfiguration

With stateless autoconfiguration in IPv6, addresses can be uniquely and automatically generated. Unlike DHCPv6 (Dynamic Host Configuration Protocol version six) which is used in IPv6 stateful autoconfiguration, the owner and status of addresses don't need to be maintained by a DHCP server. Every IPv6 device is able to generate its own and unique IP address automatically when IPv6 is initiated on its interface. It combines the prefix and the interface ID (generated from its own Ethernet MAC address, see Interface ID and EUI-64) to form a complete IPv6 address.

When IPv6 is enabled on a device, its interface automatically generates a link-local address (beginning with fe80).

When the interface is connected to a network with a router and the Zyxel Device is set to automatically obtain an IPv6 network prefix from the router for the interface, it generates ¹ another address which combines its interface ID and global and subnet information advertised from the router. This is a routable global IP address.

DHC Pv6

The Dynamic Host Configuration Protocol for IPv6 (DHCPv6, RFC 3315) is a server-client protocol that allows a DHCP server to assign and pass IPv6 network addresses, prefixes and other configuration information to DHCP clients. DHCPv6 servers and clients exchange DHCP messages using UDP.

^{1.} In IPv6, all network interfaces can be associated with several addresses.

Each DHCP client and server has a unique DHCP Unique IDentifier (DUID), which is used for identification when they are exchanging DHCPv6 messages. The DUID is generated from the MAC address, time, vendor assigned ID and/or the vendor's private enterprise number registered with the IANA. It should not change over time even after you reboot the device.

Identity Association

An Identity Association (IA) is a collection of addresses assigned to a DHCP client, through which the server and client can manage a set of related IP addresses. Each IA must be associated with exactly one interface. The DHCP client uses the IA assigned to an interface to obtain configuration from a DHCP server for that interface. Each IA consists of a unique IAID and associated IP information. The IA type is the type of address in the IA. Each IA holds one type of address. IA_NA means an identity association for non-temporary addresses and IA_TA is an identity association for temporary addresses. An IA_NA option contains the T1 and T2 fields, but an IA_TA option does not. The DHCPv6 server uses T1 and T2 to control the time at which the client contacts with the server to extend the lifetimes on any addresses in the IA_NA were obtained) a Renew message. If the time T2 is reached and the server does not respond, the client sends a Rebind message to any available server (S2). For an IA_TA, the client may send a Renew or Rebind message at the client's discretion.



DHCP Re la y Agent

A DHCP relay agent is on the same network as the DHCP clients and helps forward messages between the DHCP server and clients. When a client cannot use its link-local address and a well-known multicast address to locate a DHCP server on its network, it then needs a DHCP relay agent to send a message to a DHCP server that is not attached to the same network.

The DHCP relay agent can add the remote identification (remote-ID) option and the interface-ID option to the Relay-Forward DHCPv6 messages. The remote-ID option carries a user-defined string, such as the system name. The interface-ID option provides slot number, port information and the VLAN ID to the DHCPv6 server. The remote-ID option (if any) is stripped from the Relay-Reply messages before the relay agent sends the packets to the clients. The DHCP server copies the interface-ID option from the Relay-Forward message into the Relay-Reply message and sends it to the relay agent. The interface-ID should not change even after the relay agent restarts.

Pre fix De le g a tio n

Prefix delegation enables an IPv6 router to use the IPv6 prefix (network address) received from the ISP (or a connected uplink router) for its LAN. The Zyxel Device uses the received IPv6 prefix (for example, 2001:db2::/48) to generate its LAN IP address. Through sending Router Advertisements (RAs) regularly by multicast, the Zyxel Device passes the IPv6 prefix information to its LAN hosts. The hosts then can use the prefix to generate their IPv6 addresses.

IC MPv6

Internet Control Message Protocol for IPv6 (ICMPv6 or ICMP for IPv6) is defined in RFC 4443. ICMPv6 has a preceding Next Header value of 58, which is different from the value used to identify ICMP for IPv4. ICMPv6 is an integral part of IPv6. IPv6 nodes use ICMPv6 to report errors encountered in packet processing and perform other diagnostic functions, such as "ping".

Neighbor Discovery Protocol (NDP)

The Neighbor Discovery Protocol (NDP) is a protocol used to discover other IPv6 devices and track neighbor's reachability in a network. An IPv6 device uses the following ICMPv6 messages types:

- Neighbor solicitation: A request from a host to determine a neighbor's link-layer address (MAC address) and detect if the neighbor is still reachable. A neighbor being "reachable" means it responds to a neighbor solicitation message (from the host) with a neighbor advertisement message.
- Neighbor advertisement: A response from a node to announce its link-layer address.
- Router solicitation: A request from a host to locate a router that can act as the default router and forward packets.
- Router advertisement: A response to a router solicitation or a periodical multicast advertisement from a router to advertise its presence and other parameters.

IPv6 Cache

An IPv6 host is required to have a neighbor cache, destination cache, prefix list and default router list. The Zyxel Device maintains and updates its IPv6 caches constantly using the information from response messages. In IPv6, the Zyxel Device configures a link-local address automatically, and then sends a neighbor solicitation message to check if the address is unique. If there is an address to be resolved or verified, the Zyxel Device also sends out a neighbor solicitation message. When the Zyxel Device receives a neighbor advertisement in response, it stores the neighbor's link-layer address in the neighbor cache. When the Zyxel Device uses a router solicitation message to query for a router and receives a router advertisement message, it adds the router's information to the neighbor cache, prefix list and destination cache. The Zyxel Device creates an entry in the default router list cache if the router can be used as a default router.

When the Zyxel Device needs to send a packet, it first consults the destination cache to determine the next hop. If there is no matching entry in the destination cache, the Zyxel Device uses the prefix list to determine whether the destination address is on-link and can be reached directly without passing through a router. If the address is onlink, the address is considered as the next hop. Otherwise, the Zyxel Device determines the next-hop from the default router list or routing table. Once the next hop IP address is known, the Zyxel Device looks into the neighbor cache to get the link-layer address and sends the packet when the neighbor is reachable. If the Zyxel Device cannot find an entry in the neighbor cache or the state for the neighbor is not reachable, it starts the address resolution process. This helps reduce the number of IPv6 solicitation and advertisement messages.

Multic a st Liste ner Disc overy

The Multicast Listener Discovery (MLD) protocol (defined in RFC 2710) is derived from IPv4's Internet Group Management Protocol version 2 (IGMPv2). MLD uses ICMPv6 message types, rather than IGMP message types. MLDv1 is equivalent to IGMPv2 and MLDv2 is equivalent to IGMPv3.

MLD allows an IPv6 switch or router to discover the presence of MLD listeners who wish to receive

multicast packets and the IP addresses of multicast groups the hosts want to join on its network.

MLD snooping and MLD proxy are analogous to IGMP snooping and IGMP proxy in IPv4.

MLD filtering controls which multicast groups a port can join.

MID Messages

A multicast router or switch periodically sends general queries to MLD hosts to update the multicast forwarding table. When an MLD host wants to join a multicast group, it sends an MLD Report message for that address.

An MLD Done message is equivalent to an IGMP Leave message. When an MLD host wants to leave a multicast group, it can send a Done message to the router or switch. The router or switch then sends a group-specific query to the port on which the Done message is received to determine if other devices connected to this port should remain in the group.

Example - Enabling IPv6 on Windows XP/2003/Vista

By default, Windows XP and Windows 2003 support IPv6. This example shows you how to use the ipv6 install command on Windows XP/2003 to enable IPv6. This also displays how to use the ipconfig command to see auto-generated IP addresses.

IPv6 is installed and enabled by default in Windows Vista. Use the *ipconfig* command to check your automatic configured IPv6 address as well. You should see at least one IPv6 address available for the interface on your computer.

Example - Enabling DHC Pv6 on Windows XP

Windows XP does not support DHCPv6. If your network uses DHCPv6 for IP address assignment, you have to additionally install a DHCPv6 client software on your Windows XP. (Note: If you use static IP addresses or Router Advertisement for IPv6 address assignment in your network, ignore this section.)

This example uses Dibbler as the DHCPv6 client. To enable DHCPv6 client on your computer:

- 1 Install Dibbler and select the DHCPv6 client option on your computer.
- 2 After the installation is complete, select Start > All Programs > Dib bler DHC Pv6 > C lient Install as service.
- 3 Select Start > Control Panel > Administrative Tools > Services.
- 4 Double click Dibbler- a DHC Pv6 client.

6 850 568 전 전	36 2 5 1 1 1 1 1					
we want to all	9. Services (Local)	51				
	llibhler a DICEve client	South	Other for	290	State Feet	Log Of Ro
		State Second Second and Second	$\mathbf{b} \in \mathbf{b} : \mathbf{b} \in \mathbf{b}$	 Contractory 	Minister 1	los veha
	And the day of the second second	Petropole 4	Hendessin	28.96	ie, somedic	bose Gyntern
	<u>Boars</u> the service	E Marie M ** at	Submission.	12223	Muthan E	10 di Sestanti
	- 430	The second card of the second card	Hereits it	28.35	4.50° 635	Loss Creens
	exected as we will	🚡 Shide fersal dia den	distribution of	1000	200 C	Helpine Same
	Description	figure det	Factoria a	28.35	W. Som edic	House and a
	Obbler - a contable 10 8 296.	Sector to be	$(Q_{i}, Q_{i}) \in \mathcal{O}(Q_{i})$	Materia)	W. 110.01	hav yeshari
	This wEHCIVE class, version	🏂 De villos	Dradies e c	29.30	ie, zoniedo	bios Selfera
		 Epideoide Scheide des Process 	The second second	Sec. 1	Page 4	los restrictions
	0.7.2	📲 Ruk List Swiching Schoolshifter 👘	Frankstein, 2		759.6	teor Setting
		Bellevill in a growth	Charlenda -		200 B (1)	In 2.3 Shirth

5 Click Start and then OK.

end lugite	Franking Personality	
ente une	D10%30.40	
an system	Code - a DHCT-O clark	
Negalip: Set	Citle laper della avector rector/07.2	HOVE AN
Peinto paos ab La magazi histo	e CMCPACulture of other states on eller M	an an ce d'UNA
	- the second sec	
ar shipt a	a de rele	
St. sysley e Section set of	Shinged	
- 5165		Decarte
indersder Jød	Skipel 28 Pizz	
ensie solar Stat	Plant 7% Plan	Decarra

6 Now your computer can obtain an IPv6 address from a DHCPv6 server.

Example - Enabling IPv6 on Windows 7

Windows 7 supports IPv6 by default. DHCPv6 is also enabled when you enable IPv6 on a Windows 7 computer.

To enable IPv6 in Windows 7:

- 1 Select Control Panel > Network and Sharing Center > Local Area Connection.
- 2 Select the Internet Protocol Version 6 (TCP/IPv6) checkbox to enable it.
- 3 Click **OK** to save the change.

etworking	
Connect using	
🔮 Broadcore //	etXberne Gigebit Ethernet
This convection use	Configure
🗷 📕 Cic S Pardo	
E Bite and Pr	inter Sharing for Microsoft Networks
😥 💻 İrtamat in	And Veneral (ICP/E-II) ADDOI VENON 4 (ICP/IPV4)
🖂 💻 freening in	and where the second
🖂 💻 freening in	and when the second
E - Concerner Parenter Parente	
Initial Description TCP/IP version Initial	Character (Character)

- 4 Click Close to exit the Local Area Connection Status screen.
- 5 Select Start > All Programs > Accessories > Command Prompt.
- 6 Use the ipconfig command to check your dynamic IPv6 address. This example shows a global address (2001:b021:2d::1000) obtained from a DHCP server.

APPENDIX C CustomerSupport

In the event of problems that cannot be solved by using this manual, you should contact your vendor. If you cannot contact your vendor, then contact a Zyxel office for the region in which you bought the device.

See *https://www.zyxelcom/homepage.shtml* and also *https://www.zyxelcom/about_zyxel/zyxel_worldwide.shtml* for the latest information.

Please have the following information ready when you contact an office.

Required Information

- Product model and serial number.
- Warranty Information.
- Date that you received your device.
- Brief description of the problem and the steps you took to solve it.

Corporate Headquarters (Worldwide)

Ta iwa n

- Zyxel Communications Corporation
- http://www.zyxel.com

Asia

China

- Zyxel Communications (Shanghai) Corp.
 Zyxel Communications (Beijing) Corp.
 - Zyxel Communications (Tianjin) Corp.
- https://www.zyxel.com/cn/zh/

India

- Zyxel Technology India Pvt Ltd
- https://www.zyxel.com/in/en/

Ka za khsta n

- Zyxel Kazakhstan
- https://www.zyxel.kz

Ko re a

- Zyxel Korea Corp.
- http://www.zyxel.kr

Ma la ysia

- Zyxel Malaysia Sdn Bhd.
- http://www.zyxel.com.my

Pa kista n

- Zyxel Pakistan (Pvt.) Ltd.
- http://www.zyxel.com.pk

Philip pine s

- Zyxel Philippines
- http://www.zyxel.com.ph

Singapore

- Zyxel Singapore Pte Ltd.
- http://www.zyxel.com.sg

Ta iwa n

- Zyxel Communications Corporation
- https://www.zyxel.com/tw/zh/

Tha ila nd

- Zyxel Thailand Co., Ltd
- https://www.zyxel.com/th/th/

Vie tna m

- Zyxel Communications Corporation-Vietnam Office
- https://www.zyxel.com/vn/vi

Europe

Be la rus

- Zyxel BY
- https://www.zyxel.by

Be lg ium

- Zyxel Communications B.V.
- https://www.zyxel.com/be/nl/

https://www.zyxel.com/be/fr/

Bulg a ria

- Zyxel България
- https://www.zyxel.com/bg/bg/

Czech Republic

- Zyxel Communications Czech s.r.o
- https://www.zyxel.com/cz/cs/

De nm a rk

- Zyxel Communications A/S
- https://www.zyxel.com/dk/da/

Esto nia

- Zyxel Estonia
- https://www.zyxel.com/ee/et/

Finla nd

- Zyxel Communications
- https://www.zyxel.com/fi/fi/

Fra nc e

- Zyxel France
- https://www.zyxel.fr

Gemany

- Zyxel Deutschland GmbH
- https://www.zyxel.com/de/de/

Hung a ry

- Zyxel Hungary & SEE
- https://www.zyxel.com/hu/hu/

Ita ly

- Zyxel Communications Italy
- https://www.zyxel.com/it/it/

La tvia

- Zyxel Latvia
- https://www.zyxel.com/lv/lv/

Lithua nia

- Zyxel Lithuania
- https://www.zyxel.com/lt/lt/

Ne the rlands

- Zyxel Benelux
- https://www.zyxel.com/nl/nl/

Norway

- Zyxel Communications
- https://www.zyxel.com/no/no/

Poland

- Zyxel Communications Poland
- https://www.zyxel.com/pl/pl/

Romania

- Zyxel Romania
- https://www.zyxel.com/ro/ro

Russia

- Zyxel Russia
- https://www.zyxel.com/ru/ru/

Slo va kia

- Zyxel Communications Czech s.r.o. organizacna zlozka
- https://www.zyxel.com/sk/sk/

Spain

- Zyxel Communications ES Ltd
- https://www.zyxel.com/es/es/

Sweden

- Zyxel Communications
- https://www.zyxel.com/se/sv/

Switze rla nd

- Studerus AG
- https://www.zyxel.ch/de
- https://www.zyxel.ch/fr

Turke y

- Zyxel Turkey A.S.
- https://www.zyxel.com/tr/tr/

UK

- Zyxel Communications UK Ltd.
- https://www.zyxel.com/uk/en/

Ukra ine

- Zyxel Ukraine
- http://www.ua.zyxel.com

South America

Arg e ntina

- Zyxel Communications Corporation
- https://www.zyxel.com/co/es/

Bra zil

- Zyxel Communications Brasil Ltda.
- https://www.zyxel.com/br/pt/

Colombia

- Zyxel Communications Corporation
- https://www.zyxel.com/co/es/

Ecuador

- Zyxel Communications Corporation
- https://www.zyxel.com/co/es/

South America

- Zyxel Communications Corporation
- https://www.zyxel.com/co/es/

Middle East

Isra e l

- Zyxel Communications Corporation
- http://il.zyxel.com/

Middle East

- Zyxel Communications Corporation
- https://www.zyxel.com/me/en/

North America

USA

- Zyxel Communications, Inc. North America Headquarters
- https://www.zyxel.com/us/en/

O c e a nia

Austra lia

- Zyxel Communications Corporation
- https://www.zyxel.com/au/en/

A fric a

So uth Afric a

- Nology (Pty) Ltd.
- https://www.zyxel.com/za/en/

A PPENDIX D Legal Information

Copyright

Copyright © 2021 by Zyxel and/ or its affiliates

The contents of this publication may not be reproduced in any part or as a whole, transcribed, stored in a retrieval system, translated into any language, or transmitted in any form or by any means, electronic, mechanical, magnetic, optical, chemical, photocopying, manual, or otherwise, without the prior written permission of Zyxel and/ or its affiliates.

Published by Zyxel and/ or its affiliates. All rights reserved.

Disc la im e rs

Zyxel does not assume any liability arising out of the application or use of any products, or software described herein. Neither does it convey any license under its patent rights nor the patent rights of others. Zyxel further reserves the right to make changes in any products described herein without notice. This publication is subject to change without notice.

Your use of the Zyxel Device is subject to the terms and conditions of any related service providers.

Trademarks

Trademarks mentioned in this publication are used for identification purposes only and may be properties of their respective owners.

Regulatory Notice and Statement

UNITED STATES of AMERICA



The following information applies if you use the product within USA area.

FCC EMC Statement

- This device complies with part 15 of the FCC Rules. Operation is subject to the following two conditions:
 (1) This device may not cause harmful interference, and
- (2) this device must accept any interference received, including interference that may cause undesired operation.
 Changes or modifications not expressly approved by the party responsible for compliance could void the user's authority to operate the device.
- This product has been tested and complies with the specifications for a Class B digital device, pursuant to Part 15 of the FCC Rules. These limits are designed to provide reasonable protection against harmful interference in a residential installation. This device generates, uses, and can radiate radio frequency energy and, if not installed and used according to the instructions, may cause harmful interference to radio communications. However, there is no guarantee that interference will not occur in a particular installation.
- If this device does cause harmful interference to radio or television reception, which is found by turning the device off and on, the user is
 encouraged to try to correct the interference by one or more of the following measures:
 - Reorient or relocate the receiving antenna
 - Increase the separation between the devices
 - Connect the equipment to an outlet other than the receiver's
 - Consult a dealer or an experienced radio/TV technician for assistance

FCC Caution: Any changes or modifications not expressly approved by the party responsible for compliance could void the user's authority to operate this equipment.

FCC Radiation Exposure Statement

- This device complies with FCC RF radiation exposure limits set forth for an uncontrolled environment.
- This transmitter must be at least 20 cm from the user and must not be co-located or operating in conjunction with any other antenna or transmitter. This transmitter must be at least 30 cm (WAC6553D-E) from the user and must not be co-located or operating in conjunction with any other antenna or transmitter.
- Country Code selection feature to be disabled for products marketed to the US/CANADA.
- Operation of this device is restricted to indoor use only, except for relevant user's manual mention that this device can be installed into the external environment. (WAC6553D-E is a device for outdoor use.)

BRA ZIL

The following applies if you use the product within Brazil.

Este equipamento opera em caráter secundário, isto é, não tem direito a proteção contra interferência prejudicial, mesmo de estações do mesmo tipo, e não pode causar interferência a sistemas operando em caráter primário.

CANADA

The following information applies if you use the product within Canada area.

Innovation, Science and Economic Development Canada ICES Statement

CAN ICES-3 (B)/NMB-3(B)

Innovation, Science and Economic Development Canada RSS-GEN & RSS-247 Statement

- This device contains licence-exempt transmitter(s)/receiver(s) that comply with Innovation, Science and Economic Development Canada's
 licence-exempt RSS(s). Operation is subject to the following two conditions: (1) this device may not cause interference, and (2) this device
 must accept any interference, including interference that may cause undesired operation of the device.
- This radio transmitter 2468C-11ACAP22W(WAC500H), 2468C-11ACAP22(WAC500 and NWA1123ACV3), 2468C-WAC5302DS(WAC5302D-Sv2), (2468C-NWA5123AC (NWA1123AC V2), 2468C-NWA5123AC HD), 2468C-WAC6502DE (WAC6502D-S), 2468C-WAC6503DS (WAC6303DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6302DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS (WAC6303DS), 2468C-WAC6503DS), 2468C-WAC6503DS (WAC6303DS), 2468C-MAC6503DS (WAC6303DS), 2468C-MAC6503DS (WAC6303DS), 2468C-MAC6503DS (WAC6303DS), 2468C-MAC6503DS (WAC6303DS), 2468C-MAC6503DS (WAC6303DS (WAC6303DS), 2468C-MAC6503DS (W

Antenna Information

ANTENNA MODEL	NO.	TYPE	CONNECTOR	2.4 G GAIN (dBi)	5G GAIN (dBi)	REMARK
NWA1123-ACv2	1	PIFA	UFL	3.08		
	2	PIFA	UFL	3.07		
	3	PIFA	UFL		4.06 (5150~5250 MHz)	
					3.79 (5725~5850 MHz)	
	4	PIFA	UFL		3.99 (5150~5250 MHz)	
					3.78 (5725~5850 MHz)	
NWA1123-AC HD	1	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	2	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	3	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	4	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	5	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
NWA1302-AC	1	Loop	I-PEX	5.82 (2400-2483.5 MHz)		
WAC5302D-Sv2	2	Loop	I-PEX	5.02 (2400-2483.5 MHz)		
	3	PIFA	I-PEX		5 (5150-5250 MHz) 5 (5250-5350 MHz) 5 (5470-5725 MHz) 5 (5725-5850 MHz)	
NWA5123-AC	1	PIFA	U.FL	3.08 (2400-2483.5 MHz)		
	2	PIFA	U.FL	3.07 (2400-2483.5 MHz)		
	3	PIFA	U.FL		4.06 (5150-5250 MHz) 3.91 (5725-5850 MHz)	
	4	PIFA	U.FL		3.99 (5150-5250 MHz) 3.79 (5725-5850 MHz)	
NWA5123-AC HD	1	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	2	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	3	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	4	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	5	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
WAC6502D-E		Dipole	RSMA	5	7	
WAC6502D-S		Dipole	IPEX	4	6	
WAC6503D-S		Dipole	IPEX	4	6	
WAC6553D-E		Dipole	N type	4.5	7	
WAC6103D-I	1	PIFA	U.FL	3.28		Ceiling Mounted: Antenng 1, 2, 3
	2	PIFA	U.FL	3.37		Wall Mounted:
	3	PIFA	U.FL	3.15		Antenna 1, 2, 4
	4	Dipole	U.FL	4.33		
	5	Loop	U.FL		4.38 (5150-5250 MHz) 4.23 (5725-5850 MHz)	Ceiling Mounted: Antenna 5, 6, 7
	6	Loop	U.FL		4.31 (5150-5250 MHz) 4.22 (5725-5850 MHz)	Wall Mounted: Antenna 5, 6, 8
	7	Loop	U.FL		4.38 (5150-5250 MHz) 4.36 (5725-5850 MHz)	
	8	Dipole	U.FL		5.12 (5150-5250 MHz) 5.20 (5725-5850 MHz)	

ANTENNA MODEL	NO.	TYPE	CONNECTOR	2.4 G GAIN (dBi)	5G GAIN (dBi)	REMARK
WAC5302D-S	1	Loop	I-PEX	5.82 (2400-2483.5 MHz)		
	2	Loop	I-PEX	5.02 (2400-2483.5 MHz)		
	3	PIFA	I-PEX		5 (5150-5250 MHz) 5 (5250-5350 MHz) 5 (5470-5725 MHz) 5 (5725-5850 MHz)	
WAC6303D-S	1	Direction	U.FL	1.12 (2400-2483.5 MHz)		
	2	Direction	U.FL		1.29 (5150-5250 MHz) 1.07 (5725-5850 MHz)	
WAC6552D-S	1	Direction	I-PEX	0.8 (2400-2483.5 MHz)	4.22 (5150-5250 MHz)	
SECTX-DB r2.0					5.34 (5725-5850 MHz)	
WAX650S		Direction	U.FL	0 (2400-2483.5 MHz)	3.51 (5150-5250 MHz) 4.22 (5250-5350 MHz) 4.61 (5470-5725 MHz) 4.68 (5725-5850 MHz)	
WAX510D NWA110AX	1	Dipole	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
	2	PIFA	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
	3	Dipole	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
	4	Dipole	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
NWA210AX	1	Dipole	I-PEX		U-NII-1:7.8dBi	
WAX610D					U-NII-2A:7.7dBi	
					U-NII-2C:6.8dBi	
					U-NII-3:7.2dBi	
	2	PIFA	I-PEX	5.08dBi		
	3	PIFA	I-PEX	5.56dBi	U-NII-1:7.5dBi	
					U-NII-2A:6.8dBi	
					U-NII-2C:6.5dBi	
					U-NII-3:7.6dBi	
	4	Dipole	I-PEX	6.06dBi	U-NII-1:8.19dBi	Wall Mount
					U-NII-2A:7.7dBi	
					U-NII-2C:7.14dBi	
					U-NII-3:7.6dBi	
	5	Dipole	I-PEX		U-NII-1:6.8dBi	Ceiling Mount
					U-NII-2A:7.5dBi	
					U-NII-2C:5.81dBi	
					U-NII-3:6.99dBi	
	6	Dipole	I-PEX		U-NII-1:8.3dBi	
					U-NII-2A:7.8dBi	
					U-NII-2C:7.1dBi	
					U-NII-3:7.98dBi	
WAC500H	1	PIFA	N/A	OdBi	2.5dBi	
	2	PIFA	N/A	OdBi	2.5dBi	
WAC500	1	PIFA	N/A	OdBi	OdBi	
NWA1123ACv3	2	PIFA	N/A	OdBi	0dBi	

If the product with 5G wireless function operating in 5150-5250 MHz and 5725-5850 MHz, the following attention must be paid,

The device for operation in the band 5150-5250 MHz is only for indoor use to reduce the potential for harmful interference to co-channel •

mobile satellite systems. For devices with detachable antenna(s), the maximum antenna gain permitted for devices in the band 5725-5850 MHz shall be such that the equipment still complies with the e.i.r.p. limits specified for point-to-point and non-point-to-point operation as appropriate; and

- Where applicable, antenna type(s), antenna model(s), and the worst-case tilt angle(s) necessary to remain compliant with the e.i.r.p. elevation mask requirement set forth in Section 6.2.2.3 of RSS 247 shall be clearly indicated.
- If the product with 5G wireless function operating in 5250-5350 MHz and 5470-5725 MHz, the following attention must be paid.
- For devices with detachable antenna(s), the maximum antenna gain permitted for devices in the bands 5250-5350 MHz and 5470-5725 MHz shall be such that the equipment still complies with the e.i.r.p. limit.
- L'émetteur/récepteur exempt de licence contenu dans le présent appareil est conforme aux CNR d'Innovation, Sciences et Développement économique Canada applicables aux appareils radio exempts de licence. L'exploitation est autorisée aux deux conditions suivantes : (1) l'appareil ne doit pas produire de brouillage; (2) L'appareil doit accepter tout brouillage radioélectrique subi, même si le brouillage est susceptible d'en compromettre le fonctionnement.
- brouillage est susceptible d'en compromettre le fonctionnement.
 Le présent émetteur radio 2468C-11ACAP22W(WAC500H), 2468C-11ACAP22(WAC500 and NWA1123ACV3), 2468C-WAC5302DS(WAC5302D-Sv2), (2468C-NWA5123AC (NWA5123AC V2), 2468C-NWA5123ACHD (NWA1123-AC HD), 2468C-WAC5302DS (NWA1302-AC), 2468C-NWA5123AC (NWA5123-AC), 2468C-NWA5123ACHD (NWA5123-AC HD), 2468C-WAC6502DE (WAC6502D-S), WAC6502D-E), 2468C-WAC6503DS (WAC6502D-S), 2468C-WAC6503DS (WAC6503DS) (WAC6503DS) (WAC6503DS) (WAC6503DS) (WAC6503DS) (WAC6552DS (WAC6552D-S), 2468C-WAC6503DS) (WAC6503DS), 2468C-WAC6303D-S), 2468C-WAC6103DI (WAC6103D-I), 2468C-WAC6502D-S), 2468C-WAC650SD (WAC650S), 2468C-NAC6303DS (WAC6502D-S), 2468C-WAC6502D-E), 2468C-WAC6303DS (WAC6502D-S), 2468C-WAC6303DS), 2468C-WAC6303DS (WAC6502D-S), 2468C-WAC6303DS) (WAC6502D-S), 2468C-WAC6303DS (WAC6502D-S), 2468C-WAC6303DS) (WAC6303DS), 2468C-NAC6303DS (WAC6303D-S), 2468C-WAC6103DI (WAC6103D-I), 2468C-WAC6302DS), 2468C-WAX650S (WAX650S), 2468C-11AXAP22(NWA110AX and WAX510D) a dté approuvé par Innovation, Sciences et Développement économique Canada pour fonctionner avec les types d'antenne énumérés ci dessous et ayant un gain admissible maximal. Les types d'antenne non inclus dans cette liste, et dont le gain est supérieur au gain maximal indiqué pour tout type figurant sur la liste, sont strictement interdits pour l'exploitation de l'émetteur.

Informations Antenne

MODÈLE D'ANTENNE	NB.	TYPE	CONNECTEUR	2.4 G GAIN (dBi)	5G GAIN (dBi)	REMA RQ UE
NWA1123-ACv2	1	PIFA	UFL	3.08		
	2	PIFA	UFL	3.07		
	3	PIFA	UFL		4.06 (5150~5250 MHz)	
					3.79 (5725~5850 MHz)	
	4	PIFA	UFL		3.99 (5150~5250 MHz)	
					3.78 (5725~5850 MHz)	
NWA1123-AC HD	1	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	2	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	3	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	4	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	5	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
NWA1302-AC	1	Loop	I-PEX	5.82 (2400-2483.5 MHz)		
WAC5302D-\$v2	2	Loop	I-PEX	5.02 (2400-2483.5 MHz)		
	3	PIFA	I-PEX		5 (5150-5250 MHz) 5 (5250-5350 MHz) 5 (5470-5725 MHz) 5 (5725-5850 MHz)	
NWA5123-AC	1	PIFA	U.FL	3.08 (2400-2483.5 MHz)		
	2	PIFA	U.FL	3.07 (2400-2483.5 MHz)		
	3	PIFA	U.FL		4.06 (5150-5250 MHz) 3.91 (5725-5850 MHz)	
	4	PIFA	U.FL		3.99 (5150-5250 MHz) 3.79 (5725-5850 MHz)	
NWA5123-AC HD	1	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	2	PIFA	I-PEX	3 (2400-2483.5 MHz)		
	3	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	4	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
	5	Monopole	I-PEX		4 (5150-5250 MHz) 4 (5725-5850 MHz)	
WAC6502D-E		Dipole	RSMA	5	7	
WAC6502D-S		Dipole	IPEX	4	6	
WAC6503D-S		Dipole	IPEX	4	6	
WAC6553D-E		Dipole	N type	4.5	7	
WAC6103D-I	1	PIFA	U.FL	3.28		Ceiling Mounted: Antenna 1, 2, 3
	2	PIFA	U.FL	3.37		Wall Mounted:
	3	PIFA	U.FL	3.15		Antenna 1, 2, 4
	4	Dipole	U.FL	4.33		
	5	Loop	U.FL		4.38 (5150-5250 MHz) 4.23 (5725-5850 MHz)	Ceiling Mounted Antenna 5, 6, 7
	6	Loop	U.FL		4.31 (5150-5250 MHz) 4.22 (5725-5850 MHz)	Wall Mounted: Antenna 5, 6, 8
	7	Loop	U.FL		4.38 (5150-5250 MHz) 4.36 (5725-5850 MHz)	
	8	Dipole	U.FL		5.12 (5150-5250 MHz) 5.20 (5725-5850 MHz)	

MODÈLE D'ANTENNE	NB.	TYPE	CONNECTEUR	2.4 G GAIN (dBi)	5G GAIN (dBi)	REMARQ UE
WAC5302D-S	1	Loop	I-PEX	5.82 (2400-2483.5 MHz)		
	2	Loop	I-PEX	5.02 (2400-2483.5 MHz)		
	3	PIFA	I-PEX		5 (5150-5250 MHz) 5 (5250-5350 MHz) 5 (5470-5725 MHz) 5 (5725-5850 MHz)	
WAC6303D-S	1	Direction	U.FL	1.12 (2400-2483.5 MHz)		
	2	Direction	U.FL		1.29 (5150-5250 MHz) 1.07 (5725-5850 MHz)	
WAC6552D-S	1	Direction	I-PEX	0.8 (2400-2483.5 MHz)	4.22 (5150-5250 MHz)	
SECTX-DB r2.0					5.34 (5725-5850 MHz)	
WAX650S		Direction	U.FL	0 (2400-2483.5 MHz)	3.51 (5150-5250 MHz) 4.22 (5250-5350 MHz) 4.61 (5470-5725 MHz) 4.68 (5725-5850 MHz)	
WAX510D NWA110AX	1	Dipole	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
	2	PIFA	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
	3	Dipole	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
	4	Dipole	I-PEX	0 (2400-2483.5 MHz)	4.5 (5150-5350MHz) 5.2 (5470-5725MHz) 5.5 (5725-5850MHz)	
NWA210AX	1	Dipole	I-PEX		U-NII-1:7.8dBi	
WAX610D					U-NII-2A:7.7dBi	
					U-NII-2C:6.8dBi	
					U-NII-3:7.2dBi	
	2	PIFA	I-PEX	5.08dBi		
	3	PIFA	I-PEX	5.56dBi	U-NII-1:7.5dBi	
					U-NII-2A:6.8dBi	
					U-NII-2C:6.5dBi	
					U-NII-3:7.6dBi	
	4	Dipole	I-PEX	6.06dBi	U-NII-1:8.19dBi	Wall Mount
					U-NII-2A:7.7dBi	
					U-NII-2C:7.14dBi	
					U-NII-3:7.6dBi	
	5	Dipole	I-PEX		U-NII-1:6.8dBi	Ceiling Mount
					U-NII-2A:7.5dBi	
					U-NII-2C:5.81dBi	
					U-NII-3:6.99dBi	
	6	Dipole	I-PEX		U-NII-1:8.3dBi	
					U-NII-2A:7.8dBi	
					U-NII-2C:7.1dBi	
					U-NII-3:7.98dBi	
WAC500H	1	PIFA	N/A	OdBi	2.5dBi	WAC500H
	2	PIFA	N/A	OdBi	2.5dBi	
WAC500	1	PIFA	N/A	OdBi	OdBi	WAC500
NWA1123ACv3	2	PIFA	N/A	OdBi	OdBi	NWA1123ACv3

Lorsque la fonction sans fil 5G fonctionnant en 5150-5250 MHz and 5725-5850 MHz est activée pour ce produit, il est nécessaire de porter une attention particulière aux choses suivantes

 Les dispositifs fonctionnant dans la bande de 5 150 à 5 250 MHz sont réservés uniquement pour une utilisation à l'intérieur afin de réduire les risques de brouillage préjudiciable aux systèmes de satellites mobiles utilisant les mêmes canaux;

- Pour les dispositifs munis d'antennes amovibles, le gain maximal d'antenne permis (pour les dispositifs utilisant la bande de 5 725 à 5 850 MHz) doit être conforme à la limite de la p.i.r.e. spécifiée pour l'exploitation point à point et l'exploitation non point à point, selon le cas;
- Lorsqu'il y a lieu, les types d'antennes (s'il y en a plusieurs), les numéros de modèle de l'antenne et les pires angles d'inclinaison nécessaires pour rester conforme à l'exigence de la p.i.r.e. applicable au masque d'élévation, énoncée à la section 6.2.2.3 du CNR-247, doivent être clairement indiqués.

Lorsque la fonction sans fil 5G fonctionnant en 5250-5350 MHz et 5470-5725 MHz est activée pour ce produit, il est nécessaire de porter une attention particulière aux choses suivantes.

• Pour les dispositifs munis d'antennes amovibles, le gain maximal d'antenne permis pour les dispositifs utilisant les bandes de 5 250 à 5 350 MHz et de 5 470 à 5 725 MHz doit être conforme à la limite de la p.i.r.e.

Industry Canada radiation exposure statement

This equipment complies with ISED radiation exposure limits set forth for an uncontrolled environment. This equipment should be installed and operated with a minimum distance of 20 cm between the radiator and your body.

This equipment complies with ISED radiation exposure limits set forth for an uncontrolled environment. This equipment should be installed and operated with a minimum distance of 22cm (NWA1123-AC HD, NWA5123-AC HD) between the radiator and your body.

This equipment complies with ISED radiation exposure limits set forth for an uncontrolled environment. This equipment should be installed and operated with a minimum distance of 30 cm (WAC6553D-E) between the radiator and your body.

Déclaration d'exposition aux radiations:

Cet équipement est conforme aux limites d'exposition aux rayonnements ISED établies pour un environnement non contrôlé.Cet équipement doit être installé et utilisé avec un minimum de 20 cm de distance entre la source de rayonnement et votre corps.

Cet équipement est conforme aux limites d'exposition aux rayonnements ISED établies pour un environnement non contrôlé. Cet équipement doit être installé et utilisé avec un minimum de 22 cm (NWA1123-AC HD, NWA5123-AC HD) de distance entre la source de rayonnement et votre corps.

Cet équipement est conforme aux limites d'exposition aux rayonnements ISED établies pour un environnement non contrôlé.Cet équipement doit être installé et utilisé avec un minimum de 30 cm (WAC6553D-E) de distance entre la source de rayonnement et votre corps.

Caution:

(i) the device for operation in the band 5150-5250 MHz is only for indoor use to reduce the potential for harmful interference to co-channel mobile satellite systems;

(ii) the maximum antenna gain permitted for devices in the bands 5250-5350 MHz and 5470-5725 MHz shall comply with the e.i.r.p. limit; and

(iii) the maximum antenna gain permitted for devices in the band 5725-5825 MHz shall comply with the e.i.r.p. limits specified for point-to-point and non point-to-point operation as appropriate.

(iv) Users should also be advised that high-power radars are allocated as primary users (i.e. priority users) of the bands 5250-5350 MHz and 5650-5850 MHz and that these radars could cause interference and/or damage to LE-LAN devices.

(v) WAC6553D-E is an outdoor device and only uses 5G Band 4 (5725-5850 MHz).

Ave rtisse ment:

(i) les dispositifs fonctionnant dans la bande 5150-5250 MHz sont réservés uniquement pour une utilisation à l'intérieur afin de réduire les risques de brouillage préjudiciable aux systèmes de satellites mobiles utilisant les mêmes canaux;

(ii) le gain maximal d'antenne permis pour les dispositifs utilisant les bandes 5250-5350 MHz et 5 470-5 725 MHz doit se conformer à la limite de p.i.r.e.;

(iii) le gain maximal d'antenne permis (pour les dispositifs utilisant la bande 5725-5825 MHz) doit se conformer à la limite de p.i.r.e. spécifiée pour l'exploitation point à point et non point à point, selon le cas.

(iv) De plus, les utilisateurs devraient aussi être avisés que les utilisateurs de radars de haute puissance sont désignés utilisateurs principaux (c.-àd., qu'ils ont la priorité) pour les bandes 5250-5350 MHz et 5650-5850 MHz et que ces radars pourraient causer du brouillage et/ou des dommages aux dispositifs LAN-EL.

(v) WAC6553D-E est un appareil exterieur et seulement utilise 5G Bane 4 (5725-5850 MHz).

EURO PEAN UNION and UNITED KING DOM



The following information applies if you use the product within the European Union and United Kingdom.

Declaration of Conformity with Regard to EU Directive 2014/53/EU (Radio Equipment Directive, RED) and UK regulation

- Compliance information for wireless products relevant to the EU, United Kingdom and other Countries following the EU Directive 2014/53/EU (RED) and UK regulation. And this product may be used in all EU countries (and other countries following the EU Directive 2014/53/EU) and United Kingdom without any limitation except for the countries mentioned below table:
- In the majority of the EU and other European countries, the 5GHz bands have been made available for the use of wireless local area
 networks (LANs). Later in this document you will find an overview of countries in which additional restrictions or requirements or both are
 applicable. The requirements for any country may evolve. Zyxel recommends that you check with the local authorities for the latest status of
 their national regulations for the 5GHz wireless LANs.
- If this device for operation in the band 5150-5350 MHz, it is for indoor use only.
- This equipment should be installed and operated with a minimum distance of 20 cm between the radio equipment and your body.
- The maximum RF power operating for each band as follows:

NWA1123-ACv2 and NWA5123-AC

- The band 2,400 MHz to 2,483.5 MHz is 97.95 mW,
- The band 5,150 MHz to 5,350 MHz is 199.07 mW,
- The band 5,470 MHz to 5,725 MHz is 743.02 mW.

WAC 6503D-S

- The band 2,400 MHz to 2,483.5 MHz is 99.54 mW,
- The band 5,150 MHz to 5,350 MHz is 183.65 mW,
- The band 5,470 MHz to 5,725 MHz is 941.89 mW.

WAC6502D-Eand WAC6502D-S

- The band 2,400 MHz to 2,483.5 MHz is 94.19 mW,
- The band 5,150 MHz to 5,350 MHz is 194.98 mW,
- The band 5,470 MHz to 5,725 MHz is 986.28 mW.

WAC 6553D-E

- The band 2,400 MHz to 2,483.5 MHz is 92.26 mW,
- The band 5,150 MHz to 5,350 MHz is 198.61 mW,
- The band 5,470 MHz to 5,725 MHz is 995.41 mW.

NWA1123-AC PRO and WAC6103D-I

- The band 2,400 MHz to 2,483.5 MHz is 92.68 mW,
- The band 5,150 MHz to 5,350 MHz is 192.75 mW,
- The band 5,470 MHz to 5,725 MHz is 966.05 mW.

NWA1302-AC, WAC5301D-Sv2 and WAC5302D-S

- The band 2,400 MHz to 2,483.5 MHz is 93.33 mW,
- The band 5,150 MHz to 5,350 MHz is 192.31 mW,
- The band 5,470 MHz to 5,725 MHz is 391.74 mW.

NWA1123-AC HD and NWA5123-AC HD

- The band 2,400 MHz to 2,483.5 MHz is 97.274 mW,
- The band 5,150 MHz to 5,350 MHz is 198.61 mW,
- The band 5,470 MHz to 5,725 MHz is 995.40 mW.

WAC 6303D-S

- The band 2,400 MHz to 2,483.5 MHz is 194.09 mW,
- The band 5,150 MHz to 5,350 MHz is 198.61 mW,
- The band 5,470 MHz to 5,725 MHz is 995.41 mW.

WAC 6552D-S

- The band 2,400 MHz to 2,483.5 MHz is 93.11 mW,
- The band 5,150 MHz to 5,350 MHz is 198.61 mW,
- The band 5,470 MHz to 5,725 MHz is 914.11 mW.

WAC 500H

- The band 2,400 MHz to 2,483.5 MHz is 87.7 mW,
- The band 5,150 MHz to 5,350 MHz is 174.58 mW,
- The band 5,470 MHz to 5,725 MHz is 443.61 mW.

WAC500 and NWA1123ACv3

- The band 2,400 MHz to 2,483.5 MHz is 88.5 mW,
- The band 5,150 MHz to 5,350 MHz is 181.55 mW,
- The band 5,470 MHz to 5,725 MHz is 195,43 mW.

WAX650S

- The band 2,400 MHz to 2,483.5 MHz is 91.2 mW,
- The band 5,150 MHz to 5,350 MHz is 177.01 mW,
- The band 5.470 MHz to 5.725 MHz is 899.5 mW.

WAX510D and NWA110AX

- The band 2,400 MHz to 2,483.5 MHz is 85.31 mW,
- The band 5,150 MHz to 5,350 MHz is 172.19 mW,
- The band 5,470 MHz to 5,725 MHz is 651.63 mW.

WAX610D and NWA210AX

- The band 2,400 MHz to 2,483.5 MHz is 92.47 mW.
- The band 5,150 MHz to 5,350 MHz is 177.01 mW.

• The band 5,470 MHz to 5,725 MHz is 889.2 mW.

Български (Bulgarian)	С настоящото Zyxel декларира, че това оборудване е в съответствие със съществените изисквания и другите приложими разпоредбите на Директива 2014/53/ЕС.			
(bolganan)	National Restrictions			
	 The Belgian Institute for Postal Services and Telecommunications (BIPT) must be notified of any outdoor wireless link having a range exceeding 300 meters. Please check http://www.bipt.be for more details. Draadloze verbindingen voor buitengebruik en met een reikwijdte van meer dan 300 meter dienen aangemeld te worden bij het Belgisch Instituut voor postdiensten en telecommunicatie (BIPT). Zie http://www.bipt.be voor meer gegevens. Les liaisons sans fil pour une utilisation en extérieur d'une distance supérieure à 300 mètres doivent être notifiées à l'Institut Belge des services Postaux et des Télécommunications (IBPT). Visitez http://www.ibpt.be pour de plus amples détails. 			
Español (Spanish)	Por medio de la presente Zyxel declara que el equipo cumple con los requisitos esenciales y cualesquiera otras disposiciones aplicables o exigibles de la Directiva 2014/53/UE.			
Čeština (Czech)	Zyxel tímto prohlašuje, že tento zařízení je ve shodě se základními požadavky a dalšími příslušnými ustanoveními směrnice 2014/53/EU.			
Dansk (Danish)	Undertegnede Zyxel erklærer herved, at følgende udstyr udstyr overholder de væsentlige krav og øvrige relevante krav i direktiv 2014/53/EU.			
	Na tio na l Re stric tio ns			
	 In Denmark, the band 5150 - 5350 MHz is also allowed for outdoor usage. I Danmark må frekvensbåndet 5150 - 5350 også anvendes udendørs. 			
Deutsch (German)	Hiermit erklärt Zyxel, dass sich das Gerät Ausstattung in Übereinstimmung mit den grundlegenden Anforderungen und den übrigen einschlägigen Bestimmungen der Richtlinie 2014/53/EU befindet.			
Eesti keel (Estonian)	Käesolevaga kinnitab Zyxel seadme seadmed vastavust direktiivi 2014/53/EU põhinõuetele ja nimetatud direktiivist tulenevatele teistele asjakohastele sätetele.			
Ελληνικά (Greek)	ΜΕ ΤΗΝ ΠΑΡΟΥΣΑ ΖΥΧΕΙ ΔΗΛΩΝΕΙ ΟΤΙ εξοπλισμός ΣΥΜΜΟΡΦΩΝΕΤΑΙ ΠΡΟΣ ΤΙΣ ΟΥΣΙΩΔΕΙΣ ΑΠΑΙΤΗΣΕΙΣ ΚΑΙ ΤΙΣ ΛΟΙΠΕΣ ΣΧΕΤΙΚΕΣ ΔΙΑΤΑΞΕΙΣ ΤΗΣ ΟΔΗΓΙΑΣ 2014/53/EU.			
English	Hereby, Zyxel declares that this device is in compliance with the essential requirements and other relevant provisions of Directive 2014/53/EU.			
Français (French)	Par la présente Zyxel déclare que l'appareil équipements est conforme aux exigences essentielles et aux autres dispositions pertinentes de la directive 2014/53/EU.			
Hrvatski (Croatian)	Zyxel ovime izjavljuje da je radijska oprema tipa u skladu s Direktivom 2014/53/EU.			
Íslenska (Icelandic)	Hér með lýsir, Zyxel því yfir að þessi búnaður er í samræmi við grunnkröfur og önnur viðeigandi ákvæði tilskipunar 2014/53/ EU.			
Italiano (Italian)	Con la presente Zyxel dichiara che questo attrezzatura è conforme ai requisiti essenziali ed alle altre disposizioni pertinenti stabilite dalla direttiva 2014/53/EU.			
	 Na tio nal Re stric tio ns This product meets the National Radio Interface and the requirements specified in the National Frequency Allocation Table for Italy. Unless this wireless LAN product is operating within the boundaries of the owner's property, its use requires a "general authorization." Please check http://www.sviluppoeconomico.gov.it/ for more details. Questo prodotto è conforme alla specifiche di Interfaccia Radio Nazionali e rispetta il Piano Nazionale di ripartizione delle frequenze in Italia. Se non viene installato all'interno del proprio fondo, l'utilizzo di prodotti Wireless LAN richiede una "Autorizzazione Generale". Consultare http://www.sviluppoeconomico.gov.it/ per maggiori dettagli. 			
Latviešu valoda	Ar šo Zyxel deklarē, ka iekārtas atbilst Direktīvas 2014/53/EU būtiskajām prasībām un citiem ar to saistītajiem noteikumiem.			
(Latvian)	Na tio na l Re stric tio ns			
	 The outdoor usage of the 2.4 GHz band requires an authorization from the Electronic Communications Office. Please check http://www.esd.lv for more details. 2.4 GHz frekvenèu joslas izmantoðanai ârpus telpâm nepiecieðama atïauja no Elektronisko sakaru direkcijas. Vairâk informâcijas: http://www.esd.lv. 			
Lietuvių kalba (Lithuanian)	Šiuo Zyxel deklaruoja, kad šis įranga atitinka esminius reikalavimus ir kitas 2014/53/EU Direktyvos nuostatas.			
Magyar (Hungarian)	Alulírott, Zyxel nyilatkozom, hogy a berendezés megfelel a vonatkozó alapvető követelményeknek és az 2014/53/EU irányelv egyéb előírásainak.			
Malti (Maltese)	Hawnhekk, Zyxel, jiddikjara li dan tagħmir jikkonforma mal-ħtiġijiet essenzjali u ma provvedimenti oħrajn relevanti li hemm fid-Dirrettiva 2014/53/EU.			
Nederlands (Dutch)	Hierbij verklaart Zyxel dat het toestel uitrusting in overeenstemming is met de essentiële eisen en de andere relevante bepalingen van richtlijn 2014/53/EU.			
Polski (Polish)	Niniejszym Zyxel oświadcza, że sprzęt jest zgodny z zasadniczymi wymogami oraz pozostałymi stosownymi postanowieniami Dyrektywy 2014/53/EU.			
Português (Portuguese)	Zyxel declara que este equipamento está conforme com os requisitos essenciais e outras disposições da Directiva 2014/53/ EU.			

Română (Romanian)	Prin prezenta, Zyxel declară că acest echipament este în conformitate cu cerințele esențiale și alte prevederi relevante ale Directivei 2014/53/EU.
Slovenčina (Slovak)	Zyxel týmto vyhlasuje, že zariadenia spĺňa základné požiadavky a všetky príslušné ustanovenia Smernice 2014/53/EU.
Slovenščina (Slovene)	Zyxel izjavlja, da je ta oprema v skladu z bistvenimi zahtevami in ostalimi relevantnimi določili direktive 2014/53/EU.
Suomi (Finnish)	Zyxel vakuuttaa täten että laitteet tyyppinen laite on direktiivin 2014/53/EU oleellisten vaatimusten ja sitä koskevien direktiivin muiden ehtojen mukainen.
Svenska (Swedish)	Härmed intygar Zyxel att denna utrustning står I överensstämmelse med de väsentliga egenskapskrav och övriga relevanta bestämmelser som framgår av direktiv 2014/53/EU.
Norsk (Norwegian)	Erklærer herved Zyxel at dette utstyret er I samsvar med de grunnleggende kravene og andre relevante bestemmelser I direktiv 2014/53/EU.

No te s:

1. Although Norway, Switzerland and Liechtenstein are not EU member states, the EU Directive 2014/53/EU has also been implemented in those countries.

2. The regulatory limits for maximum output power are specified in EIRP. The EIRP level (in dBm) of a device can be calculated by adding the gain of the antenna used (specified in dBi) to the output power available at the connector (specified in dBm).

List of national codes	List of	na tio na l	lcodes
------------------------	---------	-------------	--------

COUNTRY	ISO 3166 2 LETTER CODE	COUNTRY	ISO 3166 2 LETTER CODE
Austria	AT	Liechtenstein	LI
Belgium	BE	Lithuania	LT
Bulgaria	BG	Luxembourg	LU
Croatia	HR	Malta	MT
Cyprus	СҮ	Netherlands	NL
Czech Republic	CR	Norway	NO
Denmark	DK	Poland	PL
Estonia	EE	Portugal	PT
Finland	FI	Romania	RO
France	FR	Serbia	RS
Germany	DE	Slovakia	SK
Greece	GR	Slovenia	SI
Hungary	HU	Spain	ES
Iceland	IS	Sweden	SE
Ireland	IE	Switzerland	СН
Italy	IT	Turkey	TR
Latvia	LV	United Kingdom	GB

Professional installation instruction (WAC6553D-E)

Please be advised that due to the unique function supplied by this product, the device is intended for use with our interactive entertainment software and licensed third-party only. The product will be distributed through controlled distribution channel and installed by trained professional and will not be sold directly to the general public through retail store.

1 Installation personal

This product is designed for specific application and needs to be installed by a qualified personal who has RF and related rule knowledge. The general user shall not attempt to install or change the setting.

2 Installation location

The product shall be installed at a location where the radiating antenna can be kept 30 cm from nearby person in normal operation condition to meet regulatory RF exposure requirement.

3 External antenna

Use only the antennas which have been approved by Zyxel Communications Corporation. The non-approved antenna(s) may produce unwanted spurious or excessive RF transmitting power which may lead to the violation of FCC/IC limit and is prohibited.

4 Installation procedure

Please refer to user's manual for the detail.

5 Warning

Please carefully select the installation position and make sure that the final output power does not exceed the limit set force in relevant rules. The violation of the rule could lead to serious federal penalty.

Instructions d'installation professionnelle (WAC6553D-E)

Veuillez noter que l'appareil etant dedie a une fonction unique, il doit etre utilise avec notre logiciel proprietaire de divertissement interactif. Ce produit sera propose par un reseau de distribution controle et installe par des professionels; il ne sera pas propose au grand public par le reseau de la grande distribution.

1 Installation

> Ce produit est destine a un usage specifique et doit etre installe par un personnel qualifie maitrisant les radiofrequences et les regles s'y rapportant. L'installation et les reglages ne doivent pas etre modifies par l'utilisateur final.

- 2 Emplacement d'installation En usage normal, afin de respecter les exigences reglementaires concernant l'exposition aux radiofrequences, ce produit doit etre installe de facon a respecter une distance de 30 cm entre l'antenne emettrice et les personnes.
- 3 Antenn externe

Utiliser uniquement les antennes approuvees par le fabricant. L'utilisation d'autres antennes peut conduire a un niveau de rayonnement essentiel ou non essentiel depassant les niveaux limites definis par FCC/IC, ce qui est interdit.

Procedure d'installation

Consulter le manuel d'utilisation.

5 Avertissement

> Choisir avec soin la position d'installation et s'assurer que la puissance de sortie ne depasse pas les limites en vigueur. La violation de cette regle peut conduire a de serieuses penalites federales.

Safety Warnings

- Do not use this product near water, for example, in a wet basement or near a swimming pool.
- Do not expose your device to dampness, dust or corrosive liquids.
- Do not store things on the device.
- Do not obstruct the device ventilation slots as insufficient airflow may harm your device. For example, do not place the device in an
- enclosed space such as a box or on a very soft surface such as a bed or sofa.
- Do not install, use, or service this device during a thunderstorm. There is a remote risk of electric shock from lightning.
- Connect ONLY suitable accessories to the device.
- Do not open the device or unit. Opening or removing covers can expose you to dangerous high voltage points or other risks. ONLY qualified service personnel should service or disassemble this device. Please contact your vendor for further information.
- Make sure to connect the cables to the correct ports.
- Place connecting cables carefully so that no one will step on them or stumble over them.
- Always disconnect all cables from this device before servicing or disassembling. Do not remove the plug and connect it to a power outlet by itself; always attach the plug to the power adaptor first before connecting it to a power outlet
- Do not allow anything to rest on the power adaptor or cord and do NOT place the product where anyone can walk on the power adaptor or cord.
- Please use the provided or designated connection cables/power cables/adaptors. Connect it to the right supply voltage (for example, 110V AC in North America or 230V AC in Europe). If the power adaptor or cord is damaged, it might cause electrocution. Remove it from the device and the power source, repairing the power adapter or cord is prohibited. Contact your local vendor to order a new one.
- Do not use the device outside, and make sure all the connections are indoors. There is a remote risk of electric shock from liahtning CAUTION: Risk of explosion if battery is replaced by an incorrect type, dispose of used batteries according to the instruction. Dispose them at
- the applicable collection point for the recycling of electrical and electronic devices. For detailed information about recycling of this product, please contact your local city office, your household waste disposal service or the store where you purchased the product.
- This device (WAC6553D-E, WAC6552D-S) must be grounded by qualified service personnel. Never defeat the ground conductor or operate the device in the absence of a suitably installed ground conductor. Contact the qualified service personnel if you are uncertain that suitable grounding is available. The following warning statements apply, where the disconnect device is not incorporated in the device or where the plug on the power
- supply cord is intended to serve as the disconnect device,
 - For permanently connected devices, a readily accessible disconnect device shall be incorporated external to the device;
 - For pluggable devices, the socket-outlet shall be installed near the device and shall be easily accessible.
- Do not use a power adapter that has a power cable longer than 3 meters.

Environment statement

ErP (Energy-related Products) (NWA1123ACv3, WAC500, WAC500H, WAC5302D-Sv2, NWA1123-ACv2, NWA1123-AC HD, NWA5123-AC, NWA5123-AC HD, WAC6502D-E, WAC6502D-S, WAC6503D-S, WAX510D, and NWA110AX)

Zyxel products put on the EU and United Kingdom market in compliance with the requirement of the European Parliament and the Council published

Directive 2009/125/EC and UK regulation establishing a framework for the setting of ecodesign requirements for energy-related products (recast), so called

as "ErP Directive (Energy-related Products directive) as well as ecodesign requirement laid down in applicable implementing measures, power consumption has satisfied regulation requirements which are:

Network standby power consumption < 8W, and/or

Off mode power consumption < 0.5W, and/or

Standby mode power consumption < 0.5W.

For wireless setting, please refer to the chapter about wireless settings for more detail.

Disposal and Recycling Information

The symbol below means that according to local regulations your product and/or its battery shall be disposed of separately from domestic waste. If this product is end of life, take it to a recycling station designated by local authorities. At the time of disposal, the separate collection of your product and/or its battery will help save natural resources and ensure that the environment is sustainable development.

Die folgende Symbol bedeutet, dass Ihr Produkt und/oder seine Batterie gemäß den örtlichen Bestimmungen getrennt vom Hausmüll entsorgt werden muss. Wenden Sie sich an eine Recyclingstation, wenn dieses Produkt das Ende seiner Lebensdauer erreicht hat. Zum Zeitpunkt der Entsorgung wird die getrennte Sammlung von Produkt und/oder seiner Batterie dazu beitragen, natürliche Ressourcen zu sparen und die Umwelt und die menschliche Gesundheit zu schützen.

El símbolo de abajo indica que según las regulaciones locales, su producto y/o su batería deberán depositarse como basura separada de la domástica. Cuando este producto alcance el final de su vida útil, llévelo a un punto limpio. Cuando llegue el momento de desechar el producto, la recogida por separado éste y/o su batería ayudará a salvar los recursos naturales y a proteger la salud humana y medioambiental.

Le symbole ci-dessous signifie que selon les réglementations locales votre produit et/ou sa batterie doivent être éliminés séparément des ordures ménagères. Lorsque ce produit atteint sa fin de vie, amenez-le à un centre de recyclage. Au moment de la mise au rebut, la collecte séparée de votre produit et/ou de sa batterie aidera à économiser les ressources naturelles et protéger l'environnement et la santé humaine.

Il simbolo sotto significa che secondo i regolamenti locali il vostro prodotto e/o batteria deve essere smaltito separatamente dai rifiuti domestici. Quando questo prodotto raggiunge la fine della vita di servizio portarlo a una stazione di riciclaggio. Al momento dello smaltimento, la raccolta separata del vostro prodotto e/o della sua batteria aiuta a risparmiare risorse naturali e a proteggere l'ambiente e la salute umana.

Symbolen innebär att enligt lokal lagstiftning ska produkten och/eller dess batteri kastas separat från hushållsavfallet. När den här produkten når slutet av sin livslängd ska du ta den till en återvinningsstation. Vid tiden för kasseringen bidrar du till en bättre miljö och mänsklig hälsa genom att göra dig av med den på ett återvinningsställe.







以下訊息僅適用於產品銷售至台灣地區

第十二條 經型式認證合格之低功率射頻電機,非經許可,公司,商號或使用者均不得擅自變更頻率、加大功率或變更原設計之特性及功能。 第十四條 低功率射頻電機之使用不得影響飛航安全及干擾合法通信;經發現有干擾現象時,應立即停用,並改善至無干擾時方得繼續使用。 前項合法通信,指依電信法規定作業之無線電通信。低功率射頻電機須忍受合法通信或工業、科學及醫療用電波輻射性電機設備之干擾。 無線資訊傳輸設備忍受合法通信之干擾且不得干擾合法通信;如造成干擾,應立即停用,俟無干擾之虞,始得繼續使用。 無線資訊傳輸設備的製造廠商應確保頻率穩定性,如依製造廠商使用手冊上所述正常操作,發射的信號應維持於操作頻帶中。 使用無線產品時,應避免影響附近雷達系統之操作。 高增益指向性天線只得應用於固定式點對點系統。

以下訊息僅適用於產品屬於專業安裝並銷售至台灣地區

本器材須經專業工程人員安裝及設定,始得設置使用,且不得直接販售給一般消費者。

安全警告 - 為了您的安全, 請先閱讀以下警告及指示:

- 請勿將此產品接近水、火焰或放置在高溫的環境。
- 避免設備接觸任何液體 切勿讓設備接觸水、雨水、高濕度、污水腐蝕性的液體或其他水份。
- 灰塵及污物 切勿接觸灰塵、污物、沙土、食物或其他不合適的材料。
- 雷雨天氣時,不要安裝,使用或維修此設備。有遭受電擊的風險。
- 切勿重摔或撞擊設備,並勿使用不正確的電源變壓器。
- 若接上不正確的電源變壓器會有爆炸的風險。
- 請勿隨意更換產品內的電池。
- 如果更換不正確之電池型式,會有爆炸的風險,請依製造商說明書處理使用過之電池。
- 請將廢電池丟棄在適當的電器或電子設備回收處。
- 請勿將設備解體。
- 請勿阻礙設備的散熱孔,空氣對流不足將會造成設備損害。
- 請插在正確的電壓供給插座 (如:北美 / 台灣電壓 110V AC,歐洲是 230V AC)。

- 假若電源變壓器或電源變壓器的纜線損壞,請從插座拔除,若您還繼續插電使用,會有觸電死亡的風險。
- 請勿試圖修理電源變壓器或電源變壓器的纜線,若有毀損,請直接聯絡您購買的店家,購買一個新的電源變壓器。
- 請勿將此設備安裝於室外,此設備僅適合放置於室內。
- 請勿隨一般垃圾丟棄。
- 請參閱產品背貼上的設備額定功率。
- 請參考產品型錄或是彩盒上的作業溫度。
- 產品沒有斷電裝置或者採用電源線的插頭視為斷電裝置的一部分,以下警語將適用:
 - 對永久連接之設備,在設備外部須安裝可觸及之斷電裝置;
 - 對插接式之設備,插座必須接近安裝之地點而且是易於觸及的。

About the Symbols

Various symbols are used in this product to ensure correct usage, to prevent danger to the user and others, and to prevent property damage. The meaning of these symbols are described below. It is important that you read these descriptions thoroughly and fully understand the contents.

Explanation of the Symbols

SYMBOL	EXPLANATION
	Alternating current (AC):
\sim	AC is an electric current in which the flow of electric charge periodically reverses direction.
	Direct current (DC):
	DC if the unidirectional flow or movement of electric charge carriers.
	Earth; ground:
Æ	A wiring terminal intended for connection of a Protective Earthing Conductor.
	Class II equipment:
	The method of protection against electric shock in the case of class II equipment is either double insulation or reinforced insulation.

Viewing Certific ations

Go to <u>http://www.zyxel.com</u> to view this product's documentation and certifications.

Zyxel Limited Warranty

Zyxel warrants to the original end user (purchaser) that this product is free from any defects in material or workmanship for a specific period (the Warranty Period) from the date of purchase. The Warranty Period varies by region. Check with your vendor and/or the authorized Zyxel local distributor for details about the Warranty Period of this product. During the warranty period, and upon proof of purchase, should the product have indications of failure due to faulty workmanship and/or materials. Zyxel will, at its discretion, repair or replace the defective products or components without charge for either parts or labor, and to whatever extent it shall deem necessary to restore the product of equal or higher value, and will be solely at the discretion of Zyxel. This warranty shall not apply if the product has been modified, misused, tampered with, damaged by an act of God, or subjected to abnormal working conditions.

Note

Repair or replacement, as provided under this warranty, is the exclusive remedy of the purchaser. This warranty is in lieu of all other warranties, express or implied, including any implied warranty of merchantability or fitness for a particular use or purpose. Zyxel shall in no event be held liable for indirect or consequential damages of any kind to the purchaser.

To obtain the services of this warranty, contact your vendor. You may also refer to the warranty policy for the region in which you bought the device at http://www.zyxel.com/web/support_warranty_info.php.

Registration

Register your product online at www.zyxel.com to receive e-mail notices of firmware upgrades and related information.

Open Source Licenses

This product contains in part some free software distributed under GPL license terms and/or GPL like licenses. To request the source code covered under these licenses, please go to: https://www.zyxel.com/form/gpl_oss_software_notice.shtml

Index

Symbols

Numbers

802.11k **19** 802.11r **19** 802.11v **19**

Α

AC. See AP Controller access 30 access privileges 16 access users 86 see also users 86 admin users 86 multiple logins 91 see also users 86 alerts 157, 161, 162, 164, 165, 166 AP Controller 19, 22 applications MBSSID 16 Repeater 13 Assisted Roaming, see 802.11k/v

В

backing up configuration files 169 Basic Service Set see BSS Bluetooth BLE, see Bluetooth Low Energy Bluetooth Low Energy 19 boot module 174 BSS 16

С

CA and certificates 123 CA (Certificate Authority), see certificates CEF (Common Event Format) 158, 164 Certificate Authority (CA) see certificates Certificate Management Protocol (CMP) 128 Certificate Revocation List (CRL) 123 vs OCSP 137 certificates 122 advantages of 123 and CA 123 and FTP 156 and HTTPS 144 and SSH 153 and WWW 145 certification path 123, 130, 135 expired 123 factory-default 123 file formats 123 fingerprints 131, 136 importing 126 not used for encryption 123 revoked 123 self-signed 123, 127 serial number 130, 135 storage space 125, 133 thumbprint algorithms 124 thumbprints 124 used for authentication 123 verifying fingerprints 124 certification requests 127, 128 certifications viewing 250 channel 17 CLI 27, 35 button 35

messages 35 popup window 35 Reference Guide 2 cold start 42 commands 27 sent by Web Configurator 35 Common Event Format (CEF) 158, 164 configuration information 178, 193 configuration files 167 at restart 169 backing up 169 downloading 170 downloading with FTP 155 editing 167 how applied 168 lastgood.conf 169, 171 managing 168 startup-config.conf 171 startup-config-bad.conf 169 syntax 167 system-default.conf 171 uploading 172 uploading with FTP 155 use without restart 167 contact information 238 cookies 30 copyright 244 CPU usage 45, 47 current date/time 45, 139 daylight savings 141 setting manually 142 time server 142 customer support 238

D

date 139 daylight savings 141 DCS 76 DHCP 139 and domain name 139 diagnostics 178, 193 disclaimer 244 domain name 139 dual radios 17 dual-radio application 17 dynamic channel selection 76

Ε

encryption **13** ESSID **203** Extended Service Set IDentification **93**

F

Fast Roaming, see 802.11r FCC interference statement 244 file extensions configuration files 167 shell scripts 167 file manager 167 Firefox 30 firmware and restart 173 boot module, see boot module current version 45, 174 getting updated 173 uploading 173, 174 uploading with FTP 155 flash usage 45 FTP 27, 155 and certificates 156 with Transport Layer Security (TLS) 156

G

Guide CLI Reference 2

Η

HTTP over SSL, see HTTPS redirect to HTTPS **145**

NWA/WAC/WAX Series User's Guide

vs HTTPS 144 HTTPS 143 and certificates 144 authenticating clients 144 avoiding warning messages 147 example 145 vs HTTP 144 with Internet Explorer 146 with Netscape Navigator 146 HyperText Transfer Protocol over Secure Socket Layer, see HTTPS

I

IEEE 802.1x 94 interface status 46 interfaces as DHCP servers 139 interference 17 Internet Explorer 30 Internet Protocol version 6, see IPv6 IP Address 69, 189 gateway IP address 69 IP subnet 69 IPv6 230 addressing 230 EUI-64 232 global address 230 interface ID 232 link-local address 230 Neighbor Discovery Protocol 230 ping 230 prefix 230 prefix length 230 stateless autoconfiguration 232 unspecified address 231

J

Java permissions **30** JavaScripts **30**

Κ

key pairs 122

L

lastgood.conf 169, 171 LED suppression 180 LEDs 28 Locator LED 181 log messages categories 162, 164, 165, 166 debugging 66 regular 66 types of 66 logout Web Configurator 34 logs e-mail profiles 157 e-mailing log messages 68, 161 formats 158 log consolidation 162 settings 157 syslog servers 157 system 157 types of 157

Μ

MAC address range 45 management mode 20 Management, NCC 20 Management, Standalone 20 managing the device good habits 27 using FTP, see FTP MBSSID 16 memory usage 45, 48 messages CLI 35 mode, default 20 model name 44 My Certificates, see also certificates 125

Ν

NCC, see Nebula Control Center Nebula Control Center 20 Netscape Navigator 30 Network Time Protocol (NTP) 141

0

objects certificates 122 users, account user 86 Online Certificate Status Protocol (OCSP) 137 vs CRL 137 overview 42, 186

Ρ

pop-up windows power off power on **42** product registration Public-Key Infrastructure (PKI) public-private key pairs

R

radio 17 Radio Frequency monitor 12 reboot 42, 183 vs reset 183 Reference Guide, CLI 2 registration product 250 remote management FTP, see FTP WWW, see WWW reset 205 vs reboot 183 vs shutdown 184 RESET button 43, 205 restart 183 RF interference 17 RF monitor, see Radio Frequency Monitor RFC 2510 (Certificate Management Protocol or CMP) 128 Rivest, Shamir and Adleman public-key algorithm (RSA) 127 RSA 127, 135, 136 RSSI threshold 99

S

SCEP (Simple Certificate Enrollment Protocol) 128 screen resolution 30 Secure Socket Layer, see SSL serial number 44 service control and users 143 limitations 143 timeouts 143 Service Set 93 Service Set Identifier see SSID shell scripts 167 downloading 176, 193 editing 175, 192 how applied 168 managing 175, 192 syntax 167 uploading 177, 193 shutdown 43, 184 vs reset 184 Simple Certificate Enrollment Protocol (SCEP) 128 SSH 151 and certificates 153 client requirements 153 encryption methods 153 for secure Telnet 154 how connection is established 152 versions 153

with Linux 154 with Microsoft Windows 154 SSID 16 SSID profile pre-configured 16 SSID profiles 16 SSL 143 starting the device 42 startup-config.conf 171 if errors 169 missing at restart 169 present at restart 169 startup-config-bad.conf 169 station 76 status 187 stopping the device 42 supported browsers 30 syslog 158, 164 syslog servers, see also logs system log, see logs system name 44, 139 system uptime 45 system-default.conf 171

Т

Telnet with SSH 154 time 139 time servers (default) 142 trademarks 244 Transport Layer Security (TLS) 156 troubleshooting 178, 193 Trusted Certificates, see also certificates 132

U

upgrading firmware **173** uploading configuration files **172** firmware **173**

shell scripts 175, 192 usage CPU 45, 47 flash 45 memory 45, 48 onboard flash 45 user authentication 86 user name rules 87 user objects 86 users 86 access, see also access users admin (type) 86 admin, see also admin users and service control 143 currently logged in 45 default lease time 90, 92 default reauthentication time 90, 92 lease time 89 limited-admin (type) 86 lockout 91 reauthentication time 89 types of 86 user (type) 86 user names 87

V

Vantage Report (VRPT) 158, 164 Virtual Local Area Network 72 VLAN 72 introduction 72 VRPT (Vantage Report) 158, 164

W

warm start 42 warranty 250 note 250 WDS 13 Web Configurator 26, 30 access 30 requirements 30 supported browsers 30

255

WEP (Wired Equivalent Privacy) 94 wireless channel 203 wireless client 76 Wireless Distribution System (WDS) 13 wireless LAN 203 wireless network example 75 overview 75 wireless profile 93 layer-2 isolation 93 MAC filtering 93 radio 93 security 93 SSID 93 wireless security 16, 203 wireless station 76 Wizard Setup 49 WLAN interface 17 WPA2 94 WWW 144 and certificates 145 see also HTTP, HTTPS 144

Ζ

ZDP 22 ZON Utility 22 Zyxel Discovery Protocol 22